FRANCHISE DISCLOSURE DOCUMENT



ALL Franchising, LLC A Delaware limited liability company 8100 E. Indian School Road, Suite 201 Scottsdale, Arizona 85251 Phone: (800) 267-7816

www.AssistedLivingLocators.com

ALL Franchising, LLC offers franchises for the operation of a home-based business that assists seniors and their families in locating assisted living facilities, memory care communities, nursing homes, senior care homes and independent living senior communities that meet their geographic, financial and clinical needs and preferences.

The total investment necessary to begin operation of an Assisted Living Locators franchise ranges from \$74,635 to \$94,510. This includes \$63,395 that must be paid to us.

This Disclosure Document summarizes certain provisions of your franchise agreement and other information in plain English. Read this Disclosure Document and all accompanying agreements carefully. You must receive this Disclosure Document at least 14 calendar days before you sign a binding agreement with, or make any payment to, the franchisor or an affiliate in connection with the proposed franchise sale. **Note, however, that no government agency has verified the information contained in this document.**

You may wish to receive your Disclosure Document in another format that is more convenient for you. To discuss the availability of disclosures in different formats, contact the franchisor by mail at 8100 E. Indian School Road, Suite 201, Scottsdale, Arizona 85251, or by phone at (800) 267-7816.

The terms of your contract will govern your franchise relationship. Don't rely on the Disclosure Document alone to understand your contract. Read all of your contract carefully. Show your contract and this Disclosure Document to an advisor, like a lawyer or accountant.

Buying a franchise is a complex investment. The information in this Disclosure Document can help you make up your mind. More information on franchising, such as "A Consumer's Guide to Buying a Franchise," which can help you understand how to use this Disclosure Document, is available from the Federal Trade Commission (the "FTC"). You can contact the FTC at 1-877-FTC-HELP or by writing to the FTC at 600 Pennsylvania Avenue, NW, Washington, DC 20580. You can also visit the FTC's home page at www.ftc.gov for additional information. Call your state agency or visit your public library for other sources of information on franchising.

There may also be laws on franchising in your state. Ask your state agencies about them.

Issuance Date: MayMay 9, 20242025

How to Use this Franchise Disclosure Document

Here are some questions you may be asking about buying a franchise and tips on how to find more information:

QUESTION	WHERE TO FIND INFORMATION
How much can I earn?	Item 19 may give you information about outlet sales, costs, profits or losses. You should also try to obtain this information from others, like current and former franchisees. You can find their names and contact information in Item 20 or EXHIBIT "E".
How much will I need to invest?	Items 5 and 6 list fees you will be paying to the franchisor or at the franchisor's direction. Item 7 lists the initial investment to open. Item 8 describes the suppliers you must use.
Does the franchisor have the financial ability to provide support to my business?	Item 21 or EXHIBIT "F" includes financial statements. Review these statements carefully.
Is the franchise system stable, growing, or shrinking?	Item 20 summarizes the recent history of the number of company-owned and franchised outlets.
Will my business be the only Assisted Living Locators business in my area?	Item 12 and the "territory" provisions in the franchise agreement describe whether the franchisor and other franchisees can compete with you.
Does the franchisor have a troubled legal history?	Items 3 and 4 tell you whether the franchisor or its management have been involved in material litigation or bankruptcy proceedings.
What's it like to be an Assisted Living Locators franchisee?	Item 20 or EXHIBIT "E" lists current and former franchisees. You can contact them to ask about their experiences.
What else should I know?	These questions are only a few things you should look for. Review all 23 Items and all Exhibits in this disclosure document to better understand this franchise opportunity. See the table of contents.

What You Need To Know About Franchising Generally

Continuing responsibility to pay fees. You may have to pay royalties and other fees even if you are losing money.

Business model can change. The franchise agreement may allow the franchisor to change its manuals and business model without your consent. These changes may require you to make additional investments in your franchise business or may harm your franchise business.

Supplier restrictions. You may have to buy or lease items from the franchisor or a limited group of suppliers the franchisor designates. These items may be more expensive than similar items you could buy on your own.

Operating restrictions. The franchise agreement may prohibit you from operating a similar business during the term of the franchise. There are usually other restrictions. Some examples may include controlling your location, your access to customers, what you sell, how you market, and your hours of operation.

Competition from franchisor. Even if the franchise agreement grants you a territory, the franchisor may have the right to compete with you in your territory.

Renewal. Your franchise agreement may not permit you to renew. Even if it does, you may have to sign a new agreement with different terms and conditions in order to continue to operate your franchise business.

When your franchise ends. The franchise agreement may prohibit you from operating a similar business after your franchise ends even if you still have obligations to your landlord or other creditors.

Some States Require Registration

Your state may have a franchise law, or other law, that requires franchisors to register before offering or selling franchises in the state. Registration does not mean that the state recommends the franchise or has verified the information in this document. To find out if your state has a registration requirement, or to contact your state, use the agency information in EXHIBIT "A".

Your state also may have laws that require special disclosures or amendments be made to your franchise agreement. If so, you should check the State Specific Addenda. See the Table of Contents for the location of the State Specific Addenda.

Special Risks to Consider About This Franchise

Certain states require that the following risk(s) be highlighted:

- 1. <u>Out-of-State Dispute Resolution</u>. The franchise agreement requires you to resolve disputes with the franchisor by litigation only in Arizona. Out-of-state litigation may force you to accept a less favorable settlement for disputes. It may also cost more to litigate with the franchisor in Arizona than in your own state.
- 2. <u>Sales Performance Required</u>. You must maintain minimum sales performance levels. Your inability to maintain these levels may result in loss of any territorial rights you are granted, termination of your franchise, and loss of your investment.
- 3. **Spousal Liability.** Your spouse must sign a document that makes your spouse liable for all financial obligations under the franchise agreement even though your spouse has no ownership interest in the franchise. This guarantee will place both your and your spouse's marital and personal assets, perhaps including your house, at risk if your franchise fails.
- 4. <u>Mandatory Minimum Payments</u>. You must make minimum royalty or advertising fund payments, regardless of your sales levels. Your inability to make the payments, may result in termination of your franchise and loss of your investment.

Certain states may require other risks to be highlighted. Check the "State Specific Addenda" (if any) to see whether your state requires other risks to be highlighted.

THE STATE OF MICHIGAN PROHIBITS CERTAIN UNFAIR PROVISIONS THAT ARE SOMETIMES IN FRANCHISE DOCUMENTS. IF ANY OF THE FOLLOWING PROVISIONS ARE IN THESE FRANCHISE DOCUMENTS, THE PROVISIONS ARE VOID AND CANNOT BE ENFORCED AGAINST YOU.

Each of the following provisions is void and unenforceable if contained in any document relating to a franchise:

- (a) A prohibition on the right of a franchisee to join an association of franchisees.
- (b) A requirement that a franchisee assent to a release, assignment, novation, waiver, or estoppel which deprives a franchisee of rights and protections provided in this act. This shall not preclude a franchisee, after entering into a franchise agreement, from settling any and all claims.
- (c) A provision that permits a franchisor to terminate a franchise prior to the expiration of its term except for good cause. Good cause shall include the failure of the franchisee to comply with any lawful provision of the franchise agreement and to cure such failure after being given written notice thereof and a reasonable opportunity, which in no event need be more than 30 days, to cure such failure.
- (d) A provision that permits a franchisor to refuse to renew a franchise without fairly compensating the franchisee by repurchase or other means for the fair market value at the time of expiration of the franchisee's inventory, supplies, equipment, fixtures, and furnishings. Personalized materials which have no value to the franchisor and inventory, supplies, equipment, fixtures, and furnishings not reasonably required in the conduct of the franchise business are not subject to compensation. This subsection applies only if: (i) The term of the franchise is less than 5 years and (ii) the franchisee is prohibited by the franchise or other agreement from continuing to conduct substantially the same business under another trademark, service mark, trade name, logotype, advertising, or other commercial symbol in the same area subsequent to the expiration of the franchise or the franchisee does not receive at least 6 months advance notice of franchisor's intent not to renew the franchise.
- (e) A provision that permits the franchisor to refuse to renew a franchise on terms generally available to other franchisees of the same class or type under similar circumstances. This section does not require a renewal provision.
- (f) A provision requiring that arbitration or litigation be conducted outside this state. This shall not preclude the franchisee from entering into an agreement, at the time of arbitration, to conduct arbitration at a location outside this state.
- (g) A provision which permits a franchisor to refuse to permit a transfer of ownership of a franchise, except for good cause. This subdivision does not prevent a franchisor from exercising a right of first refusal to purchase the franchise. Good cause shall include, but is not limited to:
- (i) The failure of the proposed transferee to meet the franchisor's then current reasonable qualifications or standards.
 - (ii) The fact that the proposed transferee is a competitor of the franchisor or

subfranchisor.

- (iii) The unwillingness of the proposed transferee to agree in writing to comply with all lawful obligations.
- (iv) The failure of the franchisee or proposed transferee to pay any sums owing to the franchisor or to cure any default in the franchise agreement existing at the time of the proposed transfer.
- (h) A provision that requires the franchisee to resell to the franchisor items that are not uniquely identified with the franchisor. This subdivision does not prohibit a provision that grants to a franchisor a right of first refusal to purchase the assets of a franchise on the same terms and conditions as a bona fide third party willing and able to purchase those assets, nor does this subdivision prohibit a provision that grants the franchisor the right to acquire the assets of a franchise for the market or appraised value of such assets if the franchisee has breached the lawful provisions of the franchise agreement and has failed to cure the breach in the manner provided in subdivision (c).
- (i) A provision which permits the franchisor to directly or indirectly convey, assign, or otherwise transfer its obligations to fulfill contractual obligations to the franchisee unless provision has been made for providing the required contractual services.

If the franchisor's most recent financial statements are unaudited and show a net worth of less than \$100,000.00, the franchisee may request the franchisor to arrange for the escrow of initial investment and other funds paid by the franchisee until the obligations, if any, of the franchisor to provide real estate, improvements, equipment, inventory, training or other items included in the franchise offering are fulfilled. At the option of the franchisor, a surety bond may be provided in place of escrow.

THE FACT THAT THERE IS A NOTICE OF THIS OFFERING ON FILE WITH THE ATTORNEY GENERAL DOES NOT CONSTITUTE APPROVAL, RECOMMENDATION, OR ENDORSEMENT BY THE ATTORNEY GENERAL.

Any questions regarding this notice should be directed to:

State of Michigan
Department of Attorney General
CONSUMER PROTECTION DIVISION
Attention: Franchise Section
G. Mennen Williams Building, 1st Floor
525 West Ottawa Street
Lansing, Michigan 48913
Telephone Number: (517) 373-7117

TABLE OF CONTENTS

ITEM 1_	FRANCHISO	OR AND ANY PARENTS, PREDECESSORS AND AFFILIATES	1
ITEM 2	BUSINESS I	EXPERIENCE	7
ITEM 3	LITIGATIO	N	9
ITEM 4	BANKRUPT	CCY	<u></u> 11
ITEM 5	INITIAL FE	ES	11
ITEM 6	OTHER FEE	ES	12
ITEM 7	ESTIMATEI	ES D INITIAL INVESTMENT	19
ITEM 8	RESTRICTION	ONS ON SOURCES OF PRODUCTS AND SERVICES	21
ITEM 9	FRANCHISI	EE'S OBLIGATIONS	24
ITEM 10	FINANCINO		25
ITEM 11	_FRANCHIS(OR'S ASSISTANCE, ADVERTISING, COMPUTER SYSTEMS AND	
	TRAINING.		<u></u> 25
ITEM 12	TERRITORY	Y	<u></u> 38
ITEM 13	TRADEMAI	RKS	<u></u> 43
ITEM 14	PATENTS, O	COPYRIGHTS, AND PROPRIETARY INFORMATION	<u></u> 46
ITEM 15		ON TO PARTICIPATE IN THE ACTUAL OPERATION OF THE FRANCH	
	BUSINESS.		<u></u> 47
		ONS ON WHAT THE FRANCHISEE MAY SELL	
		TERMINATION, TRANSFER, AND DISPUTE RESOLUTION	
ITEM 18	PUBLIC FIC	GURES	<u></u> 47
ITEM 19	FINANCIAL	PERFORMANCE REPRESENTATIONS	52
ITEM 20	OUTLETS A	AND FRANCHISEE INFORMATION	<u></u> 56
ITEM 21	FINANCIAL	L STATEMENTSS	<u></u> 65
ITEM 22	CONTRACT	?S	<u></u> 65
ITEM 23	RECEIPT		<u></u> 65
EXHIBIT	"A"	STATE AGENCIES AND ADMINISTRATORS	
EXHIBIT	"B"	FRANCHISORS AGENT FOR SERVICE OF PROCESS	
EXHIBIT	"C"	FRANCHISE AGREEMENT	
EXHIBIT	"D"	TABLE OF CONTENTS OF BRAND STANDARDS MANUAL	
EXHIBIT	"E"	LIST OF FRANCHISEES	
EXHIBIT	"F"	FINANCIAL STATEMENTS	
EXHIBIT	"G"	OTHER AGREEMENTS	
-EXHIBIT	Γ "G"-1	FRANCHISEE DISCLOSURE QUESTIONNAIRE	
-EXHIBIT	Γ "G"-2	GENERAL RELEASE	
-EXHIBIT	Γ "G"-3	ALL-IN USER AGREEMENT AND PRIVACY POLICY	
-EXHIBIT		HIPAA BUSINESS ASSOCIATE AGREEMENT	
-EXHIBIT		POWER OF ATTORNEY – TELEPHONE AND INTERNET	
-EXHIBIT		SAMPLE CLIENT CONTRACT	
-EXHIBIT	Γ "G"-7	FRANCHISE RESALE AGREEMENT	
EXHIBIT		STATE ADDENDA AND AGREEMENT RIDERS	
EXHIBIT	"I"	STATE EFFECTIVE DATES	
EXHIBIT	"T"	RECEIPTS	

ITEM 1 FRANCHISOR AND ANY PARENTS, PREDECESSORS AND AFFILIATES

To simplify the language in this Disclosure Document, "we," "us" and "the Company" mean ALL Franchising, LLC - the franchisor. "You" means the person who buys an Assisted Living Locators franchise - the franchisee, and includes your partners if you are a partnership, your shareholders if you are a corporation, and your members if you are a limited liability company. For purposes of this Disclosure Document, an "ALL Business" refers to an Assisted Living Locators business, including any Assisted Living Locators business operated by us, our affiliates or a franchisee.

Corporate Information

We are a Delaware limited liability company that was organized on August 22, 2022. Our principal business address is 8100 E. Indian School Road, Suite 201, Scottsdale, Arizona 85251. Our phone number is (800) 267-7816. We also do business under the name "Assisted Living Locators". Our agent for service of process is disclosed in EXHIBIT "B" to this Disclosure Document.

Our Business Experience

The Assisted Living Locators system founder, Angela Olea, opened the first ALL Business in Arizona in 2003. We began offering franchised ALL Businesses in May 2006. We currently operate businesses of the type being offered under this disclosure document. We currently engage in no business activities other than offering franchises for ALL Businesses, administering the franchise system and operating company-owned ALL Businesses. We have never offered franchises in any other line of business.

On August 23, 2022, we merged with CALLRN Franchise LLC, a Delaware limited liability company formed on February 3, 2006, with a principal business address of 7330 E. Earll Drive, Suite E, Scottsdale, Arizona 85251 and we are the surviving entity.

Parents, Predecessors and Affiliates

We do not have any predecessors. Our direct and indirect parent companies include:

PARENT COMPANIES			
Company Name	Principal Business Address	Direct or Indirect Parent	
Evive Brands, LLC ("Evive")	Same as ours	Direct	
EHC Holding Company, LLC ("EHCH")	630 Fifth Avenue, Suite 400 New York, New York 10111	Indirect	
Riverside Micro-Cap Fund VI-A, L.P.*	45 Rockefeller Center 630 Fifth Avenue, Suite 400 New York, New York 10111	Indirect	

^{*} Riverside Micro-Cap Fund VI-A, L.P. is part of The Riverside Company, which is a global private equity firm focused on investing in and acquiring growing businesses. Riverside Micro-Cap Fund VI-A, L.P. indirectly acquired our franchise system on August 24, 2022.

Our affiliate ALL Licensing, LLC is a Nevada limited liability company formed on August 22, 2022, with a principal business address of 8100 E. Indian School Road, Suite 201, Scottsdale, Arizona 85251. ALL Licensing, LLC owns the "Assisted Living Locators" service marks and has licensed them to us (see Item 13).

We do not have any affiliates that provide goods or services to our franchisees. As further described in the table below, we have 3 affiliates that offer franchises in other lines of business. None of these affiliates have operated an ALL Business.

AFFILIATE FRANCHISING COMPANIES				
Franchised Business	Name of Affiliate Franchisor	Principal Place of Business	Period of Time Franchises Offered by Affiliate	Number of Open Franchisees in US (as of 12/31/ 2023 2024)
1. Grasons ⁴	B & P Burke, LLC	Same as ours	November 2014 to present	50 <u>61</u>
2. Executive Home Care ²	Executive Home Care Franchising, LLC	Same as ours	June 2012 to present	18 <u>22</u>
3. Brothers Gutters-3	Brothers Parsons Franchising LLC	Same as ours	November 2023 to present	110 <u>355</u>

- 1. Grasons is a business that specializes in estates sale and business liquidation services under the trade name GRASONS®.
- 2. Executive Home Care is a business that provides in-home comprehensive care and non-medical services to home care clients, as well as supplemental healthcare staffing services to institutional clients under the service mark EXECUTIVE HOME CARE®.
- 3. Brothers Gutters is a business that provides gutter installation, maintenance, cleaning, repair, and related services and products under the service mark THE BROTHERS THAT JUST DO GUTTERS[®].

We do not have any other affiliates that offer franchises in this or any other line of business.

Other Riverside Company Portfolio Franchise Companies

Through various private equity funds managed by The Riverside Company, the following portfolio companies of The Riverside Company offer franchises in the US:

EverSmith Brands

OTHER PORTFOLIO COMPANIES OF THE RIVERSIDE COMPANY (EVERSMITH BRANDS)			
Franchised Business	Name of Franchisor	Principal Place of Business	Number of Open Franchisees in US (as of 12/31/20242022)
1. US Lawns	U.S. Lawns, Inc.	6700 Forum Drive, Suite 150 Orlando, Florida 32821	208 <u>210</u>
2. milliCare Floor & Textile Care	milliCare Franchising, LLC	1515 Mockingbird Lane 6700 Forum Drive, Suite 410150 Charlotte, Orlando, Florida 32821 North Carolina 28209	56 <u>59</u>
3. Kitchen Guard	Kitchen Guard Franchising, Inc.	1515 Mockingbird Lane 6700 Forum Drive, Suite 410150 Charlotte, North Carolina 28209 Orlando, Florida 32821	<u>0</u> <u>5</u>
4. Prism Specialties	Restoration Specialties Franchise Group, LLC	6700 Forum Drive, Suite 150 Orlando, Florida 32821	<u>93</u>
5. The Seals	The Seals Franchising, LLC	6700 Forum Drive, Suite 150 Orlando, Florida 32821	<u>4</u>

- 1. US Laws is a business that offers outdoor commercial property and landscaping services mark US LAWNS.
- 2. milliCare is a business that offers cleaning and maintenance of floor coverings and interior finishes and related services under the service mark MILLICARE FLOOR & TEXTILE CARE.

- 3. Kitchen Guard is a business that offers commercial kitchen exhaust system cleaning, inspection, maintenance, and restoration services under the service mark KITCHEN GUARD.
- 4. Prism Specialties is a business that offers electronic, art, textile and document recovery, repair and restoration services under the service mark PRISM SPECIALTIES.
- 5. The Seals is a business that sells and installs gaskets for refrigeration door units, freezer doors, oven doors, hardware and cutting board under the service mark THE SEALS.

Head-To-Toe Brands

OTHER PORTFOLIO COMPANIES OF THE RIVERSIDE COMPANY (HEAD-TO-TOE BRANDS)			
Franchised Business	Name of Franchisor	Principal Place of Business	Number of Open Franchisees in US (as of 12/31/20242022)
1. Bishops	BCC Franchising, LLC	Terminal Tower 50 Public Square 29th Floor Cleveland, 550 Reserve Street, Suite 380 Southlake, Texas 76092 Ohio 44113	4 <u>2</u> 4 <u>0</u>
2. Frenchies Modern Nail Care	Frenchies, LLC	2679 West Main #363 Littleton, Colorado 80120550 Reserve Street, Suite 380 Southlake, Texas 76092	23
3. Lash Lounge	The Lash Franchise Holdings, LLC	4370 Varsity Drive 550 Reserve Street, Suite G.380 Ann Arbor, Michigan 48108 Southlake, Texas 76092	127 <u>140</u>

- 1. Bishops is a business that offers haircuts, coloring, and barber services under the service <u>mark BISHOPS.</u>
- 2. Frenchies is a business that offers hand and foot care under the service mark FRENCHIES MODERN NAIL CARE.
- 3. Lash Lounge is a business that offers permanent and temporary eyelash and eyebrow extensions and other eye enhancing services under the service mark LASH LOUNGE.

Best Life Brands

OTHER PORTFOLIO COMPANIES OF THE RIVERSIDE COMPANY (BEST LIFE BRANDS)			
Franchised Business Name of Franchisor		Principal Place of Business	Number of Open Franchisees in US (as of 12/31/20242022)
1Blue Moon Estate Sales	Blue Moon Franchise Systems, LLC	900 Wilshire Drive, Suite 102 Troy, Michigan 48084	109 <u>124</u>
2. Boost Home Healthcare	Boost Franchise Systems, LLC	900 Wilshire Drive, Suite 102 Troy, Michigan 48084	6
3. ComForCare Home Care	ComForCare Franchise Systems, LLC	900 Wilshire Drive, Suite 102 Troy, Michigan 48084	230 <u>248</u>
4CarePatrol	CarePatrol Franchise Systems, LLC	900 Wilshire Drive, Suite 102 Troy, Michigan 48084	172 <u>201</u>

OTHER PORTFOLIO COMPANIES OF THE RIVERSIDE COMPANY (BEST LIFE BRANDS)			
Franchised Business	Name of Franchisor	Principal Place of Business	Number of Open Franchisees in US (as of 12/31/20242022)
5Next Day Access	Next Day Access, LLC	3150 Stage Post 900 Wilshire Drive. Suite 101 102 Bartlett, Tennessee 38133 Troy. Michigan 48084	27 <u>50</u>

- 1. Blue Moon is a business that sells personal property as well as the provision of consignment sales for those who are downsizing, relocating, or are deceased under the service mark BLUE MOON ESTATE SALES.
- 2. Boost is a business that offers intermittent care ordered by a doctor and performed by a home health aide and other licensed healthcare providers to patients of all ages with acute and chronic long-term complex health conditions within the patient's residence or within health care facilities under the mark BOOST HOME HEALTHCARE.
- 3. ComForCare is a business that offers (a) companionship and personal/domestic care services, and other special needs services, primarily on a non-medical basis, for seniors and people of all ages so that they may remain in their residences, (b) supplemental healthcare staffing services for persons who need this kind of assistance in their home or a facility in which they reside, and (c) private duty nursing services, all under the mark COMFORCARE HOME CARE.
- 4. CarePatrol is a business that offers senior living placement, referral, and consulting services to families under the mark CAREPATROL.
- 5. Next Day is a business that offers ramps and other products and accessories that enhance the life of physically disabled or challenged persons under the mark NEXT DAY ACCESS.

Threshold Brands

OTHER PORTFOLIO COMPANIES OF THE RIVERSIDE COMPANY (THRESHOLD BRANDS)			
Franchised Business	Name of Franchisor	Principal Place of Business	Number of Open Franchisees in US (as of 12/31/20242022)
1. Maid Pro	Maid Pro Franchise, LLC	77 North Washington Street Boston, Massachusetts 02114	238 <u>237</u>
2. Patio Patrol	FlyFoe, LLC d/b/a Patio Patrol	77 North Washington Street Boston, Massachusetts 02114	7
3 2. Men in Kilts	Men in Kilts US, LLC	77 North Washington Street Boston, Massachusetts 02114	20 <u>23</u>
4 <u>3</u> . Pestmaster	Pestmaster Franchise Network, LLC	9716 South Virginia Street, Suite E Reno, Nevada 89511	52 <u>57</u>
54. USA insulation	USA Insulation Franchise, LLC	17700 Saint Clair Avenue Cleveland, Ohio 44110	100 <u>109</u>
65. Granite Garage Floors	Granite Garage Floors Franchising, LLC	110 Mansell Circle, Suite 375 Roswell, Georgia 30075	44 <u>55</u>
7 <u>6</u> . Mold Medics	Mold Medics Franchising, LLC	811 Washington Avenue Carnegie, Pennsylvania 15106	4 <u>6</u>

0	OTHER PORTFOLIO COMPANIES OF THE RIVERSIDE COMPANY (THRESHOLD BRANDS)			
Franchised Business	Name of Franchisor	Principal Place of Business	Number of Open Franchisees in US (as of 12/31/ <u>2024</u> 2023)	
87. Sir Grout	Sir Grout Franchising, LLC	77 North Washington Street Boston, Massachusetts 02114	62 <u>71</u>	
98. Miracle Method	Miracle Method, LLC	4310 Arrowswest Drive Colorado Springs, Colorado 80907	194 201 (2 Master franchises)	
10.9. Plumbing Paramedics	PHP Franchise, LLC	750 E. 150 th Street Noblesville, Indiana 46060	<u> 515</u>	
11. Heating + Air Paramedics	PHP Franchise, LLC	750 E. 150 th Street Noblesville, Indiana 46060	<u>514</u>	

- 1. Maid Pro is a business that offers home cleaning services for residential and commercial customers under the mark MAID PRO.
- 2. Patio Patrol is a business that offers residential and commercial mosquito, wasp, fly, tick control and other general pest control services under the mark PATIO PATROL.
- 3.2. Men in Kilts is a business that offers window cleaning, gutter cleaning, pressure washing, siding cleaning, snow removal and other related services under the mark MEN IN KILTS.
- 4.3. Pestmaster is a business that offers structural and agricultural pest control and related services under the mark PESTMASTER.
- 5.4. USA Insulation is a business that offers residential insulation services under the mark USA INSULATION.
- 6.5. Granite Garage Floors is a business that sells and installs residential garage floor coating systems under the mark GRANITE GARAGE FLOORS.
- 7.6. Mold Medics is a business that offers mold remediation, air duct cleaning, radon testing and mitigation services, and other services and products under the mark MOLD MEDICS.
- 8.7. Sir Grout is a business that offers grout and tile cleaning, sealing, caulking and restoration services and other services under the mark SIR GROUT.
- 9.8. Miracle Method is a business that offers refinishing and restoration of bathtubs, sinks, showers, tiles, countertops and similar surfaces under the mark MIRACLE METHOD.
- 10.9. Plumbing Paramedics is a business that offers plumbing service franchises under the mark PLUMBING PARAMEDICS.
- <u>11.10.</u> Heating + Air Paramedics is a business that offers heating and air conditioning installation and service franchises under the mark HEATING + AIR PARAMEDICS.

Description of Franchised Business

If we award you a franchise, you will establish and operate a home-based business that assists seniors and their families in locating assisted living facilities, memory care communities, nursing homes, senior care homes and independent living senior communities ("Senior Communities") that meet their geographic, financial and clinical needs and preferences (your "Business"). Although the scope of an ALL Business is currently limited to providing placement and related services, we may expand the scope of the business to include the sale of related

products or merchandise in future.

Your clients will include families in need of senior housing options. You will conduct an assessment of each client's unique situation and needs (for example, budget, health care needs, religious affiliations, mobility requirements, social needs, etc.). Based on the results of the assessment, you will provide your client with information regarding Senior Communities that may be a good match for their needs. Once your client selects a Senior Community, you will help facilitate the placement.

As a franchisee, one of your important responsibilities will be inspecting, evaluating and establishing relationships with Senior Communities that are located within the geographic area we designate (your "<u>Territory</u>"). You must sign a Referral Network Participation Agreement (a "<u>Marketing Agreement</u>") with each Senior Community to whom you refer potential clients (some Senior Communities will not sign Marketing Agreements or may require that you utilize their form of Marketing Agreement). Under the Marketing Agreement, you agree to: (a) provide certain marketing services on behalf of the Senior Community; and (b) provide information about the Senior Community to your clients whose needs match the characteristics and attributes of the Senior Community.

From time to time, a franchisee may need to work with another franchisee to place a client with a Senior Community located in the other franchisee's territory. We developed policies and procedures for these situations, referred to as our "Referral Program". Our Referral Program (described in more detail in Item 12) establishes a framework for ALL Businesses to work together in a collaborative manner to achieve the best result for the client.

In order for you to serve as a more complete and valuable resource for your clients, we encourage you to identify and establish referral relationships with reputable providers of complementary senior care services, such as real estate agents, downsizing specialists, home health experts, financial planners, financing companies and elder-care attorneys ("Senior Service Providers"). Depending on your client's needs, you may refer them to an appropriate Senior Service Provider for purposes of providing them with the additional assistance they need. Senior Service Provides may also have clients in need of placement services that they can refer to you.

In order to generate a consistent flow of new clients, we strongly encourage you to establish referral relationships with hospitals, insurance companies, HMO's, transitional care companies and other similar types of businesses or institutions that are in a position to refer clients in need of placement services ("Referral Sources"). You must focus your efforts on establishing relationships with Referral Sources that are located within your Territory. If you establish a relationship with a Referral Source, such as an Employee Assistance Program, that refers 10 or more leads per month and at least 20% of the leads involve placements outside of your Territory, we may require that you transition the Referral Source to us, in which case we may, but need not, pay you compensation for leads that are placed with Senior Communities located outside of your Territory.

Your clients do not pay you for your services. Instead, when you place a client with a Senior Community in your Territory, the Senior Community pays you a fee. The fee may be calculated as a percentage of the fees that the Senior Community charges or it may be a flat fee. Similarly, when you refer a client to a Senior Service Provider, the Senior Service Provider may pay you a referral fee under the terms of the Referral Services Agreement. Some Senior Service Providers choose not to pay referral fees.

Each ALL Business is granted a license to use: (a) certain logos, service marks and trademarks, including the service mark "Assisted Living Locators®" (collectively, the "Marks"); and (b) the distinct system we developed for the operation of an ALL Business (the "System"), the distinctive characteristics of which include our:

- logo
- trade secrets
- client assessment tools and techniques
- marketing strategies
- lead generation programs
- referral programs

• operating system

"). The operational aspects of an Assisted Living Locators franchise are contained within our confidential Brand Standards Manual (the "Manual").

If we award you a franchise, you must sign the form of franchise agreement attached to this Disclosure Document as <u>EXHIBIT "C"</u> (the "<u>Franchise Agreement</u>"). You must develop, open and operate your Business in accordance with Franchise Agreement.

Market and Competition

The target market for ALL Businesses consists of seniors (65 years and older) and their families (45-65 years). The market for assisted living placement services is growing and, based on current statistics and projections, will continue to grow due to the increasing number of seniors in need of assisted living, senior housing and care services. The market for senior placement services is moderately competitive. You will compete with other businesses that offer referral and senior placement services. Examples include local senior-assisted living placement agencies, hospice, non-medical home health care agencies, as well as nursing homes and other out-of-home senior living facilities. Some of our competitors operate through franchise systems. We will assist you in locating and understanding your local competitors after you sign the Franchise Agreement.

Laws and Regulations

You must comply with all federal and state licensing and regulatory requirements relating to the operation of your Business.

Some states have laws that prohibit or restrict the manner by which assisted living facilities can pay persons that refer or place clients with the facility. For example, Nevada prohibits assisted living facilities from paying persons that refer or place clients with the facility. However, Nevada law does not prohibit the client, or the family of the client, from paying the person providing the referral or placement services. Texas and Oklahoma have laws that prohibit charging a percentage fee and arranging for transportation of the senior (but not necessarily the senior's family members) to see a facility. Some states only allow an assisted living facility to pay a person for referring or placing clients with the facility if certain conditions or exemption criteria are met. For example, some states only allow for payment of referral fees if the conditions for the referral services exemption are met under the Medicare or Medicaid anti-kickback law or safe harbor regulations (i.e., the federal anti-kickback statute that prohibits the direct payment of referral fees for federal healthcare programs). In states that have laws similar to those discussed in this paragraph, we will alter our System, if needed, to accommodate these laws.

You may also need to comply with the Health Insurance Portability and Accountability Act of 1996 and Title XIII of the American Recovery and Reinvestment Act of 2009. You must sign the form of HIPAA Business Associate Agreement that we specify from time to time. Our current form of HIPAA Business Associate Agreement is attached to this Disclosure Document as EXHIBIT "G"-4.

In addition to the industry-specific laws above, you must comply with all laws that apply to businesses generally, including laws governing discrimination and sexual harassment in the work place, minimum wage, smoking in public areas, EEOC and OSHA standards, worker's compensation and unemployment insurance and withholding and payment of federal and state income taxes, social security taxes and sales and service taxes.

There may be other local, state and/or federal laws or regulations pertaining to your Business with which you must comply. We strongly suggest that you investigate these laws before buying this franchise.

ITEM 2 BUSINESS EXPERIENCE

Chairman-and-Director: Tom Silk John Deignan

Mr. Tom Silk Deignan has served as our Chairman and Director since February 2024. Since February September 2024, he has held the same positions with: (a) our parent, Evive; (b) our affiliates, Executive Home Care Franchising, LLC, B & P Burke, LLC and Brothers Parsons Franchising LLC; and (c) Eversmith Brands, Head to Toe Brands and Threshold Brands. From September 2022 to present, he has served as Chairmanan Operating

<u>Partner for TES Solutions The Riverside Company in Cleveland, Ohio.</u> <u>Between 2013 and From May 2017 to August 2022, he held various positions with WorkStride 2024, he served as President and CEO of Baker Hill in New York, New York including: (a) Indianapolis, Indiana.</u>

<u>Chief Executive Officer</u> (October 2019 to August 2022); (b) Chief Customer Officer (July 2017 to October 2019); and (c) Executive Vice President and Member of the Board of Directors (2013 to October 2019).

Chief Executive Officer: : Ryan Parsons

Mr. Parsons has served as our Chief Executive Officer since November 2023. Since November 2023, he has held the same position with our parent Evive and our affiliates B & P Burke, LLC, Executive Home Care Franchising, LLC and Brothers Parsons Franchising LLC. From July 2014 to November 2023, he served as cofounder and Vice President of our affiliate, Brothers Parsons Franchising LLC, and its predecessor, The Brothers Franchising, Corp. Since 2002, Mr. Parsons has also served as the Vice President of Brothers Parsons HV LLC f/k/a The Brothers That Just Do Gutters HV, Inc., which owns and operates the original The Brothers That Just Do Gutters business located in Poughkeepsie, New York.

Brand President and Founder: Angela Olea, RN, CFE: Felicia Sanders

Angela Olea Felicia Sanders has held the following positions with us in Scottsdale, Arizona: (a) served as our Brand President since December 2024. From July 2024 to November 2024, she served as a Consultant for Zenoti in Seattle, Washington. From August 2022; (b) to June 2024, she served as VP of Operations for Woodhouse Spas LLC in Lakewood, Colorado. From July 2016 to June 2022, she served as Director of Operations for Massage Envy Franchising LLC in Scottsdale, Arizona.

Vice President of CALLRN Franchise LLC Operations: Angelica Olea

Angelica Olea has served as our Vice President of Operations from: (a) February 20062025 to present; and (b) March 2022 to August 2022; and (b) Chief Executive Officer of CALLRN Franchise LLC from July 2013 to August 2022. From September 2003 to January 2020, Ms. Olea served as President and Chief Executive Officer of Certified Assisted Living Locators, LLC, in Scottsdale, Arizona.

Director of Operations: Angelica Olea

Angelica Olea has been. She served as our Director of Operations since August 24, 2022, having previously served as our VP of Operations from March 2022. from August 2022 to February 2025. From February 2018 to March 2022, she served as CALLRN Franchise LLC's Franchise Support Manager. From January 2013 to January 2020, she was employed by Certified Assisted Living Locators, LLC, in Scottsdale, Arizona as: (a) Franchise Support Manager (February 2018 to January 2020); and (b) Administrative Assistant (January 2013 to February 2018).

President, Secretary and Manager: ——Caroline Ouoveser

Ms. Quoyeser has served as our Manager, President and Secretary since November 2021. She has held the same positions with our parent Evive (since February 2023) and our affiliates B & P Burke, LLC (since November 2021), Executive Home Care Franchising, LLC (since November 2021) and Brothers Parsons Franchising LLC (since November 2023). Ms. Quoyeser joined The Riverside Company in June 2016 as a Summer Analyst in its Santa Monica, California office. Since that time, she has held the following positions: (a) Summer Analyst (June 2016 to August 2016); (b) Private Equity Analyst (June 2017 to June 2019); (c) Senior Associate (June 2021 to January 2023); and (d) Assistant Vice President (January 2023 to present).

Senior-Vice President of Training and Development and Manager: Ken Parsons

Mr. Parsons has served as our Manager and Senior Vice President of Training and Development since November 2023. Since November 2023, he has also held the same positions with our parent Evive and our affiliates B & P Burke, LLC, LLC, Executive Home Care Franchising, LLC and Brothers Parsons Franchising LLC. From July 2014 to November 2023, he served as co-founder and President of our affiliate, Brothers Parsons Franchising LLC, and its predecessor, The Brothers Franchising, Corp. Since its founding in 1999, Mr. Parsons has also served as the President of Brothers Parsons HV LLC f/k/a The Brothers That Just Do Gutters

HV, Inc., which owns and operates the original The Brothers That Just Do Gutters business located in Poughkeepsie, New York.

Vice President and Manager: L. Joseph Lee

Mr. Lee has served as our Manager and Vice President and Manager since November 2021. He has held the same positions with our parent Evive (since February 2023) and our affiliates B & P Burke, LLC (since November 2021), Executive Home Care Franchising, LLC (since November 2021) and Brothers Parsons Franchising LLC (since November, 2023). Mr. Lee joined The Riverside Company in March 2006 as a Principal in its Cleveland, Ohio office. He has served as a Senior Partner since April 2013.

Chief Financial Officer: -Gregory Esgar

Mr. Esgar has served as our Chief Financial Officer since May 2022. He has held the same position with our parent Evive (since August 2023) and our affiliates B & P Burke, LLC (since May 2022), Executive Home Care Franchising, LLC (since May 2022) and Brothers Parsons Franchising LLC (since November 2023). From April 2018 to May 2022, Mr. Esgar served as Chief Financial Officer for Prose Franchising in Phoenix, Arizona.

Chief Growth Officer:—<u>Jason Wiedder</u>

Mr. Wiedder has served as our Chief Growth Officer since December 2021. He has held the <u>same</u> positions with our parent Evive (since February 2023) and our affiliates B & P Burke, LLC (since December 2021), Executive Home Care Franchising, LLC (since December 2021) and Brothers Parsons Franchising LLC (since November, 2023). From March 2018 to December 2021, he served as Vice President of Franchise Development for Always Best Care located in Roseville, California.

Vice President of Operations: Franchise Support and Technology: Michael deHart

Mr. deHart has served as our Vice President of Operations Franchise Support and Technology since February 2025. From May 2023 to February 2025, he served as our Vice President of Operations. From February 2020 to May 2023, he served as Director of Sales Operations for Gravitec Systems in Poulsbo, Washington. From January 2013 to February 2020, he served at Regional Sales Directors for Numotion in Seattle, Washington.

Director of Training:— Bethany Harley

Ms. Harley has served as our Director of Training since April 2023. From January 2019 to April 2023, she served as CORE & ADRD Trainer for Train the Trainer, LLC in Yemassee, South Carolina.

ITEM 3 LITIGATION

Concluded Matters

<u>CALLRN Franchise, LLC d/b/a Assisted Living Locators v. Vasia Corporation, Aaron Mishkin and Valerie Clark, Case No. 01-22-0000-9785</u>

On March 3, 2022, we filed a demand for arbitration against former franchisees Aaron Mishkin ("<u>Mishkin</u>") and Valerie Clark ("<u>Clark</u>") and their corporate entity, Vasia Corporation (collectively, "<u>Respondents</u>"). On February 9, 2022, we terminated four (4) Franchise Agreements with Mishkin and Clark due to their failure to timely pay the minimum monthly Royalty Fee and Technology Fee. We filed the arbitration demand after

Respondents' payment was returned for insufficient funds and Respondents failed and refused to reimburse us for the past due fees. The arbitration demand alleged that Respondents materially breached the Franchise Agreements and sought recovery of the past due fees, future unpaid Royalty Fees for the remainder of the unexpired initial terms of the Franchise Agreements in the amount of the minimum monthly fees due under Section 7.2 of the Franchise Agreements, together with our reasonable attorneys' fees and costs pursuant to Section 13.4 of the Franchise Agreements and A.R.S. § 12-341.01.

On March 28, 2022, Respondents filed their answering statement in which they deny owing the past due and future unpaid fees. On June 21,2022, Respondents filed counterclaims alleging: violation of the California Franchise Investment Law; unfair competition and business practices; violation of the Arizona Consumer Fraud Act; fraudulent inducement; constructive fraud post-sale; breach of fiduciary duty;, breach of the implied covenant of good faith and fair dealing/breach of duty of candor; breach of the implied covenant of good faith and fair dealing/breach of duty of loyalty; tortious bad faith breach of contract; negligence; and unjust enrichment. Respondents based their counterclaims on alleged misrepresentations in Item 19 of our Franchise Disclosure Document based on the methodology for disclosing Gross Invoiced Revenues Revenue and Gross Collected Revenues Revenue. Respondents sought rescission of their four (4) Franchise Agreements, an unspecified amount of compensatory damages (including return of their franchise fees and investment) and an award of their costs and reasonable attorneys' fees.

On May 18, 2023, we and Respondents entered a settlement agreement under which we agreed, without any other exchange of consideration, to repurchase Respondents' four Assisted Living Locators franchise territories for \$28,125 each, for a total payment of \$112,500.00, in exchange for the reconveyance of the territories, mutual releases and dismissals of all claims and counterclaims asserted in the arbitration hearing, with prejudice, and with each party to bear its own attorneys' fees and costs.

<u>CALLRN Franchise LLC and Assisted Living Locators Licensing, LLC v. JL2 Holdings, Inc., Lara Dickerson, Assisted Living Michelle Cohen, LLC and Michelle Cohen, Case No. 2:20-cv-01091-MTL</u>

On June 3, 2020, we and our affiliate, ALL Licensing, LLC (formerly known as Assisted Living Locators Licensing, LLC), filed a lawsuit against a former franchisee and her independent contractor in the United States District Court for the District of Arizona. The Defendants included the franchisee entity, JL2 Holdings, Inc. ("JL2H"), and its owner, Lara Dickerson ("Dickerson"), as well as the franchisee's independent contractor, Michelle Cohen ("Cohen") and her entity, Assisted Living Michelle Cohen, LLC ("ALMC").

On May 29, 2020, we terminated 3 Franchise Agreements with JL2H due to its operation of unauthorized websites and social media and repeated refusal to submit monthly Gross Revenues Revenue reports. We and our affiliate filed the lawsuit after discovering that Defendants were operating a competing business and continuing to use our Marks following our termination of the Franchise Agreements. The Complaint alleged that Defendants' operation of a competing business violated the post-termination noncompetition covenants imposed under the Franchise Agreements and Brand ProtectionConfidentiality Agreements. The Complaint also alleged trademark infringement based on Defendants continued use of our Marks. We sought preliminary and permanent injunctive relief and damages.

Defendants subsequently brought a counterclaim alleging that our termination of the Franchise Agreements breached the terms of a mutual release contained within a standardized amendment (the "Amendment") that we entered into with JL2H and numerous other franchisees. The purpose of the Amendment was to provide franchisees with certain financial relief to mitigate the financial hardships they faced as a result of the COVID-19 pandemic. Specifically, defendants claimed that: (i) we waived their defaults when we signed the Amendment; and (ii) our subsequent termination of the Franchise Agreements on the basis of those defaults was unlawful. The defendants sought an unspecified amount of damages, attorneys' fees and court costs.

On June 23, 2020, the Court granted our request for injunctive relief as to defendants Cohen and ALMC. Specifically, the court prohibited the defendants from: (i) competing with us for a 2-year period; and (ii) continuing to use our Marks. On October 2, 2020, defendants Dickerson and JL2H entered into a Settlement Agreement with us to resolve the dispute. Pursuant to the Settlement Agreement: (i) we, through our insurer, agreed to repurchase JL2H's 3 franchised territories and reimburse certain expenses in the total sum of \$46,190; (ii) JL2H and Dickerson agreed to return 2 telephone numbers associated with our trademarks and proprietary

information; (iii) JL2H agreed to comply with the post-termination noncompetition covenant, other than placement services rendered by Dickerson for friends and family without charge; and (iv) the parties dismissed the lawsuit with prejudice and signed mutual general releases, with each party bearing its own costs and attorneys' fees.

In addition to the foregoing, the Government Injunction

The following Settlement Order is currently effective against our affiliate:

On January 6, 2016, our affiliate, The Brothers Franchising, Corp., entered into a Settlement Order, case number SEC-2015-00056, with the Commonwealth of Virginia, State Corporation Commission, Division of Securities and Retail Franchising and paid a penalty of \$2,000 related to an unregistered sale of a franchise territory in Virginia. The Brothers Franchising, Corp. also paid \$500 for the cost of investigation. The sale was part of a 2015 transaction where the affiliate sold a Virginia territory to the father of our affiliate's founders prior to being registered in the Commonwealth of Virginia. Our affiliate agreed to not violate the act in the future.

Except for the three3 actions listed above, no litigation is required to be disclosed in this Item.

ITEM 4 BANKRUPTCY

No bankruptcy information is required to be disclosed in this Item.

ITEM 5 INITIAL FEES

You must pay us the following nonrefundable initial fees.

Initial Franchise Fee

When you sign the Franchise Agreement, you pay us a nonrefundable \$49,900 initial franchise fee. However, if you obtain SBA funding you pay us 10% of the initial franchise fee when you sign the Franchise Agreement and the 90% remaining balance 20 days after you sign the Franchise Agreement.

We currently offer the following discounts:

Type of Discount*	Discount	Qualifications for Discount
Veterans Discount	10% discount	Person holding at least a 51% interest in the franchise is an honorably discharged veteran of any branch of the United States military and provides Form DD-214.
Diversity Discount	5% discount	Person holding at least a 51% interest in the franchise is a member of a recognized minority group as designated by the SBA.
Registered Nurse Discount	10% discount	Person holding at least a 51% interest in the franchise is a registered nurse holding such license received through the National Council Licensure Examination (NCLEX-RN).
Multi-Unit Discount	Franchise 1 – No discount Franchise 2 – \$10,000 discount Franchise 3 and up – \$15,000 discount	You must (a) purchase 2 or more Franchises from us at the same time, (b) sign Franchise Agreements for all of the Franchises at the same time, (c) pay full \$49,900 initial franchise fee for 1st Franchise and discounted initial franchise fee for each additional Franchise at same time.
Small-Territory Discount	TBD	In rare circumstances, we may agree to sell a franchise for a territory smaller than our standard territory for a reduced initial franchise fee. This fee would be determined on a case-by-case basis.

^{*} _The Veterans Discount, Diversity Discount and Registered Nurse Discount may not be combined (if you qualify for multiple discounts, you receive the higher 10% discount). If you qualify for the Multi-Unit Discount or Small-Territory Discount, and you also qualify for the Veterans Discount, Diversity Discount

and/or Registered Nurse Discount, you will receive the Multi-Unit Discount or Small-Territory Discount plus an additional 5% (Diversity Discount) or 10% (Veterans Discount and/or Registered Nurse Discount) off the first initial franchise fee. Under the Multi-Unit Discount or the Small-Territory Discount, you will not receive any discount on the second or any subsequent franchise, nor will you receive more than a 10% discount even if you qualify for multiple discounts.

In $2024\overline{2023}$, we collected initial franchise fees ranging from \$25,000 to \$89,000.

<u>118,790 (for multi-territory purchase).</u> The initial franchise fee in uniformly imposed except for the discounts disclosed above.

Business Set-Up Fee

When you sign the Franchise Agreement, you pay us a nonrefundable \$10,000 business set-up fee. The business set-up fee covers the costs to provide you with a variety of services and materials, including the following:

- the pre-opening initial training program we conduct for your Managing Owner and high-level managers at our corporate headquarters
- a 2-day onsite training program we conduct in your Territory within the 12-month period after you open your Business (we only provide this training for the 1st franchised territory you purchase, and we reserve the right in our sole discretion to designate a substitute training or coaching program if we believe it would be more appropriate for your success and/or to postpone this training to take place at a date that is more than 12 months after you open your Business)
- access to support from an operations representative prior to initial training
- tuition for one individual to attend Dementia Care Certification training
- setting up a local landing page to promote your Business (which will be linked to our corporate website)
- setting up your social media platforms

The business set-up fee is uniformly imposed. However, in <u>20242023</u> we-reduced: (a) waived the business set-up fee for resale and/or multi-territory purchases where the franchisee was purchasing a 2nd, 3rd or 4th territory; and (b) discounted the business set-up fee to \$7,500 for one franchisee that was also a franchisee of an affiliate franchise system and \$5,000 for alla few franchisees that were purchasing a 2nd territory.

Marketing Fee

When you sign the Franchise Agreement, you pay us a nonrefundable \$3,000 marketing fee that we use to market and promote your Business during the ensuing 60-day period. The marketing fee also covers the initial supply of marketing materials we provide to you (estimated to be a 150 to 180-day supply). We may retain up to 20% of the marketing fee to compensate us for the marketing and advertising services we provide on your behalf. At our option, we may either directly perform the marketing and advertising services ourselves or we may outsource these functions to one or more suppliers of our choosing. At any time, we may require that you spend this amount directly with third-party suppliers on approved marketing and advertising (rather than paying us). The marketing fee is uniformly imposed. -However, in 20242023 we waived the initial marketing fee for one; (a) resales where the Business was already established; and (b) purchases of a 2nd territory where the franchisee that purchased through a resale-already owned and operated a pre-existing territory.

ALL-IN – Initial Setup Fee

We will grant you a license to use Assisted Living Locators Integrated Network ("<u>ALL-IN</u>"), which is our proprietary business management database. When you sign the Franchise Agreement, you pay us a nonrefundable \$495 one-time initial setup fee for ALL-IN (you pay us monthly licensing fees after opening). <u>In 2024 we waived the fee for a few existing franchisees who were acquiring existing/operational territories.</u> The initial setup fee is otherwise uniformly imposed.

ITEM 6 OTHER FEES

Type of Fee ¹	AMOUNT AMOUNT 2	DUE DATE	REMARKS
THEOFIE	TIMOUNI MICOUNI	DUEDATE	KENTAKIN

Type of Fee ¹	AMOUNT AMOUNT 2	DUE DATE	REMA	RKS
Royalty Fee	Greater of (a) 8% of Gross Collected Revenues ² Revenue or (b) the minimum royalty fee	20 th day of month for prior month's operations	See Note 3. You must sen your Gross Collected Rev royalties after your open listed on a Certificate of we issue to you). The min as follows: Months after Opening 1 (full or partial) & 2 3 through 12 13 through 24 25 through 36 Rest of term	enue. You begin paying ing date (i.e., the date Completion of Training
Referral Fee (Corporate Leads Program)	Amount we specify (, but will not to exceed 30% of Gross Collected Referral Proceeds) 4	Same as royalty fee	Imposed for Corporate I See Note 3- for additional	
Marketing Fee	[2 nd through 5 th Month After Opening] \$1,000 per month for 2 nd through 5 th full month after opening date, and \$500 per month for remainder of term, which you pay to us	1 st day of each month	See Note 5. Until the 5 th mpay the fee to us and warket your Business (we services). After the 5 th month you with third-party supplier you to pay us any deficit apply to market your B spend the required amount. We may increase the maximum of 10% per year.	we apply the funds to e may keep 20% for our spend the fee directly s, but we may require ency (which we would usiness) if you fail to t. harketing fee (up to a
	[Rest of Term] \$500 per month, which you pay directly to third- party suppliers			
Brand and System Development-Fund Fee	Greater of (a) 2% of Gross Collected Revenues 2 Revenue or (b)-minimum brand fund fee (the minimum brand fund contribution.fee is currently \$300 per month but we may increase it to \$500 per month on 30 days' prior notice)	Same as royalty fee	See Note 6.We currently system development fund brand fund fees after you have no voting right administration of the brand placement of advertising brand fund fee. We may maximum combined to marketing fees and brand not required to do so. We these fees.	d. You begin paying us our opening date. You as pertaining to the nd fund, the creation or , or the amount of the ty place a cap on the total expenditure for I fund fees, but we are
Ongoing Training & Assistance Fees	Varies, then current fee[Initial Training] \$1,500 per person (only applies to initial training after you open) 4	Prior to training (for initial training program) 10 days after invoice (for other training or assistance)	See Note 7.Imposed if Managing Owner or mar after you open or (b) a petraining after failing the fi	nager to initial training erson must retake initial

Type of Fee ¹	AMOUNT AMOUNT 2	DUE DATE	REMARKS
	[Onsite Training] \$1,250 per day with \$2,500 minimum for up to 2 days.4	10 days after invoice	Imposed if you request, and we agree, to provide onsite training or assistance in your Territory. No fee for 2-day onsite training we provide within 12 months after opening as part of initial training.
	[Other Training] \$125 per hour 4	10 days after invoice	Imposed if you request any training or assistance beyond what we are required to provide under the Franchise Agreement (including remote training or support). Does not apply to onsite training or assistance (instead the \$1,250 per day training fee listed above applies).
Conference Registration Fee	\$7501,000 per conference for 2 attendees plus \$350 for each additional attendee	60 days prior to conference or as otherwise invoiced	See Note 8. We may hold conferences to discuss matters affecting franchisees. Attendance is mandatory unless we designate attendance as. If you fail to attend a required conference you must pay a \$1,000 non-attendance fee (we will send you a copy of materials distributed at the conference). You are also responsible for all Travel Expenses.
Technology Fee	Varies Up to \$400 per month for each block of 4 users (current estimate is \$304 per month for ALL-IN and Google Workspace – see Note 5 for details)	Varies	See Note 9. Includes amounts you pay us or our affiliate for Technology Systems, including (a) amounts paid for proprietary items, (b) amounts we collect from you and remit to third-parties and (c) an administrative fee for managing the technology platform and negotiating/managing relationships with third-party licensors. It does not include amounts you pay to third parties.
Call Center Services	Then current fee, estimated between \$150Up to \$500 per month (not currently ehargedimposed)	1 st day of each month	You are required to utilizemust contract with a third-party call center for purposes of answeringto answer client calls and providing provide related client service and support, including outside of normal business hours. In the future, If we maychoose to operate athe call center ourselves in which case, you will be required to must pay this fee to us for call center services.
Product Purchases	Varies depending on item purchased	10 days after invoice	We <u>and our affiliates</u> may, <u>but need not</u> , <u>be a supplier serve as System suppliers</u> for eertain items purchased by franchisees, such as marketing material, equipment and operating supplies. We may also be a supplier for certaingoods and services, you must purchase. If we supply any goods or services this occurs, we will provide you with a price list upon request.
New Product or Supplier Testing	Cost of testing	10 days after invoice	This covers the costs of testing new products or inspecting new suppliers you propose.
Renewal Fee	\$7,500	At time you sign Renewal Agreementrenewal agreement	None.Imposed if you renew your franchise rights by signing a renewal Franchise Agreement.

Type of Fee ¹	AMOUNT AMOUNT 2	DUE DATE	REMARKS
Transfer Fee	\$10,000 (\$(reduced to \$2,500 if buyer is an existing franchisee)	Before transfer Transfer	Payable when you transfer or sell your franchise. No charge if franchise transferred to an entity you control or for certain transfers of ownership interests between existing owners. You pay the transfer fee for all Transfers other than Permitted Transfers. If our broker finds the buyer is found by a broker we engage, you must also reimburse us for all commissions we pay to the broker. If you sign a Franchise Resale Agreement, you must pay us both thea transfer fee and the Broker Commission or a franchise resale assistance service fee.
Collection Fee	Up to 35% of gross amounts collected on your behalf	As incurred	Due when we collect payment on your behalf for customers who are delinquent in their payment of 90 days or more past the contract term.
Broker Commission Franchise Resale Assistance Service Fee	Either Broker Commission(a) commission or other amount we payowe to Broker or a broker (if a broker finds the buyer) or (b) \$10,000 (if no broker is involved)	At time of transfer.	We offer an optional service to provide franchise sales assistance tohelps franchisees interested in sellingsell their business. Franchisees who ouse this service, you must sign a Franchise Resale Agreement with us (the current form is attached as EXHIBIT "G"-7) and either (a) reimburse us for all amounts we pay third-party brokers, referral agents, sales agents, development agents or lead generation sources (a "Broker") relating to the sale (a "Broker Commission") or (b) pay us a \$10,000 resale assistance us the service fee if buyer is not procured by a Broker to whom a Broker Commission is owed relating to the sale.
Fines Noncompliance Fee	\$1,000 per day if you fail for fail ure to perform background check on employee; up to \$2,500 per incident for other violations	Upon demand	PayableImposed if you fail to comply withbreach a mandatory standard or operating procedure (including timely submission of required reports or violation of Franchisee Code of Conduct) and do notfail to cure the non compliance within the time period we require. We may impose an additional fine forfee every 48 hours the non compliance issuebreach remains uncured after we impose the initial finefee. We will deposit all finesnoncompliance fees into the brand fund.
Default Reimbursements	All costs we incur to cure your default	10 days after invoice	If you fail to cure a breach of the Franchise Agreement or our brand standards in the time period we require, we may take steps to cure on your behalf and you must reimburse us for our costs (examples include failure to pay suppliers, maintain insurance or meet quality standards).
Audit Fee	\$2,500 plus actual cost of audit costs incurred (including travel and lodging expenses Travel Expenses for audit team)	10 days after invoice	Payable only Imposed if thean audit (a) reveals you understated any amount owed to us by at least 2% or (b) is necessary because you fail to send us required information or reports in a timely manner; or (b) reveals you understated any amount owed to us by 2% or more.
Late Fee	Lesser of 18% of amount past due or highest rate allowed by applicable law	10 days after invoice	None. Default interest is limited to 10% per annum in California.

Type of Fee ¹	AMOUNT AMOUNT 2	DUE DATE	REMARKS
Indemnification	Amount of our damages, losses or expenses	10 days after invoice	You must indemnify and reimburse—us for any damages, our losses from joint employer claims or expenses—we incur—due to the your operation of your the Business, your or breach of the Franchise Agreement or joint employer claims.
Attorneys' Fees and Costs	Amount of attorneys' fees and costs we incur	Upon demand	You must reimburse us for all attorneys' fees and other costs we incur relating to your breach of any term of the Franchise Agreement or any other related agreement with us or our affiliates.
Reimbursement of Insurance Costs	Amount of expenses we incur (including premiums)	10 days after invoice	If you fail to obtain and maintain the insurance we require, and we elect to do so on your behalf, you must reimburse us.
Mandatory Conference Attendance Fee	Up to \$2,000 per conference, plus cost of travel and lodging.	On the date we specify, which will be in advance of the conference date.	We have the right to require you (or your Managing Owner) to attend our Mandatory Conference, which generally will be held annually. You must pay us the Mandatory Conference Attendance Fee in advance of the conference date, on the date we specify. You are responsible for all of your travel and living expenses.

Notes:

- 1. Nature and Manner of Payment: All fees are imposed by and payable to us. All fees are nonrefundable and uniformly imposed. You must sign an ACH Authorization Form (attached to the Franchise Agreement as ATTACHMENT "D"), permitting us to electronically debit your designated bank account for all amounts owed to us and our affiliates (other than the initial franchise fee). You must deposit all Gross Collected Revenue Revenue into the bank account and ensure sufficient funds are available for withdrawal before each due date. You must deposit all revenue you receive in the designated bank account within 3 business days after receipt, including cash, checks and credit card receipts. You must pay us all taxes that are imposed upon us or that we are required to collect and pay by reason of the furnishing of products, intangible property (including trademarks) or services to you.
- <u>2. You must provide us with monthly reports of your Definitions: The following defined terms have the meanings given to them below:</u>
 - 2. "Gross Collected Revenues. "Gross Collected Revenues Revenue" means all gross sums that you collect in any way relating to your Business, such as revenues revenue from the sale or provision of goods or services, proceeds from business interruption insurance and referral fees that you collect, including referral fees paid by Referral Sources, Senior Care Providers, other ALL Businesses (such as fee splits under our Referral Program) or us (such as fee splits relating to a Regional Account, which is defined in Item 12). Gross Collected Revenues Revenue includes any Gross Collected Referral Proceeds (defined in Note 4 below) that you collect. With respect to fees generated through our Referral Program, Gross Collected Revenues Revenue only includes your share of the placement fee that you collect. Gross Collected Revenues Revenue excludes the following amounts (which you may deduct from your calculation of Gross Collected Revenues Revenue):
 - any referral fee you pay to us for a Corporate Lead
 - any placement fee you collect but pay to another ALL Business under our Referral Program
 - any sales or use tax you collect and pay to the applicable taxing authority
 - the wholesale price paid by you for any merchandise or other retail items that you sell (Gross Collected Revenues Revenue includes the difference between the retail price you receive from the sale of the item and the wholesale price you paid for the item)
- 3. You must begin paying us royalties after your opening date. Your "opening date" is the date listed on the

"Certificate of Completion of Training" that we issue to you, regardless of when you actually begin providing senior placement services. The "minimum royalty fee" varies based on the number of months that have elapsed from your opening date, as further described in the table below:

Months after Opening Date	Minimum Monthly Royalty Fee
Months 1 (full or partial) and 2	\$0 per month
Months 3 through 12	\$500 per month
Months 13 through 24	\$800 per month
Months 25 through 36	\$1,100 per month
Months 37 through end of term	\$1,400 per month

"Gross Collected Referral Proceeds" means all gross sums you collect from Corporate Leads we refer to you including placement fees and referral fees paid by Senior Service Providers for referrals of Corporate Leads.

"Permitted Transfer" means a Transfer: (a) between existing owners; or (b) by the owners to a new business entity that is 100% owned and controlled by the transferring owners. It does not include a Transfer described in (a) or (b) that results in the Managing Owner owning less than 20% of the franchised business

"Technology Systems" means all information and communication technology systems that we designate, including computer systems, point-of-sale systems, telecommunications systems, security systems and similar systems, together with the associated hardware, software (including cloud-based software) and related equipment, software applications, mobile apps and third-party services relating to the establishment, use, maintenance, monitoring, security or improvement of these systems.

"Transfer" means a transfer or assignment of: (a) the Franchise Agreement; (b) an ownership interest in the entity that is the "franchisee"; or (c) the Business you conduct under the Franchise Agreement.

"Travel Expenses" means all travel, meals, lodging, local transportation and other living expenses incurred:

(a) by us and our trainers, field support personnel, auditors or other representatives to visit your Territory; or

(b) by you or your personnel to attend training programs or conferences.

- 4.3. Referral Fee: If you place a Corporate Lead we refer to you, you must pay us a referral fee in the amount we specify from time to time. We may designate different referral fees for different types of Corporate Leads. For example, some referral fees may consist of a flat fee while others may be calculated as a percentage of the placement fee you collect. In no event will the referral fee exceed 30% of the gross sums that Gross Collected Referral Proceeds you collect from the Corporate Lead (referred to as "Gross Collected Referral Proceeds"). We reserve the right to collect the referral fee on all Gross Collected Referral Proceeds you generate from the Corporate Lead, including Gross Collected Referral Proceeds that consist of referral fees you receive from Senior Service Providers to whom you refer the Corporate Lead. You do not pay us a referral fee for any lead we send that does not qualify as a "Corporate Lead" (see "Corporate Leads Program" in Item 11 for more information). If we send you a Corporate Lead and you place the Corporate Lead under our Referral Program, our referral fee will come "off the top" and you and the other ALL Business will split the remainder of the placement fee in accordance with the terms of our Referral Program.
- 5. You pay us a monthly marketing fee that we use to market and promote your Business. We may retain up to 20% of the marketing fee to compensate us for the advertising and marketing services we provide. At any time, we may require that you spend this amount directly with third party suppliers on approved marketing and advertising (rather than paying us). Under current policy, you only pay us marketing fees through the 5th full month after opening (after the 5th month you spend the marketing fee directly with third party suppliers of advertising and marketing materials and services). If you fail to spend the required amount, we may require that you immediately pay us the deficiency amount which we will use to promote your Business. We may increase the marketing fee upon at least 30 days' prior notice, but any such increase will not exceed 10% in any 12 month period.
- 6. We have established and administer a brand and system development fund to promote public awareness of our brand and improve our System. You have no voting rights pertaining to the administration of the fund,

the creation and placement of the marketing materials or the amount of the required contribution. Your obligation to pay us brand fund contributions commences on your opening date. As of the issuance date of this Disclosure Document, the "minimum brand fund contribution" is \$300 per month. We may increase the minimum brand fund contribution (up to a maximum of \$500 per month) upon 30 days' prior notice to you. Your contributions to the brand fund are in addition to the marketing fees you must pay us. From time to time, we may implement policies that place a "cap" on the maximum total expenditures a franchisee may be required to pay for marketing fees and/or contributions to the brand fund, although we are under no obligation to do so.

7. Before you open, we will provide our pre-opening initial training program for your Managing Owner and initial high level managers in exchange for the business set up fee. You must pay us an additional fee of \$1,500 per person for: (a) any person that retakes initial training after failing the first attempt; and (b) any new Managing Owner or manager that attends our initial training program after you open.

You may request that we provide onsite training or assistance in your Territory. If we agree to do so, you must pay us a fee of \$2,500 per visit, which covers our training/assistance fee and our costs for meals, travel, and lodging for up to a 2 day visit. If we agree to visit your Territory for more than 2 days as part of a single visit, we may charge you an additional \$1,250 for each additional day we visit. We do not charge you the \$2,500 fee for the initial 2 day onsite training program that we conduct in your Territory within the 12 month period after you open your Business. We only provide a 2 day onsite training program (or substitute training or coaching program that we designate) for the 1st franchised territory that you purchase.

You may request that we provide additional training or support beyond what we are required to provide under the Franchise Agreement. For example, you may request that we assist you in negotiating a Marketing Agreement with a Senior Community. If we agree to provide additional training or support you request (either remotely or at our corporate headquarters), we may charge you our then current fee, currently \$125 per hour. Any additional training or support you request that is provided in your Territory (i.e., "onsite") will be subject to our then current onsite training fee, currently \$2,500, instead of the hourly fee.

- 8. We may hold periodic conferences to discuss business and operational issues affecting ALL Businesses. Attendance is mandatory. In addition to the conference registration fees, you are responsible for all expenses and costs the conference attendees incur, including wages, travel and living expenses. If you fail to attend a mandatory conference, we may charge you a \$1,000 non attendance fee (we will make available to you any written conference materials distributed to conference attendees).
- 4. You must acquire and utilize all information and communication technology systems that we specify from time to time (the "Technology Systems"). The "technology fee" includes all amounts you pay us or our affiliates relating to the Technology Systems, including amounts paid for proprietary items and amounts we collect from you and remit to third party suppliers based on your use of their systems, software, technology or services. The amount of the technology fee may change based on changes to the Technology Systems or the prices charged by third party suppliers with whom we enter into master agreements. Although we do not currently do so, we reserve the right to include within the technology fee a commercially reasonable administrative fee to compensate us for the time, money and resources we invest in administering the technology platform and the associated constituent components, negotiating with third party licensors and managing those relationships, and collecting and remitting technology fees owed to third party licensors on behalf of franchisees under master license arrangements. The technology fee does not include any amounts you pay directly to third party suppliers for any component of the Technology Systems. As of the issuance date of this Diselosure Document, Training Fees: We may increase these fees by not more than 10% per year. If we choose not to apply a fee increase in a given year, any unapplied increase may be carried forward and applied in a subsequent year.
- 9.5. Technology Fee: As of the issuance date of this Disclosure Document we charge a technology fee that covers the following technology-related goods and services and includes the following initial and ongoing fees:

TECHNOLOGY FEES*				
Licensed Product or Software	Description of Licensed Product or Software	Initial Fee	Ongoing Fee	
ALL-IN	Our proprietary business management software you will use to operate your Business	\$495 initial setup fee (due when you sign Franchise Agreement)	\$199250 per month for each block of 4 user licenses (due on 1st day of month)	
Google Workspace (formerly known as "G Suite")	Collection of cloud computing, productivity and collaboration tools, software and products developed by Google, including your required email addresses/accounts	None	\$2527 per month per account (due on 1st day of month)	

^{*} All fees described in the table above comprise our current "technology fee". We anticipate most franchisees will require 1 block license for ALL-IN and 2 Google Workspace accounts. We may increase the ALL-IN license fee on an annual basis. We may also increase the monthly fee for Google Workspace and other components of the technology fee upon 30 days' prior notice. We reserve the right to substitute different third party providers and you will be required to implement such changes upon written notice and will be solely responsible for all fees associated with such change.

ITEM 7 ESTIMATED INITIAL INVESTMENT

YOUR ESTIMATED INITIAL INVESTMENT				
Type of Expenditure	AMOUNT 1	METHOD OF PAYMENT	WHEN DUE	TO WHOM PAYMENT IS TO BE MADE
Initial Franchise Fee	\$49,900	Lump sum	At time you sign Franchise Agreement	Us
Business Set-Up Fee	\$10,000	Lump sum	At time you sign Franchise Agreement	Us
Food, Lodging & Travel Initial Training Expenses (1 person while training)	\$1,000 to \$1,500	As incurred	During training	Hotels, restaurants and airlines
Real Estate, Buildout and Improvements		Not Applicable – See Note 2		
Furniture & Furnishings (for home office)	\$0 to \$500	As incurred	Before opening	Suppliers
Technology Systems ³	\$0 to \$3,000	Lump sum	Before opening	Suppliers
Technology Fees ⁴ (pre-opening)	\$495	Lump sum	Before opening	Us
Marketing Fees ⁵ (pre-opening)	\$3,000	Lump sum	Before opening	Us
Utility Deposits	\$0 to \$300	As incurred	Before opening	Utility companies
Certified Senior Advisor (CSA) Certification Fee ⁶	\$990 to \$1,090	As incurred	Before opening or within 6 months after opening	Supplier
Dementia Care Certification Fee ⁷	\$0 to \$125	As incurred	Before opening or within 3 months after opening	Supplier
Business Licenses	\$50 to \$1,000	As incurred	Before opening	Government agencies
Professional Fees	\$300 to \$3,000	Lump sum	Before opening	Lawyers & accountants

YOUR ESTIMATED INITIAL INVESTMENT				
TYPE OF EXPENDITURE	AMOUNT 1	METHOD OF PAYMENT	WHEN DUE	TO WHOM PAYMENT IS TO BE MADE
Vehicle ⁸ (title, registration, deposit & 3 lease payments)	\$0 to \$5,000	Installments	Before opening and 3 months after opening	Supplier
Vehicle "Wraps" (Optional)	\$0 to \$2,800	Lump sum	Before opening	Supplier
Insurance ⁹ (12 months' premium)	\$1,700 to \$2,600	Lump sum	Before opening	Supplier
Additional Funds ¹⁰ (3 months)	\$7,200 to \$10,200	As incurred	As incurred	Suppliers, employees and us
Total Estimated Initial Investment 11	\$74,635 to \$94,510			

Notes:

- 1. <u>Financing & Refunds:</u> We do not offer direct or indirect financing for any of these items. None of the fees payable to us are refundable. We are unaware of any fees payable to third party suppliers that are refundable.
- 2. <u>Real Estate:</u> Assisted Living Locators is a home-based business. Most franchisees do not purchase or lease separate commercial office space for purposes of operating the business. However, a few elected to do so. Since we do not have any standards or specifications for offices or commercial facilities, we cannot estimate the costs. If you purchase or lease separate commercial office space, your initial investment will increase.
- 3. <u>Technology Systems</u>: This includes your initial cost to purchase and set up your Technology Systems, including your smart phone and computer system. It does not include the initial or ongoing technology fees you pay us (these fees are disclosed in a separate line item). We estimate the cost to purchase a new smart phone and computer system (including required software but excluding the \$495 ALL-IN setup fee) could range from \$1,000 to \$3,000. The cost varies depending on the specific computer and smart phone purchased. In many cases, franchisees already have a suitable smart phone and/or computer system that satisfies our requirements. The low estimate assumes you already own a suitable smart phone and computer system while the high estimate assumes you purchase a new smart phone and computer system.
- 4. Technology Fees: This estimate includes the \$495 initial setup fee you must pay for ALL-IN.
- 5. Marketing Fees: You pay us a marketing fee that we use to market and promote your Business. The first payment is \$3,000 and is due when you sign the Franchise Agreement. The 2nd payment is \$1,000 and is due the 1st day of the 2nd full month of operation. The 3rd payment is \$1,000 and is due the 1st day of the 3rd full month of operation. We do not require that franchisees spend any additional amounts on marketing during the initial 3-month period (other than the required contribution to the brand fund).
- 6. <u>CSA Certification Fee:</u> You must obtain your Certified Senior Advisor certification within 6 months after opening. We have negotiated a discounted fee for the certification program.
- 7. <u>Dementia Care Certification Fee:</u> You must obtain your Dementia Care certification within 3 months after opening. The business set-up fee includes the cost for 1 person to attend thethis certification program. If you send more than 1 person to Dementia Care certification training, you must pay the licensor an additional fee of \$125 per person. The low estimate assumes you send 1 person while the high estimate assumes you send a 2nd person.
- 8. <u>Vehicle:</u> Due to the nature of this business, you may need to travel to Senior Communities and client locations. At times you will "tour" Senior Communities with your clients. As a result, you must have access to a late-model vehicle that reasonably accommodates at least 2 passengers. We do not have any other standards or specifications for your vehicle. Most franchisees use their personal vehicle for the Business. The low estimate assumes you already own or lease an acceptable vehicle that you will use in the Business, in

which case you will not incur any additional expense. The high estimate assumes you lease a separate vehicle for the Business, and includes an estimated deposit/administrative fee, title and registration fees, and 3 monthly lease payments. The cost of leasing a vehicle varies widely depending on the type of vehicle, whether the vehicle is new or used, the amount of your deposit, whether you trade in another vehicle (and if so, the value of that vehicle), your credit rating and leasing rates available in your area.

9. <u>Insurance:</u> You must purchase the insurance policies we require through our designated insurance broker. In most cases, the initial 12-month premium for our required insurance is \$1,750. However, the actual premium may be somewhat lower or higher in a few states. You are free to purchase any additional insurance you deem appropriate. The high estimate in the table above assumes you choose to purchase additional insurance beyond our minimum requirements.

10. Additional Funds: This estimates your expenses during the first 3 months of operation, including:

- payroll costs (excluding any wage or salary paid to you)
-); post-opening technology fees (\$226304 per month, including (a) \$199 per month estimate for ALL-IN monthly licensing fee and (b) \$40 per month for 2 Google Workspace accounts)
- •—); post-opening marketing fees (\$1,000 per month starting 2nd month after opening)
- •—); phone expenses
- **utilities**
- •___gas
- •—<u>; and other miscellaneous expenses and required working capital</u>
- <u>10..</u> These figures are estimates based on the past experience of our franchisees and our affiliate in opening and operating ALL Businesses.
- 11. You may have additional expenses starting your Business. Budget: We strongly recommend you hire an accountant, business advisor or other professional to assist you in developing a budget for the construction development, opening and operation of your Business.

ITEM 8 RESTRICTIONS ON SOURCES OF PRODUCTS AND SERVICES

Source-Restricted Purchases and Leases - Generally

We require that you purchase or lease certain "source-restricted" goods and services for the development and ongoing operation of your Business. By "source-"Source-restricted," we mean that" means the good or service must meet our specifications and/or must be purchased from an approved or designated supplier (in some cases, an exclusive designated supplier, which may be us or an affiliate). Our specifications and list of approved and designated suppliers are contained in the Manual. We will notify you within 30 days of any changes to our specifications or list of approved or designated suppliers. We may notify you of these changes in various ways, including written or electronic correspondence, verbal or telephonic notification, amendments or updates to the Manual, bulletins, or other means of communication.

Supplier Criteria

Our criteria for evaluating a supplier supplier include standards for (a) quality, delivery, function, performance, design, appearance and price of the product or service as well as the and (b) dependability, production capabilities, reputation and financial viability strength of the supplier. Upon your request, we will provide you with any objective specifications pertaining to our evaluation of a supplier, although certain important subjective criteria (e.g., product appearance, design, functionality, etc.) are important to our evaluation but cannot be described in writing.

If you wantwish to purchase or lease a source-restricted item from a non-approved supplier, you must send us:

(a) a written request for approval; (b) product samples for testing purposes; and submit any(c) all additional

information that we request. We may require that you send us samples from the The supplier for testing. We may also require that we be allowed must agree to comply with our minimum insurance, indemnification and confidentiality requirements for system suppliers and allow us to periodically inspect the supplier's facilities their facility. We will notify you of our approval or disapproval decision within 30 days after we receive your request for approval plus all additional required information and product samples that we require. We may, at our option, re-inspect the facilities and products of any. We may periodically reinspect approved supplier products and suppliers and revoke our approval if thea product or supplier fails to meet any of our thencurrent criteria. You must reimburse us for all costs that we incur in reviewing a proposed supplier and testing theto evaluate products, and suppliers you propose.

Current Source--Restricted Items

As described below in more detail, we currently require that you purchase or lease the following source restricted goods and services: vehicle; Technology Systems; insurance policies; marketing materials and services; website development services; and call center services. We estimate that 65% of the total purchases and leases that will be We estimate 65% of the total purchases and leases required to establish your Business and 40% of your ongoing operating expenses will consist of source—restricted goods or services.—, as further described below:

Vehicle

You must purchase, lease or have access to a late-model vehicle that can reasonably accommodate at least 2 passengers. You must insure this vehicle according to our requirements. The vehicle must be kept in good condition and regularly cleaned. You can purchase or lease your vehicle from any supplier of your choosing. If you choose to purchase and install "wraps," the "wraps" must comply with our standards and specifications and must be purchased from a supplier that we designate or approve. At all times, the Managing Owner (and each other person associated with your Business that drives your clients) must hold a valid driver's license.

__Technology Systems

Your Technology Systems (including hardware, software, equipment, software applications, mobile apps and similar items) must meet our standards and specifications. Certain components of your Technology Systems must be purchased from approved or designated suppliers while other components may be purchased from any supplier of your choosing. We may also require that certain services relating to the establishment, use, maintenance, monitoring, security or improvement of your Technology Systems be purchased from approved or designated suppliers. Currently, we require that you license our proprietary business management database, ALL-IN, exclusively from us and we also require that you license and use QuickBooks (online version). ALL-IN is owned by our affiliate, Assisted Living Locators Licensing, LLC ("ALL Licensing"). ALL Licensing granted us a license to use (and the right to sublicense to our franchisees the right to use) the ALL-IN platform. You must also license your Google Workplace accounts exclusively through us. We reserve the right to substitute different third party Technology Systems providers and you will be required to implement such changes upon written notice and will be solely responsible for all fees associated with such change.

Insurance Policies

You must obtain the insurance coverage that we require from time to time (whether in the Franchise Agreement or in the Manual). Currently, we require that you purchase these policies exclusively through our designated insurance broker. The required coverage currently includes:

General Liability Occurrence coverage form

•	Each Claim Limit BI / PD	\$1,000,000
•	General Aggregate Limit of Liability	\$3,000,000
•	Non-Owned / Hired Auto Liability	\$1,000,000
	Professional Liability Occurrence	
	coverage form	
•	Each Professional Incident	\$1,000,000
•	Professional Aggregate Limit	\$3,000,000

You must also purchase worker's compensation and employer liability insurance as required by law, and any other limits and coverage that we periodically require. We strongly recommend, but do not require, that you purchase Sexual Abuse/Molestation coverage and Cyber Liability coverage. If you lease a vehicle for your Business, you are required to have a commercial auto policy. If you elect to continue your personal auto insurance policies, a "Business Use Endorsement" is required to be added to your automobile insurance policy with liability limits of at least \$250,000/\$500,000 split limits or \$500,000 Combined Single Limit. You must also purchase a \$1,000,000 personal umbrella policy. Proof of coverage is required annually depending on which type of vehicle you are using for your business. The required coverage and policies are subject to change. All insurance policies must be endorsed to: (a) name us as an additional insured; (b) contain a waiver by the insurance carrier of waive all subrogation rights against us; and (c) provide that we receive with at least 10 days' prior written notice of the termination, expiration, cancellation or modification of the policy. You are free tomay purchase any additional insurance that you deem appropriate.

Marketing Materials and Services

All marketing materials must comply with our standards and specifications. We must approve your marketing materials before you use them. You must purchase all branded marketing materials only from us or other suppliers we designate or approve. You must purchase your initial supply of marketing materials exclusively from us. We currently require you to purchase additional marketing materials through an online store we created through a promotional vendor. We may also require that you utilize a designated or approved supplier to conduct social media and/or other marketing on your behalf. We are the exclusive supplier of certain initial and ongoing marketing and promotional services that we conduct in exchange for the business set-up fee and marketing fees.

Call Center Services

You are required to utilizemust use a third-party call center for purposes of answeringto answer client calls and providing provide related client service and support, including outside of normal business hours ("Inbound Call Services"). We may also require that you to utilizeuse a third-party call center for purposes of client growth and engagement, including reaching out to develop leads and maintain client relationships ("Outbound Call Services"). You may must utilize thea third-party call center we designate or approve for Inbound Call Services and/or Outbound Call Services. In the future, we may operate a call center ourselves in which case you may be required to pay us for call center services. You may propose the use of another call center. We have the right to approve or disapprove of your request in our reasonable discretion.

Purchase Agreements

From time to time, we may negotiate relationships with suppliers to enable our affiliates and franchisees to purchase certain items at discounted prices. If we succeed, you may purchase these items at the discounted prices we negotiate (less any rebates or other consideration paid to us). Alternatively, we may purchase items in bulk and resell them to you at our cost plus a reasonable markup.

<u>, including favorable pricing terms, for the benefit of franchisees.</u> As of the issuance date of this Disclosure Document, we have negotiated purchase agreements (including pricing terms) <u>with suppliers</u> for the following goods and services:

- QuickBooks
- marketing materials and office supplies
- lead generation services
- insurance policies
- CSA certification program
- Dementia care certification program
- Universal Background Screen
- various in-home care service providers (these providers may pay you a referral fee for each referred client that ends up purchasing the services of that provider)

We may purchase items in bulk and resell them to you at our cost plus a reasonable markup. Currently there are no purchasing cooperatives but we may establish them in the future. You do not receive any material benefits for using designated or approved suppliers other than having access to any discounted pricing we negotiate. There are no purchasing cooperatives, although we may establish them in the future.

Franchisor Revenue from Source-Restricted Purchases

We are currently the exclusive designated supplier for the following goods and services:

- Your initial supply of marketing materials we provide in exchange for the marketing fee
- Certain initial and ongoing marketing and promotional services and business development tools we provide in exchange for the business set-up fee and marketing fees
- Certain initial and onsite training we provide in exchange for the business set-up fee
- Certain support we provide prior to initial training in exchange for the business set-up fee
- Our proprietary business management database, ALL-IN (including the separate website that may be created utilizing ALL-IN)
- The Google Workspace accounts you must utilize

We generate <u>revenues</u> from these purchases. We may designate ourselves <u>or our affiliate</u> as an approved or designated supplier for other <u>itemsgoods or services</u> in the future. <u>There are no approved or designated suppliers in which any of our officers own an interest.</u> No person affiliated with us is currently an approved (or the only approved) supplier.

We may receive rebates, payments or other material benefits from suppliers based on <u>franchiseeyour</u> purchases and <u>weleases. We</u> have no obligation to pass <u>them onthese amounts through</u> to <u>our franchiseesyou</u> or use them in any particular manner. We currently receive a credit equal to 10% of all purchases of marketing materials made by franchisees through the online store that we created with a promotional supplier. This credit is applied against purchases of marketing goods and services made by the brand and system development fund. We also may retain up to 20% of the marketing <u>feesfee paid to us</u> in exchange for the marketing and advertising services we provide on your behalf.

No persons affiliated with us are currently an approved (or the only approved) supplier. As a result, none of our affiliates derived any revenues Our total revenue during the fiscal year ended December 31, 2024 was \$4,766,961. During that year, we generated \$595,287 in revenue as a result of 2023, based on franchisee purchases and leases from suppliers. There are no approved or designated suppliers in which any of our officers owns an interest.

Our total revenues during the fiscal year ended December 31, 2023, were \$3,721,509.06. During that year, we received a total of \$519,073.20 in revenue based on franchisee purchases and leases from suppliers (consisting of \$244,415.00278,000 in business set-up fees and \$274,658.20317,287 in technology fees paid to us), which represents 13.912.49% of our total revenues for thetat year.

ITEM 9 FRANCHISEE'S OBLIGATIONS

This table lists your principal obligations under the franchise and other agreements. It will help you find more detailed information about your obligations in these agreements and other items in this Disclosure Document.

OBLIGATION	SECTION IN FRANCHISE AGREEMENT	DISCLOSURE DOCUMENT ITEM
aSite selection and acquisition/lease	Section 7.1	Item 7 & Item 11
bPre-opening purchases/leases	Section 6.2, 7.2, 11.13 & 16.1	0,<u>Item 5</u> , Item 7, Item 8 & Item 11
cSite development and other pre- opening requirements	Section 7.2 & 7.3	Item 6, Item 7 & Item 11
dInitial and ongoing training	Section 5	Item 6 & Item 11

OBLIGATION	SECTION IN FRANCHISE AGREEMENT	DISCLOSURE DOCUMENT ITEM
eOpening	Section 7.3	Item 11
fFees	Section 4.2, 5, 6.2, 6.3, 6.4, 6.5, 6.6, 6.10, 6.11, 8.4, 10, 11.15, 11.19, 14, 16.1, 17.2, 20.2	Oltem 5 & Item 6
gCompliance with standards and policies/Operating Manuals	Section 6.1, 7.2, 10.3, 11 & 18.1	Item 11
hTrademarks and proprietary information	Section 18	Item 13 & Item 14
iRestrictions on products/services offered	Section 11.3	Item 16
jWarranty and client service requirements	Section 11.18	Not Applicable
kTerritorial development and sales quotas	Section 12	Item 12
lOngoing product/service purchases	Section 11.13	Item 8
mMaintenance, appearance and remodeling requirements	Section 7.2 & 11.14	Item 11
nInsurance	Section 16.1	Item 6 & Item 7 & Item 8
oAdvertising	Section 10	Item 6, Item 7 & Item 11
pIndemnification	Section 19	Item 6
qOwner's participation/-management/ staffing	Section 8	Item 11 & Item 15
rRecords/reports	Section 16.2 & 16.3	Item 6
sInspections/audits	Section 17	Item 6 & Item 11
tTransfer	Section 20	Item 17
uRenewal	Section 4	Item 17
v. Post termination obligations	Section 22	Item 17
w. Non-competition covenants	Section 15	Item 17
x. Dispute resolution	Section 23	Item 17
y. Franchise Owner Agreement (brand protection covenants, transfer restrictions and financial assurance for owners and spouses)	ATTACHMENT "C"	Item 15

ITEM 10 FINANCING

We do not offer direct or indirect financing. We do not guarantee any of your notes, leases or obligations.

ITEM 11 FRANCHISOR'S ASSISTANCE, ADVERTISING, COMPUTER SYSTEMS AND TRAINING

Except as listed below, we are not required to provide you with any assistance.

Before you open your Business, we will:

- 1. License you the Marks necessary to begin operating your Business. (Section 2)
- 2. Provide telephonic guidance regarding site selection if you choose to establish a separate office and request our help identifying a suitable location. See Section below entitled "Site Development" for additional

- information. (Sections 7.1)
- 3. Provide access to our Manual, which will help you establish and operate your Business. See Section below entitled "Manual" for additional information. (Section 6.1)
- 3. The Manual contains approximately 274 pages. Because our Manual is in electronic format, we have attempted to approximate the number of "pages" in the Manual based upon the "quantity" of content in the materials. The Table of Contents is attached to this Disclosure Document as EXHIBIT "D". (Sections 6.1, 11.2 & 25.8)
- 4. Provide the written specifications for the goods and services you must purchase to establish your Business, as well as a written list of approved and/or designated suppliers for purposes of acquiring these goods and services. We do not deliver or install any items that you purchase. (Section 11.2)
- 5. Provide an initial supply of marketing materials (estimated to be a 150 to 180 day supply). See Section below entitled "Local Advertising" for additional information. (Section 10.2)
- 6. Provide a sample contract for purposes of engaging clients for placement services (a "<u>Client Agreement</u>"). Our current sample Client Agreement is attached to this Disclosure Document as <u>EXHIBIT "G"-6</u>. You may customize our sample Client Agreement for your use, or you may prepare your own form of agreement in compliance with the laws of your state. (Section 11.4)
- 7. Provide the form Marketing Agreement you will use to contract with Senior Communities. This form is attached to the Franchise Agreement as <u>ATTACHMENT "B"</u>. (Section 11.5)
- 8. Provide a sample referral services agreement (a "<u>Referral Services Agreement</u>") that establishes the framework for referrals between ALL Businesses and Senior Service Providers and, when permitted, provides for the payment of referral fees as an additional income stream. We recommend you sign a Referral Services Agreement with Senior Service Providers to whom you refer clients. You may customize our sample agreement for your use. (Section 11.6)
- 9. Conduct certain marketing and promotional activities to promote the opening of your Business. See Section below entitled "Local Advertising" for additional information. (Section 10.2(a))
- 10. Assign to you any Marketing Agreements that we have previously entered into with Senior Communities that are located within your Territory (other than Marketing Agreements associated with Regional Accounts). (Section 11.5)
- 11. Provide an initial training program. See Section below entitled "Training Program" for additional information. (Section 5)

During the operation of your Business, we will:

- 1. Send one or more of our representatives to your Territory to provide 2 days of onsite training and assistance within the 12-month period after you open your Business (we only provide this onsite training for the 1st franchise territory that you purchase). In our discretion, we may: (a) substitute any method of training we deem appropriate in place of the 2 days of "onsite training" (for example, we may substitute a virtual training program); and/or (b) postpone the 2 days of onsite training (or other method of training we deemed appropriate) to a later date we deem appropriate, including a date that is more than 12 months after you open your Business. See Section below entitled "Training Program" for additional information. (Section 6.5)
- 2. Provide access to ALL-IN, which is our proprietary business management database. We may discontinue ALL-IN at any time in our sole discretion. See Section below entitled "Computer System" for additional information. (Section 6.2)
- 3. Provide our guidance and recommendations on ways to improve the marketing and operation of your Business based upon our evaluation of your Business and reports you provide to us. (Section 6.6)
- 4. Provide periodic access to 1 or more registered nurses for consultations regarding case assessments and recommendations during the hours that we designate. We may discontinue providing access to registered

- nurses at any time in our sole discretion. (Section 6.6)
- 5. Provide periodic access to 1 or more of our representatives to answer questions you have regarding your Business during the hours that we designate. (Section 6.6)
- 6. Provide our recommendations regarding the fees you charge Senior Communities. (Section 11.5)
- 7. Provide periodic training programs. See Section below entitled "Training Program" for additional information. (Section 5)
- 8. Maintain a corporate website to promote our brand (www.assistedlivinglocators.com). We will also provide you with a local landing page to promote your Business, which will be linked to our corporate website. See Section below entitled "Local Advertising" for additional information. (Section 6.3)
- 9. Conduct certain ongoing marketing and promotional activities to promote your Business. We may provide these services directly or outsource them to 1 or more third-party suppliers of our choosing. (Section 10.2(b))
- 10.-Administer the brand and system development fund. See Section below entitled "Brand and System Development Fund" for additional information. (Section 10.1)

During the operation of your Business, we may, but need not:

- 1. Administer our Corporate Leads Program and refer Corporate Leads to you. See Section below entitled "Corporate Leads Program" for additional information (Section 6.6)
- 2. At no additional extra charge, refer you leads we generate through: (a) our website (www.assistedlivinglocators.com), including web leads (unless funded and controlled via a stand-alone landing page designated to support corporate funded web leads) and leads that call our corporate telephone number; (b) our call center; (c) any digital program or platform paid for by the brand and system development fund; or (d) any other digital program or platform that we designate as not being part of our Corporate Leads Program. We have sole discretion in determining the franchisee to whom we refer these leads and we are not required to refer any leads to you. (Section 6.7)
- 3. Arrange for a third party to administer a call center answering service to answer client calls. You must pay the then-current fee charged by the call center, which may include an initial set up fee and ongoing monthly fees. Our preferred vendor currently charges a \$99 one-time setup fee plus a monthly fee ranging from \$50 to \$350 based on usage. (Section 6.11)
- 4. Negotiate purchase agreements with suppliers to allow you to purchase certain goods or services at discounted prices. We may also purchase items in bulk at discounted prices and resell them to you at our cost plus shipping and a reasonable markup. (Section 6.10)
- 5. Hold periodic conferences to discuss business and operational issues affecting ALL Businesses, including industry changes, new services or programs, marketing strategies and the like. (Section 5.7)
- 6. Upon your request, provide additional training or assistance (either at our headquarters or in your Territory). See Section below entitled "Training Program" for additional information. (Section 5)
- 7. Prepare newsletters for Assisted Living Locators clients. (Section 6.12)
- 8. Create a franchise advisory council. See Section below entitled "Advisory Council" for additional information. (Section 13)

Training Program (Section 5 & (i))

Overview |

Initial Training

We will provide an initial training program for your Managing Owner and initial high-level managers. Your other owners and employees may not attend initial training without our prior permission. The initial training

program must be completed to our satisfaction. If the Managing Owner fails the first attempt, he or she may retake the training program but you must pay us an additional \$1,500 training fee. Regardless of whether the Managing Owner passes or fails the first attempt, he or she must successfully complete initial training within 60 days after you sign the Franchise Agreement. We have the option to terminate your Franchise Agreement if your Managing Owner either (a) fails the second attempt or (b) fails to successfully complete training within 60 days after you sign the Franchise Agreement.

The initial training program includes approximately 5 days of training at our corporate headquarters located in Scottsdale, Arizona (or at any other location we designate). We reserve the right to substitute online/virtual training or include a mix of online/virtual training and live training at our corporate headquarters. The training materials will consist of the Manual and certain workbooks. You will not be charged an additional fee for any of the training materials. Currently, we intend to offer the initial training program at least monthly assuming sufficient demand.

- Training Topics

The Our initial training program consists of currently covers the following topics:

TRAINING PROGRAM

SUBJECT	HOURS OF CLASSROOM TRAINING	HOURS ON THE JOB TRAINING	LOCATION
Administration	9	0	Corporate Office (Scottsdale, Arizona)
Production	6.5	0	Corporate Office (Scottsdale, Arizona)
Sales and Marketing	9	5.75	Corporate Office (Scottsdale, Arizona) and in the field
TOTAL Total	<u>24.5</u>	<u>5.75</u>	

Training Materials

The training materials will consist of the Manual and certain workbooks. You will not be charged an additional fee for any of the training materials.

Instructors

Angela Olea is in charge of our training program. Angela Olea is our founder serves as our President and Chief Executive Officer. She has been with us since our formation in 2006. From 2003 to January 2020, she served as the President/CEO of our affiliate, Certified Assisted Living Locators, LLC, which operated our affiliate owned ALL Business in Arizona until the business (but not the entity) was sold to a franchisee at the end of 2019.

Our other instructors generally have between 5 and 10 years of experience in the assisted living placement industry. In addition to our corporate instructors, we periodically recruit established franchisees that have demonstrated mastery of our System to provide certain initial and/or ongoing training to other franchisees

CSA Certification Course

Within 6 months after opening, you must complete and achieve a Certified Senior Advisor certification through the Society of Certified Senior Advisors. The Certified Senior Advisor ("CSA") education course and certification focuses on helping professionals understand the complexity of what it means to be a senior. The intent of the CSA certification is to give credibility to your Business and provide you with the confidence you need to compete in this industry. The CSA credential certification is accredited by the National Commission for Certifying Agencies ("NCCA").

The course can be taken in a classroom or on-line setting. The classroom training is 3 full days with the exam on the morning of the 4^{th} day. The on-line course is a self-paced web-course that must be completed within 6 months of starting the course. The exams for both the classroom course and the on-line course are closed-book,

proctored exams taken at a test center. The fee for the course varies depending on whether you take the classroom or on-line course. You will receive a fee discount for certain course elements due to your status as an Assisted Living Locators franchisee.

CSA CERTIFICATION COURSE			
SUBJECT	HOURS OF CLASSROOM OR ONLINE TRAINING ⁽¹⁾	HOURS OF ON-THE- JOB TRAINING	Location =
Aging	3.3	0	Classroom or On Line Online
Aging with Family and Community	3.6	0	Classroom or On Line Online
Health and Mental Health	3.3	0	Classroom or On Line Online
Financial Literacy	4.7	0	Classroom or On Line Online
Medicare, Medicaid and Social Security	2.3	0	Classroom or On Line Online
The Community of Certified Senior Advisors	0.8	0	Classroom or On Line Online
Practice Exam	0.8	0	Classroom or On Line Online
Learning Exercises/Case Studies	2.1	0	Classroom or On Line Online
Self-Study expectation prior to the course	30	0	Self-Study on your own
TOTAL	51	0	

Notes:

- 1. The class offered by Society of Certified Senior Advisors is for 9 hours per day. We have subtracted 1 hour per day for lunch and 1 hour for breaks throughout the day and arrived at 21 hours of classroom training. Because the class is offered by a third-party, we cannot guarantee you will receive the exact number of training hours listed, but we believe this an accurate estimate.
- 2. You may complete this training online. If you choose to do so, you may instead attend a live class in person (class locations and scheduling vary).

The instructors provided by the Society of Certified Senior Advisors for the course may change, but a current list is available at http://www.csa.us/educationandtraining/instructors/. You are responsible for all costs and expenses relating to your CSA certification, including the cost of the course and travel-Expenses you incur attendingto attend the course or a proctored exam, including any costs you incur to retake the exam if you do not pass on your first attempt. You and any of your personnel we designate must maintain a CSA certification throughout the term of your Franchise Agreement and you are responsible for any costs associated with ongoing recertification training.

Dementia Care Certification Course

Within 3 months after opening, you must complete and achieve a Dementia Care Education ("DCE") certification through the Dementia Care Education. The DCE course and certification focuses on helping professionals understand the complexity of dementia. The intent of the DCE certification is to give credibility to your Business and provide you with the confidence you need to compete in this industry.

The course can be taken in a classroom or on-line setting. The classroom training is 1.5 days with the exam on the afternoon of the 2nd day. Currently the initial fee for 1 attendee is included in the business set-up fee you pay to us. The fee for additional participants is \$125 per person. Below is a description of the course:

DCE CERTIFICATION COURSE					
SUBJECT	HOURS OF CLASSROOM OR ONLINE TRAINING	HOURS OF ON-THE- JOB TRAINING	LOCATION		
Dementia Care Education Basic Curriculum and & Assessment Tool	12	0	Classroom or On Line Online		
TOTAL	12	0			

You and any of your personnel we designate must maintain a DCE certification throughout the term of your Franchise Agreement and you are responsible for any costs associated with ongoing recertification training (currently \$99 plus \$25 administrative fee).

Instructors

Training Materials

The training materials will consist of the Manual and certain workbooks. You will not be charged an additional fee for any of the training materials.

Instructors

Angela Olea has overall accountability for our training program. Angela Olea is our founder and currently serves as our President and Chief Executive Officer. She has been with us since our formation in 2006. From 2003 to January 2020, she served as the President/CEO of our affiliate, Certified Assisted Living Locators, LLC, which operated our affiliate owned ALL Business in Arizona until the business (but not the entity) was sold to a franchisee at the end of 2019. She has a total of 21 years of experience in the assisted living placement industry.

Training is provided by our trainers Jennifer Tonna and Bethany Harley. and Angelica Olea. Bethany Harley, our Director of Training, has been with us since December 2023 and has 15 years' experience in the assisted living 16 years of experience in the assisted living healthcare industry. Angelica Olea has been part of our system since 2013. During this time, she has served in various roles including Director of Operations, VP of Operations, Franchise Support Manager and Administrative Assistant. She has 12 years of experience in the assisted living healthcare industry. Jennifer Tonna has been with us since

2022 and has over 21 years of experience in the assisted living placement industry. The training team also includes departmental experts, franchise business consultants and franchise support members, all of whom will have a minimum of 3 years of experience in the relevant field. In addition, certain current or former franchisees who have mastered all or part of the Assisted Living Locators System may provide portions of initial and/or ongoing training to new franchisees.

Our instructors do not conduct Stage 4 training, which is conducted by unaffiliated industry organizations.

Ongoing Training

Within the first 12 months after you open your Business, we will send a representative to your Territory to provide 2 days of onsite training and assistance. As part of this informal training program, our representative will monitor and evaluate your operations, provide you with suggestions for improvement and answer questions from you and your staff. We only provide this onsite training and assistance for the 1st franchised territory you purchase. -We reserve the right in our sole discretion to designate a substitute training or coaching program if we believe it would be more appropriate for your success and/or to postpone this training to take place at a date that is more than 12 months after you open your Business.

From time to time, we may require that your Managing Owner and high-level managers attend system-wide refresher or additional training courses. Attendance is required, but we will not require attendance at more than 1 system-wide refresher or additional training course during any 12-month period.

If you appoint a new Managing Owner or high-level manager, that person must attend and successfully complete our initial training program before assuming responsibility for the management of your Business.

You may request that we provide additional training or support beyond what we are required to provide under the Franchise Agreement. For example, you may request that we assist you in negotiating a Marketing Agreement with a Senior Community.

You may also request that we provide onsite training or assistance in your Territory. We are not required to provide this additional training.

We also operate a series of structured coaching/training programs designed to help new and experienced franchisees improve their overall performance. You may be asked to participate in one or more of these programs based on your performance. In some instances, we may charge you a fee for participation (the fee varies depending on whether the program is conducted onsite or at our headquarters as discussed in the Section below.

Training Fees and Costs

You must pay us a \$10,000 business set-up fee which covers, among other things:

- thea pre-opening initial training program we conduct for your Managing Owner and high-level managers
- a 2-day onsite training program we conduct in your Territory within 12 months after you open (or any substitute training or coaching program that we designate if we feel it would be more appropriate for conductive to your success)

You must pay us an additional initial training fee of \$1,500 per person for: (a) any new Managing Owner or manager that attends our initial training program after you open; and (b) any person that retakes initial training after failing the first a prior attempt.

We do not currently charge any fee to attend system-wide refresher or additional training courses but reserve the right to do so in the future.

We may charge you an onsite training fee, <u>currently of \$1,250 per day (with a \$2,500, minimum fee for up to 2 days)</u> if we provide onsite training or assistance in your Territory (excluding the 2-day training covered by the business set-up fee). This fee covers a 2-day visit. If we agree to visit your Territory for more than 2 days as part of a single visit, we may charge you an additional fee, currently \$1,250, for each additional day we visit. We reserve the right to increase or otherwise modify these fees at any time by not more than 10% per year.

If you request that we provide any additional training or support beyond what we are required to provide under the Franchise Agreement (either remotely or from our headquarters), we may charge you our then current fee, (currently \$125 per hour. Any __ subject to increase of 10% per year). However, if you request additional onsite training or support that you request that is provided in your Territory (i.e., "onsite") will be subject to the then current onsite training \$1,250 daily fee (currently \$2,500) applies instead of the hourly fee.

You must pay the fee charged by the organization providing the CSA certification course at the time you sign up (currently ranges from \$990 to \$1,090 depending on whether you choose the online or classroom course).

You are responsible for all <u>expenses wages</u> and <u>costs-Travel Expenses incurred by you and your trainees incurred</u> for training, <u>including wages</u>, <u>travel and living expenses</u>. If you transfer your franchise, we may charge the new franchisee our then-current initial training fee.

Manual (Section 6.1, 11.2 & 25.8)

We will provide you with access to our Manual in text or electronic form for the term of your Franchise Agreement. The Manual may include, among other things:

- a description of the authorized goods and services you may offer and sell
- specifications, operating procedures that we prescribe from time to time for ALL Businesses
- reporting and insurance requirements

- policies and procedures pertaining to our Corporate Leads Program, Referral Program and any other programs we establish in the future
- policies and procedures pertaining to permissible extra-territorial marketing and business activities
- policies and procedures for collaboration and cooperation between ALL Businesses
- policies and procedures relating to data ownership, protection, sharing and use
- a list of (a) the goods and services (or specifications for goods and services) you must purchase to develop and operate your Business and (b) any designated or approved suppliers for these goods and services

The Manual is designed to establish and protect our brand standards and the uniformity and quality of the goods and services offered by our franchisees. All mandatory provisions contained in the Manual are binding on you. We can modify the Manual at any time, but the modifications will not alter your status or fundamental rights under the Franchise Agreement. Any modification to the Manual is effective at the time we notify you of the change. However, we may provide you with a reasonable period of time to implement certain changes resulting from modifications to the Manual (for example, implementing new software of technology). The Manual contains a total of approximately 274 pages. Because our Manual is in electronic format, we have attempted to approximate the number of "pages" in the Manual based upon the "quantity" of content in the materials. A copy of the Table of Contents to the Manual is attached to this Disclosure Document as EXHIBIT "D".

Site Development (Section 7.1 & 7.2)

Assisted Living Locators is a home-based business. We do not impose any standards or have any requirements for your home office.

We do not anticipate that many franchisees will purchase or lease separate commercial office space for purposes of operating the business. However, a few of our franchisees have elected to do so. If you choose to utilize separate commercial office space, we do not impose any standards or specifications except that it must present a professional appearance and must be located in your Territory. You do not need our approval of your office location or your lease. There are no restrictions on your ability to relocate your office within your Territory. We do not select the site for your office and we do not purchase the premises and lease it to you.

Most of your business activities will be conducted from your vehicle. Because safety is of the utmost importance to our brand, we may require that you attend a driver's safety course of our choosing at your cost if we receive complaints regarding unsafe driving by you or your employees. At all times, the Managing Owner (and each other person driving clients) must hold a valid driver's license.

Computer System (Section 6.4, 11.13, 11.14, 11.15, 16.3 & 17.1)

You must purchase and use all Technology Systems (as defined in Note 9 in Item 6)—we designate from time to time. Our current Technology Systems consists of the following items:

- Scanner/Fax/Copier
- High Speed Internet Connection
- Smart Phone (with capability to identify business calls)
- Desktop, Laptop or Tablet
- Internet Fax Program (Efax, MyFax or equivalent)
- Quickbooks (online version)
- ALL-IN business management database
- Google Workspace
- Phone system
- Current Version of Microsoft Office (or equivalent)

You may already have these items, other than ALL-IN. If you do not have any of these itemsthem, we estimate

your cost to purchase the Technology Systems will range from \$1,000 to \$3,000 (excluding the \$495 one-time initial setup fee for ALL-IN). We may change the components of the Technology Systems from time to time, including your computer system.

You will use your computer system to access and utilize ALL-IN and communicate with us, your clients and your employees electronically. You must license and utilize ALL-IN, which is our proprietary business management database. You will use this program to:

- manage and store data regarding your about clients, Senior Communities, Referral Sources and Senior Care Providers
- invoice Senior Communities for placements
- track and report client leads, placements and Gross Collected Revenues Revenue

ALL-IN will serve as your business database. You must follow all policies and procedures in the Manual with respect to the use of ALL-IN and your entry of data. You must enter all new clients into the database by the next business day regardless of how you obtained the referral and regardless of whether the outcome is predetermined.

ALL-IN will collect data regarding your clients (prospective leads and existing clients), Senior Communities, Referral Sources, Senior Care Providers, placements, fees and other operational data. We will have independent unlimited access to all data entered into ALL-IN and there are no contractual limits imposed on our access.

You must pay us a \$495 one-time initial setup fee for ALL-IN and an ongoing monthly fee of \$199250 per month (\$2,3883,000 per year), which includes a license for up to 4 users. If you have more than 4 users, you must pay us an additional \$199250 per month that covers a maximum of 4 additional users. We may increase the ALL-IN monthly licensing fee on an annual basis; (up to a maximum of \$400 per month per block of 4 users). The monthly licensing fee is included as part of your technology fee. In exchange for the monthly fee, we will include all required maintenance, support and updates for ALL-IN. Your Managing Owner, and all of your other managers and employees that we designate, must sign the ALL-IN User Agreement and Privacy Policy that is attached to this Disclosure Document as EXHIBIT "G"-3.

You must license Google Workspace accounts from us. The cost of each Google Workspace account is currently \$2527 per account per month (\$300324 per year). We expect most franchisees will need 2 accounts. Google Workspace is a collection of cloud computing, productivity and collaboration tools, software and products developed by Google, including your required email addresses/accounts. You must exclusively use the email addresses provided through the Google Workspace platform for all communications with us, clients, Senior Communities, Referral Sources, Senior Care Providers, suppliers and other persons relating to your Business. You may not use any email address provided through the Google Workspace platform for any purpose unrelated to your Business. We will own the email addresses and the account (with the right to monitor the communications) but will allow you to use them during the term of your franchise. We reserve the right to substitute different third party providers and you will be required to implement such changes upon written notice and will be solely responsible for all fees associated with such change.

Except as otherwise disclosed above, neither we nor any other party has any obligation to provide ongoing maintenance, repairs, upgrades or updates to your Technology Systems. We are not aware of any optional or required maintenance, updating, upgrading or support contracts relating to your Technology Systems. However, we strongly recommend that you purchase a maintenance and support contract for your computer that would cover any necessary repairs or other technical issues that could disrupt your Business operations. Since you may purchase any type of computer that you desire, we do not know the cost of any associated maintenance and support contract that might be available for you to purchase. However, in most cases we would expect the annual cost to range from \$150 to \$250 for a typical computer.

You must maintain your Technology Systems in good working order at your cost. During the term of your franchise, you may be required to upgrade or update your Technology Systems to conform to our then-current specifications. There are no contractual limitations on the frequency or cost of these updates or upgrades. We will have independent access to the data collected on your computer system as well as all information in the facility database that is licensed we license to you. You must grant permission for us to link your QuickBooks

Online account to our master QuickBooks Online account. We will have full and independent access to all information you enter into your QuickBooks Online account and we may retrieve, download, analyze and store this information and data at any time and for any purpose.

We may change the components of the Technology Systems from time to time, including your computer system.

As <u>further detailed_discussed</u> in Item 6, you <u>must</u>-pay us a technology fee for certain software, technology and related services that we provide. As of the issuance date of this Disclosure Document, <u>Ourour</u> current technology fee includes the following:

TECHNOLOGY FEES*					
Licensed Product or Software Description of Licensed Product or Software		Initial Fee	Ongoing Fee		
ALL IN	Our proprietary business management software you will use to operate your Business	\$495 initial setup fee (due when you sign Franchise Agreement)	\$199250 per month for each block of 4 user licenses (due on 1st day of month)		
Google Workspace (formerly known as "G Suite")	Collection of cloud computing, productivity and collaboration tools, software and products developed by Google, including your required email addresses/accounts	None	\$2527 per month per account (due on 1st day of month)		

^{*-}_All of the fees described in the table above comprise our current "technology fee". We anticipate most franchisees will require 1 block license for ALL IN and 2 Google Workspace accounts. These fees are subject to increase up to a maximum of \$400 per month per block of 4 users.

Corporate Leads Program (Section 6.6)

We have established and currently administer a program that is designed to generate client leads for ALL Businesses (our "Corporate Leads Program"). Under our Corporate Leads Program, we: (a) attempt to establish relationships with Referral Sources; and (b) develop digital marketing programs or platforms that are designed to generate leads.

When you obtain a lead through our Corporate Leads Program, you will enter (or we will enter for you) the lead into ALL-IN. The first correct instance of the lead appearing in the ALL-IN platform will determine ownership of the lead and the terms associated with the placement. If we refer a lead to you before you or another franchisee correctly enters the same lead into ALL-IN, or the lead is sent by a Referral Source with whom we have entered into a referral agreement, then the lead is deemed a "Corporate Lead".

If we refer a Corporate Lead to you, you have the option, but not the obligation, to accept the Corporate Lead for purposes of providing placement services. When you place a Corporate Lead, you must pay us the referral fee described in Item 6 of this Disclosure Document (in addition to royalty fee and other fees imposed upon the revenues revenue you collect). We only charge the referral fee for "Corporate Leads". "Corporate Leads" do not include (and you do not pay us a referral fee relating to) any lead that is generated through:

- our website (www.assistedlivinglocators.com), including web leads (unless funded and controlled via a stand-alone landing page designated to support corporate funded web leads) and leads that call our corporate telephone number
- our call center
- any digital program or platform paid for by the brand and system development fund
- any other digital program or platform we designate as not being part of our Corporate Leads Program

We retain complete discretion in determining whether to refer a Corporate Lead and the franchisee to whom we refer athe Corporate Lead. We are not required to refer any Corporate Leads to you.

Brand and System Development Fund (Section 10.1)

We administer a brand and system development fund to promote public awareness of our brand and to improve

our System. We may use the fund to pay for any of the following in our discretion:

- developing maintaining, distributing or administering, directing, preparing, or reviewing advertising and marketing materials, promotions and programs
- conducting and administering promotions, contests or giveaways
- expanding public awareness of the Marks
- public and consumer relations and publicity
- brand development
- sponsorships
- <u>and</u> charitable and nonprofit donations and events
- research and development of technology, products and services
- website development and search engine optimization
- development, maintenance and promotion of an ecommerce platform
- development and implementation of quality control programs, including the use of mystery shoppers or customer satisfaction surveys
- conducting market research
- changes and improvements to the System
- the fees and expenses of anycharged by advertising agency agencies we engage to assist in producing or conducting advertising or provide marketing efforts services
- collecting and accounting for contributions to the brand fund
- <u>fees and preparing and distributing financial accountings of the brand fund</u>
- any other programs or activities that we deem appropriate to promote or improve the System
- direct and indirect labor, reimbursing us for administrative, overhead and other expenses incurred by us and/or our affiliates in relation to any of these activities (we incur to administer the brand fund, including salary, benefits and other compensation of any of our, and any of our affiliate's, officers, directors, employees or independent contractors based uponpaid to our personnel for time spent working on any brand fund Brand Fund matters described above)

We will direct and have complete control and discretion over all advertising programs paid for by the fund, including the creative concepts, content, materials, endorsements, frequency, placement and media used for the programs. The fund will not be used for payingto pay for advertisements principally directed at selling additional franchises, although consumer advertising may include notations such as "franchises available" and one or more pages on our website may promote the franchise opportunity. We are not required to spend any amount of money from the fund on advertising in your Territory.

On a monthly basis, you must contribute to thepay us a brand fund an amount fee equal to the greater of: (a) 2% of Gross Collected Revenues; Revenue or (b) the minimum brand fund contribution. The minimum brand fund contribution is— (up to \$500 per month, but currently \$300 per month, but we may increase the minimum contribution (up to a maximum of \$500 per month) upon 30 days' prior notice to you.). At this time, we do not apply the minimum brand fund contribution on a "per territory" basis for franchisees that own multiple franchised territories. However, we reserve the right to do so at any time. We will deposit into the fund: (a) all fund contributions paid by you and other franchisees; and (b) all fines we collect from ALL Businesses. Currently, any ALL Business owned by us or our affiliate will contribute to the fund on the same basis as our franchisees. However, we may change this policy at any time, in which case any ALL Business owned by us or our affiliate would not be required to contribute to the fund. Except as stated in this paragraph, we

We have no obligation to expend our own funds or resources for any marketing activities in your area. <u>Under current policy, however, company-owned ALL Businesses contribute to the brand fund on the same basis as franchisees. However, we may change this policy at any time, in which case company-owned ALL Businesses need not contribute to the brand fund.</u>

All monies deposited into the <u>brand</u> fund that are not used in the fiscal year in which they accrue will be utilized in the following fiscal year. Any surplus of monies in the fund may be invested and we may lend money to the <u>fund</u> if there is a deficit. The <u>fund</u> is not audited. An unaudited financial accounting of the <u>operations of the fundBrand Fund contributions and expenditures</u> will be prepared annually and made available to you upon request. During the fiscal year ended December 31, <u>20242022</u>, we spent the marketing funds in the following manner:

Allocation of Marketing Expenditures (20242023)					
Use of Funds Production Media Placement Administrative Expenses Other					
Percentage Allocation	84.5 <u>4%</u>	<u>13.1</u> <u>61%</u>	2.4 23%	0 <u>12%</u>	

We assume no direct or indirect liability or obligation to you with respect to the maintenance, direction or administration of the fund. The fund is not a trust-and we. We have no fiduciary obligations or liability to you with respect to our administration of the fund. We reserve the right tomay discontinue the fund at any time in our sole discretion upon at least-brand fund on 30 days' prior-notice. We may spend in any calendar year more or less than the total contributions to the fund in that year and may cause the fund to invest any surplus for future use by the fund. We may borrow from ourselves, our affiliates or other lenders on behalf of the fund to cover deficits of the fund, and such amounts shall be treated as a loan. We will keep records of any such borrowing described above.

Local Advertising (Section 10.2 & 10.3)

At the time you sign the Franchise Agreement, you must pay us a \$10,000 business set-up fee which covers, among other things:

- <u>(a)</u> setting up a local landing page to promote your Business, which will be linked to our corporate website
- ; and (b) setting up your social media platforms

In addition to the business set-up fee, you must pay us a \$3,000 marketing fee when you sign the Franchise Agreement. We use this fee to pay for your initial supply of marketing materials and to market and promote your Business prior to opening and during the 60-day period after your sign the Franchise Agreement. Some of these marketing activities may include: (a) preparing lists of Senior Communities and Referral Sources for you to contact; (b) issuing press releases; and (c) developing a marketing plan for your Territory. You are not required to spend any additional money on marketing prior to the opening of your Business beyond the business set-up fee and initial \$3,000 marketing fee.

Commencing with the 2nd full month after your opening date, you must begin paying us a monthly marketing fee. The amount of the monthly marketing fee is: (a) \$1,000 per month for month 2 through month 5 after your opening date; and (b) \$500 per month thereafter for the remainderrest of the term of your Franchise Agreement. At our option, we may either directly perform the marketing and advertising services ourselves or we may outsource these functions to 1 or more third-party suppliers of our choosing. Starting with the 6th full month after opening, you must directly spend the marketing fee with third-party suppliers of advertising and marketing materials and services. We may increase the marketing fee upon at least 30 days' prior notice, but any such increase will not exceed 10% in any 12-month period. At any time following written notice to you (including prior to your 6th full month after opening), we may require that you spend the required marketing fee directly with third-party suppliers on marketing and advertising we approve (rather than paying the marketing fee to us).

You must also participate, at your own expense, in all advertising, promotional and marketing programs we require. Except for the business set-up fee and marketing fees described above, we are not required to spend any amount of money on advertising in your Territory.

Once your initial supply of marketing materials has been used, you may purchase additional advertising and marketing materials from us at our then-current pricing. We may use the marketing fees and/or brand and system development fund to pay for the creation and distribution of these materials, in which case there will be no additional charge. We may make these materials available over the Internet (in which case you must arrange

for printing the materials and paying all printing costs). Alternatively, we may enter into relationships with third-party suppliers who will create the advertising or marketing materials for your purchase. We will provide reasonable marketing consulting, guidance and support throughout the franchise term on an as needed basis.

You may create advertising for your own use, provided we approve it in advance. You may not use any advertising materials that we have not approved. You must submit to us any advertising materials you prepare or modify. We have 15 days to review We will provide reasonable marketing consulting, guidance and either approve or reject support throughout the materials. Our failure to approve the materials within the 15 day period will constitute our disapproval of the materials franchise term on an as needed basis.

You are encouraged to market your Business through approved social media channels, subject to the following requirements:

- you may only conduct social media utilizing the social media platforms that we approve
- you must strictly comply with our then-current social media policy
- you must immediately remove any post we disapprove (even if it complies with our social media policy)
- we may require that you utilize a supplier we designate for social media marketing services
- we will own the social media account but allow you to use it during the term of your franchise

We have established a corporate website (www.assistedlivinglocators.com) that will include a zip locator and a link to your landing page. Your landing page will include information relating to your specific business location and select content that we provide on our website. We reserve the right to discontinue our website and/or your local landing page (if applicable) at any time upon 30 days' prior notice to you (we do not need to provide prior notice if you are in default), in which case we have no obligation to compensate you or refund any fees to you. Any electronic or digital marketing you conduct must be conducted through an approved or designated supplier or must be reviewed and approved by us in advance.

Except to the extent provided above, you may not, without our prior approval:

- develop, host, create or otherwise maintain a website or other online or digital presence relating to your Business, including any website bearing any of our Marks
- establish a Google My Business account, or similar permanent web accounts, utilizing any of our Marks (including in conjunction with your Territory name) (if we permit this, we must be named an administrator on the account)
- utilize the Internet to conduct digital or online advertising or engage in ecommerce
- utilize social media to conduct advertising or marketing

You are not required to participate in an advertising cooperative. However, on occasion franchisees do pool funds together on a voluntary basis to purchase regional advertising. We do not have a franchisee advertising counsel to advise us on advertising and similar matters. However, the franchise advisory council (discussed below) may provide non-binding input on marketing matters.

Franchise Advisory Council (Section 13)

We have established a Franchise Advisory Council (the "FAC") to provide us with suggestions to improve the System, including matters such as marketing, operations and new product or service suggestions. We consider all FAC suggestions in good faith but are not bound by them. The FAC has been established and operates according to rules and regulations set forth in the FAC Bylaws, including procedures governing the selection of representatives of the FAC to communicate with us on matters raised by the FAC. Under current FAC Bylaws, the FAC is composed of 6 franchisee representatives (1 from each designated geographic region) and 1 representative from the franchisor to serve as a facilitator. Each representative is voted in by franchisees in the applicable region and serves a 2-year term (except 3 of the 6 original FAC representatives will serve a 1-year term to allow for a staggered transition of new representatives each year). All franchisees in good standing with us are eligible to be selected as a representative. All franchisees in good standing may cast 1 vote on all matters on which franchisees are authorized to vote, regardless of the number of franchises or territories owned by the franchisee. The FAC bylaws may be amended from time to time by the majority vote of the then-current

representatives and corporate facilitator in attendance at the FAC meeting. Except to the extent restricted by the then-current FAC Bylaws: (a) any ALL Business operated by us or our affiliate may serve as a representative on the FAC and vote on all matters for which other franchisees in the region may vote; and (b) we reserve the right to change or dissolve the FAC. However, under current FAC Bylaws only franchisees may serve as representatives and vote and we may not change or dissolve the FAC except through the amendment process established in the Bylaws.

Opening Requirements (Section 7.3)

We anticipate a typical franchisee expect most franchisees will open his or hertheir ALL Business within 2 months after signing the Franchise Agreement. Factors that may affect this time include: identification of a suitable location; securing financing; completion of training; obtaining insurance; obtaining required licenses, permits and bonding; and complying with local laws and regulations. You may not open your Business prior to receipt of Certificate of Completion of Training from us. Your Business is "open" at the time we issue the Certificate of Completion of Training.

You may not open your Business prior to receipt of Certificate of Completion of Training from us: We will not issue a Certificate of Completion of Training until all of the following conditions are met:

- the Managing Owner successfully completes initial training
- you purchase all required insurance and provide us with evidence of coverage
- you obtain all required licenses, permits and other governmental approvals
- you obtain any required bonding
- you obtain access to a suitable vehicle to be used with your Business

Your Business is "open" at the time we issue the Certificate of Completion of Training. Unless we agree to the contrary, your Business must be open within 90 days after signing the Franchise Agreement. We may terminate the Franchise Agreement if you fail to open by this deadline.

ITEM 12 TERRITORY

Location of Your Business

The Franchise Agreement grants you the right to operate a single ALL Business solely within the Territory we designate. You may operate your Business from a home office or you may choose to operate from separate commercial office space. If you operate from a separate commercial office, your office must be located in your Territory. You may relocate your office anywhere within your Territory without out approval, but you must notify us of the relocation in advance. If you operate from a home office, there are no restrictions on the location of your office or your ability to relocate. You do not need our approval of the location of your office.

If you operate from a home office and your home is not located in your Territory, you must provide a physical business address in your Territory. -This address must be a physical location, must not be a PO Box or mailbox in a remote location, and must be precise to ensure customers can locate the business.

Your Territory

We will grant you a Territory that includes one or more adjacent zip codes (as determined by the United States Postal Service). In determining zip codes for a territory, we consider the following criteria:

- the total number of assisted living properties, senior retirement apartments, hospitals, nursing homes and rehabilitation centers in the area (most territories include approximately 50 of these properties, but some include less)
- economic and demographic information, including total population (most territories include a population between 300,000 and 500,000, but some include less)

There is no minimum sized territory. The boundaries of your Territory will be described in <u>ATTACHMENT</u> "A" to the Franchise Agreement at least 7 days before you sign it.

Territorial Protections

You will not receive an exclusive territory. You may face competition from other franchisees, from outlets that we own, or from other channels of distribution or competitive brands that we control.

Although we do not grant exclusive territories, we do grant you the territorial protections discussed below.

A. Marketing Agreements

We will not enter into, or authorize any other ALL Business to enter into, a Marketing Agreement with any Senior Community located in your Territory, except as otherwise permitted below with respect to Regional Accounts and Acquisitions (each defined below). If we entered into a Marketing Agreement with a Senior Community in your Territory before you sign the Franchise Agreement, we will assign the Marketing Agreement to you when you open (unless the Marketing Agreement is associated with a Regional Account).

B. Targeted Marketing

We will not engage in, or authorize any other ALL Business to engage in, "Targeted Marketing" within your Territory, except as otherwise permitted below with respect to Acquisitions. "<u>Targeted Marketing</u>" is any advertising or marketing that is directed into a specific territory for the primary purpose of soliciting clients, Senior Communities and/or Referral Sources within that territory. Examples include:

- direct mail that is intended to be distributed to addresses located within a given territory
- telephone calls made to persons or businesses known to reside or be located within a given territory
- television or radio advertising intended to be published or played within a given territory
- advertising in a newspaper, magazine or other publication distributed to subscribers residing within a given territory
- social media marketing intended to be delivered to computers, smart phones or other digital devices within a given territory
- direct face-to-face marketing conducted from any physical location within a given territory, such as being an exhibitor at an expo that takes place within the territory

Marketing and advertising that is distributed, circulated or received within both the territory of the person engaging in the advertising as well as the territory of another ALL Businesses is not deemed to be "Targeted Marketing" if: (a) the person engaging in the advertising uses reasonable efforts to limit circulation or distribution to areas within that person's territory; and (b) the majority of the recipients of the advertising are located within the territory of the person engaging in the advertising and there is only incidental circulation or distribution outside of that person's territory. Targeted Marketing also does not include any marketing we conduct as part of our administration of the brand and system development fund or other system-wide marketing or advertising efforts. The Manual may include more details describing the circumstances in which advertising or marketing is deemed to be "Targeted Marketing".

Other ALL Businesses are free to contract with clients, Referral Sources and Senior Care Providers that are located or reside within your Territory as long as they did not solicit the clients, Referral Sources or Senior Care Providers utilizing Targeted Marketing within your Territory (unless you previously authorized them to engage in the Targeted Marketing).

C. Placements

We will not place, or authorize any other ALL Business to place, clients with Senior Communities that are located in your Territory except as otherwise permitted below with respect to Regional Accounts, Corporate Leads, Acquisitions and our Referral Program.

Limitations on Your Territorial Protections

We and our affiliates reserve all rights not expressly granted to you under the Franchise Agreement. Your territorial rights and protections described above are subject to the following exceptions and limitations:

A. Regional Accounts

A "<u>Regional Account</u>" is a Senior Community that has facilities located within multiple territories. We have the exclusive right to enter into Marketing Agreements with Regional Accounts, including Regional Accounts with 1 or more Senior Communities in your Territory. You may not enter into a Marketing Agreement with a Regional Account without our prior written consent, which we may withhold in our sole discretion.

If you enter into a Marketing Agreement with a Senior Community that (at the time of contracting) only has facilities within your Territory but subsequently opens additional facilities in 1 or more territories owned by other ALL Businesses, we reserve the right to take any of the following actions in our sole discretion:

- (a) Negotiate and enter into a new Marketing Agreement with the Senior Community and convert them to a Regional Account (our Marketing Agreement would be substituted for your Marketing Agreement). If we convert the Senior Community to a Regional Account, we may negotiate an alternative fee structure, but we will attempt to retain any favorable fee structure you negotiated under your original Marketing Agreement for future placements made in your Territory.
- (b) Limit the scope of your Marketing Agreement to Senior Communities located within your Territory and allow other ALL Businesses to negotiate and enter into separate Marketing Agreements for Senior Communities located in their territories.
- (c) Allow you to retain the Marketing Agreement and manage the relationship as a Regional Account, in which case any placements made at Senior Communities that are part of the Regional Account but located outside of your Territory will be governed by the terms of our Referral Program. At any time, we may revoke our authorization for you to manage the Regional Account and either: (i) transition the Regional Account to us (in accordance with clause (a) above); or (ii) require separate Marketing Agreements with each ALL Business that owns a territory in which 1 or more of the Senior Communities are located (in accordance with clause (b) above).

When you place a client with a Senior Community that is located in your Territory but is part of a Regional Account owned by us, we will either collect the placement fee and remit 100% of the proceeds to you (but only if you cooperate with us and assist the client with the placement) or we may allow you to directly invoice and collect the placement fee (depending on the preference of the Senior Community). However, we reserve the right to treat these placements as part of our Referral Program and split the placement fee with you in accordance with the Referral Program fee sharing provisions (although we have never done so). In addition, if the client is a Corporate Lead referred by us, we may charge you our standard referral fee.

B. Referral Program

Other ALL Businesses may place clients with Senior Communities located in your Territory through our Referral Program. The ALL Business that establishes the initial relationship with the client is referred to as the "Referring Owner" and the ALL Business that owns the territory in which the Senior Community is located is referred to as the "Territory Owner".

In order for the Referring Owner to place a client with a Senior Community located in the territory of the Territory Owner, the Referring Owner must first notify the Territory Owner of the possible placement. This notice must be in the form and include all information prescribed by the Manual. Upon receipt of this notice, the Referring Owner must elect to either:

- (a) provide the placement services utilizing the information provided by the Territory Owner regarding the Senior Community; or
- (b) turn the client over to the Territory Owner and allow the Territory Owner to provide the placement services.

In both situations, only the Territory Owner can collect the placement fee from the Senior Community (unless the Territory Owner transfers billing responsibility to the Referring Owner). Once the placement is complete, the Territory Owner will collect the placement fee and split the fee as follows:

- (a) 70% is retained kept by the owner that provides the placement services
- Referring Owner; and (b) 30% is retained kept by the other owner

<u>Territory Owner.</u> The Referring Owner and Territory Owner may agree upon a different fee split based upon the respective services provided by each party, but any different fee split must be agreed upon in writing before the placement occurs.

ALL Businesses must cooperate with each other relating to placements made through the Referral Program. If the Referring Owner elects to turn the client over to the Territory Owner for placement services, but the Territory Owner is unable or refuses to provide the services, we may provide the placement services. If this occurs, the Territory Owner must collect the placement fee from the Senior Community and immediately remit the entire fee to us. We will then pay 30% of the fee to the Referring Owner and keep the rest. Failure to cooperate with another ALL Business relating to a placement under the Referral Program is a default under the Franchise Agreement.

Details of the Referral Program are contained in the Manual. We can modify the Referral Program at any time. However, we will not modify the fee split unless 51% of all participating ALL Businesses vote in favor of the modification. A "participating" ALL Business is any ALL Business (including any ALL Business owned by us or our affiliates) that actually casts a vote at the time and in the manner we establish for the vote. Each ALL Business is entitled to 1 vote, regardless of how many franchises and/or territories are owned and operated by the owner of that business (either directly or through affiliated entities).

C. Corporate Leads

If we refer <u>you</u> a Corporate Lead to <u>you</u> and you <u>choose not to accept decline</u> the Corporate Lead or <u>you</u> fail to contact or service the Corporate Lead in the manner we require, we may provide, or authorize another ALL Business to provide, placement services for the Corporate Lead, including for placements made at Senior Communities located within your Territory. If this happens, you will not be entitled to any portion of the placement fee. We may require that you collect the placement fee from the Senior Community and immediately remit the entire fee to us.

D. Acquisitions

We and our affiliates reserve the right to acquire, or be acquired by, another business or chain that may sell competitive or identical goods or services (in either case, an "Acquisition"), and those businesses may be converted into ALL Businesses operating under the Marks. Any such acquired or converted ALL Business may be located within your Territory and engage in any and all activities within your Territory that you are authorized to conduct. You will not be entitled to any compensation for placements made, or other goods or services sold, by any such acquired or converted ALL Business within your Territory, except you will receive a split of any placement fee involving a placement made with a Senior Community with whom you have entered into a Marketing Agreement (any such placement would be governed by the terms of our Referral Program). Additionally, we and our affiliates reserve the right to merge with, acquire or be acquired by any business of any kind under other systems and/or other marks, which businesses may offer, sell, operate or distribute and/or license others to offer, sell, operate and distribute goods and services the same as, similar to, or different from those offered by your ALL Business inside or outside your Territory.

E. Additional Franchises

We reserve the right to develop and operate and grant rights, franchises and licenses to others to develop and operate ALL Businesses under the Marks on any terms and conditions that we in our sole discretion deem appropriate provided that we comply with the Territorial Protections set forth above.

Alternative Channels of Distribution

Although we do not currently do so, we and our affiliates reserve the right to sell, or license others to sell, competitive or identical goods or services (under the Marks or different trademarks) through alternative channels of distribution (such as through the Internet, catalog sales, telemarketing or other direct marketing), including within your Territory. You are not entitled to any compensation for sales that take place through

alternative channels of distribution.

Restrictions on Your Sales and Marketing Activities

Unless we provide our written approval, you may not market or sell goods or services through alternative channels of distribution (such as the Internet, catalog sales, telemarketing or other direct marketing) either within or outside your Territory.

You may not place a client with a Senior Community located in a territory owned by another ALL Business except in compliance with our Referral Program.

Your advertising and marketing must be primarily directed within your Territory. You may not engage in any Targeted Marketing within a territory owned by another ALL Business without first obtaining written approval from the territory owner.

You may engage in Targeted Marketing to solicit clients, Senior Communities and Referral Sources within areas outside your Territory (an "Expansion Territory") only if each of the following conditions is met:

- the Expansion Territory is adjacent to your Territory
- no portion of the Expansion Territory is included within a territory owned by another ALL Business
- you are reasonably capable of consistently providing placement services in the Expansion Territory in accordance with our standards, policies and procedures

You have no territorial rights or protections whatsoever with respect to an Expansion Territory. We may revoke your authorization to conduct Targeted Marketing within an Expansion Territory at any time for any reason (or for no reason). If we revoke our authorization, you must: (a) immediately discontinue all Targeted Marketing within the Expansion Territory; and (b) upon request, assign any Marketing Agreements you entered into with Senior Communities located in the Expansion Territory either to us or to another ALL Business we designate.

You may not enter into Marketing Agreements with Senior Communities located outside your Territory without our prior written approval. If we allow you to do so, you must agree to any conditions we impose and our approval will be limited to adjacent territories that are not owned by another ALL Business. We may revoke our authorization at any time in our sole discretion. If you sign a Marketing Agreement with a Senior Community located in an area outside your Territory that subsequently becomes part of a territory owned by another ALL Business: (a) you must assign the Marketing Agreement to that ALL Business (and assist with transitioning the associated account to the new ALL Business); and (b) you may not engage in any further placements involving that Senior Community following the date that the new ALL Business commences operations (which is deemed to be the date of completion of initial training for new franchisees) except under the terms of our Referral Program.

You may contract with a client that resides in a territory owned by another ALL Business as long as you do not acquire the client as a result of Targeted Marketing you conduct in that territory. For example, you may contract with a client that resides in a territory owned by another ALL Business if:

- you solicit the client or the client's family within your Territory
- the client is referred to you by one of your other clients or a third person not associated with you
- we refer the client to you
- another ALL Business refers the client to you
- you obtain the client while located within the other territory as long as the client is obtained through incidental activities and not through Targeted Marketing
- the client is a Corporate Lead <u>andthat</u> we <u>authorizeallow</u> you to service <u>the client</u> because the territory owner declined <u>to service</u> the Corporate Lead or failed to contact or service the Corporate Lead in the required time or manner

Your marketing activities are also subject to the additional restrictions described in Item 11 under the Section entitled "Local Advertising". There are no other restrictions on your right to solicit clients, Senior Communities, Referral Sources or Senior Care Providers, whether from inside or outside of your Territory.

Minimum Performance Requirements

After your Business has been open for 12 months, you must complete a minimum of 18 new client placements during each subsequent 6-month period (the "Minimum Performance Requirements"). Your failure to comply with this requirement constitutes an event of default under the Franchise Agreement, in which case we have the right to terminate the Franchise Agreement or modify or eliminate your territorial protections, including by taking any or all of the following actions:

- reducing the size of your Territory with a corresponding reduction in your Minimum Performance Requirements
- permitting other ALL Businesses to provide placement services involving Senior Communities and clients located within your Territory
- establishing, or authorizing another person to establish, an ALL Business within your Territory

Your territorial protections do not depend on achieving any other sales volume, market penetration, or other contingency.

Additional Franchises and Territories

You are We do not granted any grant options, rights of first refusal or similar rights to acquire additional territories or franchises.

Competitive Businesses Under Different Marks

We and our affiliates reserve the right to establish, own or operate, and grant others the right to establish, own or operate a business offering the same or similar services as those offered through an ALL Business, or different services under different trademarks (other than the Marks) inside and outside the Territory. Currently, neither we nor any affiliate of ours intends to operate or franchise another business under a different trademark that sells products or services similar to the products or services offered at by an ALL Business.

ITEM 13 TRADEMARKS

We grant you the right to operate a franchise under the name "Our affiliate Assisted Living Locators" Licensing, LLC ("ALL Licensing") owns the following Marks that are registered with the United States Patent and Trademark Office (USPTO):

REGISTERED MARKS

logo shown on

<u>Mark</u>	Registration Number	Registration Date (Renewal Date)	Register (Principal or Supplemental)
ASSISTED LIVING LOCATORS	<u>3263817</u>	July 10, 2007 (November 28, 2016)	Supplemental
ASSISTED LIVING LOCATORS	<u>4699430</u>	March 10, 2015 (June 25, 2024)	<u>Principal</u>
MATCHING SENIORS WITH CARING PROVIDERS FOR OVER A DECADE	<u>5077408</u>	November 8, 2016 (November 7, 2023)	<u>Principal</u>
ASSISTED LIVING LOCATORS	6075037	<u>June 9, 2020</u>	<u>Principal</u>
AssistedLiving Locators	<u>3292322</u>	<u>September 11, 2007</u> (November 28, 2016)	<u>Principal</u>
Assisted Living	<u>4930127</u>	April 5, 2016 (September 20, 2022)	<u>Principal</u>

<u>Mark</u>	Registration Number	Registration Date (Renewal Date)	Register (Principal or Supplemental)
Assisted Living Locators	6091748	June 30, 2020	<u>Principal</u>
Assisted Living Locators Your local serilor fiving advisor	<u>6091747</u>	June 30, 2020	<u>Principal</u>
Assisted Living Locators	<u>6091746</u>	June 30, 2020	<u>Principal</u>
Assisted Living Locators Tour local tensor living achiever	6091745	June 30, 2020	<u>Principal</u>

ALL Licensing also applied to register the cover page of following Mark with the USPTO based on actual use:

<u>Unregistered Marks</u>				
<u>Mark</u>	<u>Serial Number</u>	Application Date		
	<u>98965938</u>	<u>January 15, 2025</u>		

We do not have a federal registration for the Mark in the table immediately above. Therefore, this Disclosure Document. By Mark does not have many legal benefits and rights as a federally registered trademark, we mean trade names, trademarks, service marks, and logotypes used to identify. If our right to use this Mark is challenged, you may have to change to an alternative trademark, which may increase your expenses.

All required affidavits have been filed and all required renewals for the trademark registrations listed above have been obtained.

Business or the products or services you provide. We may change the trademarks you may use from time to time including by discontinuing use of the Marks listed in this Item 13). If this happens, we will reimburse you for your tangible costs of compliance (e.g., changing brochures, business cards, etc.). We have no other liability to you for changing the Marks.

The following trademarks are registered on the United States Patent and Trademark Office:

You must notify us immediately if you discover an infringing use (or challenge to your use) of the Marks. We will take the action we deem appropriate. We are not required to take any action if we do not feel it is warranted. You may not control any proceeding or litigation involving our Marks.

Except as disclosed above, the Franchise Agreement does not require that we: (a) protect your right to use the Marks; (b) protect you against claims of infringement or unfair competition arising out of your use of the Marks; or (c) participate in your defense or indemnify you for expenses or damages you incur if you are a party to an administrative or judicial proceeding involving our Marks or if the proceeding is resolved in a manner unfavorable to you.

There are currently no: (a) effective material determinations of the Patent and Trademark Office, the Trademark Trial and Appeal Board, the trademark administrator of this state or any court; (b) pending infringements, oppositions or cancellations; (c) pending material litigation matters involving any of the Marks; or (d) infringing uses we are aware of that could materially affect your use of the Marks.

REGISTERED MARKS				
Mark	Registration Number	Registration Date (Renewal Date)		
AssistedLiving Locators	(Principal Register) 3,292,322	September 11, 2007 (November 28, 2016)		
ASSISTED LIVING LOCATORS (word mark)	(Supplemental Register) 3,263,817	July 10, 2007 (November 28, 2016)		
ASSISTED LIVING LOCATORS (word mark)	(Principal Register) 4,699,430	March 10, 2015		
Assisted iving	(Principal Register) 4,930,127	April 5, 2016		
Matching Seniors with Caring Providers for Over a Decade	(Principal Register) 5,077,408	November 8, 2016		
ASSISTED LIVING LOCATORS	(Principal Register) 6,075,037	June 9, 2020		
Assisted Living Locators	(Principal Register) 6,091,748	June 30, 2020		
Assisted Living Locators Your local serior living advisor	(Principal Register) 6,091,747	June 30, 2020		
Assisted Living Locators	(Principal Register) 6,091,746	June 30, 2020		
Assisted Living Locators Your local senior living advisor	(Principal Register) 6,091,745	June 30, 2020		

All required affidavits have been filed and all required renewals for the trademark registrations listed above have been obtained.

The Marks above are owned by our affiliate, ALL Licensing, LLC ("ALL Licensing"). Pursuant to a License Agreement dated May 4, 2020 (the "License Agreement"), we were granted the right to use the Marks in the Assisted Living Locators System and to sublicense the Marks to our franchisees. The term of the License

Agreement automatically renews annually unless it is terminated in accordance with its terms. ALL Licensing is permitted to terminate the License Agreement only if we declare bankruptcy or become insolvent, if we and ALL Licensing mutually agree to terminate the License Agreement or if we breach ALL Licensing's quality control standards and fail to cure the breach within a 60-day cure period. If the License Agreement is terminated, the agreement states that all sublicenses granted by us to our franchisees will continue in full force and effect until the expiration or termination of the applicable franchise agreement. Except as discussed above, no No other agreements limit our right to use or sublicense the use of the Marks.

You must follow our rules when using the Marks. You cannot use our name or mark as part of a corporate name or with modifying words, designs or symbols unless you receive our prior written consent. You may not use our Marks relating to the sale of any product or service that we have not authorized.

You must notify us immediately when you learn about an infringing or challenging use of the Marks. We will take the action we think appropriate, but we are not required to take any action if we do not feel it is warranted. We may require your assistance, but you may not control any proceeding or litigation relating to our Marks. You must not directly or indirectly contest our or ALL Licensing's right to the Marks.

Except as disclosed above, we are not required under the Franchise Agreement to: (a) protect your right to use the Marks or protect you against claims of infringement or unfair competition arising out of your use of the Marks; or (b) participate in your defense or indemnify you for expenses or damages you incur if you are a party to an administrative or judicial proceeding involving our marks or if the proceeding is resolved in a manner that is unfavorable to you.

There are no: (a) currently effective material determinations of the Patent and Trademark Office, the Trademark Trial and Appeal Board, the trademark administrator of this state or any court; (b) pending infringements, oppositions or cancellations; or (c) pending material litigation involving any of the Marks. We do not know of any infringing uses that could materially affect your use of the Marks.

ITEM 14 PATENTS, COPYRIGHTS, AND PROPRIETARY INFORMATION

No patents or pending patent applications are material to the franchise.

Although we have not filed an application for copyright registration for theour Manual, our website, our proprietary business management database (ALL-IN), or our marketing materials, we do claim a copyright to these items.

During the term of your Franchise Agreement, we will allow you to use our proprietary information relating to the development, marketing and operation your Business, including, methods, techniques, assessments, specifications, procedures, policies, marketing strategies and information comprising the System and the Manual. We own all of your We own all ideas, improvements, inventions, marketing materials and other concepts you develop relating to an ALL Business. We also own all business records regarding your Business operations, including data regarding your clients, Senior Communities, Referral Sources and Senior Care Providers, but will allow. These business records and data are confidential and proprietary. We license you the right to use these business records and the associated data during the term of theyour Franchise Agreement. All ideas, improvements, inventions, marketing materials, and other concepts you develop relating to the operation of your Business will be owned by us You must comply with all applicable data protection laws and our data processing and data privacy policies in the Manual.

You must maintain the confidentiality of We provide access to our proprietary confidential information through the Manual, training programs and other periodic support and guidance. You may use it only this information solely for purposes of developing, marketing and operating your Business in strict accordance compliance with the terms of the Franchise Agreement and the Manual. You Manual. All information in the Manual is confidential. You may not disclose our confidential information to anyone other than your employees, on a need-to-know basis, without our prior permission. All your employees and representatives must sign the Confidentiality Agreement attached to the Franchise Agreement as ATTACHMENT "E" before you give them access to our confidential information.

You must promptly tellnotify us when if you learn about discover an unauthorized use of our proprietary

information or copyrighted materials. We are not obligated required to act, but will respond to this information as we deem appropriate. You may not control any proceeding or litigation allegation to unauthorized use of any of our proprietary information or copyrighted materials. We have no obligation to indemnify you for any expenses or damages arising from you incur as a result of any such proceeding or litigation involving our proprietary information. There are no infringements that are known byto us at this time.

ITEM 15 OBLIGATION TO PARTICIPATE IN THE ACTUAL OPERATION OF THE FRANCHISE BUSINESS

You must designate an owner who will be primarily responsible for the daily management and supervision of the Business (the "Managing Owner"). The Managing Owner must: (a) be approved by us; (b) successfully complete all training programs we require; (c) dedicate full-time efforts to the Business; and (d) at all times hold at least a 20% ownership interest in the business. We do not require that any owners (other than the Managing Owner) personally participate in the management or operation of your Business. If you are an entity, each owner (i.e., each person holding a direct or indirect ownership interest in the entity) and the spouse of each owner must sign the Franchise Owner Agreement attached to the Franchise Agreement as ATTACHMENT "C".

- be approved by us
- successfully complete all training programs we require
- dedicate full-time efforts to the Business
- at all times hold at least a 20% ownership interest in the franchise (or the entity that is the franchisee under the Franchise Agreement).

You may hire managers to assist the Managing Owner with the daily management and supervision of your Business. Any person you hire as a manager must:

- •—_successfully complete all training programs we require
- <u>and</u> sign the Brand Protection Confidentiality Agreement attached to the Franchise Agreement as ATTACHMENT "E"

_At all times, the Managing Owner must monitor and supervise the activities of each manager to ensure the Business is operated in accordance with the Franchise Agreement and the Manual. The Managing Owner must remain actively involved with the management and supervision of the Business on a full-time basis even if you hire managers. We do not require that the manageryour managers own anyan equity interest in the franchise.

If you are an entity, each owner (i.e., each person holding an ownership interest in you) and the spouse of each owner must sign the Franchise Owner Agreement attached to the Franchise Agreement as ATTACHMENT "C".

ITEM 16 RESTRICTIONS ON WHAT THE FRANCHISEE MAY SELL

We must approve all goods and services you offer or sell. You must offer all goods and services we require. You may not offer, sell or provide any goods or services we have disapproved. We may change the goods and services you offer, sell and provide at any time and you must comply with the change.

Except as we may otherwise approve, you may only enter into Marketing Agreements with Senior Communities located in your Territory. You may not contract with clients that reside outside of your Territory except as otherwise provided in Item 12 of this Disclosure Document in the Section entitled "Restrictions on Your Sales and Marketing Activities." You may not place clients with Senior Communities located within a territory owned by another ALL Business except in compliance with our Referral Program.

ITEM 17 RENEWAL, TERMINATION, TRANSFER, AND DISPUTE RESOLUTION

This table lists certain important provisions of the franchise and related agreements. You should read these provisions in the agreements attached to this Disclosure Document.

THE FRANCHISE RELATIONSHIP

Provision	SECTIONS SECTION IN AGREEMENT	SUMMARY		
aLength of the franchise term	Section 4.1	Term is equal to 10 years.		
bRenewal or extension of the term	Section 4.1 & 4.2	If you meet our conditions for renewal, you can enter into 4 consecuti successor franchise agreements. Each renewal term will be 5 years, for total maximum term of 30 years. You and we may agree to furth renewals but neither of us can be required to do so (subject to state law)		
cRequirements for you to renew or extend	Section 4.1 & 4.2	You must: not be in default; give us timely notice; sign our then-current form of franchise agreement—and related documents (e.g., Franchise Owner Agreement, Brand Protection Agreement, etc.); sign a general release (subject to state law); and pay renewal fee. (subject to state law). You may not renew if we withdraw from the market in which your Territory is located. If you renew, you may be required to sign a contract with materially different terms and conditions than the original contract.		
dTermination by you	Not applicable	None except as subject to state law. You can terminate under any grounds permitted by law.		
eTermination by us without cause	Section 21.4	We can terminate without cause if you and we mutually agree to terminate provide your written consent.		
f Termination by us with cause	Section 21.1, 21.2 & 21.3	We can terminate if you default.		
g"Cause" defined - curable defaults	Section 21.2 & 21.3	You have the following cure periods: (a) 10 days to cure any monetary default. You havefor financial defaults; and (b) 30 days to cure for any other default (other than defaults default described below under "noncurable defaults").		

	THE FRANCHISE RELATIONSHIP			
Provision	SECTIONS SECTION IN AGREEMENT	SUMMARY		
h"Cause" defined - non-curable defaults	Section 21.1 & 21.2 Section 21.1 & 21.2	The Franchise Agreement will automatically terminate without notice or an opportunity to cure if: you, a Managing Owner or any of your guarantors become insolvent, makes a general assignment for the benefit of creditors, or is adjudicated a bankrupt, unless otherwise restricted by the relevant bankruptcy laws; or a petition in bankruptcy for liquidation, reorganization, or other proceeding is filed by or against you, a Managing Owner or any of your guarantors; or a receiver is appointed or a bill in equity or other proceeding for the appointment of a receiver of you, a Managing Owner or any of your guarantors or other custodian for the Business or assets is allowed; or the real or personal property of you, a Managing Owner or any of your guarantors is attached or levied upon by any sheriff, marshal, or constable, or scized, taken over or foreclosed by a creditor, lienholder or lessor; or a final judgment against you remains unsatisfied for 30 days (unless a supersedes or other appeal bond has been filed). The Franchise Agreement automatically terminates without opportunit to cure for the following defaults: insolvency; bankruptcy; seizure of assets; or similar events or circumstances. We may, in our sole discretion, terminate the Franchise Agreement upo written notice; without opportunity to cure; for any of the following reasons defaults: failure to successfully complete training—or; open it timely manner; insolvency; abandonment of franchise; failure to maintain required license or permit; conviction of certain types of crimes or subject of certain administrative actions; failure to comply with violation of material law; commission of act that may adversely affect reputation of System or Marks; health or safety hazards; multiple incidents of dangerous or unlawful driving; material misrepresentations; 2' underreporting of any amount dueamounts owed to us by at least 2% unauthorized transfers; unauthorized use of our intellectual property violation breach of brand protection covenant; breach of, Franchise Own		
i Your obligations on termination/ non-renewal	Section 22.1	Obligations include: remove wraps or Marks from your vehicle(s); cease use of intellectual property; cease use of ALL-IN and delete copies from your computer; return Manual and branded materials; assign telephorn numbers, email addresses, listings, domain names, associated/linked pages, websites, listings, etc. (you must sign the Power of Attorney Telephone and Internet, attached as EXHIBIT"G"-5 , which authorizes use to transfer these items); assign business records; assign Marketin Agreements, Client Agreements and Referral Services Agreements and transition them to us or another franchisee we designate; cease a communications with clients, Senior Communities and Referral Source cancel fictitious names; and pay amounts due (also see "r", below). We have the right to collect any revenues revenue owed to you under Marketing Agreements, deduct all amounts owed to us and our affiliate and pay the balance to you.		
j Assignment of contract by us	Section 20.1	No restriction on our right to assign.		
k"Transfer" by you – definition	Section 20.2 & Attachment A (definition of "Transfer")	Includes transfer of contract or assets, or ownership change.		

THE FRANCHISE RELATIONSHIP			
Provision	SECTIONS SECTION IN AGREEMENT	SUMMARY	
l Our approval of transfer by you	Section 20.2, 20.3 & Attachment A (definition of "Permitted Transfer")	If certain conditions are met, you You may transfer to a newly formed entity owned by you, orengage in certain instances, to an existing owner, a Permitted Transfer (defined in Note 2 in Item 6) without our approval. We have the right to must approve all other transfers Transfers but will not unreasonably withhold approval.	
m Conditions for our approval of transfer	Section 20.2	Transferee must: meet our qualifications; successfully complete training (or commit to do so); obtain all-required licenses and permits; agrees in writing to assume your obligations under any agreements contracts relating to the Business; and sign then-current form of franchise agreement for remainder of the term (or, at our option, take assignment of existing franchise agreement). assume your Franchise Agreement. You must: be in compliance with Franchise Agreement; pay us-transfer fee; and sign a general release (subject to state law).	
n. Our right of first refusal to acquire your business	Section 20.5	We have the right to match any bona fide, arms-length offer for your business.	
o. Our option to purchase your business	Not Applicable	Not Applicable	
p. Your death or disability	Section 20.4	Within 180 days, franchise must be assigned by estate to an assignee in compliance with conditions for other transfers. We may designate manager to operate the Business prior to transfer Transfer.	
q. Non-competition covenants during the term of the franchise	Section 15.2 & 15.3	No involvement in competing business; comply with non-solicitation and non-disclosure covenants. Non-competition provisions are subject to state law.	
r. Non-competition covenants after the franchise is terminated or expires	Section 15.2, 15.4 & 22.1	No involvement for 2 years in competing business operating within the county or counties in which your Territory is located; comply with non-solicitation and non-disclosure covenants; cease use of intellectual property. Non-competition provisions are subject to state law.	
s Modification of the agreement	Section 25.3 & 25.8	Requires writing signed by both parties (except for unilateral changes to Manual or unilateral reduction of scope of restrictive covenants by us). Other modifications primarily to comply with various states laws.	
t Integration/ merger clause	Section 25.8	Only terms of the Franchise Agreement and its attachments to Franchise Agreement—are binding (subject to state law). Any representations or promises made outside the Disclosure Document and Franchise Agreement may not be enforceable. Nothing in the Franchise Agreement or any related agreements is intended to disclaim any of the representations we made in this Disclosure Document. No statement, questionnaire, or acknowledgment signed or agreed to by a franchisee in connection with the commencement of the franchise relationship shall have the effect of (a) waiving any claims under any applicable state franchise law, including fraud in the inducement, or (b) disclaiming reliance on any statement made by any franchisor, franchise seller, or other person acting on behalf of the franchisor. This provision supersedes any other term of any document executed in connection with the franchise.	

THE FRANCHISE RELATIONSHIP			
Provision	SECTIONS SECTION IN AGREEMENT	SUMMARY	
uDispute resolution by arbitration or mediation	Section 23 -& 11.16	You must first bring any claim or dispute between you and us to our President and provide us with 30 days' notice and opportunity to cure. You must exhaust this internal dispute resolution procedure before you may bring your dispute before a third party. If the matter isn't resolved, at our option, any claim, dispute or disagreement, including any matter pertaining to the interpretation of the Franchise Agreement or issues relating to the offer and sale of the franchise or the relationship between the parties (a "Dispute") to mediation under the auspices of the American Arbitration Association ("AAA"), in accordance with AAA's Commercial Mediation Rules then in effect. The mediation will take place in the county in which we maintain our principal place of business at the time the Dispute arises (currently, Maricopa County, Arizona) and the parties irrevocably waive any objection to such venue. (subject to state law). Subject to state law, all disputes must be mediated before litigation, except for certain disputes involving our intellectual property, compliance with restrictive covenants or suits to collect amounts owed.	
v. Choice of forum	Section 23	Subject to state law, all mediation and litigation must take place in county where we maintain our principal place of business at time dispute arises (currently, Maricopa County, Arizona).	
w Choice of law	Section 25.1	Subject to state law, Arizona law governs.	

ITEM 18 PUBLIC FIGURES

We do not use any public figures to promote our franchise.

ITEM 19 FINANCIAL PERFORMANCE REPRESENTATIONS

The FTC's Franchise Rule permits a franchisor to provide information about the actual or potential financial performance of its franchised and/or franchisor-owned outlets, if there is a reasonable basis for the information, and if the information is included in the Disclosure Document. Financial performance information that differs from that included in the Hem 19 may be given only if: (1) a franchisor provides the actual records of an existing outlet you are considering buying; or (2) a franchisor supplements the information provided in this tem 19, for example, by providing information about possible performance at a particular location or under particular circumstances.

Defined Terms

System Statistics & Subsets Utilized

The For purposes of this FPR, the following terms have the meanings given to them below:

"Gross Collected Revenue" means all gross sums collected by the ALL Business, such as revenue from the sale or provision of goods or services, proceeds from business interruption insurance and referral fees that you collect, including referral fees paid by Referral Sources, Senior Care Providers, other ALL Businesses (such as fee splits under our Referral Program) or us (such as fee splits relating to a Regional Account). Gross Collected Revenue includes any Gross Collected Referral Proceeds (defined in Item 6, Note 2). With respect to fees generated through our Referral Program, Gross Collected Revenue only includes the franchisee's share of the placement fee that is collected. Gross Collected Revenue excludes:

- any referral fee the franchisee pays us for a Corporate Lead
- any placement fee the franchisee collects but pays to another ALL Business under our Referral Program
- any sales or use tax the franchisee collects and pays to the applicable taxing authority
- presents certain financial performance information for the 12 month the wholesale price paid by the franchisee for any merchandise or other retail items that it sells (Gross Collected Revenue includes the difference between the retail price received from the sale of the item and the wholesale price that the franchisee paid for the item)

The total Gross Collected Revenue may be materially less than the total Gross Invoiced Revenue.

"Gross Invoiced Revenue" means the total amount of revenue invoiced, including (a) amounts that were not collected during the calendar year, or a reduction in fee due to a short-term stay or move-out, (b) amounts that could not be collected due to a client's death or for any other reason. The term Gross Invoiced Revenue does not include any sales or use taxes. The total amount of Gross Collected Revenue may be materially less than the total amount of Gross Invoiced Revenue.

"Measuring Period" means the period of time commencing January 1, 20242023 and ending December 31, 2023 (the "Measuring Period") for each franchisee that met certain qualifying criteria. We did not include data for any ALL Business owned by us or our affiliates. 2024.

In order to be included ("Qualifying Franchisee" means a "qualifying franchisee"), the franchisee must have that has: (a) been open and operating for the entire 2023 calendar year Measuring Period; and (b) been operated and managed by 1 or more owners of the franchise on a full-time basis. The financial performance representation does not include data for any franchised outlets that are not managed by at least 1 franchise owner on a full-time basis (i.e., passive ownership). In the past, we allowed passive ownership where the franchise owner was not actively involved with the business on a full-time basis and either (a) hired a third-party manager to manage the business or (b) operated the business on a part-time basis while holding another job. We no longer allow passive ownership under our current franchise model. Therefore, the results from those franchisees have been excluded since their method of operation is materially different than the franchise offered under this Disclosure Document.

System Statistics & Subsets Utilized

The following presents certain financial performance information for the Measuring Period for each franchisee that met certain qualifying criteria. We did not include data for any ALL Business owned by us or our affiliates.

As of December 31, 2023 2024, there were 118 127 franchisees open and operating (with a total of 145 161 territories), and 83 87 of these franchisees are qualifying franchisees. Qualifying Franchisees. Of the 118 127 franchisees that were open as of December 31, 2023, 35 2024, 40 were not qualifying franchisees for the following reasons:-

- 2330 franchisees (with a total of 2437 territories) were excluded because they opened during the Measuring Period and therefore were not open for the entire Measuring Period
- <u>12</u>11 franchisees (with a total of <u>14</u>15 territories) were excluded because they were operated on a part-time basis and/or due to passive ownership

The table below summarizes the franchisee statistics and the number of qualifying franchisees Qualifying Franchisees (each referred to as an "outlet") whose data has been included:

System Statistics						
Franchisees in Operation (January 1, 20232024)	Franchisees in Operation -(December 31, 20232024)	Franchisees that Opened During 2023 2024	Franchisees that Left System During 20232024	Franchisees in Operation All of 20232024	Passive Ownership Franchisees in Operation All of 20232024	Qualifying Franchisees
102 <u>114</u>	118 <u>127</u>	23 <u>29</u>	<u>810</u>	95 98	12 <u>11</u>	83 <u>87</u>

^{*—} There are nois 1 excluded outlets in this Financial Performance Representation that closed during 20232024 after being open less than 12 months.

There are no material differences between the operations of the ALL Businesses operated by the franchisees whose data has been included and the franchised business offered under this Disclosure Document.-

For purposes of this financial performance representation, each franchisee is considered a separate "outlet," although some franchisees operate multiple territories. The 8387 franchisees whose data was included operate a total of 110109 territories as of December 31, 20232024. We have created the following subsets based on the number of territories owned as of December 31, 20232024, as reflected in the table on the following page:

<u>Subset</u>	Number of Territories Owned**	Number of Franchisees	<u>Total Number of Territories</u>
<u>1</u>	All Franchisees-	83 <u>87</u>	110 109
<u>2</u>	Franchisees Operating 1 Territory	61 <u>69</u>	61 <u>69</u>
<u>3</u>	Franchisees Operating 2 Territories	-18 * <u>15</u>	36 <u>30</u>
<u>4</u>	Franchisees Operating 3 Territories	3 <u>2</u>	<u>96</u>
<u>5</u>	Franchisees Operating 4 Territories	<u>1</u>	<u>4</u>

^{*} The determination of the number of territories owned was made as of the end of the 2023/2024 calendar year. For example, if a franchisee began 2023/2024 with 2 territories, closed a territory in June 2023/2024 and ended 2023/2024 owning 1 territory, the franchisee would be listed as a Franchisee Operating 1 Territory. In 2023/2024, we had 21 franchisees (whose data was included in this Financial Performance Representation) change the number of territories owned during 2023/2024, including 21 franchisees that began 2023/2024 with 32 territories, and each closed 1 territory during the year, ending the year with 21 territories each. These franchisees are reported as operating 21 territories, and we have excluded the closed territories from the financial performance representation.

Defined Terms

For purposes of this financial performance representation, the following terms have the meanings given to them below:

"Gross Invoiced Revenues" means the total amount of revenues invoiced, including (a) amounts that were not collected during the calendar year, or a reduction in fee due to a short-term stay or move-out, (b) amounts that could not be collected due to a client's death or for any other reason. The term Gross Invoiced Revenues does not include any sales or use taxes. The total amount of Gross Collected Revenues may be materially less than the total amount of Gross Invoiced Revenues.

"Gross Collected Revenues" means all gross sums collected by the franchised ALL Business, such as revenues from the sale or provision of goods or services, proceeds from business interruption insurance and referral fees collected, including referral fees paid by Referral Sources, Senior Care Providers, other ALL Businesses (such as fee splits under our Referral Program) or us (such as fee splits relating to a Regional Account). Gross Collected Revenues includes any Gross Collected Referral Proceeds (defined in Item 6, Note 4). With respect to fees generated through our Referral Program, Gross Collected Revenues only includes the franchisee's share of the placement fee collected by the franchisee. Gross Collected Revenues excludes the following amounts:

- any referral fee the franchisee pays us for a Corporate Lead
- <u>any placement fee the franchisee collects but pay to another ALL Business under our Referral Program</u>
- any sales or use tax the franchisee collects and pays to the applicable taxing authority
- the wholesale price paid by the franchisee for any merchandise or other retail items that it sells (Gross Collected Revenues includes the difference between the retail price received from the sale of the item and the wholesale price that the franchisee paid for the item)

The total amount of Gross Collected Revenues may be materially less than the total amount of Gross Invoiced Revenues.

Financial Performance Representation-

The following table presents the Gross Invoiced Revenues generated during 2023 the Measuring Period by the qualifying franchisees described above:

2023 CDOSS INVOICED REVENUES

			######################################											
	(Sui	BSETS BASED ON	Number of Ter	RITORIES OWNE	(D)									
2024 GROSS INVOICED REVENUE (SUBSETS BASED ON NUMBER OF TERRITORIES OWNED)														
Subset (Number of franchisees)	Number & Percent that Achieved/Surpassed Average													
Subset 1 (83(87 franchisees)	All	\$ 0 24,797	\$1, 024,867 <u>060,71</u> <u>9</u>	\$ 148,221 <u>172,64</u> <u>4</u>	\$ 209,684 <u>229,88</u> <u>4</u>	26 of 83 (31 <u>32/87</u> (<u>37</u> %)								
Subset 2 (61 franchisees69 Franchisees)	1	\$ 0 24,797	\$ 555,290 736,803	\$ 133,863 <u>141,93</u> <u>0</u>	\$ 177,256 <u>209,69</u> <u>8</u>	23 of 6126/69 (38%)								
Subset 3 (18 franchisees 15 Franchisees)	2	\$ 51,972 48,146	\$1, 024,867 <u>060,71</u> <u>9</u>	\$ 150,532 205,29 <u>&</u>	\$ 262,124 <u>291,92</u> <u>6</u>	6 of 18 (33 <u>3/15</u> (20%)								
Subset 4 (3 Franchisees2	3	\$136, 384 <u>991</u>	\$ 778,692 <u>763,540</u>	\$ 767,106 <u>450,26</u> <u>5</u>	\$ 560,727 <u>450,26</u> <u>5</u>	<u>1/</u> 2 of 3 (66 <u>(50</u> %)								

2023 GROSS INVOICED REVENUES

(SUBSETS BASED ON NUMBER OF TERRITORIES OWNED)

2024 GROSS INVOICED REVENUE

(SUBSETS BASED ON NUMBER OF TERRITORIES OWNED)

Subset (Number of franchisees)	Number of Territories Owned		Highest	Median	Average	Number & Percent that Achieved/Surpassed Average
<u>franchisees</u>)						
Subset 5 (1 Franchisees franchisee)	4	\$ 190,765 <u>251,33</u> <u>8</u>	\$ 190,765 <u>251,338</u>	\$ 190,765 <u>251,33</u> <u>8</u>	\$ 190,765 <u>251,33</u> <u>8</u>	1 -of - <u>/</u> 1 (100%)

The following table presents the Gross Collected Revenues generated during 2023 the Measuring Period by the qualifying Franchisees Qualifying Franchisees described above:

	20232024 GROSS COLLECTED REVENUES REVENUE (SUBSETS BASED ON NUMBER OF TERRITORIES OWNED)										
Subset (Number of franchisees)	Number of Territorie s Owned	Lowest	Highest	Median	Average	Number & Percent that Achieved/Surpass ed Average					
Subset 1 (83 (87 franchisees)	All	\$ 0 17,914	\$ 920,942 <u>901,54</u> <u>0</u>	\$ 137,300 140,45 <u>0</u>	\$ 189,067 <u>199,43</u> <u>4</u>	25 of 83 (30 <u>37/87</u> (43%)					
Subset 2 (61 franchisees69 Franchisees)	1	\$ 0 17,914	\$ 522,981 <u>621,80</u> <u>7</u>	\$ 133,863 <u>130,54</u> <u>1</u>	\$ 177,256 <u>179,96</u>	20 of 60 (3326/69 (38%)					
Subset 3 (18 franchisees 15 Franchise es)	2	\$4 0,462 60,813	\$ 920,942 <u>901,54</u> <u>0</u>	\$ 133,110 <u>211,11</u> <u>3</u>	\$247,296 <u>265,30</u>	7 of 18 (39 <u>4/15</u> (<u>27</u> %)					
Subset 4 (3 Franchisees2 franchisees)	3	\$ 119,545 <u>129,09</u> <u>6</u>	\$ <mark>817,262</mark> 695,88	\$ <mark>538,479</mark> 412,48	\$ <mark>491,762</mark> 412,48	<u>1/</u> 2 -of 3 (66 _(50%)					
Subset 5 (1 Franchisees franchisee)	4	\$ 139,221 <u>128,31</u> <u>2</u>	\$ 139,221 <u>128,31</u> <u>2</u>	\$ 139,221 <u>128,31</u>	\$ 139,221 <u>128,31</u> <u>2</u>	1 -of ∕1 -(100%)					

Notes:-

- 1. In making the above financial performance representation, we have relied upon reports submitted to us by our franchisees. Neither we nor any independent certified public accountant has independently audited or verified the information.
- 2. The financial performance representation is based on the historical results from the franchisees described above.
- 3. The financial performance representation does not include any expense information. As a franchisee, you will incur a variety of expenses, such as payroll expenses (if you have employees or independent

contractors), marketing expenses, gas, vehicle expenses, utilities as well as the initial and ongoing fees imposed under the Franchise Agreement.

Some Assisted Living Locators businesses have earned this amount. Your individual results may differ. There is no assurance that you will earn as much.

Written substantiation for this financial performance representation will be made available to you upon your reasonable written request.

Other than the preceding financial performance representation, we do not make any financial performance representations. We also do not authorize our employees or representatives to make any such representations either orally or in writing. If you are purchasing an existing outlet, however, we may provide you with the actual records of that outlet. If you receive any other financial performance information or projections of your future income, you should report it to the franchisor's management by contacting Angela Angelica Olea at 8100 E. Indian School Road, Suite 201, Scottsdale, Arizona 85251, (800) 267-7816, the Federal Trade Commission, and the appropriate state regulatory agencies.

ITEM 20 OUTLETS AND FRANCHISEE INFORMATION

	TABLE 1 - SYSTEM-WIDE OUTLET SUMMARY FOR YEARS 20212022 TO 20232024										
Outlet Type	Year	Outlets at the Start of the Year	Outlets at the End of the Year	Net Change							
	2021 <u>2022</u>	129 <u>131</u>	131 <u>134</u>	+ 2 <u>3</u>							
Franchised	<u>2023</u> 2022	131 <u>134</u>	134 <u>145</u>	+ <u>3</u> 11							
	2023 <u>2024</u>	134 <u>145</u>	145 <u>161</u>	+ 11 <u>16</u>							
	<u>2022</u> 2021	<u>63</u>	3	-3 <u>0</u>							
Company-Owned	<u>2023</u> 2022	3	3	0							
	2023 <u>2024</u>	3	<u>34</u>	<u>0+1</u>							
	<u>2022</u> 2021	135 <u>134</u>	133 <u>137</u>	<u>-1±3</u>							
Total Outlets	<u>2023</u> 2022	13 4 <u>137</u>	137 <u>148</u>	+ <u>3</u> 11							
	2023 <u>2024</u>	137 <u>148</u>	<u>148</u> 165	+ 11 <u>17</u>							

TABLE 2 - TRANSFERS OF OUT	ETS FROM FRANCHISEES TO NEW OWNERS (OTHER THAN THE FRANCHISOR)
	<u>for Years 2021 2022 to 2023 2024</u>	
State	Year	Number of Transfers
	<u>2022</u> 2021	<u>01</u>
Arizona	<u>2023</u>	<u>1</u>
	<u>2024</u>	<u>4</u>
<u>Colorado</u>	2022	<u> 40</u>
	2023	<u> 40</u>
	<u>2024</u>	<u>0</u>
	<u>2022</u> 2021	<u>03</u>
Dlavida	2022	3
Florida	2023	2
	<u>2024</u>	<u>1</u>
	<u>2022</u> 2021	0
Georgia	2022	0
	2023	2

TABLE 2 - TRANSPERS OF CUILE	TS FROM FRANCHISEES TO NEW OWNER FOR YEARS 20212022 TO 2023 2024	SACOTIER THAN THE PRANCHISOR)
State	Year	Number of Transfers
	<u>2024</u>	<u>O</u>
	<u>2022</u> 2021	1
	2022	1
Maryland	2023	0
	<u>2024</u>	<u>0</u>
	<u>2022</u> 2021	1
	2022	1
Massachusetts	2023	0
	<u>2024</u>	<u>0</u>
	<u>2022</u> 2021	0
Minnesota	<u>2023</u>	<u>1</u>
	<u>2024</u>	<u>0</u>
North Carolina	2022	0
	<u>2023</u>	<u>0</u>
	2021 <u>2024</u>	1
	<u>2022</u>	<u>0</u>
South Carolina	<u>2023</u>	<u>0</u>
	<u>2024</u>	<u>1</u>
	<u>2022</u>	<u>0</u>
<u>Texas</u>	<u>2023</u>	<u>0</u>
	<u>2024</u>	<u>1</u>
	<u>2022</u> 2021	<u>01</u>
Washington	2022	1
	2023	0
Total	2021 <u>2024</u>	<u>21</u>
<u>Total</u>	2022	7
	2023	6
	2024	9

Table 3 - Status of Franchised Outlets¹
For Years 2021 to 2023 2022-2024

_

¹ Note that if there were multiple events applicable to a single outlet, only the last event should be reported. For example, a franchisee abandons the franchise (i.e., ceased operations), the franchisor terminates the franchise (i.e., a termination), and following termination, the franchisor reacquires the franchise by purchasing the assets (i.e., a reacquisition). Only the event last in time (i.e., reacquisition) should be reported in the chart.

State	Year	Outlets at Start of Year	Outlets Opened ²	Terminations	Non- Renewals	Reacquired by Franchisor	Ceased Operations - Other Reasons	Outlets at End of Year
	2022 20 21	2	0	0	0	0	0	2
Alabama	2023 20	2	0	0	0	0	0	2
	2023 <u>20</u> 24	2	0	<u>01</u>	0	0	0	<u>21</u>
	2022 20 21	10	0	0	0	0	0	10
Arizona	2023 20	10	0	0	0	0	0	10
	2023 <u>20</u> 24	10	0 1	0	0	0	0	10 <u>11</u>
<u>Arkansas</u>	<u>2022</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>
Alkansas	<u>2023</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>O</u>
California	2021 <u>20</u> 24	16 <u>0</u>	<u>01</u>	<u>0</u> 1	0	<u>4</u> <u>0</u>	<u>20</u>	13 <u>0</u>
<u>California</u>	2022	13	4	4	0	0	2	11
	2023	11	2	1	0	0	0	12
Colorado	2021 <u>20</u> 24	5 <u>12</u>	<u> 45</u>	<u>01</u>	0	0	0	6 <u>16</u>
<u>Colorado</u>	2022	6	0	0	0	0	0	6
	2023	6	0	0	0	0	0	6
	<u>2024</u>	<u>6</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>6</u>
Connecticut	2022 ₂₀ 21	1	<u>01</u>	0	0	0	0	<u>42</u>
	2022	4	1	0	θ	0	θ	2
	2023	2	0	0	0	0	0	2
Delaware	2021 <u>20</u> 24	0 <u>2</u>	0	0	0	0	0	0 2_
<u>Delaware</u>	2022	0	1	0	0	0	0	1
	2023	1	0	0	0	0	0	1
	<u>2024</u>	<u>1</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	1
District of Columbia	2022 20 21	1	0	0	0	0	0	1
	2022	4	θ	0	θ	0	θ	4
	2023	1	0	0	0	0	0	1
Florida	2021 <u>20</u> 24	<u>81</u>	<u>90</u>	0	0	0	<u> 40</u>	16 <u>1</u>

² This should include both new outlets and existing company-owned outlets that a franchisee purchased from the franchisor. The number of existing company-owned outlets that are sold to a franchisee should also be reported in Table 3 in column called "Outlets Sold to Franchisee."

				Status of France Years <mark>2021 to 2</mark> 0		TS ¹		
State	Year	Outlets at Start of Year	Outlets Opened ²	Terminations	Non- Renewals	Reacquired by Franchisor	Ceased Operations - Other Reasons	Outlets at End of Year
<u>Florida</u>	2022	16	0	0	0	0	2	14
	2023	14	4	0	0	0	0	18
Georgia	2021 <u>20</u> 24	<u>218</u>	1	0 <u>2</u>	0	0	0	<u>317</u>
<u>Georgia</u>	2022	3	0	0	0	0	0	3
	2023	3	1	0	0	0	0	4
	<u>2024</u>	<u>4</u>	<u>1</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>5</u>
Idaho	2022 <mark>20</mark> 21	1	0	0	0	0	0	1
	2022	1	0	θ	0	0	0	1
	2023	1	0	0 1	0	0	0	<u> 40</u>
	2024	1	0	<u>1</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>
	2022 20 21	<u> 42</u>	<u>40</u>	0	0	0	0	2
Illinois	202320 22	2	0	0	0	0	0	2
	2023 <u>20</u> 24	2	€2	0	0	0	0	2 <u>4</u>
Indiana	2022 20 21	3 <u>2</u>	0	<u> 10</u>	0	0	0	2
	2022	2	θ	θ	0	0	0	2
	2023	2	0	0	0	0	0	2
	2024	<u>2</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>2</u>
	<u>2022</u> 20 21	3	0	0	0	0	0	3
Kansas	2023 20 22	3	0	0	0	0	0	3
	2023 <u>20</u> 24	3	0	0	0	0	0	3
Kentucky	2022 <mark>20</mark> 21	0	θ <u>1</u>	0	0	0	0	<u> 01</u>
	2022	θ	1	0	0	0	θ	1
	2023	1	0	0	0	0	0	1
	2024	<u>1</u>	<u>0</u>	<u>1</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>
Louisiana	<u>2022</u> 20 21	1	0	0	0	0	0	1
	2022	4	0	θ	0	0	0	4
	2023	1	0	0	0	0	0	1
	2024	<u>1</u>	<u>1</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>2</u>
Maine	<u>2022</u> 20 21	1	0	0	0	0	0	1

Page 59

Franchise Disclosure Document (2025)

				STATUS OF FRANC YEARS 2021 to 2 0		<u>TS¹</u>		
State	Year	Outlets at Start of Year	Outlets Opened ²	Terminations	Non- Renewals	Reacquired by Franchisor	Ceased Operations - Other Reasons	Outlets at End of Year
	2022	4	0	0	0	0	0	4
	2023	1	0	0	0	0	0	1
	<u>2024</u>	<u>1</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>1</u>
Maryland	2022 20 21	7	0	<u>01</u>	0	<u>01</u>	<u>01</u>	7 <u>4</u>
	2022	7	θ	1	θ	1	1	4
	2023	4	0	0	0	0	0	4
	<u>2024</u>	<u>4</u>	<u>2</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>6</u>
Massachusetts	2022 20 21	3	0 2_	0	0	0	0	<u>35</u>
	2022	3	2	θ	θ	θ	θ	5
	2023	5	0	0	0	0	0	5
	<u>2024</u>	<u>5</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>5</u>
Michigan	2022 ₂₀ 21	8	<u> 40</u>	<u>40</u>	0	0	0	8
	2022	8	θ	0	0	0	θ	8
	2023	8	1	0	0	0	0	9
	<u>2024</u>	<u>9</u>	<u>1</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>10</u>
Minnesota	2022 20 21	<u>01</u>	<u> 40</u>	0	0	0	0	1
	2022	4	0	0	0	0	0	4
	2023	1	2	0	0	0	0	3
	<u>2024</u>	<u>3</u>	<u>2</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>5</u>
Mississippi	2022 20 21	0	<u>01</u>	0	0	0	0	<u>01</u>
	2022	θ	1	θ	θ	θ	θ	1
	2023	1	0	0	0	0	0	1
	<u>2024</u>	1	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	1
Missouri	2022 <mark>20</mark> 21	1	0	0	0	0	0	1
	2022	1	θ	0	θ	0	0	1
	2023	1	1	0	0	0	0	2
	<u>2024</u>	<u>2</u>	<u>2</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>4</u>
Nebraska	2022 20 21	2	0	0	0	0	0	2
	2022	2	θ	0	θ	0	θ	2
	2023	2	0	0	0	0	0	2
	<u>2024</u>	<u>2</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>2</u>

				STATUS OF FRANC YEARS 2021 TO 20		<u>TS¹</u>		
State	Year	Outlets at Start of Year	Outlets Opened ²	Terminations	Non- Renewals	Reacquired by Franchisor	Ceased Operations - Other Reasons	Outlets at End of Year
New Hampshire	$\frac{202220}{21}$	2	0	0	0	0	0	2
	2022	2	0	0	0	0	0	2
	2023	2	0	0	0	0	0	2
	<u>2024</u>	<u>2</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>2</u>
New Jersey	2022 20 21	3	<u>02</u>	0	0	0	0	<u>35</u>
	2022	3	2	0	θ	0	θ	5
	2023	5	1	0	0	0	0	6
	<u>2024</u>	<u>6</u>	<u>1</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>7</u>
New York	2022 20 21	2	0	0	0	0	0	2
	2022	2	θ	0	0	0	0	2
	2023	2	0	0	0	0	0	2
	<u>2024</u>	<u>2</u>	<u>3</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>5</u>
North Carolina	<u>2022</u> 20 21	7 <u>6</u>	0	<u> 10</u>	0	0	0 2_	<u>64</u>
	2022	6	0	θ	0	0	2	4
	2023	4	0	1	0	0	0	3
Ohio	2021 <u>20</u> 24	<u>83</u>	<u>01</u>	<u>40</u>	0	0	<u> 10</u>	<u>64</u>
<u>Ohio</u>	2022	6	1	0	0	0	0	7
	2023	7	0	0	0	0	0	7
	<u>2024</u>	<u>7</u>	<u>1</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>8</u>
Oregon	2022 20 21	<u>32</u>	<u>01</u>	0	0	<u> 10</u>	0	<u>23</u>
	2022	2	1	0	0	0	0	3
	2023	3	0	1	0	0	0	2
	<u>2024</u>	<u>2</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>1</u>	<u>0</u>	<u>1</u>
Pennsylvania	2022 20 21	2	<u>01</u>	0	0	0	0	<u>23</u>
	2022	2	1	0	0	0	0	3
	2023	3	1	0	0	0	0	4
	<u>2024</u>	<u>4</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>4</u>
Rhode Island	2022 20 21	0	<u>01</u>	0	0	0	0	<u> </u>
	2022	θ	1	0	0	0	0	1
	2023	1	0	0	0	0	0	1
	<u>2024</u>	1	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>1</u>

				STATUS OF FRAN YEARS <mark>2021 TO 2</mark> 0		<u>TS¹</u>		
State	Year	Outlets at Start of Year	Outlets Opened ²	Terminations	Non- Renewals	Reacquired by Franchisor	Ceased Operations - Other Reasons	Outlets at End of Year
South Carolina	2022 20 21	5	0	0	0	0	θ <u>1</u>	<u>54</u>
	2022	5	θ	θ	0	0	1	4
	2023	4	0	0	0	0	0	4
	<u>2024</u>	<u>4</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>4</u>
Tennessee	2022 20 21	2	<u>40</u>	0	0	0	<u> 10</u>	2
	2022	2	0	0	0	0	0	2
	2023	2	0	0	0	0	0	2
Fexas	2021 <u>20</u> 24	10 2	<u>20</u>	0	0	0	<u>30</u>	9 2
Texas	2022	9	0	0	0	0	1	8
	2023	8	1	1	0	0	0	8
Jtah	2021 <u>20</u> 24	<u>18</u>	<u> 40</u>	<u>12</u>	0	0	0	<u> 16</u>
<u>Utah</u>	2022	1	1	0	0	0	1	1
	2023	1	0	0	0	0	0	1
	<u>2024</u>	<u>1</u>	<u>1</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>2</u>
Virginia	2022 20 21	<u>32</u>	<u>01</u>	0	0	0	<u> 40</u>	2 <u>3</u>
	2022	2	1	θ	θ	0	0	3
	2023	3	0	0	0	0	0	3
	<u>2024</u>	<u>3</u>	<u>2</u>	<u>2</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>3</u>
Washington	2022 20 21	3	<u>01</u>	0	0	0	0	<u>34</u>
	2022	3	1	0	θ	0	0	4
	2023	4	0	0	0	0	0	4
	<u>2024</u>	<u>4</u>	<u>2</u>	2	<u>0</u>	<u>0</u>	<u>0</u>	<u>4</u>
Wisconsin	2022 20 21	2	0	0	0	0	0	2
	2022	2	0	0	0	0	0	2
	2023	2	1	0	0	0	0	3
Totals	2021 <u>20</u> 24	129 <u>3</u>	<u>180</u>	<u>50</u>	0	<u>20</u>	<u>90</u>	<u>1313</u>
<u>Totals</u>	2022	131	19	5	0	1	10	134
	2023	134	15	4	0	0	0	145
	<u>2024</u>	<u>145</u>	<u>30</u>	<u>13</u>	<u>0</u>	<u>1</u>	<u>0</u>	<u>161</u>

^{*} In some cases, a franchisee may own a territory that encompasses a portion of 2 states. In those situations, we

list the outlet within the state in which the majority of the territory (i.e., the largest population) is located.

TABLE 4 - STATUS OF COMPANY-OWNED OUTLETS FOR YEARS 2021 2022 TO 2023 2024								
State	Year	Outlets at Start of Year	Outlets Opened	Outlets Reacquired From Franchisee	Outlets Closed	Outlets Sold to Franchisee	Outlets at End of Year	
California	2022 202 1	<u>0</u> 1	0	<u> 10</u>	0	0	1	
	2023 202 2	1	0	0	0	0	1	
	2023 <u>202</u> <u>4</u>	1	0	0	0	0	1	
<u>Georgia</u> Colorad 0	2022 <mark>202</mark>	1	0	0	0	<u> 40</u>	<u>01</u>	
	2023 202 2	<u>0</u> 1	0	0	0	0	<u>0</u> 1	
	2023	0	θ	θ	0	0	1	
Georgia	2021 <u>202</u> <u>4</u>	1	0	0	0	0	1	
Oregon	2022	1	0	0	0	0	1	
	2023	1	0	0	0	0	1	
Minnesota	2021 <u>202</u> <u>4</u>	1	0	0	0	<u> 40</u>	<u>01</u>	
Washington	2022	0	0	0	0	0	0	
	2023	0	0	0	0	0	0	
North Carolina	2021 <u>202</u> <u>4</u>	<u> 40</u>	₩2	0	<u> 10</u>	<u> 01</u>	θ <u>1</u>	
	2022	0	0	θ	0	0	0	
	2023	0	0	0	0	0	0	
Oregon	2021	0	0	1	0	0	1	
	2022	4	0	0	0	0	1	
	2023	1	θ	θ	θ	θ	1	
Tennessee	2021	4	0	0	4	0	0	
	2022	θ	θ	θ	θ	θ	θ	
	2023	θ	θ	θ	θ	θ	θ	
Texas	2021	4	0	θ	1	θ	0	
	2022	θ	θ	θ	θ	θ	θ	
	2023	θ	θ	θ	θ	θ	θ	
Totals	2022 <mark>202</mark> 1	<u>63</u>	0	<u>20</u>	<u>3</u> <u>0</u>	<u>20</u>	3	
	2022	3	θ	θ	θ	θ	3	
	2023	3	0	0	0	0	3	
	<u>2024</u>	<u>3</u>	<u>1</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>4</u>	

	TABLE 5 - PROJECTED OPENINGS AS OF DECEMBER 31, 2023 2024						
State	Franchise Agreements Signed But Outlet Not Opened	Projected New Franchised Outlets in the Next Fiscal Year	Projected New Company- Owned Outlets in the Next Fiscal Year				
California	0	0 2	0				
Connecticut	0	0	0				
Delaware	0	0	0				
Florida	<u> 10</u>	<u>42</u>	0				
Georgia	<u> 40</u>	1	0				
<u>Illinois</u>	<u>0</u>	2	<u>0</u>				
<u>Indiana</u>	<u>0</u>	<u>1</u>	<u>0</u>				
Maryland	<u>0</u> 1	<u>01</u>	0				
<u>Michigan</u>	<u>0</u>	<u>1</u>	<u>Q</u>				
New York	<u>1</u>	<u>0</u>	<u>0</u>				
North Carolina	0	<u>01</u>	0				
Ohio	0	<u>01</u>	0				
Oklahoma	0	0	0				
<u>Oregon</u>	<u>0</u>	<u>1</u>	<u>0</u>				
Pennsylvania	0	0	0				
South Carolina	<u>0</u>	<u>1</u>	<u>0</u>				
<u>Tennessee</u>	<u>0</u>	<u>1</u>	<u>0</u>				
Texas	0	<u>01</u>	0				
Virginia	0	0	0				
Washington	1	1	0				
Totals	3	<u>3</u> 17	0				

Our fiscal year ends on December 31* A list of all current Assisted Living Locators franchisees is attached to this Disclosure Document as <u>EXHIBIT "E"</u> (Part A), including their names and the addresses and telephone numbers of their outlets as of December 31, <u>20242022</u>. In addition, <u>EXHIBIT "E"</u> (Part B) lists the name, city and state, and the current business telephone number (or, if unknown, the last known home telephone number) of every franchisee who had an outlet transferred, terminated, canceled, not renewed, or otherwise voluntarily or involuntarily ceased to do business under the franchise agreement during our most recently completed fiscal year or who has not communicated with us within 10 weeks of the issuance date of this Disclosure Document. If you buy this franchise, your contact information may be disclosed to other buyers when you leave the franchise system.

In the last 3 fiscal years, some franchisees have signed confidentiality agreements with us. In some instances, current and former franchisees sign provisions restricting their ability to speak openly about their experience with us. You may wish to speak with current and former franchisees but be aware that not all such franchisees will be able to communicate with you.

We have created the following Franchise Advisory Council:

Franchise Advisory Council 8100 E. Indian School Road, Suite 201, Scottsdale, Arizona 85251 Scottsdale, Arizona 85251 (800) 267-7816

Attn: Angela Angelica Olea angela angelica @assisted living locators.com

Except for the Franchise Advisory Council listed above, there are no (a) trademark-specific franchisee organizations associated with the franchise system being offered that we have created, sponsored or endorsed or (b) independent franchisee organizations that have asked to be included in this Disclosure Document.

ITEM 21 FINANCIAL STATEMENTS

Our fiscal year ends on December 31st. Attached to this Disclosure Document as Exhibit F-1 are the audited consolidated financial statements for EHC Holding Company, LLC, our parent, for the fiscal years ended December 31, 2024, December 31, 2023 and December 31, 2022, and . Our parent, EHC Holding Company, LLC, has guaranteed our audited financial statements for the years ended December 31, 2020 2021.performance with you. In addition, an unaudited balance sheet as of May 31, 2024, April 30, 2025 and profit and loss statement of operations and statements of cash flows from January 1, 20242025 through May 31, 2024April 30, 2025 for EHC Holding Company, LLC, our parent, are attached to this Disclosure Document as Exhibit F-2. A copy of the guaranty of performance is attached as Exhibit F-3.

Our parent, EHC Holding Company, LLC, has guaranteed our performance with you. A copy of the guaranty of performance is attached as <u>Exhibit F-3</u>. EHC Holding Company, LLC has not been in business for three or more years and cannot therefore provide all of the financial statements that would otherwise been required in this Item. Our fiscal year end is December 31 of each year.

ITEM 22 CONTRACTS

Attached to this Disclosure Document (or the Franchise Agreement attached to this Disclosure Document) are copies of the following franchise and other contracts or agreements proposed for use or in use in this state:

Exhibits to Disclosure Document

EXHIBIT "C"	Franchise Agreement
-------------	---------------------

EXHIBIT "G"-1 Franchisee Disclosure Questionnaire (Questionnaire may not be signed or used if the

 $franchisee \ resides \ within, or \ the \ franchised \ business \ will \ be \ located \ within, a \ franchise$

registration state)

EXHIBIT "G"-2 General Release

EXHIBIT "G"-3 ALL-IN User Agreement and Privacy Policy

EXHIBIT "G"-4 HIPAA Business Associate Agreement

EXHIBIT "G"-5 Power of Attorney – Telephone and Internet

EXHIBIT "G"-6 Sample Client Contract

EXHIBIT "G"-7 Franchise Resale Agreement

EXHIBIT "H" State Addenda and Agreement Riders

EXHIBIT "I" State Effective Dates

Attachments to Franchise Agreement

ATTACHMENT "B" Form of Referral Network Participation Agreement

ATTACHMENT "C" Franchise Owner Agreement

ATTACHMENT "D" ACH Authorization Form

ATTACHMENT "E" Brand Protection Confidentiality Agreement

ITEM 23 RECEIPT

<u>EXHIBIT "J"</u> to this Disclosure Document are detachable receipts. You are to sign both, keep one copy and return the other copy to us.

EXHIBIT "A"

TO DISCLOSURE DOCUMENT

STATE AGENCIES AND ADMINISTRATORS

CALIFORNIA

Commissioner of Financial Protection & Innovation Department of Financial Protection & Innovation 320 West 4th Street, #750 Los Angeles, CA 90013 (213) 576-7500 1-866-275-2677

HAWAII

Commissioner of Securities of the State of Hawaii 335 Merchant Street, Room 203 Honolulu, Hawaii 96813 (808) 586-2722

Agents for Service of Process:
Commissioner of Securities of the State of Hawaii
Department of Commerce and Consumer Affairs

Business Registration Division 335 Merchant Street, Room 203 Honolulu, Hawaii 96813 (808) 586-2722

ILLINOIS

Illinois Attorney General Chief, Franchise Division 500 South Second Street Springfield, IL 62706 (217) 782-4465

INDIANA

Secretary of State
Securities Division
Room E-018
302 West Washington Street
Indianapolis, IN 46204
(317) 232-6681

MARYLAND

Office of the Attorney General Securities Division 200 St. Paul Place Baltimore, Maryland 21202 (410) 576-6360

MICHIGAN

Franchise Section Consumer Protection Division 525 W. Ottawa Street, G. Mennen Williams Building, 1st Floor Lansing, MI 48913 (517) 335-7567

MINNESOTA

Commissioner of Commerce Director of Registration 85 Seventh Place East, #280 St. Paul, Minnesota 55101-3165 (651) 539-1500

NEW YORK

New York Attorney General Investor Protection Bureau Franchise Section 28 Liberty Street New York, NY 10005 (212) 416-8222 Agents for Service of Process: New York Department of State One Commerce Plaza

Albany, NY 12231 NORTH DAKOTA

North Dakota Securities
Department
State Capitol, 5th Floor, Dept 414
600 East Boulevard Avenue
Bismarck, North Dakota 58505
(701) 328-4712

99 Washington Avenue, 6th Floor

RHODE ISLAND

Department of Franchise Regulation 1511 Pontiac Avenue, John O. Pastore Complex, Bldg 69-1 Cranston, Rhode Island 02920 (401) 462-9527

SOUTH DAKOTA

Department of Labor and

Regulation
Division of Insurance Securities
Regulation
124 S. Euclid, Suite 104
Pierre, South Dakota 57501

VIRGINIA

(605) 773-3563

State Corporation Commission
Division of Securities and Retail
Franchising
1300 East Main Street, 9th Floor
Richmond, Virginia 23219
(804) 371-9051
Agents for Service of Process:
Clerk of the State Corporation
Commission
1300 East Main Street, 1st Floor
Richmond, Virginia 23219

WASHINGTON

Department of Financial Institutions Securities Division 150 Israel Road SW Tumwater, WA 98501 (360) 902-8760

WISCONSIN

Department of Financial Institutions Division of Securities 201 W Washington Avenue, Suite 500, Madison, WI 53703 (608) 261-9555

EXHIBIT "B"

TO DISCLOSURE DOCUMENT

FRANCHISOR'S AGENT FOR SERVICE OF PROCESS

ALL Franchising, LLC A Delaware limited liability company 8100 E. Indian School Road, Suite 201 Scottsdale, Arizona 85251

In states listed in EXHIBIT "A", the additional agent for Service of Process is listed in EXHIBIT "A"

EXHIBIT "C"

TO DISCLOSURE DOCUMENT

FRANCHISE AGREEMENT

[See Attached]



ASSISTED LIVING LOCATORS FRANCHISE AGREEMENT

FRANCHISEE:	
DATE:	

TABLE OF CONTENTS

1.	DEFINITIONS		1			
2.		HISE				
3.	TERRITORIAL RIGHTS AND LIMITATIONS					
4.	TERM AND RENEV	WAL	5			
5.	TRAINING AND CONFERENCES					
6.		OR ASSISTANCE.				
7.	ESTABLISHING YO	OUR BUSINESS	10			
8.	MANAGEMENT A	ND STAFFING.	11			
9.		NTITY				
10.	ADVERTISING & N	MARKETING	12			
11.	OPERATING STAN	DARDS.	15			
12.		RMANCE REQUIREMENTS				
13.		SORY COUNCIL				
14.	FEES		21			
15.	BRAND PROTECTION COVENANTS					
16.	YOUR OTHER RES	PONSIBILITIES	24			
17.	INSPECTION AND	AUDIT	26			
18.	INTELLECTUAL P	ROPERTY	27			
19.	INDEMNITY		28			
20.	TRANSFERS		29			
21.	TERMINATION		30			
22.	POST-TERM OBLIG	GATIONS	32			
23.	DISPUTE RESOLU'	ΓΙΟΝ	33			
24.	OMITTED		36			
25.	GENERAL PROVIS	IONS	34			
ATTACE	IMENTS					
ATTACH	MENT "A"	Deal Terms				
ATTACH	MENT "B"	Form of Referral Network Participation Agreement				
ATTACH	ACHMENT "C" Franchise Owner Agreement					
ATTACH	CHMENT "D" ACH Authorization Form					

 $\underline{ \textbf{Brand Protection}}\underline{ \textbf{Confidentiality}} \ \textbf{Agreement}$

Definitions

ATTACHMENT "E" ATTACHMENT "F"

ASSISTED LIVING LOCATORS FRANCHISE AGREEMENT

This As	siste	ed Living	Locators	s Franchis	se Agı	reement (thi	s " <u>Agr</u>	eement") is	entered	into as of			
202 (the	"Effectiv	e Date")	between	ALL	Franchising	g, LLC	, a Delawa	re limite	d liability	company	(" <u>we</u> "	oı
" <u>us</u> ") ar	nd			, a(n)			(" <u>y</u>	<u>'ou</u> '').					

- **DEFINITIONS**. Capitalized terms used in this Agreement are defined either in the body of this Agreement or in <u>ATTACHMENT "F"</u>. For capitalized terms that are defined in the body of this Agreement, <u>ATTACHMENT "F"</u> lists the Sections of this Agreement in which such terms are defined. For purposes of this Agreement, the term "<u>ALL Business</u>" refers to an Assisted Living Locators business, and includes any Assisted Living Locators business operated by us, our affiliates, you or another Assisted Living Locators franchisee.
- 2. GRANT OF FRANCHISE. We hereby grant you a license to own and operate an Assisted Living Locators business (your "Business") using our Intellectual Property solely within the geographic area identified in ATTACHMENT "A" (your "Territory"). As an Assisted Living Locators franchisee, you will operate a home-based business that assists seniors and their families with evaluating and locating Senior Communities (defined below) that meet their geographic, financial and clinical needs and preferences. In connection with your Business, you will:
 - (i) identify, inspect, evaluate, monitor and establish relationships with assisted living facilities, memory care communities, nursing homes, senior care homes and independent living senior communities (collectively, "Senior Communities") that are located within your Territory;
 - (ii) identify and establish referral relationships with hospitals, insurance companies, HMO's, transitional care companies and other similar types of businesses or institutions that are in a position to refer clients in need of senior placement services ("Referral Sources");
 - (iii) identify, evaluate and establish referral relationships with providers of various senior care services (other than placement services) that may be necessary or helpful to your clients, including real estate agents, downsizing specialists, home health experts, financial planners, financing companies and elder-care attorneys ("Senior Service Providers"):
 - (iv) contract with your clients and conduct an assessment of each client's unique situation and needs to identify Senior Communities that may be a good match for their needs;
 - (v) facilitate the placement of each client with the Senior Community that best meets their needs, either directly or in conjunction with another ALL Business through our Referral Program described in Section 11.11; and
 - (vi) when appropriate, refer your clients to one or more Senior Service Providers that offer goods or services that you reasonably believe may be necessary of helpful to your clients.

You must establish and operate your Business in strict compliance with all of the terms and conditions of this Agreement. We reserve all rights not expressly granted to you.

3. TERRITORIAL RIGHTS AND LIMITATIONS.

- **3.1.** <u>Territorial Rights.</u> You are not granted an exclusive Territory. However, you will receive the territorial protections described below for the duration of the Term, unless we modify such protections pursuant to <u>Section 12</u> as a result of your breach of the minimum performance requirements:
 - (a) <u>Marketing Agreements</u>. We will not enter into, or authorize any other ALL Business to enter into, a Marketing Agreement with any Senior Community located in your Territory except as otherwise permitted by <u>Section 3.2(a)</u> (Regional Accounts) and <u>Section 3.2(d)</u> (Acquisitions). If we have entered into a Marketing Agreement with a Senior Community in your Territory prior to the Effective Date, we will assign the Marketing Agreement to you at the time you commence operating your Business.

- Notwithstanding the foregoing, we have no obligation to assign any Marketing Agreement to you if such Marketing Agreement is associated with a Regional Account.
- (b) Targeted Marketing. We will not engage in, or authorize any other ALL Business to engage in, "Targeted Marketing" within your Territory, except as otherwise permitted by <u>Section 3.2(d)</u> (Acquisitions). "<u>Targeted Marketing</u>" is any advertising or marketing that is directed into a specific territory for the primary purpose of soliciting clients, Senior Communities and/or Referral Sources within that territory. Marketing and advertising that is distributed, circulated or received within both the territory of the person engaging in the advertising as well as the territory of another ALL Businesses is not deemed to be "Targeted Marketing" if: (a) the person engaging in the advertising uses reasonable efforts to limit the circulation or distribution of the advertising to areas within that person's territory; and (b) the majority of the recipients of the advertising are located within the territory of the person engaging in the advertising and there is only incidental circulation or distribution outside of that person's territory. "Targeted Marketing" also does not include any marketing that we conduct as part of our administration of the brand and system development fund or other systemwide marketing or advertising efforts. The Manual may further describe the circumstances in which advertising or marketing is deemed to be "Targeted Marketing." Other ALL Businesses are free to contract with clients, Referral Sources and Senior Care Providers that reside or are located within your Territory as long as they do not, without your prior written permission, solicit the clients, Referral Sources or Senior Care Providers utilizing Targeted Marketing within your Territory.
- (c) <u>Placements</u>. We will not place, or authorize any other ALL Business to place, clients with Senior Communities that are located in your Territory except as otherwise permitted by <u>Section 3.2(a)</u> (Regional Accounts), <u>Section 3.2(b)</u> (Referral Program), Section 3.2(c) (Corporate Leads) and Section 3.2(d) (Acquisitions).
- **3.2.** <u>Limitations on Territorial Rights.</u> We and our affiliates reserve all rights not expressly granted to you under this Agreement. Your territorial rights and protections described in <u>Section 3.1</u> above are subject to the following exceptions and limitations:
 - (a) Regional Accounts. We have the exclusive right to enter into Marketing Agreements with businesses that have Senior Communities in multiple territories (a "Regional Account"), including Regional Accounts that have one (1) or more Senior Communities located in your Territory. You may not enter into a Marketing Agreement with a Regional Account without our prior written consent, which we may withhold in our sole discretion. If you enter into a Marketing Agreement with a Senior Community that (at the time of contracting) only has facilities within your Territory but subsequently opens additional facilities in one (1) or more territories owned by other ALL Businesses, we reserve the right to take any of the following actions in our sole discretion:
 - (i) Negotiate and enter into a new Marketing Agreement with the Senior Community and convert them to a Regional Account, in which case your Marketing Agreement would be terminated and replaced by our Marketing Agreement. If we convert the Senior Community to a Regional Account, we may negotiate an alternative fee structure, but we will attempt to retain any favorable fee structure that you negotiated under your original Marketing Agreement for future placements that you make in your Territory. If you place a client with a Senior Community in your Territory that we manage as part of a Regional account, we reserve the right to treat the placement as part of our Referral Program and share the placement fee with you in accordance with the Referral Program fee sharing provisions. However, as of the Effective Date, our current practice in these situations is to either collect the placement fee and remit 100% of the proceeds to you or allow you to directly invoice and collect

- the placement fee depending on the preference of the Senior Community (subject to our right to charge you the referral fee pursuant to <u>Section 6.6</u> if the client was referred to you as a Corporate Lead).
- (ii) Limit the scope of your Marketing Agreement to the Senior Community or Senior Communities that are located within your Territory and allow other ALL Businesses to negotiate and enter into separate Marketing Agreements for the Senior Communities that are located in their territories.
- (iii) Authorize you to retain the Marketing Agreement and manage the relationship as a Regional Account, in which case any placements made at Senior Communities that are part of the Regional Account but located outside of your Territory would be governed by the terms of our Referral Program. At any time, we may revoke our authorization for you to manage the Regional Account and either: (a) transition the Regional Account to us (in accordance with clause (i) above); or (b) require separate Marketing Agreements with each ALL Business that owns a territory in which one (1) or more of the Senior Communities are located (in accordance with clause (ii) above).
- (b) Referral Program. Other ALL Businesses may place clients with Senior Communities located in your Territory through our Referral Program described in Section 11.11. You and all other ALL Businesses must cooperate with each other relating to placements made through the Referral Program. If another ALL Business seeks to place a client at a Senior Community in your Territory and elects to turn the client over to you, we may provide the placement services if you are unable or refuse to provide the placement services. If this occurs: (i) you must collect the placement fee from the Senior Community and immediately remit the entire fee to us: (ii) we will share a portion of the placement fee with the other ALL Business that initially established the client relationship; and (iii) you will not be entitled to any portion of the placement fee.
- (c) <u>Corporate Leads</u>. If we refer a Corporate Lead to you and you choose not to accept the Corporate Lead or you fail to contact or service the Corporate Lead in the manner we require, we may provide, or authorize another ALL Business to provide, placement services for the Corporate Lead, including for placements made at Senior Communities located within your Territory, and you will not be entitled to any portion of the placement fee. If this occurs, we may require that you collect the placement fee from the Senior Community and immediately remit the entire fee to us.
- (d) Acquisitions. We and our affiliates reserve the right to engage in an Acquisition, even if a competitive business of the acquired or acquiring company is located within your Territory and convert the same to an ALL Business operating under our Marks. Any such acquired or converted ALL Business may be located within your Territory and engage in any and all activities within your Territory that you are authorized to conduct, including, without limitation, engaging in Targeted Marketing within your territory, entering into Marketing Agreements with Senior Communities in your Territory and placing clients with Senior Communities in your Territory. You will not be entitled to any compensation for placements made, or other goods or services sold, by any such acquired or converted ALL Business within your Territory, except that you will receive a split of any placement fee involving a placement made with a Senior Community with whom you have entered into a Marketing Agreement (any such placement would be governed by the terms of our Referral Program). Additionally, we and our affiliates reserve the right to merge with, acquire or be acquired by any business of any kind under other systems and/or other marks, which businesses may offer, sell, operate or distribute and/or license others to offer, sell, operate and distribute goods and services the same as, similar to, or different from those offered by your ALL Business inside or outside your Territory.

- (e) Additional Franchises. We reserve the right to develop and operate and grant rights, franchises and licenses to others to develop and operate ALL Businesses under the Marks on any terms and conditions that we in our sole discretion deem appropriate provided that we comply with the territorial protections set forth in Section 3.1. Additionally, we and our affiliates reserve the right to establish, own or operate, and grant others the right to establish, own or operate a business offering the same or similar services as those offered through an ALL Business, or different services under different trademarks (other than the Marks) inside and outside the Territory.
- (f) <u>Alternative Channels of Distribution</u>. We and our affiliates reserve the right to sell or grant franchises or licenses to third parties to sell, competitive or identical goods or services in your Territory, under the Marks or under different trademarks, through alternative channels of distribution, such as through the Internet, catalog sales, telemarketing or other direct marketing. You are not entitled to any compensation for sales that take place through alternative channels of distribution.
- **3.3.** Extra-Territorial Operations. You may not operate your Business, or conduct certain activities related to your Business, outside of your Territory, except as expressly permitted by this Agreement.
 - (a) Marketing Agreements. You may not enter into Marketing Agreements with Senior Communities that are located outside of your Territory without our prior written approval. If we allow you to do so, you must agree to any conditions that we impose and our approval will only be for adjacent territories that are not owned by another ALL Business. We may revoke our authorization at any time in our sole discretion. If you sign a Marketing Agreement with a Senior Community located in an area outside of your Territory and that area subsequently becomes part of a territory owned by another ALL Business, then (i) you must assign the Marketing Agreement to such ALL Business; and (ii) and you may not engage in any further placements involving that Senior Community following the date that the new ALL Business commences operations in such territory (which is deemed to be the date of completion of initial training for new franchisees) except under the terms of our Referral Program. You must cooperate and help facilitate the transition of the Senior Community to the ALL Business that acquired the territory.
 - (b) Targeted Marketing. You may not engage in any Targeted Marketing within a territory owned by another ALL Business without the prior written approval of the territory owner. You may engage in Targeted Marketing to solicit clients, Senior Communities and Referral Sources within an area that is outside your Territory (an "Expansion Territory") only if: (i) the Expansion Territory is adjacent to your Territory; (ii) the Expansion Territory is not included within a territory owned by another ALL Business (either in whole or in part); and (iii) you are reasonably capable of consistently providing placement services in the Expansion Territory in accordance with our standards, policies and procedures. You will have no territorial rights or protections whatsoever with respect to an Expansion Territory. We retain sole discretion to revoke your authorization to conduct Targeted Marketing within an Expansion Territory at any time for any reason (or for no reason). If we revoke our authorization, you must: (i) immediately discontinue all Targeted Marketing within the Expansion Territory; and (ii) upon our request, assign any Marketing Agreements you entered into with Senior Communities located in the Expansion Territory either to us or to another ALL Business that we designate.
 - (c) <u>Placements</u>. You may not place a client with any Senior Community that is located in a territory owned by another ALL Business except in compliance with our Referral Program.

Clients. You may contract with a client that resides in a territory owned by another ALL Business as long as you do not acquire the client as a result of Targeted Marketing that you conduct in that territory. For example, you may contract with a client that resides in a territory owned by another ALL Business if: (i) you solicit the client or the client's family within your Territory; (ii) the client is specifically referred to you by one of your other clients or a third person not associated with you; (iii) we refer the client to you; (iv) another ALL Business refers the client to you; (v) you obtain the client while located within the other territory as long as the client is obtained through incidental activities and not through Targeted Marketing; or (vi) the client is a Corporate Lead and we authorize you to service the client because the territory owner declined to service the Corporate Lead or failed to contact or service the Corporate Lead in the time or manner that we required.

4. TERM AND RENEWAL.

- 4.1. Generally. The term of this Agreement will begin on the Effective Date and expire 10 years thereafter (the "Term"). If this Agreement is the initial franchise agreement for your Business, you may enter into a maximum of four (4) successor franchise agreements (each, a "Successor Agreement") as long as you meet the conditions for renewal specified below. The Successor Agreement shall be the current form of franchise agreement that we use in granting Assisted Living Locators franchises as of the expiration of the Term or renewal term, as applicable. The terms and conditions of the Successor Agreement may vary materially and substantially from the terms and conditions of this Agreement. Each renewal term will be five (5) years, for a maximum total term of 30 years. You will have no further right to operate your Business following the expiration of the final renewal term unless we grant you another franchise or agree to further renewals in our sole discretion. If this Agreement is a Successor Agreement, the renewal provisions in your original franchise agreement will dictate the length of the Term of this Agreement as well as your remaining renewal rights, if any.
- **4.2. Renewal Requirements.** In order to enter into a Successor Agreement, you and the Owners (as applicable) must:
 - (i) notify us in writing of your desire to enter into a Successor Agreement not less than 210 days nor more than 270 days before the expiration of the Term or renewal term, as applicable;
 - (ii) not be in default under this Agreement or any other agreement with us or any affiliate of ours at the time you send the renewal notice or the time you sign the Successor Agreement;
 - (iii) sign the Successor Agreement and all ancillary documents that we require franchisees to sign;
 - (iv) sign a General Release;
 - (v) pay us a \$7,500 renewal fee; and
 - (vi) take any additional action that we reasonably require. You may not renew if we have notified you that we have withdrawn from the market in which your Territory is located and no longer operate, or grant others the right to operate, ALL Businesses in such market.
- **4.3. Interim Term.** If you do not sign a Successor Agreement after the expiration of the Term and you continue to accept the benefits of this Agreement, then at our option, this Agreement may be treated either as: (a) expired as of the date of the expiration with you then operating without a franchise to do so and in violation of our rights; or (b) continued on a month-to-month basis (the "Interim Term") until either party provides the other party with 30 days' prior written notice of the party's intention to terminate the Interim Term. In the latter case, all of your obligations will

remain in full force and effect during the Interim Term as if this Agreement had not expired, and all obligations and restrictions imposed on you upon the expiration or termination of this Agreement will be deemed to take effect upon the termination of the Interim Term.

Except as otherwise permitted by this <u>Section 4</u>, you have no right to continue to operate your Business following the expiration of the Term.

5. TRAINING AND CONFERENCES

- **5.1. Initial Training Program.** The Managing Owner and all of your high-level managers must attend and successfully complete our initial training program within 60 days after the Effective Date. Your other Owners and employees may not attend initial training without our prior permission. The Managing Owner must successfully complete our initial training program on either the first or second attempt (and in each case within 60 days after the Effective Date). The business set-up fee that you pay pursuant to Section 14.2 covers, among other things, the costs of the pre-opening initial training program. However, you must pay us an additional fee of \$1,500 per person for any person that must retake initial training after failing the first attempt. This fee is due prior to training.
- **Initial Training For New Owners/Managers.** If you hire a new high-level manager or appoint a new Managing Owner after we conduct our pre-opening initial training program, the new manager or Managing Owner, as applicable, must attend and successfully complete our then-current initial training program. You must pay us a fee of \$1,500 per person for: (a) any new Managing Owner or manager that attends our initial training program after you open; and (b) any person that retakes initial training after failing the first attempt. This fee is due prior to training.
- **Periodic Training.** We may offer periodic refresher or additional training courses for your Managing Owner and high-level managers. Attendance at these training programs is mandatory. We will not require attendance at more than one (1) periodic training course during any 12-month period. We currently do not charge an additional fee for this training but reserve the right to do so during the Term.
- **Onsite Training and Assistance.** Upon your written request, we may provide onsite training or assistance in your Territory. The business set-up fee that you pay pursuant to Section 14.2 covers, among other things, the cost of the initial two (2) days of onsite training we provide within the first 12 months after you open (this two-day onsite training is only provided for the first (1st) franchised territory that you purchase and we reserve the right, in our sole discretion, to (a) substitute a different training or coaching program if we feel it would be more appropriate for your success and/or (b) postpone this training to take place at a date that is more than 12 months after you open your Business). If we provide any additional onsite training or assistance, you must pay us a fee of \$2,500 per visit. The \$2,500 fee covers a two-day visit. If we agree to visit your Territory for more than two (2) days as part of a single visit, we may charge you an additional \$1,250 for each additional day that we visit. This fee is due 10 days after invoicing.
- **Additional Training Upon Request.** Upon your written request, we may, but need not, provide additional assistance or training beyond that which we are required to provide under this Agreement. If we provide such additional training or support, we may charge you an additional hourly fee of (currently \$125 per hour, subject to increase up to 10% per year). This fee is due 10 days after invoicing. Any additional training or support that you request that is provided onsite in your Territory will be subject to our then current onsite training fee, currently \$2,500, described in Section 5.4 (currently \$2,500, subject to increase by up to 10% per year) instead of the hourly fee described in this Section.
- **5.6.** Certifications. You agree to obtain and maintain all certifications that we require from time to time that are relevant to your Business. You and any of your personnel we designate must

maintain all certifications that we require throughout the Term and you are responsible for any costs associated with ongoing recertification training.

- business issues and operational and general business concerns affecting matters relevant to ALL Businesses. Attendance at these conferences is mandatory. We may charge you a conference registration fee of \$7501,000 per conference for two (2) attendees plus an additional \$350 for each additional attendee. If you fail to attend a required conference, we may charge you a \$1,000 non attendance fee and make available to you any written conference materials distributed to conference attendees. The conference registration fee is due 60 days prior to the conference or as otherwise invoiced by us. Attendance is mandatory unless: (a) we designate attendance as optional; or (b) we waive your obligation to attend based on showing of good cause. In order to request a waiver, you must complete our prescribed form of Non-Attendance Request and submit the completed form to us for our review and consideration. If you fail to attend a required conference without an approved waiver from us, you must pay us a \$1,000 non-attendance fee.
- **Expenses.** You are responsible for all expenses and ad costs that your trainees incur for training or attending conferences, including wages, travel and living expenses. We do not pay for or cover any of these costs. If you transfer your franchise, we may charge the new franchisee our then-current initial training fee.

6. OTHER FRANCHISOR ASSISTANCE.

- **Manual.** During the Term, we will provide you with access to our confidential Brand Standards Manual (the "Manual") in text or electronic form. The Manual will help you establish and operate your Business. The information in the Manual is confidential and proprietary and may not be disclosed to third parties without our prior approval.
- 6.2. **ALL-IN**. We will license you the right to utilize Assisted Living Locators Integrated Network ("ALL-IN"), which is our proprietary business management database. You and your managers must execute our current form of ALL-IN User Agreement and Privacy Policy, which we may change from time to time. You must utilize ALL-IN to perform the specific functions described in the Manual, such as: managing and storing data regarding your clients, Senior Communities, Referral Sources and Senior Care Providers; invoicing Senior Communities for placements; and tracking and reporting client leads, placements and Gross Collected Revenues Revenue. You must follow all policies and procedures in the Manual with respect to the use of ALL-IN and your entry of data. You must pay us: (a) a nonrefundable \$495 initial setup fee for ALL-IN upon execution of this Agreement; and (b) a nonrefundable monthly licensing fee of \$199250 per month (covers up to four (4) users) plus an additional \$199250 per month if you have more than four (4) users (covers a maximum of four (4) additional users). We reserve the right to increase the monthly licensing fee on an annual basis, (not to exceed \$400 per month per block of four (4) users). The monthly license fee is due on the 1st day of each month. The initial setup fee and monthly license fee are deemed part of the technology fee imposed under Section 11.15(d).
- **Website.** We will have established and maintain our primary corporate website for our brand (www.assistedlivinglocators.com). As part of this website, we will include a link to a local landing page that will include the information about your Business that we deem appropriate. We will set up your landing page as part of the services we provide in exchange for the business set-up fee. We may modify the content of and/or discontinue our corporate website (including your local landing page, if applicable) at any time in our sole discretion. We will control all content on your landing page, if applicable. We may terminate or suspend your local landing page at any time without prior notice if you are in default under this Agreement, the Manual or the Franchisee Code of Conduct. At all times, we will own the website (including your landing page) and domain name.

- 6.4. Email Addresses and Telephone Number. You must license and use Google Workspace. As part of your monthly technology fee, you must pay us \$2527 per month per Google Workspace account. We may increase this fee upon 30 days' prior notice. As part of the Google Workspace platform, you will receive Assisted Living Locators email addresses for use with your Business. You must exclusively use these email addresses for all communications with us, clients, Senior Communities, Referral Sources, Senior Care Providers, suppliers and other persons relating to your Business. You may not use any other email address for any purpose unrelated to your Business. You must save all emails on the Assisted Living Locators server that are sent to any email address that we provide to you. We may, but need not, provide you with a telephone number and account at no additional charge. You must exclusively list the telephone number that we provide to you as the primary phone number for your Business. If we do not provide you with a telephone number, you must notify us of the telephone number you will use for your Business. You may only list the telephone number we provide to you (or the telephone number that you designate to us as your dedicated telephone number for your Business) on all digital and print marketing materials. We will own your Business telephone number and account but you must directly pay the associated monthly and periodic telephone charges. If you list any phone number, other than your dedicated Business telephone number, in any digital or print marketing materials, then we will also own that telephone number and account. We will own the email addresses, telephone number(s), and the associated accounts, but will allow you to use them during the Term solely in connection with the operation of your Business.
- days to provide onsite training and assistance. As part of this informal training program, our representative will monitor and evaluate your operations, provide you with suggestions for improvement and answer questions from you and your staff. The onsite training program will take place within the first 12 months after you open your Business. Our costs and fees for this training are included in the business set-up fee paid by you pursuant to Section 14.2. If we provide any additional onsite training or assistance, we may charge you the onsite training and assistance fee in accordance with Section 5.4. In our discretion, we may: (a) substitute any method of training we deem appropriate in place of the two (2) days of "onsite training" referenced in this Section (for example, we may substitute a virtual training program); and/or (b) postpone the onsite training (or other method of training we deemed appropriate) referenced in this Section to a later date we deem appropriate, including a date that is more than 12 months after you open your Business.
- Corporate Leads Program. We have established and administer a program referred to as our 6.6. "Corporate Leads Program." Under our Corporate Leads Program, we will: (a) attempt to establish relationships with Referral Sources; and (b) develop digital marketing programs or platforms that are designed to generate leads. When you obtain a lead through our Corporate Leads Program, you will enter (or we will enter for you) the lead into ALL-IN. The first correct instance of the lead appearing in the ALL-IN platform will determine ownership of the lead and the terms associated with the placement. If we refer a lead to you before you or another franchisee correctly enters the same lead into ALL-IN, or the lead is sent by a Referral Source with whom we have entered into a referral agreement, then the lead is deemed a "Corporate Lead." If we refer a Corporate Lead to you, you have the option, but not the obligation, to accept the Corporate Lead for purposes of providing placement services. If you place a Corporate Lead that we refer to you, you must pay us a referral fee in the amount that we specify from time to time. We may designate different referral fees for different types of Corporate Leads. For example, some referral fees may consist of a flat fee while others may be calculated as a percentage of the placement fee you collect. In no event will the referral fee exceed 30% of the Gross Collected Referral Proceeds that you collect in connection with the Corporate Lead. We reserve the right to collect the referral fee on all Gross Collected Referral Proceeds that you generate from the Corporate Lead, including Gross Collected Referral Proceeds that consist of referral fees that you receive from Senior Service Providers to whom you refer the Corporate Lead. Referral fees are due on each royalty payment due date with

respect to Gross Collected Referral Proceeds collected by you during the immediately preceding month. Corporate Leads do not include (and you do not pay us a referral fee relating to) any lead that is generated through:

- (i) our website (www.assistedlivinglocators.com), including web leads (unless funded and controlled via a stand-alone landing page designated to support corporate funded web leads) and leads that call our corporate telephone number;
- (ii) our call center;
- (iii) any digital program or platform paid for by the brand and system development fund; or
- (iv) any other digital program or platform we designate as not being part of our Corporate Leads Program.

If we send you a Corporate Lead and you place the Corporate Lead under our Referral Program, our referral fee will come "off the top" and you and the other ALL Business will split the remainder of the placement fee in accordance with terms of our Referral Program. We retain complete discretion in determining whether to refer a Corporate Lead and the franchisee to whom we refer a Corporate Lead. We do not represent or guaranty that we will refer any Corporate Leads to you.

- **6.7. Other Leads.** From time to time, we may, but need not, refer leads to you that contact us through:
 - (i) our website (www.assistedlivinglocators.com), including web leads (unless funded and controlled via a stand-alone landing page designated to support corporate funded web leads) and leads that call our corporate telephone number;
 - (ii) our call center;
 - (iii) any digital program or platform paid for by the brand and system development fund; or
 - (iv) any other digital program or platform we designate as not being part of our Corporate Leads Program.

We have sole discretion in determining the franchisee to whom we will refer these leads. If we refer a lead to you pursuant to this Section (i.e., a lead that is not a Corporate Lead), you are required to provide placement services for the lead but we will not charge you the referral fee. We retain complete discretion in determining whether to refer a lead and the franchisee to whom we refer a lead. We do not represent or guaranty that we will refer any leads to you.

- 6.8. Guidance, Consultations and Support. Based upon our period evaluations of your Business and reports that you submit to us, we will provide our guidance and recommendations on ways to improve the marketing and/or operation of your Business. We will also provide you with access to: (a) one or more of our representatives to answer general questions you have regarding the operation of your Business; and (b) one or more registered nurses for consultations regarding case assessments and recommendations. We may limit the days and times during which these representatives and registered nurses will be available to you. You understand that these individuals may have limited availability due to prior commitments or high demand for their services. We reserve the right, at any time, to limit the number of hours each month that you may speak with our representatives and/or registered nurses. We also reserve the right to discontinue providing you with access to registered nurses at any time. The guidance, consultations and support described in this Section will be provided remotely via email, conference calls, video calls, or in any other manner we periodically designate.
- **6.9.** Marketing Assistance. As further described in Section 10.1 and Section 10.2, we will administer the brand and system development fund and provide you with other marketing assistance during the Term.

- **Purchase Agreements.** We may, but need not, negotiate purchase agreements with suppliers to obtain discounted prices for us and our franchisees. If we succeed in negotiating a purchase agreement, we will arrange for you to be able to purchase the goods directly from the supplier at the discounted prices that we negotiate (subject to any rebates the supplier pays to us). We may also purchase certain items from suppliers in bulk and resell them to you at our cost plus shipping fees and a reasonable markup.
- **6.11.** Call Center. You are required to utilize a third-party call center for purposes of answering client calls and providing related client service and support, including outside of normal business hours ("Inbound Call Services"). We may also require you to utilize a third-party call center for purposes of client growth and engagement, including reaching out to develop leads and maintain client relationships ("Outbound Call Services"). You may utilize the third-party call center we designate for Inbound Call Services and/or Outbound Call Services. You must pay the monthly fees imposed by the company operating the call center. At any time, we may discontinue the call center or we may begin operating the call center ourselves, in which case we may charge you commercially reasonable fees for administering the call center. (not to exceed \$500 per month).
- **6.12.** Newsletters. From time to time, we may, but need not, prepare newsletters for distribution to Assisted Living Locators clients.

7. ESTABLISHING YOUR BUSINESS

- **Office.** An Assisted Living Locators business is a home-based business. You are not required to lease or purchase a separate commercial office in connection with your Business. You may relocate your home-office at any time upon written notice to us. If you do plan to lease commercial office space and request our help identifying a suitable location, we will provide you with telephonic guidance regarding site selection. Any commercial office you establish must be located in your Territory and must present a professional appearance. You may relocate your commercial office anywhere within your Territory as long as you provide us with at least 30 days prior written notice. If you operate from a home office and your home is not located in your Territory, you must provide a physical business address in your Territory. -This address must be a physical location, must not be a PO Box or mailbox in a remote location, and must be precise to ensure customers can locate the business.
 - 7.2. <u>Vehicle</u>. Most of your business activities will be conducted from your vehicle. As such, you must own, lease or have regular access to a late-model vehicle that can reasonably accommodate at least two (2) passengers. The vehicle must be kept in good condition and regularly cleaned. If you choose to install "wraps" on your vehicle, your wraps must comply with our standards and specifications and must be purchased from an approved or designated supplier. Because safety is of the utmost importance to our brand, we may require that you attend a driver's safety course of our choosing at your cost if we receive complaints regarding unsafe driving by you or your employees. At all times, the Managing Owner (and each other person associated with your Business that drives your clients) must hold a valid driver's license.
 - **7.3.** Opening. You must open your Business within 90 days after the Effective Date. You may not open your Business before:
 - (i) the Managing Owner successfully completes the initial training program;
 - (ii) you purchase all required insurance and provide us with evidence of coverage;
 - (iii) you obtain all required licenses, permits and other governmental approvals;
 - (iv) you obtain any required bonding;
 - (v) you obtain access to a suitable vehicle to be used with your Business.

Upon successful completion of initial training by the Managing Owner, we will issue to you a Certificate of Completion of Training, at which point your Business will be open. You must fulfil all of your other pre-opening obligations prior to completion of initial training. BY VIRTUE OF OPENING YOUR BUSINESS, YOU ACKNOWLEDGE THAT WE HAVE FULFILLED ALL OF OUR PRE-OPENING OBLIGATIONS TO YOU.

8. MANAGEMENT AND STAFFING.

- 8.1. Owner Participation. You acknowledge that a major requirement for the success of your Business is the active, continuing, and substantial personal involvement and hands-on supervision by your Managing Owner. The Managing Owner must at all times be actively involved in the operation of the Business on a full time basis. Any new Managing Owner that we approve must successfully complete the initial training program. At all times during the Term, the Managing Owner must hold at least a 20% ownership interest in the franchise or the Entity that is the franchisee under this Agreement, as applicable.
- **Managers.** You may hire one (1) or more managers to assist the Managing Owner with the daily management and supervision of your Business, but only if: (a) the manager successfully completes the initial training program; (b) the manager signs a **Brand ProtectionConfidentiality** Agreement; and (c) the Managing Owner agrees to assume responsibility for the daily management and supervision of your Business if the manager is unable to perform his or her duties due to death, disability, termination of employment, or for any other reason, until such time that you obtain a suitable replacement manager. At all times, the Managing Owner must dedicate commercially reasonable efforts to monitoring and supervising the activities of each manager to ensure the Business is managed and operated in accordance with this Agreement and the Manual. The Managing Owner must remain actively involved with the management and supervision of the Business on a full-time basis even if you hire managers.
- **8.3. Employees.** You must determine appropriate staffing levels for your Business to ensure full compliance with this Agreement and our system standards. You may hire, train and supervise employees to assist you with the proper operation of the Business. You must pay all wages, commissions, fringe benefits, worker's compensation premiums and payroll taxes (and other withholdings required by law) due for your employees. These employees will be employees of yours and not of ours. We do not control the day to day activities of your employees or the manner in which they perform their assigned tasks. You must inform your employees that you exclusively supervise their activities and dictate the manner in which they perform their assigned tasks. In this regard, you must use your legal business entity name (not our Marks or a fictitious name) on all employee applications, paystubs, pay checks, employment agreements, time cards, and similar items. We also do not control the hiring or firing of your employees. You have sole responsibility and authority for all employment related decisions, including employee selection and promotion, hours worked, rates of pay and other benefits, work assignments, training and working conditions. We will not provide you any advice or guidance on these matters. However, because the safety and wellbeing of our clients is of the utmost concern, you must conduct a background check on your employees before they may have any involvement with the Business. You must ensure that the results of the background check do not indicate that the employee presents a significant risk to the safety or wellbeing of your clients. You must require that your employees review and sign the acknowledgment form we prescribe that explains the nature of the franchise relationship and notifies the employee that you are his or her sole employer. You must also post a conspicuous notice for employees in the back-ofthe-house area explaining your franchise relationship with us and that you (and not we) are the employee's sole employer. We may prescribe the form and content of this notice.
- **8.4.** <u>Interim Manager</u>. We have the right, but not the obligation, to designate an individual of our choosing (an "<u>Interim Manager</u>") to manage your Business if either: (a) your Managing Owner ceases to perform the responsibilities of a Managing Owner (whether due to retirement, death, disability, or for any other reason) and you fail to find an adequate replacement Managing

Owner within 30 days; or (b) you are in material breach. The Interim Manager will cease to manage your Business at such time that you hire an adequate replacement Managing Owner who has completed training or you cure the material breach, as applicable. The Interim Manager will have no liability to you except for gross negligence or willful misconduct. We will have no liability to you for the activities of an Interim Manager unless we are grossly negligent in appointing the Interim Manager.

9. FRANCHISEE AS ENTITY. If you are an Entity, you represent that Part A of <u>ATTACHMENT "A"</u> includes a complete and accurate list of all of your Owners. Upon our request, you must provide us with a resolution of the Entity authorizing the execution of this Agreement, a copy of the Entity's organizational documents and a current Certificate of Good Standing (or the functional equivalent thereof). You represent that the Entity is duly formed and validly existing under the laws of the state of its formation or incorporation. If you are an Entity, all Owners (whether direct or indirect) and their spouses must sign a Franchise Owner Agreement, the current form of which is attached as ATTACHMENT "C".

10. ADVERTISING & MARKETING.

10.1. Brand and System Development Fund.

- (a) <u>Administration</u>. We have established and maintain a brand and system development fund to promote public awareness of our brand and to improve our System. We may use the fund to pay for any of the following in our sole discretion:
 - (i) developing, maintaining, administering, directing, preparing, or reviewing advertising and marketing materials, promotions and programs;
 - (ii) conducting and administering promotions, contests or giveaways;
 - (iii) improving public awareness of the Marks;
 - (iv) public and consumer relations and publicity;
 - (v) brand development;
 - (vi) sponsorships;
 - (vii) charitable and non-profit donations and events;
 - (viii) research and development of technology, products and services;
 - (ix) website development and search engine optimization;
 - (x) development of an ecommerce platform;
 - (xi) development and implementation of quality control programs, including the use of mystery shoppers or customer satisfaction surveys;
 - (xii) conducting market research;
 - (xiii) changes and improvements to the System;
 - (xiv) the fees and expenses of any advertising agency we engage to assist in producing or conducting advertising or marketing efforts;
 - (xv) collecting and accounting for contributions to the fund;
 - (xvi) preparing and distributing financial accountings of the fund;
 - (xvii) any other programs or activities that we deem appropriate to promote or improve the System; and

(xviii) direct or indirect labor, administrative, overhead and other expenses incurred by us and/or our affiliates in relation to any of these activities, including salary, benefits and other compensation of any of our (and any of our affiliate's) officers, directors, employees or independent contractors based upon time spent working on any brand fund matters described above.

We have sole discretion in determining the content, concepts, materials, media, endorsements, frequency, placement, location and all other matters pertaining to any of the foregoing activities. Any surplus of monies in the fund may be invested and we may lend money to the fund if there is a deficit. The fund is not a trust and we have no fiduciary obligations to you with respect to our administration of the fund. A financial accounting of the operations of the fund, including deposits into and disbursements from the fund, will be prepared annually and made available to you upon request. In terms of marketing activities paid for by the fund, we do not ensure that these expenditures in or affecting any geographic area are proportionate or equivalent to the fund contributions by franchisees operating in that geographic area or that any franchisee benefits directly or in proportion to their fund contributions. We may discontinue the fund at any time in our discretion upon at least 30 days' prior notice. We may spend in any calendar year more or less than the total contributions to the fund in that year and may cause the fund to invest any surplus for future use by the fund. We may borrow from ourselves, our affiliates or other lenders on behalf of the fund to cover deficits of the fund, and such amounts shall be treated as a loan. We will keep records of any such borrowing described above.

(b) <u>Contributions</u>. On each royalty payment due date, you must pay us a brand and system development fund contribution equal to the greater of: (i) 2% of Gross Collected Revenues Revenue; or (ii) the minimum brand fund contribution. The minimum brand fund contribution is currently \$300 per month, but we may increase the minimum contribution (up to a maximum of \$500 per month) upon 30 days' prior notice to you. If you own multiple franchise territories, we may apply the minimum brand fund contribution on a "per territory" basis, such that you must pay a separate minimum brand fund contribution for each territory that you own (if less than 2% of Gross Collected Revenues Revenue from the applicable territory). We will deposit into the fund all fund contributions paid by you and other franchisees and all fines that we collect from franchisees. The brand and system development fund contribution is in addition to, and not credited towards, the marketing fees you must pay us pursuant to Section 10.2.

10.2. Marketing Assistance From Us.

- (a) <u>Initial Marketing Materials and Support</u>. In partial consideration for the business set-up fee paid pursuant to <u>Section 14.2</u> as well as the \$3,000 initial marketing fee paid pursuant to Section 10.2(b) we will:
 - (i) provide you with an initial supply of marketing materials (estimated to be a 150 to 180 day supply) before you open your Business;
 - (ii) conduct certain activities to market and promote the opening of your Business and assist you with the development and launch of your Business, such as preparing lists of Senior Communities and Referral Sources for you to contact, issuing press releases and developing a marketing plan for your Territory;
 - (iii) set up your local landing page to promote your Business; and
 - (iv) set up your social media platforms.

- (b) Ongoing Marketing Support. In addition to your required contribution to the brand and system development fund, you must pay us a marketing fee that we will use to advertise and promote your Business. You must pay us an initial marketing fee of \$3,000 upon execution of this Agreement, which we will use to market and promote your Business during the 60-day period after execution of this Agreement. Commencing with the second (2nd) full month after your opening date, you must begin paying us a monthly marketing fee, which will be due on the first (1st) day of each month. The amount of the monthly marketing fee is: (i) \$1,000 per month for months two (2) through month five (5) after your opening date; and (ii) \$500 per month thereafter for the remainder of the Term. At our option, we may either directly perform the marketing and advertising services ourselves or we may outsource these functions to one (1) or more third-party suppliers of our choosing. Commencing with month six (6) after your opening date, you must directly spend the marketing fee with third-party suppliers of advertising and marketing materials and services rather than pay the marketing fee to us. If you fail to spend the required amount, we may require that you immediately pay us the deficiency amount which we will use to promote your Business. We may increase the marketing fee upon at least 30 days' prior notice, but any such increase will not exceed 10% in any 12-month period. At any time following written notice to you (including prior to month six (6) after opening), we may require that you spend the required marketing fee directly with third-party suppliers on marketing and advertising that we approve (rather than paying the marketing fee to us).
- (c) Other Marketing Support. We may create and make available to you additional advertising and other marketing materials for your purchase. We may use the brand and system development fund or the marketing fee to pay for the creation and distribution of these materials, in which case there will be no additional charge. We may make these materials available over the Internet (in which case you must arrange for printing the materials and paying all printing costs). Alternatively, we may enter into relationships with third-party suppliers who will create the advertising or marketing materials for your purchase. We will provide reasonable marketing consulting, guidance and support throughout the Term on an as-needed basis.

10.3. Your Marketing Activities.

- (a) <u>Generally</u>. We must approve all of your advertising in accordance with <u>Section 10.3(c)</u>. You agree to participate at your own expense in all advertising, promotional and marketing programs that we require.
- (b) <u>Standards for Advertising</u>. All advertisements and promotions that you create or use must be completely factual and conform to the highest standards of ethical advertising and comply with all federal, state and local laws. You must ensure that your advertisements and promotional materials do not infringe upon the intellectual property rights of others. All of your advertising and marketing must be primarily directed towards clients, Senior Communities and Referral Sources in your Territory. You may not engage in Targeted Marketing outside of your Territory except as otherwise provided in Section 3.3(b).
- (c) <u>Approval of Advertising</u>. Before you use them, we must approve all advertising and promotional materials that we did not prepare or previously approve (including materials that we prepared or approved and you modify). We will be deemed to have disapproved the materials if we fail to issue our approval within 15 days after receipt. You may not use any advertising or promotional materials that we have disapproved (including materials that we previously approved and later disapprove). Any electronic or digital marketing that you conduct must be conducted through an approved supplier or must be reviewed and approved by us in advance.

- (d) <u>Social Media</u>. You may advertise and market your Business using social media, provided that:
 - (i) you only utilize social media platforms we approve;
 - (ii) you strictly comply with our social media policy;
 - (iii) you immediately remove any post we disapprove (even if the post complies with our social media policy);
 - (iv) you utilize any supplier we designate for social media marketing services; and
 - (v) we will own the social media account but allow you to use it during the Term.
- (e) <u>Internet and Websites</u>. Without our prior approval, which we may withhold in our sole discretion, you may not:
 - (i) establish a Google My Business account, or similar permanent web accounts, utilizing any of our Marks (including in conjunction with your Territory name) (if we permit this, we must be named an administrator on the account);
 - (ii) develop, host, create or otherwise maintain a website or other online or digital presence in connection with your Business (other than the landing page we provide), including any website bearing our Marks; or
 - (iii) utilizing the Internet to conduct digital or online advertising or engage in ecommerce (except as electronic or digital marketing permitted by <u>Section</u> 10.3(c)).

11. OPERATING STANDARDS.

- **11.1.** Generally. You agree to operate your Business: (a) in a manner that will promote the goodwill of the Marks; and (b) in full compliance with all of our standards and the terms of this Agreement and the Manual.
- **11.2. Brand Standards Manual**. You agree to establish and operate your Business in accordance with the Manual. The Manual may contain, among other things:
 - (i) a description of the authorized goods and services you may offer and sell;
 - (ii) specifications, operating procedures and quality standards for products, services and procedures that we prescribe from time to time for ALL Businesses
 - (iii) reporting and insurance requirements;
 - (iv) policies and procedures pertaining to our Corporate Leads Program, Referral Program and any other programs we establish in the future;
 - (v) policies and procedures pertaining to permissible extra-territorial marketing and business activities;
 - (vi) policies and procedures for collaboration and cooperation between ALL Businesses;
 - (vii) policies and procedures relating to data ownership, protection, sharing and use; and
 - (viii) a list of (a) the goods and services (or specifications for goods and services) you must purchase to develop and operate your Business and (b) any designated or approved suppliers for these goods and services.

The Manual is designed to establish and protect our brand standards and the uniformity and quality of the goods and services offered by ALL Businesses. We can modify the Manual at any time. The modifications will become binding at the time we notify you of the modification

- (subject to any "grace period" we provide to implement the change). All mandatory provisions contained in the Manual (whether they are included now or in the future) are binding on you.
- 11.3. Authorized Goods and Services. You agree to offer all goods and services that we require from time to time in our commercially reasonable discretion. You may not offer any other goods or services in connection with your Business without our prior written permission. You may not use your Business or permit your Business to be used for any purpose other than offering the goods and services that we authorize. We may, without obligation to do so, add, modify or delete authorized goods and services, and you must do the same upon notice from us. Our addition, modification or deletion of one or more goods or services shall not constitute a termination of the franchise or this Agreement.
- 11.4. Clients. You may obtain clients from your own marketing efforts, Referral Sources and from our marketing efforts. You may, but need not, enter into a written contract with each client (a "Client Agreement"). We will provide you with a sample Client Agreement that you may customize for your use. You must conduct an assessment of each client's unique situation and needs utilizing the client assessment processes and tools that we provide in order to identify Senior Communities that may be a good match for their needs. You must also provide all services that we require to facilitate the placement of each client with the Senior Community that best meets their needs.
- 11.5. <u>Senior Communities</u>. You must identify, inspect, evaluate and monitor Senior Communities that are located within your Territory in the manner described in the Manual. With respect to each Senior Community, you must prepare initial and periodic reports in the format that we prescribe from time to time. You must sign a Referral Network Participation Agreement in the form attached to this Agreement as <u>ATTACHMENT "B"</u> (a "<u>Marketing Agreement</u>") with each Senior Community to whom you may refer potential clients. Upon your request, we will provide you with our suggestions and recommendations for the fees that you charge pursuant to the Marketing Agreements. However, you are free to determine your own fee structure.
- 11.6. <u>Senior Service Providers</u>. We strongly encourage you identify, evaluate and establish referral relationships with reputable Senior Service Providers. Depending upon your client's needs, you shall serve as a resource by referring them, when appropriate, to Senior Service Providers offering goods or services that may be necessary or desirable. We recommend that you sign a referral services agreement (a "<u>Referral Services Agreement</u>") with each Senior Service Provider to whom you refer clients to facilitate payment of referral fees. We will provide you with a sample Referral Services Agreement that you may customize for your use. You may generate additional revenue streams from referral fees paid by Senior Service Providers under Referral Services Agreements, although some Senior Service Providers do not pay referral fees. Senior Service Providers may also refer clients to you in need of placement services.
- 11.7. <u>Referral Sources</u>. We strongly encourage you to identify and establish referral relationships with Referral Sources. If you establish a relationship with a Referral Source, such as an Employee Assistance Program, that refers 10 or more leads per month and at least 20% of the leads involve placements outside of your Territory, we reserve the right to require that you transition the Referral Source to us, in which case we may, but need not, pay you compensation for leads that are placed with Senior Communities located outside of your Territory.
- 11.8. Business Agreements. You must exclusively utilize the form of Marketing Agreement that we prescribe from time to time for purposes of establishing relationships with Senior Communities except to the extent that: (i) a Senior Community requires that you utilize its form of agreement; or (ii) a Senior Community refuses to sign a Marketing Agreement (either as a matter of corporate policy or for any other reason). We must approve any form of Marketing Agreement prescribed by a Senior Community before you sign it. We reserve the right to modify our form of Marketing Agreement from time to time, and you must immediately commence utilization of the modified form after receipt of notice from us. We may also modify our sample Referral

Services Agreement and Client Agreement at any time. Prior to utilization, we strongly encourage you to provide your attorney with a copy of our form of Marketing Agreement (including all future modifications), sample Referral Services Agreement (if you choose to use it) and sample Client Agreement (if you choose to use it) to review for compliance with the laws and regulations applicable in your jurisdiction. You must notify us of all changes to our form of Marketing Agreement that are recommended or required by your attorney, and we must approve such changes prior to implementation. We have not engaged local counsel in each state to review our form agreements for compliance with the laws of such states. We make no representation that our form agreements comply with the laws of any particular state.

- 11.9. Sales Lead Management. If we refer a lead to you (regardless of whether or not the lead is a Corporate Lead), you must contact the lead within the minimum period of time, and in accordance with our standards, specified in the Manual. If you fail to do so, we may pursue and place the lead ourselves (or we may authorize another ALL Business to place the lead) without being deemed to violate your territorial protections and you will not be entitled to any compensation relating to the placement.
- 11.10. Client Assessments and Placements. Your clients will include families in need of senior housing options. You must conduct an assessment of each client's unique situation and needs in compliance with our standards and procedures. Based on the results of the assessment, you will provide your client with information regarding Senior Communities that may be a good match for their needs and facilitate the placement of your client with the Senior Community they select. If the Senior Community is located in a territory operated by another ALL Business, you must coordinate your placement efforts with such other ALL Business in compliance with our Referral Program. You may not charge your clients for placement services without our prior written approval.
- 11.11. Referral Program. From time to time, you may be required to work with another ALL Business in order to place a client with a Senior Community located in the territory assigned to such other ALL Business. We have developed policies and procedures for these situations, referred to as our "Referral Program". Our Referral Program establishes a framework for ALL Businesses to work together in a collaborative manner to achieve the best result for client. Under the Referral Program, you and other ALL Businesses are free to enter into a written agreement identifying a mutually agreed upon fee split. In the absence of a prior written fee split arrangement, the placement fee will be shared according to the then-current fee sharing provisions of our Referral Program. You and all other ALL Businesses must comply with our Referral Program. The Referral Program, and our policies and procedures pertaining to the Referral Program, are set forth in the Manual. We reserve the right to modify the Referral Program from time to time. However, we will not modify the fee split unless 51% of all participating ALL Businesses vote in favor of the modification. A "participating" ALL Business is any ALL Business that actually casts a vote at the time and in the manner that we establish for the vote. Each ALL Business is entitled to one (1) vote, regardless of how many franchises and/or territories are owned and operated by such ALL Business.
- 11.12. <u>Data Entry.</u> If required by the Manual, you must input certain prescribed data and information into ALL-IN regarding each of your clients, Senior Communities, Referral Sources, Senior Care Providers, as well each transaction or placement that you consummate. You must enter the data in the manner and within the period of time prescribed by the Manual. Without limiting the generality of the foregoing, you must enter all new clients into the database no later than the close of business on the next business day regardless of how you obtained the referral and regardless of whether the outcome is predetermined. We will have independent access to all data entered into ALL-IN. All such data shall be owned by us but licensed to you for purposes of operating your Business during the Term.
- **11.13. Suppliers and Purchasing.** You agree to purchase or lease all products, supplies, equipment, services and other items specified in the Manual from time to time. If required by the Manual,

you agree to purchase certain goods and services only from suppliers designated or approved by us (which may include, or be limited exclusively to, us or our affiliate). You acknowledge that our right to specify the suppliers that you may use is necessary and desirable so that we can control the uniformity and quality of goods and services used, sold or distributed in connection with the development and ongoing operation of ALL Businesses, maintain the confidentiality of our trade secrets, obtain discounted prices for our franchisees if we choose to do so, and protect the reputation and goodwill associated with the System and the Marks. If we receive rebates or other financial consideration from these suppliers based upon franchisee purchases, we have no obligation to pass these amounts on to you or to use them for your benefit. If you want us to approve a supplier that you propose, you must send us a written notice specifying the supplier's name and qualifications and provide any additional information that we request. We will approve or reject your request within 30 days after we receive your notice and all additional information (and samples) that we require. We shall be deemed to have rejected your request if we fail to issue our approval within the 30-day period. You must reimburse us for all costs and expenses that we incur in reviewing a proposed supplier within 10 days after invoicing.

11.14. Equipment Maintenance and Changes. You agree to maintain all of your equipment in good condition and promptly replace or repair any equipment that is damaged, worn-out or obsolete. We may require that you change your equipment, which may require you to make additional investments. You acknowledge that our ability to require franchisees to make significant changes to their equipment is critical to our ability to administer and change the System and you agree to comply with any such required change within the time period that we reasonably prescribe.

11.15. Technology Systems.

- (a) Generally. You must acquire and utilize all information and communication technology systems that we specify from time to time, including, without limitation, computer systems, telecommunications systems, security systems, and similar systems, together with the associated hardware, software (including cloud-based software) and related equipment, software applications, mobile apps, and third-party services relating to the establishment, use, maintenance, monitoring, security or improvement of these systems (collectively referred to as the "Technology Systems"). The Technology Systems may relate to matters such as purchasing, pricing, accounting, order entry, inventory control, security, information storage, retrieval and transmission, customer information, customer loyalty, marketing, communications, copying, printing and scanning, or any other business purpose that we deem appropriate. We may require that you, at your expense, acquire new or substitute Technology Systems, and/or replace, upgrade or update existing Technology Systems, upon reasonable prior notice.
- (b) <u>Use and Access</u>. You must utilize your Technology Systems in accordance with the Manual. You may not load or permit any unauthorized programs or games on your Technology Systems. You must ensure that your employees are adequately trained in the use of the Technology Systems. You agree to take all steps necessary to enable us to have independent and unlimited access to the operational data collected through your Technology Systems, including information regarding your Gross Collected Revenues Revenue for purposes of calculating fees owed. Upon our request, you agree to provide us with the user IDs and passwords for your Technology Systems, including upon termination or expiration of this Agreement. Without limiting the generality of the foregoing, you must grant permission for us to link your QuickBooks Online account to our master QuickBooks Online account. We will have full and independent access to all information you enter into your QuickBooks Online account and we may retrieve, download, analyze and store this information and data at any time and for any purpose.
- (c) <u>Disruptions</u>. You are solely responsible for protecting against computer viruses, bugs, power disruptions, communication line disruptions, internet access failures, internet

- content failures, date-related problems, and attacks by hackers and other unauthorized intruders. Upon our request, you must obtain and maintain cyber insurance and business interruption insurance for technology disruptions.
- (d) Fees and Costs. You are responsible for all fees, costs and expenses associated with acquiring, licensing, utilizing, updating and upgrading the Technology Systems. Certain components of the Technology Systems must be purchased or licensed from third party suppliers. We and/or our affiliate may develop proprietary software, technology or other components of the Technology Systems that will become part of our System. If this occurs: (i) you agree to pay us (or our affiliate) commercially reasonable licensing, support and maintenance fees; and (ii) upon our request, you agree to enter into a license agreement with us (or our affiliate) in a form that we prescribe governing your use of the proprietary software, technology or other component of the Technology Systems. We also reserve the right to enter into master agreements with third-party suppliers relating to any components of the Technology Systems and then charge you for all amounts that we must pay to these suppliers based upon your use of the software, technology, equipment, or services provided by the suppliers. The "technology fee" includes all amounts that you must pay us or our affiliates relating to the Technology Systems, including amounts paid for proprietary items and amounts that we collect from you and remit to third-party suppliers based on your use of their systems, software, technology or services. The amount of the technology fee may change based upon changes to the Technology Systems or the prices charged by third-party suppliers with whom we enter into master agreements. Although we do not currently do so, we reserve the right to include within the technology fee a commercially reasonable administrative fee to compensate us for the time, money and resources we invest in administering the technology platform and the associated constituent components, negotiating with third-party licensors and managing those relationships, and collecting and remitting technology fees owed to third-party licensors on behalf of franchisees under master license arrangements. The technology fee does not include any amounts that you directly pay to third party suppliers for any component of the Technology Systems. The technology fee is due on the dates that we specify from time to time. Our current technology fee covers the following technology-related goods and services and includes the following initial and ongoing fees:

TECHNOLOGY FEES*				
Licensed Product or Software	Description of Licensed Product or Software	Initial Fee	Ongoing Fee	
ALL IN	Our proprietary business management software you will use to operate your Business	\$495 initial setup fee (due when you sign Franchise Agreement)	\$199250 per month for each block of 4 user licenses (due on 1st day of month)	
Google Workspace (formerly known as "G Suite")	Collection of cloud computing, productivity and collaboration tools, software and products developed by Google, including your required email addresses/accounts	None	\$2527 per month per account (due on 1st day of month)	

^{*} All of the fees described in the table above comprise our current "technology fee." We may increase the ALL-IN license fee on an annual basis. We may also increase the monthly fee for Google Workspace and other components of the technology fee upon 30 days' prior

notice. We reserve the right to substitute different third-party providers and you will be required to implement such changes upon written notice and will be solely responsible for all fees associated with such change.

- 11.16. Code of Conduct. You agree to operate your Business in a manner consistent with the Code of Conduct established by us from time to time. Code of Conduct violations may be identified by us or another franchisee. All disputes regarding alleged Code of Conduct violations shall be resolved in accordance with the dispute resolution provisions in Section 23 unless you and we mutually agree in writing to resolve such dispute through binding arbitration conducted by a peer group of Assisted Living Locators franchisees established specifically for purposes of dealing with potential Code of Conduct violations (the "Peer Group"). If binding arbitration is conducted by a Peer Group, all parties agree to honor the decision rendered by the Peer Group and hereby waive any right to appeal the Peer Group's decision. You and we shall hold the Peer Group harmless for the decision rendered, and the Peer Group will render a binding and legally enforceable decision that may be confirmed by a court of competent jurisdiction. Attorney's will not participate in the Peer Group arbitration process. The party filing the arbitration with the Peer Group must bear the cost of the arbitration fees or costs. However, the Peer Group will have the discretion to award to the prevailing party all costs of the proceedings, including the cost of the Peer Group participation and any fines or additional costs. Peer Group arbitration is available only for Code of Conduct disputes, and only for a maximum of two (2) identical disputes (meaning a dispute involving the same parties and same nature of the dispute). A third (3rd) Code of Conduct dispute between the same parties involving the same nature of dispute shall be resolved in accordance with the dispute resolution provisions in Section 23. Additional details regarding the policies and procedures governing Code of Conduct arbitration proceedings may be set forth in the Code of Conduct.
- 11.17. Hours of Operation. Your Business must remain open for business not later than 9:00 a.m. and close not earlier than 5:00 p.m. local time, Monday through Friday, except on Thanksgiving Day, Christmas Day, New Year's Day and other national holidays, unless otherwise agreed to by us. You must establish specific days and hours of operation and submit those hours to us for approval. You understand that your clients may require assistance outside of normal business hours. Accordingly, you must ensure that at all times that you are available to assist clients outside your normal business hours. As long as you contract with a call center answering services that we designate or approve and pay the associated fees, you will be deemed to be in compliance with your obligation to answer phones outside of normal business hours.
- **11.18.** <u>Client Complaints</u>. If you receive a client complaint, you must follow the complaint resolution process that we specify to protect the goodwill associated with the Marks.
- 11.19. Failure to Comply with Standards. You acknowledge the importance of every one of our standards and operating procedures to the reputation and integrity of the System and the goodwill associated with the Marks. If we notify you of a failure to comply with any breach of our standards or operating procedures (including failure to comply with submit required reports in a timely manner or violation of our Franchisee Code of Conduct) and you fail to correct the non-compliance cure within the time period of time that we require, then, prescribe, we may (in addition to anyour other remedies available to us under this Agreement, we may impose a fine noncompliance fee of up to \$2,500 per occurrence; provided, however that. We may impose a separate \$2,500 fee every 48 hours the same noncompliance issue remains uncured after we may impose a fine of \$1,000 per daythe initial fee. Notwithstanding the above, if you fail to perform complete a required background check on an employee and fail to complete, we may impose a noncompliance fee of \$1,000 per day until the background check within theis completed. Any noncompliance fees we collect are paid in consideration of us refraining from exercising our contractual right to terminate this Agreement. If you fail to cure a breach before the expiration of the cure period of time that we require (the \$1,000 per day fine will cease at such time that you complete the background check and provide us with sufficient evidence of same). We may impose an additional fine for every 48 hours the same non-compliance issue

remains uncured following our imposition(if any) and we take steps to cure the breach (for example, obtaining required insurance coverage on your behalf or paying amounts you owe to system suppliers), then you must reimburse all costs and expenses we directly or indirectly incur in connection with our efforts to cure the default. Your payment of the initial fine noncompliance fees and default expense reimbursements does not preclude us from terminating this Agreement pursuant to Section 21 if the default continues after we collect these amounts. We will deposit all fines any noncompliance fees we collect into the brand and system development fund.

- 12. MINIMUM PERFORMANCE REQUIREMENTS. Commencing with the expiration of the 12th month after you open your Business, you must complete a minimum of 18 new client placements during each subsequent six (6) month period. Your failure to comply with these minimum performance requirements constitutes an event of default under this Agreement, in which case we have the right to elect, in our sole discretion, whether to terminate this Agreement or modify or eliminate the territorial protections afforded to you, including by taking any one or more of the following actions:
 - (i) reducing the size of your Territory, with a corresponding reduction in your minimum performance requirements; and/or
 - (ii) permitting other ALL Businesses to provide placement services involving Senior Communities and clients located within your Territory; and/or
 - (iii) establishing, or licensing or franchising others to establish, another Assisted Living Locators business to operate within your Territory.
- 13. FRANCHISE ADVISORY COUNCIL. We may, but need not, create a franchise advisory council (the "FAC") to provide us with suggestions to improve the System, including matters such as marketing, operations and new product or service suggestions. We will consider all suggestions from the advisory council in good faith, but we are not bound by any such suggestions. The FAC will be established and operated according to a set of FAC bylaws, including procedures governing the selection of representatives of the FAC who will communicate with us on matters raised by the advisory council. You will have the right to be elected as an FAC representative as long as you are in good standing with us (but subject to any representative election requirements set forth in the FAC bylaws). Except as otherwise prohibited by the FAC bylaws, we reserve the right to change or dissolve the FAC in our discretion.

14. FEES

- **14.1.** <u>Initial Franchise Fee.</u> You agree to pay us a nonrefundable initial franchise fee in the amount set forth in Part B of <u>ATTACHMENT "A"</u> in one lump sum at the time you sign this Agreement.
- **14.2.** Business Set-Up Fee. You agree to pay us a nonrefundable \$10,000 business set-up fee in one lump sum at the time you sign this Agreement.
- **14.3.** Royalty Fee. Commencing with the date you open your Business, you must pay us a nonrefundable monthly royalty fee equal to the greater of: (a) 8% of your Gross Collected Revenues Revenue; or (b) the minimum royalty fee. The royalty fee is due on the 20th day of each month for Gross Collected Revenues Revenue generated during the immediately preceding month. We reserve the right to change the royalty fee due date from time to time upon at least 30 days' prior notice to you. The "minimum royalty fee" varies based on the number of months that have elapsed from your opening date, as further described in the table below:

Months after Opening Date	Minimum Royalty Fee
Months 1 (full or partial) and 2	\$0 per month
Months 3 through 12	\$500 per month

Months 13 through 24	\$800 per month
Months 25 through 36	\$1,100 per month
Months 37 through end of Term	\$1,400 per month

- **14.4.** Collection Fee. If we collect payment on your behalf for customers who are delinquent in their payment of 90 days or more past the contract term, we may retain up to 35% of gross amounts collected on your behalf.
- **14.5.** Other Fees and Payments. You agree to pay all other fees, expense reimbursements and other amounts specified in this Agreement in a timely manner as if fully set forth in this Section 14. You also agree to promptly pay us an amount equal to all taxes levied or assessed against us based upon goods or services that you sell or based upon goods or services that we furnish to you (other than income taxes that we pay based on amounts that you pay us under this Agreement).
- 14.6. Late Fee. If any sums due under this Agreement have not been received by us when due (or there are insufficient funds in your Account to cover any sums owed to us when due) then, in addition to those sums, you must pay us interest on the amounts past due at the rate equal to the lesser of 18% per annum (pro-rated on a daily basis), or the highest rate permitted by your State's law. If no due date has been specified by us, then interest begins to run 10 days after we bill you. We will not impose a late fee for any amounts paid pursuant to Section 14.7 if, but only to the extent that, sufficient funds were available in your Account to be applied towards the payments at the time the payments became due and payable. However, we may impose a late fee for any amounts that we are unable to reasonably determine due to your failure to furnish us with a report required by Section 16.3 within the required period of time or record sales in a timely manner, in which case we may assess a late fee on the entire amount that was due and payable. You acknowledge that this Section 14.6 shall not constitute our agreement to accept the late payments after same are due, or a commitment by us to extend credit to or otherwise finance the operation of your Business.
- 14.7. Method of Payment. You must complete and send us an ACH Authorization Form allowing us to electronically debit a banking account that you designate (your "Account") for: (a) all fees payable to us pursuant to this Agreement (other than the initial franchise fee); and (b) any amounts that you owe to us or any of our affiliates for the purchase of goods or services. We will debit your Account for these payments on or after the due date. Our current form of ACH Authorization Form is attached to this Agreement as ATTACHMENT "D". You must sign and deliver to us any other documents that we or your bank may require to authorize us to debit your Account for these amounts. You must immediately notify us of any change in your banking relationship, including changes in account numbers. You must deposit into the Account all revenues that revenue you generate from the operation of your Business no later than three (3) business days after receipt (including cash, checks and credit card receipts). You must make sufficient funds available for withdrawal by electronic transfer before each due date. If there are insufficient funds in your Account to cover all amounts that you owe, any excess amounts that you owe will be payable upon demand, together with any late charge imposed pursuant to Section 14.6.
- **14.8.** <u>Application of Payments</u>. We have sole discretion to apply any payments from you to any past due indebtedness of yours or in any other manner we feel appropriate.

15. BRAND PROTECTION COVENANTS.

15.1. Reason for Covenants. You acknowledge that the Intellectual Property and the training and assistance that we provide would not be acquired except through implementation of this Agreement. You also acknowledge that competition by you, the Owners or persons associated

with you or the Owners (including family members) could seriously jeopardize the entire franchise system because you and the Owners have received an advantage through knowledge of our day-to-day operations and Know-how related to the System. Accordingly, you and the Owners agree to comply with the covenants described in this Section to protect the Intellectual Property and our franchise system.

15.2. Our Know-how. You and the Owners agree to:

- (i) refrain from using the Know-how in any business or capacity other than the operation of your Business pursuant to this Agreement;
- (ii) maintain the confidentiality of the Know-how at all times;
- (iii) refrain from making unauthorized copies of documents containing any Know-how;
- (iv) take all reasonable steps we require to prevent unauthorized use or disclosure of Knowhow; and
- (v) stop using the Know-how immediately upon the expiration, termination or Transfer of this Agreement (and any Owner who ceases to be an Owner before the expiration, termination or Transfer of this Agreement must stop using the Know-how immediately at the time he or she ceases to be an Owner).
- **15.3.** <u>Unfair Competition During Term</u>. You and your Owners agree not to unfairly compete with us during the Term by engaging in any of the following activities ("Prohibited Activities"):
 - (i) owning, operating or having any other interest (as an owner, partner, director, officer, employee, manager, consultant, shareholder, creditor, representative, agent or in any similar capacity) in any Competitive Business, other than owning an interest of five percent (5%) or less in a publicly traded company that is a Competitive Business;
 - (ii) diverting or attempting to divert any business from any ALL Business;
 - (iii) inducing a client of any ALL Business to transfer their business to any person other than an ALL Business;
 - (iv) inducing any Senior Community that you know, or have reason to know, has entered into a Marketing Agreement with any ALL Business (including with you), to enter into an agreement for placement services with any person other than an ALL Business; or
 - (v) inducing any Referral Source that you know, or have reason to know, has entered into a referral agreement or established a referral relationship with any ALL Business (including with you), to refer clients in need of senior living placement services to any person other than an ALL Business.
- 15.4. Unfair Competition After Term. During the Post-Term Restricted Period, you and your Owners agree not to engage in any Prohibited Activities. Notwithstanding the foregoing, you and your Owners may have an interest in a Competitive Business during the Post-Term Restricted Period as long as the Competitive Business does place clients with Senior Communities that are located within the Restricted Territory. If you or an Owner engages in a Prohibited Activity during the Post-Term Restricted Period (other than having an interest in a Competitive Business that is permitted under this Section), then the Post-Term Restricted Period applicable to you or the non-compliant Owner, as applicable, shall be extended by the period of time during which you or the non-compliant Owner, as applicable, engaged in the Prohibited Activity.
- **15.5.** <u>Immediate Family Members</u>. The Owners acknowledge that they could circumvent the purpose of <u>Section 15</u> by disclosing Know-how to an immediate family member (i.e., spouse, parent, sibling, child, or grandchild). The Owners also acknowledge that it would be difficult for

us to prove whether the Owners disclosed the Know-how to family members. Therefore, each Owner agrees that he or she will be presumed to have violated the terms of <u>Section 15</u> if any member of his or her immediate family engages in any Prohibited Activities during the Term or Post-Term Restricted Period or uses or discloses the Know-how. However, the Owner may rebut this presumption by furnishing evidence conclusively showing that the Owner did not disclose the Know-how to the family member.

- 15.6. Employees and Others Associated with You. You must ensure that all of your managers, officers, directors, partners, members, independent contractors and other persons associated with you or your Business who may have access to our Know-how, sign a Brand ProtectionConfidentiality Agreement. You must use your best efforts to ensure that these individuals comply with the terms of the Brand Protection AgreementsConfidentiality Agreement and you must immediately notify us of any breach that comes to your attention. You agree to reimburse us for all reasonable expenses that we incur in enforcing a Brand ProtectionConfidentiality Agreement, including reasonable attorneys' fees and court costs.
- 15.7. Covenants Reasonable. You and the Owners acknowledge and agree that: (a) the terms of this Agreement are reasonable both in time and in scope of geographic area; (b) our use and enforcement of covenants similar to those described above with respect to other ALL Businesses benefits you and the Owners in that it prevents others from unfairly competing with your Business; and (c) you and the Owners have sufficient resources and business experience and opportunities to earn an adequate living while complying with the terms of this Agreement.

 YOU AND THE OWNERS HEREBY WAIVE ANY RIGHT TO CHALLENGE THE TERMS OF THIS SECTION 15 AS BEING OVERLY BROAD, UNREASONABLE OR OTHERWISE UNENFORCEABLE.
- 15.8. **Injunctive Relief.** Nothing contained in this Agreement will prevent us from applying to and obtaining from any court having jurisdiction a writ of attachment, a temporary injunction, preliminary injunction and/or other emergency relief available to safeguard and protect our interests, without bond, against conduct or threatened conduct that will cause us loss or damages, under the usual equity rules, including the applicable rules for obtaining restraining orders and preliminary and permanent injunctions. If a court requires the filing of a bond notwithstanding the preceding sentence, the parties agree that the amount of the bond shall not exceed \$1,000. None of the remedies available to us under this Agreement are exclusive of any other, but may be combined with others under this Agreement, or at law or in equity, including injunctive relief, specific performance and recovery of monetary damages. Any claim, defense or cause of action that you or an Owner may have against us, regardless of cause or origin, cannot be used as a defense against our enforcement of this Agreement. If injunctive relief is granted, your only remedy will be the court's dissolution of the injunctive relief. If the injunctive relief was wrongfully issued, you expressly waive all claims for damages you incurred as a result of the wrongful issuance.

16. YOUR OTHER RESPONSIBILITIES

- **16.1. Insurance.** For your protection and ours, you agree to maintain the following insurance policies:
 - (i) General Liability Occurrence coverage form in the minimum amount of \$1,000,000 for each claim limit BI / PD, \$3,000,000 general aggregate limit of liability and \$1,000,000 non-owned / hired auto liability;
 - (ii) Professional Liability Occurrence coverage form in the minimum amount of \$1,000,000 for each professional incident and \$3,000,000 professional aggregate limit;
 - (iii) worker's compensation insurance and employer's liability insurance as required by law; and
 - (iv) any other insurance that we specify in the Manual from time to time.

If you lease a vehicle for your Business you are required to have a commercial auto policy. If you elect to continue your personal auto insurance policies, a "Business Use Endorsement" is required to be added to your automobile insurance policy with liability limits of at least \$250,000/\$500,000 split limits or \$500,000 Combined Single Limit. As of the Effective Date, we require that you purchase our required policies through our designated insurance broker, Eaton-Provident Group, LLC. We also highly recommend that you purchase Sexual Abuse/Molestation coverage and Cyber Liability coverage in addition to the coverage outlined above. You agree to provide us with proof of coverage on demand. All insurance policies must be issued by insurance carriers rated A or better by Alfred M. Best & Company, Inc. All carriers must be licensed and admitted in the state in which you operate your Business. All insurance policies must endorsed to: (a) name "ALL Franchising, LLC, 8100 E. Indian School Road, Suite 201, Scottsdale, Arizona 85251" as an additional insured; (b) contain a waiver by the insurance carrier of all subrogation rights against us; and (c) provide that we receive 10 days prior written notice of the termination, expiration, cancellation or modification of the policy. If any of vour policies fail to meet these criteria, then we may disapprove the policy and you must immediately find additional coverage with an alternative carrier satisfactory to us. The description of operations box on the certificate must read: Senior Referral Agency. You must obtain and keep in force a Commercial Liability policy of insurance protecting you and us as an additional insured against claims arising out of actions of yours. You must obtain all required insurance and provide us with copies of certificates of insurance listing the policies and liability limits, prior to completion of our initial training program. Thereafter, you must provide us within copies of certificates of insurance: (a) on each anniversary of your opening date; and (b) within 30 days of your purchase or renewal of any insurance policy. Upon 10 days' notice to you, we may increase the minimum protection requirement as of the renewal date of any policy and require different or additional types of insurance at any time, including excess liability (umbrella) insurance, to reflect inflation, identification of special risks, changes in law or standards or liability, higher damage awards or other relevant changes in circumstances. If you fail to maintain any required insurance coverage, we have the right to obtain the coverage on your behalf (which right shall be at our option and in addition to our other rights and remedies in this Agreement), and you must promptly sign all applications and other forms and instruments required to obtain the insurance and pay to us, within 10 days after invoicing, all costs and premiums that we incur. You are free to purchase any additional insurance that you deem appropriate.

- **Books and Records.** You agree to prepare and maintain at your office for at least five (5) years after their preparation, complete and accurate books, records, accounts and tax returns pertaining to your Business, including, without limitation, all Business Records. You must maintain, and upon our request furnish to us by e-mail, mail or facsimile, a written list of all of your clients, Referral Sources, Senior Communities and Senior Service Providers. You must send us copies of your books and records within seven (7) days of our request.
- 16.3. Reports. No later than the 20th day of each month, you must prepare and provide to us monthly statements of your Gross Collected Revenues Revenue. At any time that we require you to spend the marketing fees directly with third-party suppliers, you must prepare and provide to us monthly statements of your expenditures on local advertising that were incurred during the prior month (which shall be accompanied by copies of receipts for such expenditures). You also agree to prepare all other reports that we require in the form and manner that we require. You agree to send us a copy of any report required by this Section upon request. If you miscalculate Gross Collected Revenues Revenue, you must notify us of the error no later than the end of the next Gross Collected Revenues Revenue reporting period. Otherwise, you will not be entitled to any refund or credit of any fees paid to us based upon the Gross Collected Revenues Revenue previously reported. If we require that you purchase a computer and/or automated cash management system that allows us to electronically retrieve information concerning your sales transactions, you agree that we will have the right to electronically poll your computer and/or automated cash management system to retrieve and compile information regarding the

operation of your Business. We also have the right to independently monitor email communications utilizing the email addresses that we provide to you.

- 16.4. Financial Statements. Within 90 days after the end of each calendar year, you must prepare a balance sheet for your Business (as of the end of the calendar year) and an annual statement of profit and loss and source and application of funds. In addition, no later than 30 days after the end of each calendar quarter, you must prepare and provide us with quarterly balance sheets and statements of profits and loss and source and application of funds for the immediately preceding calendar quarter. All financial statements must be: (a) verified and signed by you certifying to us that the information is true, complete, and accurate; (b) prepared on an accrual basis in compliance with Generally Accepted Accounting Principles; and (c) submitted in any format that we reasonably require. We have the right to require that your financial statements be audited by a certified public accountant if you submit to us materially inaccurate financial statements on a prior occasion. You agree to send us a copy of any financial statement required by this Section upon request. You authorize us to disclose the financial statements, reports, and operating data to prospective franchisees, regulatory agencies and others at our discretion, provided the disclosure is not prohibited by applicable law.
- 16.5. <u>Legal Compliance</u>. You must secure and maintain in force all required licenses, permits and regulatory approvals for the operation of your Business and operate and manage your Business in full compliance with all applicable laws, ordinances, rules and regulations. You must notify us in writing within two (2) business days of the beginning of any action, suit, investigation or proceeding, or of the issuance of any order, writ, injunction, disciplinary action, award or decree of any court, agency or other governmental instrumentality, which may adversely affect the operation of your Business or your financial condition. You must immediately deliver to us a copy of any inspection report, warning, certificate or rating by any governmental agency involving any health or safety law, rule or regulation that reflects your failure to fully comply with the law, rule or regulation. To ensure compliance with the Health Insurance Portability and Accountability Act of 1996 and the associated regulations, you must sign the form of HIPAA Business Associate Agreement that we specify from time to time.

17. INSPECTION AND AUDIT

- 17.1. <u>Inspections.</u> To ensure compliance with this Agreement, we or our representatives will have the right to enter your office, evaluate your operations and inspect or examine your books, records, accounts and tax returns. Our evaluation may include monitoring your operations and interactions with clients and Senior Communities. We may also contact your employees, clients, Senior Communities, Referral Sources and Senior Care Providers. We may conduct our evaluation at any time and without prior notice, but we will provide you reasonable prior notice before inspecting your home office. During the course of our inspections, we and our representatives will use reasonable efforts to minimize our interference with the operation of your Business, and you and your employees will cooperate and not interfere with our inspection. You consent to us accessing your computer system and retrieving any information that we deem appropriate in conducting the inspection.
- 17.2. Audit. We have the right, at any time, to have an independent audit made of your books and financial records. You agree to fully cooperate with us and any third parties that we hire to conduct the audit. If an audit reveals an understatement of your Gross Collected Revenues Revenue or any amount that you owe us, you agree to immediately pay to us any additional fees that you owe us together with any late fee payable pursuant to Section 14.6. Any audit will be performed at our cost and expense unless the audit: (a) is necessitated by your failure to provide the information requested or to preserve records or file reports as required by this Agreement; or (b) reveals an understatement of any amount due to us by at least two percent (2%), in which case you agree to pay us \$2,500 and also reimburse us for the cost of the audit or inspection, including without limitation, reasonable accounting and attorneys' fees and travel and lodging expenses that we or our representatives incur. The audit fee and cost

reimbursements will be due 10 days after invoicing. We shall not be deemed to have waived our right to terminate this Agreement by accepting reimbursements of our audit costs or payment of the fee.

18. INTELLECTUAL PROPERTY

- **Ownership and Use of Intellectual Property**. You acknowledge that: (a) we are the sole and exclusive owner of the Intellectual Property and the goodwill associated with the Marks; (b) your right to use the Intellectual Property is derived solely from this Agreement; and (c) your right to use the Intellectual Property is limited to a license granted by us to operate your Business during the Term pursuant to, and only in compliance with, this Agreement, the Manual, and all applicable standards, specifications and operating procedures that we prescribe from time to time. You may not use any of the Intellectual Property in connection with the sale of any unauthorized product or service or in any other manner not expressly authorized by us. Any unauthorized use of the Intellectual Property constitutes an infringement of our rights. You agree to comply with all provisions of the Manual governing your use of the Intellectual Property. This Agreement does not confer to you any goodwill, title or interest in any of the Intellectual Property.
- 18.2. Changes to Intellectual Property. We have the right to modify the Intellectual Property at any time in our sole and absolute discretion, including by changing the Marks, the System, the Copyrights or the Know-how. If we modify or discontinue use of any of the Intellectual Property, then you must comply with any such instructions from us within 30 days. If we require you to change the Marks, our sole obligation will be to reimburse you for your reasonable documented expenses of compliance, including changing signage, brochures, stationary, etc. You waive all other claims arising from or relating to any change, modification, substitution or discontinuation of the Intellectual Property. Except for the reimbursement obligation listed in this Section, we will not be liable to you for any expenses, losses or damages that you incur (including the loss of any goodwill associated with a Mark) because of any addition, modification, substitution or discontinuation of the Intellectual Property.
- 18.3. <u>Use of Marks</u>. You agree to use the Marks as the sole identification of your Business; provided, however that you must identify yourself as the independent owner of your Business in the manner that we prescribe. You may not use any Marks in any modified form or as part of any corporate or trade name or with any prefix, suffix, or other modifying words, terms, designs or symbols (other than logos licensed to you by this Agreement). You agree to: (a) prominently display the Marks on or in connection with any media advertising, promotional materials, posters and displays, receipts, stationery and forms that we designate and in the manner that we prescribe to give notice of trade and service mark registrations and copyrights; and (b) obtain any fictitious or assumed name registrations required under applicable law. You may not use the Marks in signing any contract, lease, check, negotiable instrument or other agreement or in any manner likely to confuse or result in liability to us for any indebtedness or obligation of yours.
- **18.4.** <u>Use of Know-how</u>. We will disclose the Know-how to you in the initial training program, the Manual, and in other guidance furnished to you during the Term. All data entered into ALL-IN regarding your clients, Senior Communities, Referral Sources and Senior Care Providers shall be deemed our Know-how. You agree that you will not acquire any interest in the Know-how other than the right to utilize it in strict accordance with the terms of this Agreement in the development and operation of your Business. You acknowledge that the Know-how is proprietary and is disclosed to you solely for use in the development and operation of your Business during the Term.
- **18.5.** Ownership of Business Records. You acknowledge and agree that we have access to, and will be the exclusive owner of, all of your Business Records. You are licensed to use all Business Records solely in connection with the operation of your Business. All such Business Records

are deemed part of our Intellectual Property. We may utilize your Business Records in any manner that we deem appropriate and to be in the best interest of the System.

- **18.6.** <u>Improvements</u>. If you conceive of or develop any improvements, creative ideas or additions to the marketing, method of operation or the services or products offered by an ALL Business (collectively, "<u>Improvements</u>"), you agree to promptly and fully disclose the Improvements to us without disclosing the Improvements to others. You must obtain our approval prior to using any such Improvements. Any Improvement that we approve may be used by us and any third parties that we authorize to operate an Assisted Living Locators franchise, without any obligation to pay you royalties or other fees. You must assign to us or our designee, without charge, all rights to any such Improvement, including the right to grant sublicenses. In return, we will authorize you to use any Improvements that we or other franchisees develop that we authorize for general use in connection with the operation of an ALL Business.
- 18.7. Notification of Infringements and Claims. You must immediately notify us of any:

 (a) apparent infringement of any of the Intellectual Property; (b) challenge to your use of any of the Intellectual Property; or (c) claim by any person of any rights in any of the Intellectual Property. You may not communicate with any person other than us and our counsel in connection with any such infringement, challenge or claim. We will have sole discretion to take such action as we deem appropriate. We have the right to exclusively control any litigation, Patent and Trademark Office proceeding, or other proceeding arising out of any such infringement, challenge or claim. You agree to execute any and all instruments and documents, render such assistance, and do such acts and things as may, in the opinion of our counsel, be necessary or advisable to protect and maintain our interest in any such litigation, Patent and Trademark Office proceeding or other proceeding, or to otherwise protect and maintain our interest in the Intellectual Property.
- **19. INDEMNITY.** You agree to indemnify the Indemnified Parties and hold them harmless for, from and against any and all Losses and Expenses incurred by any of them as a result of or in connection with any of the following:
 - (i) the marketing, use or operation of your Business or your performance and/or breach of any of your obligations under this Agreement, including any Claims brought by any of your clients, Senior Communities, Referral Sources or Senior Care Providers;
 - (ii) any Claim relating to taxes or penalties assessed by any governmental entity against us that are directly related to your failure to pay or perform functions required of you under this Agreement;
 - (iii) any labor, employment or similar type of Claim pertaining to your employees;
 - (iv) any actions, investigations, rulings or proceedings conducted by any state or federal agency relating to your employees, including, without limitation, the United States Department of Labor, the Equal Employment Opportunity Commission and the National Labor Relations Board: and
 - (v) any labor, employment or similar type of Claim pertaining to our relationship with you or your Owners.

You and your Owners agree to give us notice of any action, suit, proceeding, claim, demand, inquiry or investigation described above. The Indemnified Parties shall have the right, in their sole discretion to: (a) retain counsel of their own choosing to represent them with respect to any Claim; and (b) control the response thereto and the defense thereof, including the right to enter into an agreement to settle such Claim. You may participate in such defense at your own expense. You agree to give your full cooperation to the Indemnified Parties in assisting the Indemnified Parties with the defense of any such Claim, and to reimburse the Indemnified Parties for all of their costs and expenses in defending any such Claim, including court costs and reasonable attorneys' fees, within 10 days of the date of each

invoice delivered by such Indemnified Party to you enumerating such costs, expenses and attorneys' fees.

20. TRANSFERS

- **20.1.** By Us. This Agreement and the franchise is fully assignable by us (without prior notice to you) and shall inure to the benefit of any assignee(s) or other legal successor(s) to our interest in this Agreement, provided that we shall, subsequent to any such assignment, remain liable for the performance of our obligations under this Agreement up to the effective date of the assignment. We may also delegate some or all of our obligations under this Agreement to one or more persons without assigning the Agreement.
- **20.2. By You**. You understand that the rights and duties created by this Agreement are personal to you and the Owners and that we have granted the franchise in reliance upon the individual or collective character, skill, aptitude, attitude, business ability and financial capacity of you and your Owners. Because this Agreement is a personal services contract, neither you nor any Owner may engage in any Transfer other than a Permitted Transfer without our prior written approval. Any Transfer (other than a Permitted Transfer) without our approval shall be void and constitute a breach of this Agreement. We will not unreasonably withhold our approval of any proposed Transfer, provided that the following conditions are all satisfied:
 - (i) the proposed transferee is, in our opinion, an individual of good moral character, who has sufficient business experience, aptitude and financial resources to own and operate an ALL Business and otherwise meets all of our then applicable standards for franchisees;
 - (ii) you and your Owners are in full compliance with the terms of this Agreement and all other agreements with us or our affiliate;
 - (iii) all of the owners of the transferee have successfully completed, or made arrangements to attend, the initial training program (and the transferee has paid us our then-current initial training fee);
 - (iv) the transferee and its owners, to the extent necessary, have obtained all licenses and permits required by applicable law, and obtained any required bonding, in order to own and operate the Business;
 - (v) the transferee signs an agreement, in a form satisfactory to us, agreeing to discharge and guaranty all of your obligations under this Agreement and any other agreement relating to the Business, including, without limitation, Client Agreements, Marketing Agreements, Referral Services Agreements, technology contracts, and other contracts;
 - (vi) the transferee and its owners sign our then-current form of franchise agreement (unless we, in our sole discretion, instruct you to assign this Agreement to the transferee), except that: (a) the Term and renewal term(s) shall be the Term and renewal term(s) remaining under this Agreement; and (b) the transferee need not pay a separate initial franchise fee;
 - (vii) you or the transferee pay us a \$10,000 transfer fee (reduced to \$2,500 if the transferee is an existing Assisted Living Locators franchisee) to defray expenses that we incur in connection with the Transfer (if the transferee is found by a broker we engage, you must also reimburse us for all commissions we pay to the broker, which amount shall be in addition to the transfer fee);
 - (viii) you and your Owners sign a General Release for all claims arising before or contemporaneously with the Transfer;
 - (ix) we do not elect to exercise our right of first refusal described in Section 20.5; and

you or the transferring Owner, as applicable, and the transferee have satisfied any other conditions we reasonably require as a condition to our approval of the Transfer.

Our consent to a Transfer shall not constitute a waiver of any claims we may have against the transferor, nor shall it be deemed a waiver of our right to demand exact compliance with any of the terms or conditions of the franchise by the transferee.

- **20.3. Permitted Transfers.** You may engage in a Permitted Transfer without our prior approval, but you must give us at least 10 days prior written notice. You and the Owners (and the transferee) agree to sign all documents that we reasonably request to effectuate and document the Permitted Transfer.
- **20.4. Death or Disability of an Owner.** Upon the death or permanent disability of an Owner, the Owner's ownership interest in you or the franchise, as applicable, must be assigned to another Owner or to a third party approved by us within 180 days. Any assignment to a third party will be subject to all of the terms and conditions of Section 20.2 unless the assignment qualifies as a Permitted Transfer. For purposes of this Section, an Owner is deemed to have a "permanent disability" only if the person has a medical or mental problem that prevents the person from substantially complying with his or her obligations under this Agreement or otherwise operating the Business in the manner required by this Agreement and the Manual for a continuous period of at least three (3) months.
- 20.5. Our Right of First Refusal. If you or an Owner desire to engage in a Transfer, you or the Owner, as applicable, must obtain a bona fide, signed written offer from the fully disclosed purchaser and submit an exact copy of the offer to us. We will have 30 days after receipt of the offer to decide whether we will purchase the interest in your Business or the ownership interest in you for the same price and upon the same terms contained in the offer (however, we may substitute cash for any form of payment proposed in the offer). If we notify you that we intend to purchase the interest within the 30-day period, you or the Owner, as applicable, must sell the interest to us. We will have at least an additional 30 days to prepare for closing. We will be entitled to receive from you or the Owner, as applicable, all customary representations and warranties given by you as the seller of the assets or the Owner as the seller of the ownership interest or, at our election, the representations and warranties contained in the offer. If we do not exercise our right of first refusal, you or the Owner, as applicable, may complete the Transfer to the purchaser pursuant to and on the terms of the offer, subject to the requirements of Section 20.2 (including our approval of the transferee). However, if the sale to the purchaser is not completed within 120 days after delivery of the offer to us, or there is a material change in the terms of the sale, we will again have the right of first refusal specified in this Section. Our right of first refusal in this Section shall not apply to any Permitted Transfer.

21. TERMINATION

- **21.1.** <u>Automatic Termination.</u> You shall be deemed to be in default under this Agreement and all rights granted to you hereunder shall terminate automatically without notice to you upon the occurrence of any of the following:
 - (i) You, a Managing Owner or any of your guarantors become insolvent, makes a general assignment for the benefit of creditors, or is adjudicated a bankrupt, unless otherwise restricted by the relevant bankruptcy laws;
 - (ii) A petition in bankruptcy for liquidation, reorganization, or other proceeding is filed by or against you, a Managing Owner or any of your guarantors; or a receiver is appointed or a bill in equity or other proceeding for the appointment of a receiver of you, a Managing Owner or any of your guarantors or other custodian for the Business or assets is allowed; or the real or personal property of you, a Managing Owner or any of your guarantors is attached or levied upon by any sheriff, marshal, or constable, or seized,

taken over or foreclosed by a creditor, lienholder or lessor; or a final judgment against you remains unsatisfied for 30 days (unless a supersedes or other appeal bond has been filed).

- **21.2.** <u>Termination By Us Without Cure Period</u>. We may, in our sole discretion, terminate this Agreement upon written notice, without opportunity to cure, for any of the following reasons, all of which constitute material events of default under this Agreement:
 - (i) if the Managing Owner fails to satisfactorily complete the initial training program in the manner required by <u>Section 5.1</u>;
 - (ii) if you fail to open your Business within the time period required by <u>Section 7.3</u>;
 - (iii) if you abandon or fail to operate your Business for three (3) consecutive business days, unless the failure is due to an event of force majeure or another reason that we approve;
 - (iv) if a regulatory authority suspends or revokes a license or permit held by you or an Owner that is required to operate the Business, even if you or the Owner still maintain appeal rights;
 - (v) if you or an Owner (a) is convicted of or pleads no contest to a felony, a crime involving moral turpitude or any other material crime or (b) is subject to any material administrative disciplinary action or (c) fails to comply with any material federal, state or local law or regulation applicable to your Business;
 - (vi) if you or an Owner commits an act that can reasonably be expected to adversely affect the reputation of the System, or the goodwill associated with the Marks;
 - (vii) if you manage or operate your Business in a manner that presents a health or safety hazard to your customers, employees or the public;
 - (viii) if we receive reasonably satisfactory evidence that the Managing Owner and/or your employees have engaged in unsafe, reckless or unlawful driving on multiple occasions while driving clients or while driving any vehicle that bears or displays any of our Marks;
 - (ix) if you or an Owner make any material misrepresentation to us, whether occurring before or after being granted the franchise;
 - if you or an Owner fail to pay any amount owed to us or an affiliate of ours within ten (10) days after receipt of a demand for payment;
 - (xi) if you underreport any amount owed to us by at least two percent (2%), after having already committed a similar breach that had been cured in accordance with Section 21.3;
 - (xii) if you make an unauthorized Transfer;
 - (xiii) if you or an Owner make an unauthorized use of the Intellectual Property;
 - (xiv) if you or an Owner or guarantor breach any of the brand protection covenants described in Section 15;
 - (xv) if any Owner, or the spouse of any Owner, breaches a Franchise Owner Agreement;
 - (xvi) if you engage in placement services outside of your Territory unless the placement: (a) is made in compliance with our Referral Program; (b) is made after obtaining our prior written permission; or (c) is otherwise authorized under the terms of this Agreement or the Manual.
 - (xvii) if you fail to meet the minimum performance requirements described in Section 12; or

- (xviii) if we terminate any other agreement between you and us or if any affiliate of ours terminates any agreement between you and the affiliate because of your default.
- **21.3.** Additional Conditions of Termination. In addition to our termination rights in Section 21.2, we may, in our sole discretion, terminate this Agreement upon 30 days' written notice if you or an Owner fail to comply with any other provision of this Agreement (including any mandatory provision in the Manual) or any other agreement with us, unless such default is cured, as determined by us in our sole discretion, within such 30-day notice period. If we deliver a notice of default to you pursuant to this Section 21.3, we may suspend performance of any of our obligations under this Agreement until you fully cure the breach.
- **21.4.** <u>Mutual Agreement to Terminate</u>. If you and we mutually agree in writing to terminate this Agreement, you and we will be deemed to have waived any required notice period.

22. POST-TERM OBLIGATIONS.

- **22.1.** Obligations of You and the Owners. After the termination, expiration or Transfer of this Agreement, you and the Owners agree to:
 - (i) immediately cease to use the Intellectual Property;
 - (ii) pay us all amounts that you owe us;
 - (iii) comply with all covenants described in <u>Section 15</u> that apply after the expiration, termination or Transfer of this Agreement or the disposal of an ownership interest by an Owner;
 - (iv) return all copies of the Manual, or any portions thereof, as well as all signs, sign faces, brochures, advertising and promotional materials, forms, and any other materials bearing or containing any of the Marks, Copyrights or other identification relating to an ALL Business, unless we allow you to transfer such items to an approved transferee;
 - (v) cease utilizing ALL-IN and delete all copies from your computer (you must allow us access to your computer to confirm removal of ALL-IN);
 - (vi) take such action as may be required to cancel all fictitious or assumed names or equivalent registrations relating to your use of any of the Marks;
 - (vii) provide us with a copy of all of your Business Records and a list of all of your current, former and prospective clients, Referral Sources, Senior Communities and Senior Care Providers;
 - (viii) upon our request, assign all Client Agreements to us or to another ALL Business that we designate and assist us with notifying and transitioning all of your clients to us or to such other ALL Business;
 - (ix) upon our request, assign all Marketing Agreements and Referral Services Agreements to us or to another ALL Business that we designate and assist us with notifying and transitioning all of your Senior Communities, Referral Sources and Senior Care Providers to us or to such other ALL Business;
 - (x) cease all communications with clients, Senior Communities and Referral Sources except as we otherwise expressly approve in writing;
 - (xi) cease to collect any placement fees or other amounts paid by Senior Communities after the date this Agreement terminates or expires (we have the exclusive right to collect all such payments, deduct all amounts owed to us and our affiliates, and remit any remaining balance to you);
 - (xii) remove all wraps or other references to our Marks on any vehicle owned or leased by

you or your employees;

- (xiii) notify all telephone companies, listing agencies and domain name registration companies (collectively, the "Agencies") of the termination or expiration of your right to use: (a) the telephone numbers, email addresses and/or domain names (and any affiliated, associated or linked pages, websites and listings), if applicable, related to the operation of your Business; and (b) any regular, classified or other telephone directory listings associated with the Marks (you hereby authorize the Agencies to transfer such telephone numbers, domain names and listings to us and you authorize us, and appoint us and any officer we designate as your attorney-in-fact to direct the Agencies to transfer the telephone numbers, domain names and listings to us if you fail or refuse to do so); and
- (xiv) provide us with satisfactory evidence of your compliance with the above obligations within 30 days after the effective date of the termination, expiration or Transfer of this Agreement.

To facilitate our ability to ensure compliance with <u>Section (xiii)</u>, you must execute our required form of Power of Attorney – Telephone and Internet concurrently with the execution of this Agreement, or at such other time that we designate in our sole discretion.

23. DISPUTE RESOLUTION.

- 23.1 You must first bring any claim or dispute between you and us to our President and provide us with thirty (30) days' notice and opportunity to cure. You must exhaust this internal dispute resolution procedure before you may bring your dispute before a third party. This agreement to first attempt resolution of disputes internally will survive termination or expiration of this Agreement.
- 23.2 Once the internal dispute provisions of Section 23.1 have been exhausted, at our option, any claim, dispute or disagreement, including any matter pertaining to the interpretation of this Agreement or issues relating to the offer and sale of the franchise or the relationship between the parties (a "Dispute") to mediation under the auspices of the American Arbitration Association ("AAA"), in accordance with AAA's Commercial Mediation Rules then in effect. The mediation will take place in the county in which we maintain our principal place of business at the time the Dispute arises (currently, Maricopa County, Arizona) and the parties irrevocably waive any objection to such venue. Before commencing any legal action against us or our affiliates with respect to any such claim or dispute, you must submit a notice to us, which specifies, in detail, the precise nature and grounds of such claim or dispute. We will have a period of thirty (30) days following receipt of such notice within which to notify you whether we or our affiliates elect to exercise our option to submit claims or disputes to mediation. You may not commence any action against us or our affiliates with respect to any such claim or dispute in any court unless we fail to exercise our option to submit such claim or dispute to mediation, or such mediation proceedings have been terminated either: (i) as the result of a written declaration of the mediator(s) that further mediation efforts are not worthwhile; or (ii) as a result of our written declaration. Our right to mediate, as set forth herein, may be specifically enforced by us. Each party will bear its own cost of mediation and we will share mediation costs equally. This agreement to mediate will survive any termination or expiration of this Agreement. The parties agree that there will be no class action mediation.

The parties will not be required to first attempt to mediate a controversy, dispute, or claim through mediation as set forth in this Section 23.2 if such controversy, dispute, or claim concerns an allegation that a party has violated (or threatens to violate, or poses an imminent risk of violating): (1) Any federally protected intellectual property rights in the Marks, Ecommerce Program, the System, Proprietary Materials, Proprietary Software, or in any Confidential Information; (2) Any claims pertaining to or arising out of any warranty issue; (3)

Any of the restrictive covenants contained in this Agreement; or (4) Any claims to collect past due amounts owed to us or our affiliates.

- 23.3 If the Dispute is not resolved by mediation, the parties expressly agree to the jurisdiction and venue of any federal or state court of general jurisdiction in the county in which we maintain our principal place of business at the time the Dispute arises (currently, Maricopa County, Arizona) and the parties irrevocably waive any objection to such venue. If we must enforce this Agreement in a judicial proceeding, we will be entitled to reimbursement of our costs and expenses, including reasonable accounting and legal fees. In addition, if you breach any term of this Agreement or any other agreement with us or an affiliate of ours, you agree to reimburse us for all reasonable legal fees and other expenses we incur relating to such breach, regardless of whether the breach is cured prior to the commencement of any dispute resolution proceedings. Nothing contained in this Agreement shall prevent us from applying to and obtaining from any court having jurisdiction a writ of attachment, a temporary injunction, preliminary injunction and/or other emergency relief available to safeguard and protect our interests.
- 23.4 You further agree that no cause of action arising out of or under this Agreement may be maintained by you against us unless brought before the expiration of one (1) year after the act, transaction or occurrence upon which such action is based or the expiration of one (1) year after you become aware of facts or circumstances reasonably indicating that you may have a claim against us hereunder, whichever occurs sooner, and that any action not brought within this period will be barred as a claim, counterclaim, defense, or set-off. You hereby waive the right to obtain any remedy based on alleged fraud, misrepresentation, or deceit by us, including, without limitation, rescission of this Agreement, in any mediation, judicial, or other adjudicatory proceeding arising hereunder, except upon a ground expressly provided in this Agreement, or pursuant to any right expressly granted by any applicable statute expressly regulating the sale of franchises, or any regulation or rules promulgated thereunder. The parties agree that all proceedings will be conducted on an individual, not a class-wide basis, and that any proceeding between you, your Owners, and we or our affiliates or employees may not be consolidated with any other proceeding between us and any other party or entity.
- JURY TRIAL WAIVER. THE PARTIES HEREBY AGREE TO WAIVE TRIAL BY JURY IN ANY ACTION, PROCEEDING OR COUNTERCLAIM, WHETHER AT LAW OR EQUITY, REGARDLESS OF WHICH PARTY BRINGS SUIT. THIS WAIVER WILL APPLY TO ANY MATTER WHATSOEVER BETWEEN THE PARTIES HERETO WHICH ARISES OUT OF OR IS RELATED IN ANY WAY TO THIS AGREEMENT, THE PERFORMANCE OF EITHER PARTY, AND/OR YOUR PURCHASE OF A FRANCHISE OR ANY GOODS OR SERVICES FROM US OR OUR AFFILIATES.
- 23.6 Waiver of Punitive Damages. You hereby waive, to the fullest extent permitted by law, any right to or claim for any punitive, exemplary, incidental, indirect, special or consequential damages (including, without limitation, lost profits) against us arising out of any cause whatsoever (whether such cause be based in contract, negligence, strict liability, other tort or otherwise) and agree that, in the event of a dispute, your recovery is limited to actual damages. If any other term of this Agreement is found or determined to be unconscionable or unenforceable for any reason, the foregoing provisions will continue in full force and effect, including, without limitation, the waiver of any right to claim any consequential damages.

24. OMITTED.

25. GENERAL PROVISIONS

25.1. Governing Law. Except as governed by the United States Trademark Act of 1946 (Lanham Act, 15 U.S.C. §§ 1051, et seq.), this Agreement and the franchise relationship shall be governed by the laws of the State of Arizona (without reference to its principles of conflicts of law), but any law of the State of Arizona that regulates the offer and sale of franchises or

business opportunities or governs the relationship of a franchisor and its franchisee will not apply unless its jurisdictional requirements are met independently without reference to this Section.

- **Relationship of the Parties.** You understand and agree that nothing in this Agreement creates a fiduciary relationship between you and us or is intended to make either party a general or special agent, legal representative, subsidiary, joint venture, partner, employee or servant of the other for any purpose. During the Term, you must conspicuously identify yourself at your base of operations, and in all dealings with third parties, as a franchisee of ours and the independent owner of your Business. You agree to place such other notices of independent ownership on such forms, stationery, advertising, business cards and other materials as we may require from time to time. Neither we nor you are permitted to make any express or implied agreement, warranty or representation, or incur any debt, in the name of or on behalf of the other or represent that our relationship is other than franchisor and franchisee. In addition, neither we nor you will be obligated by or have any liability under any agreements or representations made by the other that are not expressly authorized by this Agreement.
- **25.3.** Severability and Substitution. Each section, subsection, term and provision of this Agreement, and any portion thereof, shall be considered severable. If any applicable and binding law imposes mandatory, non-waivable terms or conditions that conflict with a provision of this Agreement, the terms or conditions required by such law shall govern to the extent of the inconsistency and supersede the conflicting provision of this Agreement. If a court concludes that any promise or covenant in this Agreement is unreasonable and unenforceable: (a) the court may modify such promise or covenant to the minimum extent necessary to make such promise or covenant to the minimum extent necessary to make such promise or covenant enforceable.
- **Waivers.** We and you may by written instrument unilaterally waive or reduce any obligation of or restriction upon the other. Any waiver granted by us shall be without prejudice to any other rights we may have. We and you shall not be deemed to have waived or impaired any right, power or option reserved by this Agreement (including the right to demand exact compliance with every term, condition and covenant in this Agreement or to declare any breach of this Agreement to be a default and to terminate the franchise before the expiration of its term) by virtue of: (a) any custom or practice of the parties at variance with the terms of this Agreement; (b) any failure, refusal or neglect of us or you to exercise any right under this Agreement or to insist upon exact compliance by the other with its obligations under this Agreement, including any mandatory specification, standard, or operating procedure; (c) any waiver, forbearance, delay, failure or omission by us to exercise any right, power or option, whether of the same, similar or different nature, relating to other Assisted Living Locators franchisees; or (d) the acceptance by us of any payments due from you after breach of this Agreement.
- **25.5.** Approvals. Whenever this Agreement requires our approval, you must make a timely written request for approval, and the approval must be in writing in order to bind us. Except as otherwise expressly provided in this Agreement, if we fail to approve any request for approval within the required period of time, we shall be deemed to have disapproved your request. If we deny approval and you seek legal redress for the denial, the only relief to which you may be entitled is to acquire our approval. You are not entitled to any other relief or damages for our denial of approval.
- **25.6.** Force Majeure. Neither we nor you shall be liable for loss or damage or deemed to be in breach of this Agreement if our or your failure to perform our or your obligations results from any event of force majeure. Any delay resulting from an event of force majeure will extend performance accordingly or excuse performance, in whole or in part, as may be reasonable under the circumstances.

- **25.7. Binding Effect.** This Agreement is binding upon the parties to this Agreement and their respective executors, administrators, heirs, assigns and successors in interest. Nothing in this Agreement is intended, nor shall be deemed, to confer any rights or remedies upon any person or legal entity not a party to this Agreement; provided, however, that the additional insureds listed in Section 16.1 and the Indemnified Parties are intended third party beneficiaries under this Agreement with respect to Section 16.1 and Section 19, respectively.
- 25.8. Integration. THIS AGREEMENT CONSTITUTES THE ENTIRE AGREEMENT BETWEEN THE PARTIES AND MAY NOT, EXCEPT AS PERMITTED BY SECTION 11.2 AND SECTION 25.3, BE CHANGED EXCEPT BY A WRITTEN DOCUMENT SIGNED BY BOTH PARTIES. Any email correspondence or other form of informal electronic communication shall not be deemed to modify this Agreement unless such communication is signed by both parties and specifically states that it is intended to modify this Agreement. The attachment(s) are part of this Agreement, which, together with any Amendments or Addenda executed on or after the Effective Date, constitutes the entire understanding and agreement of the parties, and there are no other oral or written understandings or agreements between us and you about the subject matter of this Agreement. As referenced above, all mandatory provisions of the Manual are part of this Agreement. Any representations not specifically contained in this Agreement made before entering into this Agreement do not survive after the signing of this Agreement. This provision is intended to define the nature and extent of the parties' mutual contractual intent, there being no mutual intent to enter into contract relations, whether by agreement or by implication, other than as set forth above. The parties acknowledge that these limitations are intended to achieve the highest possible degree of certainty in the definition of the contract being formed, in recognition of the fact that uncertainty creates economic risks for both parties which, if not addressed as provided in this Agreement, would affect the economic terms of this bargain. Nothing in this Agreement is intended to disclaim any of the representations we made in the Franchise Disclosure Document. No statement, questionnaire, or acknowledgment signed or agreed to by a franchisee in connection with the commencement of the franchise relationship shall have the effect of (a) waiving any claims under any applicable state franchise law, including fraud in the inducement, or (b) disclaiming reliance on any statement made by any franchisor, franchise seller, or other person acting on behalf of the franchisor. This provision supersedes any other term of any document executed in connection with the franchise.
- 25.9. Covenant of Good Faith. If applicable law implies a covenant of good faith and fair dealing in this Agreement, the parties agree that the covenant shall not imply any rights or obligations that are inconsistent with a fair construction of the terms of this Agreement. Additionally, if applicable law shall imply the covenant, you agree that: (a) this Agreement (and the relationship of the parties that is inherent in this Agreement) grants us the discretion to make decisions, take actions and/or refrain from taking actions not inconsistent with our explicit rights and obligations under this Agreement that may affect favorably or adversely your interests; (b) we will use our judgment in exercising the discretion based on our assessment of our own interests and balancing those interests against the interests of our franchisees generally (including ourselves and our affiliates if applicable), and specifically without considering your individual interests or the individual interests of any other particular franchisee; (c) we will have no liability to you for the exercise of our discretion in this manner, so long as the discretion is not exercised in bad faith; and (d) in the absence of bad faith, no trier of fact in any or litigation shall substitute its judgment for our judgment so exercised.
- **25.10.** Rights of Parties are Cumulative. The rights of the parties under this Agreement are cumulative and no exercise or enforcement by either party of any right or remedy under this Agreement will preclude any other right or remedy available under this Agreement or by law.
- **25.11. Survival**. All provisions that expressly or by their nature survive the termination, expiration or Transfer of this Agreement (or the Transfer of an ownership interest in the franchise) shall continue in full force and effect subsequent to and notwithstanding its termination, expiration or

Transfer and until they are satisfied in full or by their nature expire, including, without limitation, <u>Section 14</u>, <u>Section 15</u>, <u>Section 17</u>, <u>Section 19</u>, <u>Section 22</u>, <u>Section 23</u> and <u>Section 25</u>.

- **25.12.** Construction. The headings in this Agreement are for convenience only and do not define, limit or construe the contents of the sections or subsections. All references to Sections refer to the Sections contained in this Agreement unless otherwise specified. All references to days in this Agreement refer to calendar days unless otherwise specified. The term "you" as used in this Agreement is applicable to one or more persons or an Entity, and the singular usage includes the plural, and the masculine and neuter usages include the other and the feminine and the possessive.
- **25.13. Time of Essence.** Time is of the essence in this Agreement and every term thereof.
- **25.14.** Counterparts. This Agreement may be signed in multiple counterparts, each of which shall be deemed an original and all of which together shall constitute but one and the same document.
- **25.15.** Notice. All notices given under this Agreement must be in writing, delivered by hand, email (to the last email address provided by the recipient) or first-class mail, to the following addresses (which may be changed upon 10 business days prior written notice):

YOU: As set forth in Part A of <u>ATTACHMENT "A"</u>

US: ALL Franchising, LLC

8100 E. Indian School Road, Suite 201

Scottsdale, Arizona 85251

Notice shall be considered given at the time delivered by hand, or one (1) business day after sending by email or comparable electronic system, or three (3) business days after placed in the mail, postage prepaid, by certified mail with a return receipt requested.

The parties to this Agreement have executed this Agreement effective as of the Effective Date first above written.

FRANCHISOR:

ALL Franchising, LLC, a Delaware limited company	liability	
By:		
YOU (If you are an entity):		YOU (If you are not an entity):
a(n)		Name:
By:		
Name:		Name:
Its:		
		Name:

ATTACHMENT "A"

TO FRANCHISE AGREEMENT

DEAL TERMS

A.	. <u>Franchisee Details</u> .									
	Name of Franchisee: [J							
	Is the franchisee one or more natur	al persons signing in their individual	capacity? Yes: No:							
	Type of Entity and State of Formation* (if applicable): []									
	* If the franchisee is a business entity, each natural person holding a direct or indirect ownership interest in the business entity, and spouse of each such person, must sign the Franchise Owner Agreement concurrently with the execution of this Agreement.									
			olding a direct or indirect ownership ble) along with a description of their							
			Direct or Indirect							
	Owner's Name	% Ownership Interest	(if indirect, include description of nature of interest)							
	N. 2 . A 11									
			J							
В.	<u>Initial Franchise Fee</u> .									
	The initial franchise fee payable by	you under this Agreement is identif	ied below (check appropriate box):							
	\$49,900 (1 st territory – no discount)									
	\$47,405 (1 st territory w	th 5% diversity discount)								
	\$44,910 (1 st territory – w	th 10% veteran or nurse discount)								
	\$39,900 (2 nd territory in r	nulti-territory purchase – no discoun	t)							
	\$34,900 (3 rd or subsequer	nt territory in multi-territory purchase	e – no discount)							

C. <u>Territory</u>.

The Ter	rritory	referenced	in the	Franchise	Agreement	shall	consist	of th	e following	geographic	area	(as
further o	lepicte	d on the ma	p attac	hed on the	following pa	ge):						

If there are any changes to the zip codes or other boundaries that define your Territory during the term of the Franchise Agreement or any renewal term, then, unless otherwise agreed to by you and us in writing, the boundaries of your Territory shall remain defined by the zip codes or other boundaries in effect as of the Effective Date and depicted on the map on the following page.

ATTACHMENT "B"

TO FRANCHISE AGREEMENT

FORM OF REFERRAL NETWORK PARTICIPATION AGREEMENT

[See Attached]

REFERRAL NETWORK PARTICIPATION AGREEMENT

This F	Referral 1	Network	Participation	Agreement (this	"Agreen	nent") is entered i	nto this	day of
			, 202 (this	"Effective Date") by and	between		,
with	a	L	principal	place	of	business	located	at
					_("Comr	nunity") and ALL	Franchising, LI	LC, dba
"Assis	sted Livi	ng Locat	ors", a Delay	vare limited liabi	lity com	pany, with its prin	cipal place of b	ousiness
located	d at 8100	E. India	n School Roa	d. Suite 201, Scot	ttsdale. A	rizona 85251 ("AL	L").	

WHEREAS, Community owns, operates or manages senior living community and wishes to participate in ALL's referral network and to engage ALL to provide certain marketing exposure services as described in this Agreement.

WHEREAS, ALL directs seniors and their families (collectively, "Clients") to appropriate care settings based specifically on their preferences and stated needs to communities that participate in ALL's referral network.

NOW, THEREFORE, the parties agree as follows:

- **Section 1. Referral Placement Service.** ALL provides referral and other information to Clients. ALL assists Clients by: (i) helping them find assisted living, Alzheimer's care, retirement communities, homecare and senior care resources by providing information and referrals based on stated needs and preferences, and (ii) offering them a consultative advisor who may further assist them with their search for senior care options. Clients are notified that ALL does not charge Clients for ALL's services and that ALL receives a fee from Community.
- Section 2. Marketing Exposure Services. In consideration of the participation fee described in Section 4, ALL shall include Community's facilities set forth in Schedule A attached hereto in ALL's referral network, and ALL shall provide Community the marketing exposure services as described herein. ALL may employ a variety of marketing exposure methods including on-line listing services through the use of its website, web affiliations, search engine optimization, search engine management and numerous other marketing exposure programs. ALL will maintain an individual webpage listing on their website for the Community so that Community's services are known to the visitors of the website. ALL places no requirements on the manner in which Community provides services to individuals referred by ALL.
- Section 3. Non-Recommendation Policy. ALL neither endorses nor recommends any particular community or facility in its referral network, including Community. The names of subscribing communities are provided to a potential Client based solely on the person's stated needs and preferences and are given as a reference and information source only. ALL provides only an information service and does not (i) perform any medical assessment of or for Client or Participating Agency, or (ii) participate in Client's and Community's decision regarding services. It shall be the sole and exclusive responsibility of Community to review the information concerning all potential clients referred by ALL and to verify and confirm the accuracy of this information as well as information provided by a Client. ALL is not a representative of or agent for either Client or Community and does not act on behalf of either. In the event Community denies the admission of a Client, Community shall immediately notify ALL of such denial and refer Client back to ALL in order for ALL to present Client with alternative community options. Community may not refer any clients introduced to Community by ALL to other facilities/communities. If Community refers clients introduced to Community to another facility/community then Community shall pay those fees that would have been earned by ALL for placing the client.

- **Section 4. Admission Fee.** In consideration of the services provided by ALL hereunder, Community agrees to pay ALL as follows:
- A. Long Term Stays. For each Client referred by ALL and each spouse of Client (whether or not referred by ALL) that moves-in the Community and stays thirty (30) days or more, an amount equal to the greater of eighty-five percent (85%) of the first month's Rent or \$2,500 for Assisted Living. Fee shall be due and payable within thirty (30) days after the move-in date of each Client (or spouse, as applicable).
- **B.** Short Term Stays. For each Client referred by ALL and each spouse of Client (whether or not referred by ALL) that moves-in a Community facility and stays less than thirty (30) days (whether as a respite-stay patient or otherwise), an amount equal to either (i) the greater of thirty percent (30%) of the Client's weekly Rent payment to Community or \$250 per week, or (ii) in the event the resident ultimately stays thirty (30) days or greater, an amount equivalent to the long term stay fee set forth in Section 4(A). Such fee shall be due and payable within thirty (30) days after the move-in date of the Client (or spouse, as applicable).

For purposes of Section 4A and 4B, "Rent" shall be deemed to include the total amounts payable by Client to Community, including amounts payable for services (including health care services) provided by Community, and excluding any discounts or promotions provided to the Client. Any amounts that become past due hereunder will be assessed late fees of the lesser of 1.5% per month (18% per annum) or the maximum percentage allowed under applicable law. Community agrees to notify ALL within seventy-two (72) hours of each Client (and spouse) move-in. The fees set forth in this Section 4 shall apply with respect to any Client referred by ALL and each spouse of Client (whether or not referred by ALL) that moves-in the Community during the twelve (12) month period following any Client contact with the Community through efforts of ALL.

For the states of **Oklahoma and Texas**, a flat rate fee for each Client referred by ALL and each spouse of Client (whether or not referred by ALL) that moves-in and stays thirty (30) days or more, an amount equal to \$2,000 for Assisted Living, \$1,500 for Independent Living, and \$2500 for Memory Care. Fee shall be due and payable within thirty (30) days after the move-in date of each Client (or spouse, as applicable). For each Client referred by ALL and each spouse of Client (whether or not referred by ALL) that moves-in the Community and stays less than thirty (30) days (whether as a respite-stay patient or otherwise), \$250 per week, or (ii) in the event the resident ultimately stays thirty (30) days or greater, an amount equivalent to the long term stay fee set forth in Section 4(A). Such fee shall be due and payable within thirty (30) days after the move-in date of the Client (or spouse, as applicable).

No Federal or State Healthcare Program Consumers: ALL's intent and practice is to ensure that no Prohibited Consumers are referred to Community. A "Prohibited Consumer" is a Consumer whose source of payment for Services is, in whole or in part, provided by Medicare, Medicaid, or similar state laws or to be provided by any federal or state funding source that prohibits the arrangement between the parties contemplated by this Agreement Community agrees that it has the independent responsibility to verify and confirm whether a Consumer is a Prohibited Consumer. ALL will not knowingly refer a Prohibited Consumer to Community. Community will not provide Services to any Prohibited Consumer referred to Community by ALL. Community represents and warrants that the Agencies do not provide federally funded services and do not accept any direct or indirect remuneration in connection with the referral of Prohibited Consumers to entities providing federally funded services. If Community decides to provide federally funded services, Community will immediately notify ALL and this Agreement will immediately and automatically terminate. If the ALL Consumer becomes a Prohibited Consumer, Community will immediately notify ALL. Community will not owe or pay ALL any Fee relating to any period for which the Consumer is a Prohibited Consumer. Should any Consumer become a Prohibited Consumer, the Fee will be refunded.

Section 5. Redundant Leads. The admission fee set forth in Section 4 will not be assessed in the event of the following: (i) the Community has been in contact with the potential Client (including a family member or other involved party) during the ninety (90) day period prior to the date such Client

was referred by Community, (ii) Community has verified such prior contact within five (5) business days of Client's referral to Community in writing to ALL with the initial referral information including the original date of referral, and specific referral source.

- **Section 6. Relationship of the Parties.** The parties hereby agree that they are at all times acting as independent contractors who have entered into this Agreement on the terms and conditions set forth in this Agreement. Nothing in this Agreement will be construed or deemed to create a relationship of employer and employee or principal agent between Community and ALL (or ALL's personnel), and ALL agrees that it shall not assert that the relationship with Community is other than an independent contractor relationship. This obligation will survive any termination of this Agreement.
- **Section 7. Policies and Conduct.** ALL and its personnel shall at all times act ethically and honestly and shall present Community and its facilities in a fair and honest manner.
- Representations and Warranties. ALL and Community each makes the following warranties and representations to the other, which each of them covenants on its own behalf are true and accurate as of the date of this Agreement and shall remain as such during its term: (i) each is a legal entity duly organized, validly existing and in good standing under applicable state law and has all requisite legal power, licenses, certifications and permits to enter into this Agreement and to perform its obligations hereunder; (ii) Community will notify ALL in writing, within 2 business days of any licensing changes, including, without limitation, any revocations, suspensions, conditions, or other limitations imposed by any government authority; (iii) this Agreement has been duly executed and delivered by each and is the legal, valid, and binding obligation of each, fully enforceable against each in accordance with its terms; and (iv) each, including each facility of Community, is duly licensed, certified, accredited or authorized, to the extent required under law, to engage in its business and operations. In addition, Community represents and warrants as of the Effective Date and throughout the term of this Agreement that all information provided by Community to ALL pertaining to Community and its facilities for purposes of the marketing exposure services is accurate and complete; (v) Community agrees that Clients referred by ALL will not be charged any portion of ALL's fees, and, subject to customary and independent negotiation between Client and Community, Client will be charged by Community at the same rates as it charges other persons not referred by ALL, including being eligible for all discounts or concessions offered by Community; (vi) Community will not seek to recover all, or a portion of, fees payable to ALL from Client; (vii) Community will comply with applicable federal, state, or local laws, regulations, rules, orders or judgments.
- **Section 9. Licenses and Permits.** Each party shall obtain and keep valid and in force at all times all licenses or permits that may be necessary to engage in its business and operations. If either party's license or permit is no longer valid and in force for any reason: (i) such party shall immediately notify the other party, and such party so notified may immediately terminate this Agreement. Community shall provide ALL copies of any current licenses, registrations, and/or renewals of Community and any facility of Community, upon ALL's request.
- **Section 10. Termination.** This Agreement shall remain in full force and effect until terminated as follows: (i) By either party, with or without cause, upon at least thirty (30) days prior written notice to the other party, (ii) in the event of material default by either party, the non-defaulting party shall provide notice setting forth the alleged default to the defaulting party, and the non-defaulting party shall have the right to terminate this Agreement immediately if the defaulting party fails to cure the alleged default within ten (10) days after receipt of the notice, or (iii) as set forth in Section 9 of this Agreement. Following termination, this Agreement shall remain in full force and effect as it applies to any referrals which are either in process prior to the thirty (30) day notice period or arise during the thirty (30) day notice period and fees shall be due for all such referrals as set forth in the placement section above.
- **Section 11. Indemnification.** Each party (hereafter, "Indemnitor") agrees to indemnify, defend and hold harmless the other party (hereafter, Indemnitee") and its affiliates, successors, shareholders, personal representatives, directors, officers, employees, partners and agents, from and against any and all claims,

demands, losses damages, penalties, liabilities, suits, judgements, costs, expenses, (including, without limitation, reasonable attorney fees, consultant fees and litigation costs incurred by Indemnitee), arising out of or resulting from, directly or indirectly, any breach of this Agreement, or the intentional or grossly negligent acts or omissions of Indemnitor or its directors, officers, managers, employees or agents, arising out of or pertaining to any actions or inactions of Indemnitor or in connection with the performance of the Agreement by Indemnitor and other regulatory or legal risk of any sort arising out of, in connection to, or resulting from, directly or indirectly, the acts or omissions of Community in its failure to comply with the Fair Labor Standards Act, the Americans with Disabilities Act, Title VII of the Civil Rights Act, the Age Discrimination in Employment Act, the Immigration Reform and Controls Act, the Family and Medical Leave Act, and any and all other federal, state and local employment laws, statutes, ordinances, rules, regulations, codes and/or orders.

- **Section 12. Marketing, Publicity & Trademarks.** Community has the right to and hereby grants ALL a non-exclusive license to incorporate the Community's name, logo, trademark, and other Community information into ALL's marketing products and services. Community hereby affirms that none of that intellectual property violates or infringes upon the rights of others, including their privacy or publicity rights.
- **Section 13. Non-discrimination.** The parties agree that there shall be no discrimination in the performance of this Agreement against any employee, resident, or other person as the result of that individual's race, color, disability, religion, sex, sexual preference, age or national origin or in violation of applicable federal, state or local law and regulation.
- **Section 14. No Third-Party Beneficiaries.** This Agreement shall not create nor be construed to create in any manner whatsoever any rights in any other person as a third-party beneficiary of this Agreement or in any other manner.
- Confidentiality of Records. ALL may provide Community with information regarding Section 15. Client. Community will treat this information as confidential and will only use it for: (i) its own internal evaluation of the potential consumer(s); and (ii) providing Services to such client(s). Community will not disclose this information to any third party and will comply with all applicable state and federal laws respecting the protection, use, and disclosure of such information. Community shall not provide ALL with any individually identifiable health information. The parties shall (iii) comply with all applicable state and federal laws respecting the confidentiality of proprietary information, data and other confidential or personal information concerning the medical, personal, or business affairs of the parties acquired hereunder or in connection herewith; (iv) keep confidential all financial, operating, proprietary or business information relating to the parties that is not otherwise public information; and (v) keep confidential any information, not described above, specified in writing by either party as confidential information. For purposes of the foregoing, information shall not be considered confidential information if such information's disclosure is compelled by court order, by applicable law or if such information was obtained from an unrelated third party not itself subject to a confidentiality requirement with respect to such information; (vi) it will treat confidentially and not transfer to any third party any Client information provided by ALL, including not re-selling or re-distributing the same.
- **Section 16. Assignment and Sub-Contracting.** Both party's rights and obligations under this Agreement may be assigned, delegated, or sub-contracted in whole or in part without the prior written consent of the other party, provided, however, that ALL may assign its obligations under this Agreement to an affiliate of ALL sharing common ownership without the consent of Community.
- **Section 17. Notices.** Any notices required or permitted to be given under this Agreement shall be in writing by mail, overnight service, facsimile or email.
- **Section 18.** Waiver of Right to Jury Trial & Miscellaneous. Each party knowingly and irrevocably waives its right to a jury trial of any claim or cause of action based upon, arising out of, or relating to the subject matter of this Agreement. The scope of this waiver is intended to be allencompassing whether claims are asserted on an individual basis or as putative representative of a class.

This Agreement may not be modified or altered except by a written instrument executed by both parties. The failure of either party to exercise in any respect any right provided for herein shall not be deemed a waiver of any rights. This Agreement, together with all the schedules attached hereto, constitutes the entire Agreement between the parties with respect to the subject matter hereof and supersedes and merges all prior proposals, understandings, and all other agreements, oral and written between the parties relating to such subject matter. It is the intent of the parties hereto that the validity, interpretation and construction of this Agreement shall be governed by the laws of the State of Arizona. Community and ALL agree that any action or proceeding relating to this Agreement must be brought in Maricopa County, AZ.

Section 19. Personal Guarantee. In connection with the transactions contemplated by the Referral Network Participation Agreement, the undersigned hereby absolutely, unconditionally, and irrevocably guarantees the prompt payment and performance (and not merely collection) of Community's obligations under Sections 4, 8, 9, 11 and 15 of the Agreement (collectively, the "Guaranteed Obligations"). This provision has the effect of rendering the undersigned a surety and the equivalent of a co-obligor with Community with respect to the Guaranteed Obligations.

The undersigned agrees that, until the Guaranteed Obligations have been satisfied in full, the undersigned shall not be released by or because of the taking, or failure to take, any action that might in any manner or to any extent vary the risks of the undersigned under this personal guarantee or that, but for this paragraph, might discharge or otherwise reduce, limit, or modify the undersigned's obligations under this personal guarantee. The undersigned waives and surrenders any defense to any liability under this personal guarantee based upon any such action. Without limiting the generality of the foregoing, the undersigned waives the benefits of any provision of law requiring that one or more beneficiaries hereunder exhaust any right or remedy, or take any action, against Community, any other guarantor of Community's obligations, or any other person and/or property.

IN WITNESS WHEREOF, the parties have executed this Agreement as of the day and year first above written.

Assisted Living Locators Referral Source	Community		
By (Print Name)		By (Print Name)	
Signature		Signature	
Title		Title	-
8100 E. Indian School Road, S Scottsdale, Arizona 85251	Suite 201		

(T) 800-267-7816; (F) 877-275-5991

Exhibit A (use if signing up more than one property)

Facilities

ATTACHMENT "C" TO FRANCHISE AGREEMENT FRANCHISE OWNER AGREEMENT

[See Attached]

FRANCHISE OWNER AGREEMENT

This Franchise Owner Agreement (this "<u>Agreement</u>") is entered into by: (i) each of the undersigned owners of Franchisee (defined below); and (ii) the spouse of each such owner, in favor of ALL Franchising, LLC, a Delaware limited liability company, and its successors and assigns ("<u>us</u>"), upon the terms and conditions set forth in this Agreement. Each signatory to this Agreement is referred to as "<u>you</u>".

- 1. **Definitions.** For purposes of this Agreement, the following terms have the meanings given to them below:
 - "ALL Business" means an Assisted Living Locators business, whether operated by us, our affiliate or a franchisee (including Franchisee).
 - "Business Records" means any and all evidence of each business transaction, and all financial, marketing, and other operating aspects of the Franchised Business, and all evidence and records with respect to customers, clients, employees, and other service professionals relating the Franchised Business including, without limitation, all databases in print, electronic or other form, including all names, addresses, phone numbers, email addresses, customer/client records, and all other records contained in the database, and all other records created and maintained by Franchisee relating to the operation of the Franchised Business, including all such data pertaining to clients, Senior Communities, Referral Sources and Senior Service Providers.
 - "Competitive Business" means any business competitive with an ALL Business that assists seniors and their families in locating and evaluating suitable assisted living facilities, memory care communities, nursing homes, senior care homes or independent living senior communities.
 - "Copyrights" means all works and materials for which we or our affiliate has secured common law or registered copyright protection and that we allow ALL Businesses to use, sell or display in connection with the marketing and/or operation of an ALL Business, whether now in existence or created in the future.

Franchise Agreement" means the Assisted Living Locators Franchise Agreement executed by Franchise ith an effective date of, 202	O
Franchised Business" means the ALL Business operated by Franchisee pursuant to the Franchise greement.	
Franchisee" means	Franchisee" means

- "Improvements" means any additions, modifications or improvements (including creative ideas) to (a) the goods or services offered at an ALL Business, (b) the method of operation of an ALL Business or (c) any marketing or promotional ideas relating to an ALL Business, whether developed by you, Franchisee or any other person.
- "Intellectual Property" means, collectively or individually, our Marks, Copyrights, Know-how, System and Improvements.
- "Know-how" means all of our trade secrets and other proprietary information relating to the development, marketing and/or operation of an ALL Business, including, but not limited to, methods, techniques, assessments, specifications, procedures, policies, marketing strategies and information comprising the System and the Manual. Know-how also includes the identities and information regarding Senior Communities and Referral Sources that are entered into ALL-IN as well as all information comprising Franchisee's Business Records.
- "Manual" means our confidential brand standards manual for the operation of an ALL Business.
- "Marketing Agreement" means any agreement between an ALL Business and a Senior Community pursuant to which the Senior Community agreed to pay a fee to the ALL Business for clients placed at such Senior Community.
- "Marks" means the logotypes, service marks, and trademarks now or hereafter involved in the operation of

an ALL Business, including "Assisted Living Locators" and the related logo, and any other trademarks, service marks or trade names that we designate for use by an ALL Business.

"Prohibited Activities" means any or all of the following: (a) owning, operating or having any other interest (as an owner, partner, director, officer, employee, manager, consultant, shareholder, creditor, representative, agent or in any similar capacity) in a Competitive Business (other than owning an interest of five percent (5%) or less in a publicly traded company that is a Competitive Business); (b) diverting or attempting to divert any business from us (or one of our affiliates or franchisees); (c) inducing a client of any ALL Business to transfer their business to any person other than an ALL Business; (d) inducing any Senior Community that you know, or have reason to know, has entered into a Marketing Agreement with any ALL Business (including with Franchisee), to enter into an agreement for placement services with any person other than an ALL Business; or (e) inducing any Referral Source that you know, or have reason to know, has entered into a referral agreement or established a referral relationships with any ALL Business (including with Franchisee), to refer clients in need of senior living placement services to any person other than an ALL Business.

"Referral Source" means any hospital, insurance company, HMO, transitional care company or other similar types of business or institution that is in a position to refer clients in need of placement services.

"Restricted Period" means the two (2) year period after the earliest to occur of the following: (a) the termination or expiration of the Franchise Agreement; (b) the date on which Franchisee assigns the Franchise Agreement to another person with respect to whom neither you nor your spouse holds any direct or indirect ownership interest; or (ci) the date on which you cease to be an owner of Franchisee or your spouse ceases to be an owner of Franchisee, as applicable; provided however, that if a court of competent jurisdiction determines that this period of time is too long to be enforceable, then the "Restricted Period" means the one (1) year period after the earliest to occur of the following: (a) the termination or expiration of the Franchise Agreement; (b) the date on which Franchisee assigns the Franchise Agreement to another person with respect to whom neither you nor your spouse holds any direct or indirect ownership interest; or (c) the date on which you cease to be an owner of Franchisee, as applicable.

"Restricted Territory" means the geographic area within: (a) the Territory granted to Franchisee under the Franchise Agreement; and (b) the county or counties in which Franchisee's Territory is located, in whole or in part; provided, however, that if a court of competent jurisdiction determines that the foregoing Restricted Territory is too broad to be enforceable, then the "Restricted Territory" means the geographic area the Territory granted to Franchisee under the Franchise Agreement.

"Senior Care Providers" means providers of various senior care services (other than placement services) that may be necessary or helpful to Franchisee's clients, including real estate agents, downsizing specialists, home health experts, financial planners, financing companies and elder-care attorneys.

"Senior Community" means an assisted living facility, nursing home, senior care home, independent living senior community or similar type of facility.

"System" means our distinct system for the operation of an ALL Business, the distinctive characteristics of which include logo, trade secrets, client assessment tools and techniques, marketing strategies, lead generation programs, referral programs, and operating system.

2. Background. In your capacity as an owner of Franchisee, or the spouse of an owner of Franchisee, you may gain knowledge of our System and Know-how. You understand that protecting the Intellectual Property is vital to our success and that of our franchisees and that you could seriously jeopardize our entire franchise system if you were to unfairly compete with us. In addition, you understand that certain terms of the Franchise Agreement apply to "owners" and not just Franchisee. You agree to comply with the terms of this Agreement In order to: (i) avoid damaging our System by engaging in unfair competition; and (ii) bind yourself to the terms of the Franchise Agreement applicable to owners.

3. Brand Protection Covenants.

- (a) Intellectual Property. You agree: (i) you will not use the Know-how in any business or capacity other than the Franchised Business operated by Franchisee; (ii) you will maintain the confidentiality of the Know-how at all times; (iii) you will not make unauthorized copies of documents containing any Know-how; (iv) you will take such reasonable steps as we may ask of you from time to time to prevent unauthorized use or disclosure of the Know-how; and (v) you will stop using the Know-how immediately if you are no longer an owner of Franchisee or your spouse is an owner of Franchisee, as applicable. You further agree that you will not use the Intellectual Property for any purpose other than the development and operation of the Franchised Business pursuant to the terms of the Franchise Agreement and Manual. You agree to assign to us or our designee, without charge, all rights to any Improvement developed by you, including the right to grant sublicenses. If applicable law precludes you from assigning ownership of any Improvement to us, then such Improvement shall be perpetually licensed by you to us free of charge, with full rights to use, commercialize, and sublicense the same.
- (b) <u>Unfair Competition During Relationship</u>. You agree not to unfairly compete with us at any time while you are an owner of Franchisee or while your spouse is an owner of Franchisee, as applicable, by engaging in any Prohibited Activities.
- (c) <u>Unfair Competition After Relationship</u>. You agree not to unfairly compete with us during the Restricted Period by engaging in any Prohibited Activities; provided, however, that the Prohibited Activity relating to having an interest in a Competitive Business will only apply with respect to a Competitive Business that places clients with Senior Communities that are located within the Restricted Territory. If you engage in any Prohibited Activities during the Restricted Period, then you agree that your Restricted Period will be extended by the period of time during which you were engaging in the prohibited activity (any such extension of time will not be construed as a waiver of your breach or otherwise impair any of our rights or remedies relating to your breach).
- (d) Immediate Family Members. You acknowledge that you could circumvent the purpose of this Agreement by disclosing Know-how to an immediate family member (i.e., parent, sibling, child, or grandchild). You also acknowledge that it would be difficult for us to prove whether you disclosed the Know-how to family members. Therefore, you agree that you will be presumed to have violated the terms of this Agreement if any member of your immediate family (i) engages in any Prohibited Activities during any period of time during which you are prohibited from engaging in the Prohibited Activities or (ii) uses or discloses the Know-how. However, you may rebut this presumption by furnishing evidence conclusively showing that you did not disclose the Know-how to the family member.
- (e) <u>Covenants Reasonable</u>. You acknowledge and agree that: (i) the terms of this Agreement are reasonable both in time and in scope of geographic area; and (ii) you have sufficient resources and business experience and opportunities to earn an adequate living while complying with the terms of this Agreement. You Hereby Waive any right to Challenge the terms of this Agreement. You Hereby Waive any right to Challenge the terms of this Agreement as Being overly broad, unreasonable or otherwise unenforceable. Although you and we both believe that the covenants in this Agreement are reasonable in terms of scope, duration and geographic area, we may at any time unilaterally modify the terms of the system protection covenants in Section 3 of this Agreement, upon written notice to you, by limiting the scope of the Prohibited Activities, narrowing the definition of a Competitive Business, shortening the duration of the Restricted Period, reducing the geographic scope of the Restricted Territory and/or reducing the scope of any other covenant imposed upon you under Section 3 of this Agreement to ensure that the terms and covenants are enforceable under applicable law
- (f) <u>Breach</u>. You agree that failure to comply with the covenants in this <u>Section 3</u> will cause substantial and irreparable damage to us and/or other ALL Businesses for which there is no adequate remedy at law. Therefore, you agree that any violation of these covenants will entitle us to injunctive relief. You agree that we may apply for such injunctive relief, without bond, but upon due notice, in addition to such further and other relief as may be available at equity or law, and the sole remedy of yours, in the event

of the entry of such injunction, will be the dissolution of such injunction, if warranted, upon hearing duly held (all claims for damages by reason of the wrongful issuance of any such injunction being expressly waived hereby). If a court requires the filing of a bond notwithstanding the preceding sentence, the parties agree that the amount of the bond shall not exceed \$1,000. None of the remedies available to us under this Section are exclusive of any other, but may be combined with others under this Agreement, or at law or in equity, including injunctive relief, specific performance and recovery of monetary damages.

- (g) Applicable State Law. If applicable state law prohibits us from imposing the brand protection covenants set forth in this Section on the spouse of an owner of Franchisee (who is not also an owner of Franchisee), then the brand protection covenants set forth in this Section 3 shall be deemed inapplicable to such spouse.
- **4. Transfer Restrictions.** If you are an owner of Franchisee, you acknowledge that we must approve all persons who hold a direct or indirect ownership interest in Franchisee. Accordingly, you agree that you will not, directly or indirectly or by operation of law, sell, assign, mortgage, pledge or in any manner transfer any direct or indirect ownership interest in Franchisee except in accordance with the terms and conditions set forth in Section 20.2 of the Franchise Agreement.
- 5. Financial Security. In order to secure Franchisee's financial obligations under the Franchise Agreement and all ancillary agreements executed by Franchisee in connection with the Franchise Agreement, including, but not limited to, any agreement for the purchase of goods or services from us or an affiliate of ours and any promissory note related to payments owed to us (collectively, the "Secured Agreements"), you, jointly and severally, personally and unconditionally: (a) guarantee to us and our successor and assigns, that Franchisee shall punctually fulfil all of its payment and other financial obligations under the Secured Agreement; and (b) agree to be personally bound by, and personally liable for, each and every monetary provision in the Secured Agreements. You waive: (1) acceptance and notice of acceptance by us of the foregoing undertakings; (2) notice of demand for payment of any indebtedness guaranteed; (3) protest and notice of default to any party with respect to the indebtedness guaranteed; (4) any right you may have to require that an action be brought against Franchisee or any other person as a condition of liability; and (5) the defense of the statute of limitations in any action hereunder or for the collection of any indebtedness hereby guaranteed. You agree that: (1) your direct and immediate liability under this guaranty shall be joint and several with Franchisee and all other signatories to this Agreement; (2) you will render any payment required under the Secured Agreements upon demand if Franchisee fails or refuses punctually to do so; (3) your liability shall not be contingent or conditioned upon pursuit by us of any remedies against Franchisee or any other person; and (4) liability shall not be diminished, relieved or otherwise affected by any extension of time, credit or other indulgence that we may grant to Franchisee or to any other person, including the acceptance of any partial payment or performance, or the compromise or release of any claims, none of which shall in any way modify or amend this guarantee, which shall be continuing and irrevocable during the term of each of the Secured Agreements and following the termination, expiration or transfer of each of the Secured Agreements to the extent any financial obligations under any such Secured Agreements survive such termination, expiration or transfer. This guaranty will continue unchanged by the occurrence of any bankruptcy with respect to Franchisee or any assignee or successor of Franchisee or by any abandonment of one or more of the Secured Agreements by a trustee of Franchisee. Neither your obligation to make payment in accordance with the terms of this undertaking nor any remedy for enforcement shall be impaired, modified, changed, released or limited in any manner whatsoever by any impairment, modification, change, release or limitation of the liability of Franchisee or its estate in bankruptcy or of any remedy for enforcement, resulting from the operation of any present or future provision of the U.S. Bankruptcy Act or other statute, or from the decision of any court or agency.
- 6. Dispute Resolution. Any dispute between the parties relating to this Agreement shall be brought in accordance with the dispute resolution procedures set forth in the Franchise Agreement. Notwithstanding the foregoing, if any of the dispute resolution procedures set forth in the Franchise Agreement conflict with any of the terms of this Agreement, the terms of this Agreement shall prevail. You acknowledge and agree that a breach of this Agreement by you shall constitute a material event of default under the Franchise Agreement, permitting us to terminate the Franchise Agreement in accordance with the

terms thereof.

7. Miscellaneous.

OWNER / SPOUSE

- (a) If either party hires an attorney or files suit against the other party relating to or alleging a breach of this Agreement, the losing party agrees to pay the prevailing party's reasonable attorneys' fees and costs incurred in connection with such breach.
- (b) This Agreement will be governed by, construed and enforced under the laws of Arizona and the courts in that state shall have jurisdiction over any legal proceedings arising out of this Agreement.
- (c) Any claim, defense or cause of action that you may have against us or against Franchisee, regardless of cause or origin, cannot be used as a defense against our enforcement of this Agreement.
- (d) Each section of this Agreement, including each subsection and portion thereof, is severable. In the event that any section, subsection or portion of this Agreement is unenforceable, it shall not affect the enforceability of any other section, subsection or portion; and each party to this Agreement agrees that the court may impose such limitations on the terms of this Agreement as it deems in its discretion necessary to make such terms reasonable in scope, duration and geographic area.
- (e) You agree that we may deliver to you any notice or other communication contemplated by this Agreement in the same manner and to the same address listed in the notice provisions of the Franchise Agreement and any such delivery shall be deemed effective for purposes of this Agreement. You may change the address to which notices must be sent by sending us a written notice requesting such change, which notice shall be delivered in the manner and to the address listed in the Franchise Agreement.

IN WITNESS WHEREOF, each of the undersigned has executed this Agreement as of the date or dates set forth below.

By:	
Name:	
Date:	
OWNER / SPOUSE	
By:	
Name:	
Date:	
OWNER / SPOUSE	
OWNER / SPOUSE By:	
By:	
By:	
By:	
By:	
By: Name: Date:	
By: Name: Date: OWNER / SPOUSE	

ATTACHMENT "D" TO FRANCHISE AGREEMENT ACH AUTHORIZATION FORM

[See Attached]

Franchisee Information:		
Franchisee Name		Business No.
Franchisee Mailing Address (stree	t)	Franchisee Phone No.
Franchisee Mailing Address (city,	state, zip)	
Contact Name, Address and Phone	e number (if different from above)	
Franchisee Fax No.		Franchisee Email Address
Bank Account Information:		
Bank Name		
Bank Mailing Address (street, city	, state, zip)	
Bank Account No.	(check one)	Bank Routing No. (9 digits)
Bank Mailing Address (city, state,	zip)	Bank Phone No.
Authorization:		
above and Franchisee authorizes t made from time to time in an ame and Franchisee as well as to cover be bound by the National Autom entries will be initiated only as a written notification from Franchise	he Bank to accept and to debit the amount of ount sufficient to cover any fees payable to F any purchases of goods or services from Fran ated Clearing House Association (NACHA) athorized above. This authorization is to rem ee of its termination in such time and in such the shall notify Franchisor of any changes to an	debit entries to Franchisee's account with the Bank listed such entries to Franchisee's account. Each debit shall be ranchisor pursuant to any agreement between Franchisor achisor or any affiliate of Franchisor. Franchisee agrees to rules in the administration of these debit entries. Debi ain in full force and effect until Franchisor has received manner as to afford Franchisor and the Bank a reasonable by of the information contained in this authorization form
Signature:		Date:
Name:		
Its:		

ATTACHMENT "E" TO FRANCHISE AGREEMENT

BRAND PROTECTION AGREEMENT

[See Attached]

BRAND PROTECTION CONFIDENTIALITY AGREEMENT

CONFIDENTIALITY AGREEMENT

This <u>Confidentiality</u> Agreement (this "<u>Agreement</u>") is entered into by the undersigned ("<u>you</u>") in favor of ALL Franchising, LLC, a Delaware limited liability company, and its successors and assigns ("<u>us</u>"), upon the terms and conditions set forth <u>in this Agreement.below.</u>

1. **Definitions.** For purposes of this Agreement, the following terms have the meanings given to them below:

<u>"ALL Business"</u> means an Assisted Living Locators business, whether operated by us, our affiliate or a franchisee (including Franchisee).

"Business Records" means any and all evidence of each business transaction, and all financial, marketing, and other operating aspects of the Franchised Business, and all evidence and records with respect to customers, clients, employees, and other service professionals relating the Franchised Business including, without limitation, all databases in print, electronic or other form, including all names, addresses, phone numbers, email addresses, customer/client records, and all other records contained in the database, and all other records created and maintained by Franchisee relating to the operation of the Franchised Business, including all such data pertaining to clients, Senior Communities, Referral Sources and Senior Service Providers.

"Confidential Information" means and includes: (a) the Know-How; (b) the terms of the Franchise Agreement and all related agreements signed by Franchisee in connection with the ALL Business, and all attachments thereto and amendments thereof; (c) the components of the System; (d) all information within or comprising the Manual; and (e) all other concepts, ideas, trade secrets, financial information, marketing strategies, expansion strategies, studies, supplier information, customer information, franchisee information, investor information, flow charts, inventions, mask works, improvements, discoveries, standards, specifications, formulae, recipes, designs, sketches, drawings, policies, processes, procedures, methodologies and techniques, together with analyses, compilations, studies or other documents that: (i) are designated as confidential; (ii) are known by you to be considered confidential by us; and/or (iii) are by their nature inherently or reasonably to be considered confidential. Confidential Information does not include any information that: (a) is now, or subsequently becomes, generally available to the public (except as a result of a breach of confidentiality obligations by you, Franchisee or Franchisee's owners, employees or other constituents); (b) you can demonstrate was rightfully in your possession, without obligation of nondisclosure, before the information was disclosed to you by us or Franchisee (or any person associated with us or Franchisee); (c) is independently developed by you without any use of, or reference to, any Confidential Information; or (d) is rightfully obtained from a third party who has the right to transfer or disclose such information to you without breaching any obligation of confidentiality imposed on such third party.

<u>"Copyrights"</u> means all works and materials for which we or our affiliate has secured common law or registered copyright protection and that we allow ALL Businesses to use, sell or display in connection with the marketing and/or operation of an ALL Business, whether now in existence or created in the future.

"Franchise Agreement"	means the	Assisted	Living L	ocators	Franchise	Agreement	executed	by	Franchisee
with an effective date of			, 202	2					

<u>"Franchised Business"</u> means the ALL Business operated by Franchisee pursuant to the Franchise Agreement.

<u>"Franchisee"</u> means the Assisted Living Locators franchisee for whom you are an officer, director, employee or independent contractor.

"Improvements Improvement" means any additions, modifications or improvements (including creative ideas) to (a) the goods or services offered at an ALL Business, (b) the method of operation of an ALL Business or (c) any marketing or promotional ideas relating to an ALL Business, whether developed by you, Franchisee or any other person.

<u>"Intellectual Property"</u> means, collectively or individually, <u>our Marks,the</u> Copyrights, <u>Know how, System and Improvements, Know-How (including all of Franchisee's Business Records and all data in the ALL-IN database), Marks and System.</u>

<u>"Know-how"</u> means all of our trade secrets and other proprietary information relating to the development, marketing and/or operation of an ALL Business, including, but not limited to, methods, techniques, assessments, specifications, procedures, policies, marketing strategies and information comprising the System and the Manual. Know-how also includes the identities and information regarding Senior Communities and Referral Sources that are entered into ALL-IN as well as all information comprising Franchisee's Business Records.

"Manual" means our confidential brand standards manual for the operation of an ALL Business.

<u>"Marks"</u> means the logotypes, service marks, and trademarks now or hereafter involved in the operation of an ALL Business, including "Assisted Living Locators" and the related logo, and any other trademarks, service marks or trade names that we designate for use by an ALL Business.

"Prohibited Activities" means any or all of the following: (a) diverting or attempting to divert any business from us (or one of our affiliates or franchisees); (b) inducing a client of any ALL Business to transfer their business to any person other than an ALL Business; (c) inducing any Senior Community that you know, or have reason to know, has entered into a Marketing Agreement with any ALL Business (including with Franchisee), to enter into an agreement for placement services with any person other than an ALL Business; (d) inducing any Referral Source that you know, or have reason to know, has entered into a referral agreement or established a referral relationships with any ALL Business (including with Franchisee), to refer clients in need of senior living placement services to any person other than an ALL Business.

<u>"Referral Source"</u> means any hospital, insurance company, HMO, transitional care company or other similar types of business or institution that is in a position to refer clients in need of placement services.

<u>"Senior Care Providers"</u> means providers of various senior care services (other than placement services) that may be necessary or helpful to Franchisee's clients, including real estate agents, downsizing specialists, home health experts, financial planners, financing companies and elder-care attorneys.

<u>"Senior Community"</u> means an assisted living facility, nursing home, senior care home, independent living senior community or similar type of facility.

<u>"System"</u> means our distinct system for the operation of an ALL Business, the distinctive characteristics of which include logo, trade secrets, client assessment tools and techniques, marketing strategies, lead generation programs, referral programs, and operating system.

- 2. Background. You are an officer, director, employee or independent contractor of Franchisee. As a result of this association, you may gain knowledge of our System and Know-how. You understand that protecting the Intellectual Property is vital to our success and that of our franchisees and that you could seriously jeopardize our entire franchise system if you were to unfairly compete with us. In order to avoid such damage, you agree to comply with the terms of this Agreement.
- 3.—<u>Know-How and Intellectual Property</u>. You agree: (a) you will not use the Know-how in any business or capacity other than the Franchised Business operated by Franchisee; (b) you will to:
 - (i) maintain the confidentiality of the Know-how and other Confidential Information at all times; (e) you will not make unauthorized copies of documents containing any Know how; (d) you will
 - (ii) take suchall reasonable steps as-we may ask of you from time to time require to prevent unauthorized use or disclosure of the Know how; and (e) you will stop using the Know how immediately if you are no longer an officer, director, employee or independent contractor of Franchisee. You further agree that you will not use the Know-how or other Confidential Information;
 - (iii) refrain from using the Intellectual Property or other Confidential Information in any capacity or for any purpose other than the performance of your duties for Franchisee and within the scope of your

- employment or other engagement with Franchisee.solely in connection with the operation of the Franchised Business;
- <u>4. Prohibited Activities during the Term.</u> You agree not to engage in refrain from making unauthorized copies of documents containing any Prohibited Activities Know-how or other Confidential Information;
- (v) refrain from directly or indirectly contacting or soliciting any Senior Community, Referral Source, Senior Care Provider, or client of an ALL Business for any business purpose unrelated to the operation of the Franchised Business; and
- (vi) immediately stop using the Intellectual Property and Confidential Information at any such time while that you are no longer an officer, director, employee or independent contractor of Franchisee.
- 5. Ownership of Phone Number. You must obtain a separate telephone number solely for use with the Franchised Business. You acknowledge and agree that we will own this phone number and the associated account but allow you to use it during the terms of the Franchise Agreement at your expense. You must immediately transfer the number to us upon the termination, expiration or transfer of the Franchise Agreement and you agree to sign any documents we require to enable us to independent assign the telephone number to us in compliance with this Section.
 - **6.** Immediate You agree to assign to us or our designee, without charge, all rights to any Improvement developed by you, including the right to grant sublicenses. If applicable law precludes you from assigning ownership of any Improvement to us, then such Improvement shall be perpetually licensed by you to us free of charge, with full rights to use, commercialize and sublicense the same.
- 4. Family Members. You acknowledge that Because you could circumvent the purpose of this Agreement by disclosing Know-how or Confidential Information to an immediate family member (i.e., spouse, parent, sibling, child, or grandchild). You also acknowledge that and it would be difficult for us to prove whether you disclosed the Know how to family members. Therefore, you agree that any such breach, you will be presumed to have violated the terms of breached this Agreement if any a member of your immediate family (a) engages in any Prohibited Activities during any period of time during which you are prohibited from engaging in the Prohibited Activities or (b) uses or discloses disclosed the Know-how or other Confidential Information. However, you may rebut this presumption by furnishing with evidence conclusively showing that you did not disclose the Know-how to the family member.

7. Breach

- 5. You agree that failure to comply with Covenants Reasonable. You acknowledge and agree that: (a) the terms of this Agreement are reasonable both in time and in scope of geographic area; and (b) you have sufficient resources and business experience and opportunities to earn an adequate living while complying with the terms of this Agreement.
- G. Breach. You agree that: (a) any failure to comply with this Agreement will is likely to cause substantial and irreparable damage to us and/or other ALL Businesses franchisees for which there is no adequate remedy at law. Therefore, you agree that any violation of the terms of this Agreement will entitle us; and (b) we are entitled to injunctive relief. You agree that we may apply for such injunctive relief, without bond, but upon due notice, in addition to such further and other relief as may be if you breach this Agreement together with any other relief available at equity or law, and the sole remedy of yours, in the event of the entry of such injunction, will be the dissolution of such injunction, we will notify you if warranted, upon hearing duly held (all claims for damages by reason of the wrongful issuance of any such injunction being expressly waived hereby) we intend to seek injunctive relief, but we need not post a bond. If a court requires the filing ofthat we post a bond notwithstanding despite our mutual agreement to the preceding sentence contrary, the parties agree that the required amount of the bond shall may not exceed \$1,000. If an injunction is wrongfully issued, you may seek dissolution of the injunction as your sole remedy. You hereby waive all claims for damages resulting from a wrongfully issued injunction. None of the remedies available to us under this Agreement are exclusive of any other, but may be combined with others under this Agreement, or at law or

in equity, including injunctive relief, specific performance and recovery of monetary damages. Any No claim, defense or cause of action that held by you may have against us, our affiliate or against Franchisee, regardless of cause or origin, cannot may be used as a defense against our enforcement of this Agreement.

7.__8. Miscellaneous.

÷

- (a) If we hire an attorney or file suit against you because you have breached for breach of this Agreement and we prevail against, you, you agree to pay must reimburse us for our reasonable attorneys' fees and costs in doing so.
- (b) This Agreement will be governed by, construed and enforced under the laws of Arizona and the courts in that state shall have jurisdiction over any legal proceedings arising out of this Agreement.
- (c) _Each section of this Agreement, including each subsection _(and portion thereof,) is severable. In the event that If any section, subsection _(or portion of this Agreement thereof) is unenforceable, it shall not affect the enforceability of any other section, subsection _(or portion; and each party to this Agreement agrees that the _thereof). A court may impose such limitations on the terms revise any provision of this Agreement as it deems in its discretion to the extent necessary to make such terms reasonable in scope, duration and geographic area, the provision enforceable.

This Brand ProtectionConfidentiality Agreement is executed as of the date or dates set forth below.

Ву:	 	 	
Name:		 	
Date:			

ATTACHMENT A

Restricted Territory

[Insert description of Franchisee's Territory]

ATTACHMENT "F"

TO FRANCHISE AGREEMENT

DEFINITIONS

- "Account" is defined in Section 14.7.
- "Acquisition" means either (a) a competitive or non-competitive company, franchise system, network or chain directly or indirectly acquiring us, whether in whole or in part, including by asset or stock purchase, change of control, merger, affiliation or otherwise or (b) us directly or indirectly acquiring another competitive or non-competitive company, franchise system, network or chain, whether in whole or in part, including by asset or stock purchase, change of control, merger, affiliation or otherwise.
- "Agencies" is defined in Section (xiii).
- "Agreement" is defined in the Introductory Paragraph.
- "ALL Business" is defined in Section 1.
- "ALL-IN" is defined in Section 6.2.
- "Brand Protection Agreement" means our form of Brand Protection Agreement, the most current form of which is attached to this Agreement as <u>ATTACHMENT</u>".
- "Business" is defined in Section 2.
- "Business Records" means any and all evidence of each business transaction, and all financial, marketing, and other operating aspects of the Business, and all evidence and records with respect to customers, clients, employees, and other service professionals relating the Business including, without limitation, all databases in print, electronic or other form, including all names, addresses, phone numbers, email addresses, customer/client records, and all other records contained in the database, and all other records created and maintained by you relating to the operation of your Business, including all such data pertaining to clients, Senior Communities, Referral Sources and Senior Service Providers.
- -"Claim" or "Claims" means any and all claims, actions, demands, assessments, litigation, or other form of regulatory or adjudicatory procedures, claims, demands, assessments, investigations, or formal or informal inquiries.
- "Competitive Business" means any business competitive with an ALL Business that assists seniors and their families in locating and evaluating suitable assisted living facilities, memory care communities, nursing homes, senior care homes or independent living senior communities.
- "Confidentiality Agreement" means our form of Confidentiality Agreement, the most current form of which is attached to this Agreement as ATTACHMENT "E".
- "Copyrights" means all works and materials for which we or our affiliate has secured common law or registered copyright protection and that we allow ALL Businesses to use, sell or display in connection with the marketing and/or operation of an ALL Business, whether now in existence or created in the future.
- "Dispute" is defined in Section 23.
- "Effective Date" is defined in the Introductory Paragraph.
- "Entity" means a corporation, partnership, limited liability company or other form of association.
- "Expansion Territory" is defined in Section 3.3(b).
- "General Release" means our current form of general release of all claims against us and our affiliates and subsidiaries, and our and their respective members, officers, directors, agents and employees, in both

their corporate and individual capacities.

"Gross Collected Referral Proceeds" means all gross sums that you collect in connection with Corporate Leads that we refer to you, including, without limitation, placement fees from placing Corporate Leads and referral fees paid by Senior Service Providers for referrals of Corporate Leads.

"Gross Collected Revenues Revenue" means all gross sums that you collect in any way relating to your Business, including, without limitation, revenues revenue from the sale or provision of goods or services, proceeds from business interruption insurance and referral fees paid to you, including referral fees paid by Referral Sources, Senior Care Providers, other ALL Businesses (such as fee splits under our Referral Program) or us (such as fee splits relating to a Regional Account). "Gross Collected Revenues Revenue" includes any Gross Collected Referral Proceeds that you collect. With respect to fees generated through our Referral Program, "Gross Collected Revenues Revenue" only includes your share of the placement fee that you collect. "Gross Collected Revenues Revenue" excludes the following amounts (which you may deduct from your calculation of Gross Collected Revenues Revenue): (a) the amount of any referral fee you pay to us for a Corporate Lead; (b) the amount of any placement fee you collect but pay to another ALL Business under our Referral Program ("Gross Collected Revenues Revenue" includes your share of the placement fee); (c) the amount of any sales or use tax that you collect and pay to the applicable taxing authority; and (d) the wholesale price paid by you for any merchandise or other retail items that you sell ("Gross Collected Revenues Revenue" includes the difference between the retail price you receive from the sale of the item and the wholesale price that you paid for the item).

"Improvements" is defined in Section 18.6.

"Indemnified Party" or "Indemnified Parties" means us and each of our past, present and future owners, members, officers, directors, employees and agents, as well as our parent companies, subsidiaries and affiliates, and each of their past, present and future owners, members, officers, directors, employees and agents.

"Intellectual Property" means, collectively or individually, our Marks, Copyrights, Know-how, System and Improvements.

"Interim Manager" is defined in Section 8.4.

"Interim Term" is defined in Section 4.3.

"Know-how" means all of our trade secrets and other proprietary information relating to the development, marketing and/or operation of an ALL Business, including, but not limited to, methods, techniques, assessments, specifications, procedures, policies, marketing strategies and information comprising the System and the Manual. Know-how also includes the identities and information regarding Senior Communities and Referral Sources that are entered into ALL-IN as well as all information comprising your Business Records.

"Losses and Expenses" means all compensatory, exemplary, and punitive damages; fines and penalties; attorneys' fees; experts' fees; court costs; costs associated with investigating and defending against Claims; settlement amounts; judgments; compensation for damages to our reputation and goodwill; and all other costs, damages, liabilities and expenses associated with any of the foregoing losses and expenses or incurred by an Indemnified Party as a result of a Claim.

"Managing Owner" means the Owner that you designate and we approve who is primarily responsible for the daily management and supervision of the Business.

"Manual" is defined in Section 6.1.

"Marketing Agreement" is defined in Section 11.5.

"Marks" means the logotypes, service marks, and trademarks now or hereafter involved in the operation of an ALL Business, including "Assisted Living Locators" and the related logo, and any other trademarks, service marks or trade names that we designate for use by an ALL Business.

"Owner" or "Owners" means any individual who owns a direct or indirect ownership interest in the franchise or the Entity that is the franchisee under this Agreement. "Owner" includes both passive and active owners.

"Peer Group" is defined in Section 11.16.

"Permitted Transfer" means: (a) a Transfer from one Owner to another Owner who was an approved Owner prior to such Transfer, other than a Transfer by an Owner who is the Managing Owner that results in the Managing Owner holding less than a 20% ownership interest in the franchise or the Entity that is the franchisee under this Agreement, as applicable; and/or (b) a Transfer to a newly established Entity for which the Owners collectively own and control 100% of the ownership interests and voting power.

"Post-Term Restricted Period" means, with respect to you, a period of two (2) years after the termination, expiration or Transfer of this Agreement; provided, however, that if a court of competent jurisdiction determines that the two-year Post-Term Restricted Period is too long to be enforceable, then the "Post-Term Restricted Period" means, with respect to you, a period of one (1) year after the termination, expiration or Transfer of this Agreement. "Post-Term Restricted Period" means, with respect to an Owner, a period of two (2) years after the earlier to occur of (a) the termination, expiration or Transfer of this Agreement or (b) the Owner's Transfer of his or her entire ownership interest in the franchise or the Entity that is the franchisee, as applicable; provided, however, that if a court of competent jurisdiction determines that the two-year Post-Term Restricted Period is too long to be enforceable, then the "Post-Term Restricted Period" means, with respect to an Owner, a period of one (1) year after the earlier to occur of (a) the termination, expiration or Transfer of this Agreement or (b) the Owner's Transfer of his or her entire ownership interest in the franchise or the Entity that is the franchisee, as applicable.

"Prohibited Activities" is defined in Section 15.3.

"Referral Services Agreement" is defined in Section 11.6.

"Referral Sources" is defined in Section (i).

"Regional Account" is defined in Section 3.2(a).

"Restricted Territory" means the geographic area within: (a) your Territory; and (b) the county or counties in which your Territory is located (in whole or in part); provided, however, that if a court of competent jurisdiction determines that the foregoing Restricted Territory is too broad to be enforceable, then the "Restricted Territory" means the geographic area within your Territory.

"Senior Communities" is defined in Section (i).

"Senior Service Providers" is defined in Section (iii).

"Successor Agreement" is defined in Section 4.1.

"System" means our distinct system for the operation of an ALL Business, the distinctive characteristics of which include logo, trade secrets, client assessment tools and techniques, marketing strategies, lead generation programs, referral programs, and operating system.

"Targeted Marketing" is defined in Section 3.1(b).

"Technology Systems" is defined in Section 11.15.

"Term" is defined in Section 4.1.

"Territory" is defined in Section 2.

"Transfer" means any direct or indirect, voluntary or involuntary (including by judicial award, order or decree), assignment, sale, conveyance, subdivision, sublicense or other transfer or disposition of the franchise (or any interest therein), the Business (or any portion thereof) or an ownership interest in an Entity that is the franchisee, including by merger or consolidation, by issuance of additional securities representing an ownership interest in the Entity that is the franchisee, or by operation of law, will or a trust upon the death of an Owner (including the laws of intestate succession).

"We" or "us" is defined in the Introductory Paragraph.

"You" is defined in the Introductory Paragraph.

EXHIBIT "D"

TO DISCLOSURE DOCUMENT

TABLE OF CONTENTS OF BRAND STANDARDS MANUAL

[See Attached]



Table of Contents

Introduction to Assisted Living Locators	7
1.1 Welcome to Assisted Living Locators	7
1.2 Assisted Living Locators History and Development	7
1.3 Assisted Living Locators Services	7
1.4 Purpose of this Manual	8
1.5 Importance of Confidentiality	8
1.6 Disclaimer	9
2. Terminology ^F	10
3. Policies Guides, Manuals and Certifications	17
3.1 Compliance & Enforcement	17
3.2 Federal Disclosure Document & Franchise Agreement	17
3.3 Brand Protection Agreement	18
3.4 Brand Guide	19
3.5 Brand Platform	20
3.6 Code of Conduct	21
3.7 Acceptable Use Policy	22
3.8 Privacy Policy	23
3.9 HIPAA Policy	24
3.10 Websites Policy	24
3.11 ALL-IN [™] Data Policy	26
3.12 Social Media Policies & Procedures	27
3.13 Training	29
3.14 Certifications	34
4. Overview of Services Provided to Franchisee	37
4.1 Before You Open Your Assisted Living Locators Franchise, We Will:	37
4.2 During the Operation of your Assisted Living Locators Franchise, We Will:	38
4.3 During the Operation of your ALL Franchise, We May, But Need Not:	39
5. Getting Started ^F	41
5.1 Forms of Business Ownership	41

5.2 Licenses & Permits	45
5.3 Business Laws	45
5.4 Taxes	47
5.5 Naming Your Entity	50
5.6 Insurance	51
5.7 Elder Abuse Policy	52
6. Required Equipment and Services	59
6.1 Vehicle	61
6.2 Computers	61
6.3 Computer System Back-up	62
6.4 Printer, Copier, Scanner, & Fax	63
6.5 High Speed Internet Connection	64
6.6 Telephone Systems	64
6.7 Website, Email, & G-Suite	67
6.8 Software	71
6.9 Equipment & Software Maintenance & Service	72
6.10 Technology Fee	73
7. Bookkeeping & Records F	75
7.1 Bookkeeping, Inspections & Audits	75
7.2 Invoicing for Services	76
7.3 Split Fee Placement and Refund Policy	82
7.4 Receiving Payment	82
7.5 Fee Adjustment	87
7.6 Split Fee Placement & Refund Policy	87
7.7 Invoicing for Revenue Reports	92
8. Administration and Employee Management ^F	95
8.1 Owner Day-to-Day Participation	95
8.2 Employee Selection Process	96
8.3 Suggested Employee Policies	112
8.4 Termination	114
9. Advertising and Marketing Your Services	122
9.1 Why Advertising and Marketing are Important	122
9.2 Defining Your Marketing Objectives	123
9.3 Begin by Marketing Around You	124
9.4 Resource Hierarchy	127

9.5 Marketing Methods	128
9.6 Marketing to Healthcare Professionals	133
9.7 Marketing Outside of Healthcare	144
9.8 Marketing to Community Resources	148
9.9 Internet Marketing	150
9.10 Marketing Limitations	155
9.11 Advertising	155
9.12 Targeted Advertising & Marketing	163
9.13 Corporate Guidance	166
10. Senior Communities ^F	167
10.1 The ALL & Senior Community Relationship	167
10.2 Senior Community Requirements	168
10.3 Building Your Inventory	169
10.4 Initial Contact	174
10.5 Initial Tour Appointment	176
10.6 Referral Network Partner Agreements	184
10.7 Community Inventory Database	185
10.7 Meeting with the Owners/Managers	193
10.8 Evaluating the Care Facility	193
10.9 Entering Information on Your Website's Facility Manager	194
10.10 Evaluating the Care Facility Review Results	199
10.11 Facility Insurance	200
11. Client Acquisition ^F	201
11.1 ALL-IN TM Entry	201
11.2 Organic Referral Sources	203
11.3 Paid Leads Generation Sources	204
11.4 Paid Advertising Programs	207
11.5 Responding to the Referral	209
11.6 Answering the Telephone	216
11.7 Tracking Referral Sources	221
11.8 Referral Thank You's	221
12. Client Information and Demographics	223
12.1 General Assessment Guidelines	223
12.2 Initial Client Demographics	225
13. Placement and Touring Guidelines	244

_ 0	13.1 The Process	244
3	13.2 Determining the Right Environment	244
-3	13.3 Identifying Client Priorities	245
0	13.4 Referring the Client to the Senior Community	246
13	13.5 Touring the Client	248
- 3	13.6 Client Placement	251
3	13.7 Issues You May Encounter Post Move-In	252
14.	Your Territory, Referral and Split Placements ^F	254
10	14.1 Activities within Your Territory	254
1	14.2 Your Operations Outside of Your Territory	254
	14.3 Referral ProgramF	255
43	14.4 Active and Passive Approach	257
d	14.5 Split-Fee Notification, Invoicing and Refund Request	257
15.	Additional Revenue Streams	259
	15.1 Private Duty Non-Medical Companies	259
G	15.2 Subscription-Based Fees	259
Ġ	15.3 Revenue Streams	259
16.	Financial Options to Help Pay for Care ^F	260
	16.1 Medicaid	260
	16.2 Long-Term Care Insurance	263
	16.3 Bridge Loan	264
13	16.4 Reverse Mortgage	264
	16.5 Life Insurance	265
13	16.6 Veteran Aid and Attendance	268
1	16.7 Private Pay	270
	APPENDIX:	271
	REFERRAL ALERT	273
	EINANCIAL ODTIONS TO HELD DAY FOR CARE OLUZ	27/

EXHIBIT "E"

TO DISCLOSURE DOCUMENT

LIST OF FRANCHISEES

$\underline{Part\ A}\ (Current\ Franchisees)$

State	City	Address	Phone	Owner Name(s)
Alabama	Birmingham	1465 Valley Trace Irondale, Alabama 35210	205-420-8046	Sarah Garcia Nourish with Freedom, LLC
Alabama	Huntsville	2714 Biscayne St SW Huntsville, AL 35805	256 542 0223	Kim Krause
Arizona ¹	Chandler-Tempe	5959 E Baseline Rd, Mesa, AZ 852063133 West Frye Road, Suite 101, PMB 1209, Chandler, Arizona 85226	480-649-0008	To-Dan and Jenni Henderson Dustin Baker Grace ful Living, LLC
Arizona ¹	Gilbert	5959 E Baseline Rd, Mesa, AZ 852063133 West Frye Road, Suite 101, PMB 1209, Chandler, Arizona 85226	480-649-0008	<u>Dan and Jenni Henderson</u> Dustin Baker Grace ful Living, LLC
Arizona ¹	Glendale	3948 E Sunnydale Dr. Queen Creek, <u>AZArizona</u> 85142	623-703-5326	Lori Sears MOZONA ALL, INC
Arizona	Mesa	18185 E. Meadow Hill Drive Scottsdale, AZArizona 85260	480-685-9822	Jay & Melissa Stuckey JMS & Associates, LLC
Arizona ¹	Mesa East	5959 E Baseline Rd, Mesa, AZ 852063133 West Frye Road, Suite 101, PMB 1209, Chandler, Arizona 85226	480-649-0008	Dan and Jenni Henderson Dustin Baker Grace-ful Living, LLC
Arizona	North Phoenix	14039 N. Canterbury Dr. Phoenix, AZArizona 85023	602-774-4342	Chad Kinney Paramount Placement, LLC
<u>Arizona</u>	North Tucson	300 S Corte Tortuga Vista, Tucson, Arizona 85745	<u>520- 882-3772</u>	Rob and Sheri Lewin 4Ever Partners, Inc.
Arizona	Phoenix	7510 West Julie Dr Glendale, AZArizona 85308	602-254-7500	Tera Johnson
Arizona	Prescott	4801 N Verde Vista Dr. Prescott Valley, <u>AZArizona</u> 86314	928-421-1872	Ashly Blackwell ALL NAZ LLC
Arizona	Scottsdale	24012 N. 76th Place Scottsdale, AZ 852553133 West Frye Road, Suite 101, PMB 1209, Chandler, Arizona 85226	480-454- 5611 <u>649-0008</u>	Rachel Wasserstrom Pamikis, LLCDan and Jenni Henderson
Arizona ¹	Surprise	3948 E Sunnydale Dr. Queen Creek, <u>AZArizona</u> 85142	623-703-5326	Lori Sears Mozona ALL, INC
California ¹	Beverly Hills	525 Alandele Ave Los Angeles, <u>CACalifornia</u> 90036	310-853-8282	Sarah Ordover Senior Living Advisors, LLC
California ¹	Covina	599 S. Barranca Suite 216 Covina, CACalifornia 91723	626-966-7111	LeeAnn Allman Allman Assisted Transitions INC
California	Encinitas- Oceanside	6389 Paseo Descanso Carlsbad, CACalifornia 92009	760-904-6017	Susan Carey Individual
<u>California</u>	Huntington Beach	2901 W Coast HWY #200 Newport Beach, CA 92663	949 405 6937	Simone Kelly

State	City	Address	Phone	Owner Name(s)
California	Palm Springs	74351 Parosella St Palm Desert, CA 922603699 Taylor Dr, Palm Springs, California 92262	760-576-3000	Kendall Forbes David Capra Individual
California	San Francisco	403 Broderick St, Apt #2 San Francisco, CA <u>California</u> 94117	415-766-0361	Brent Hensley Compassionate Directions, Inc.
California ¹	San Gabriel	599 S. Barranca Suite 216 Covina, CACalifornia 91723	626-966-7111	LeeAnn Allman Allman Assisted Transitions INC
<u>California</u>	San Mateo	403 Broderick St, Apt #2 San Francisco, California 94117	415-766-0361	Brent Hensley
<u>California</u>	Santa Anna	90 Summerland Cir Aliso Viejo, California 92656	949-480-1137	Negar Lencioni
California ¹	Santa Monica	454 ½ S Cloverdale Avenue Los Angeles, <u>CACalifornia</u> 90036	310-853-8282	Sarah Ordover, Senior Living Adviosr, Inc.
California	Santa Rosa	691 N Oak St. Ukiah, CA <u>California</u> 95482	707-536-0514	Ramona & Matt Maurer ALL North Bay LLC
<u>California</u>	South Orange County	<u>3 Thorp Spring</u> Ladera Ranch, California 92694	949-610-8977	Jay Minor
<u>California</u>	<u>Temecula</u>	<u>29801 Holsteiner Way</u> <u>Menifee, California 92584</u>	951-246-6150	<u>Liza Cerame</u>
California	Thousand Oaks	42 West Columbia Rd Thousand Oaks, <u>CACalifornia</u> 91360	805-277-3005	Angelique Sieverson Angelique for ALL, LLC
California	Torrance	2605 Graham Avenue A Redondo Beach, <u>CACalifornia</u> 90278	310-975-3168	Jennifer Weintraub
California	West Inland Empire	4633 Wheeler Ave La Verne, <u>CACalifornia</u> 91750	909-284-8888	Vincent Bonnemere Bonnemere Boys, Inc.
Colorado ¹	Boulder	471 Expedition Ln Johnstown, COColorado 80534	877-223-1426	Maureen Walker ALL NOCO Seniors, LLC
Colorado	Colorado Springs	12575 McClelland Rd Calhan, CO <u>Colorado</u> 80808	281-691-6450	Victoria Barcus Senior Services of Southern Colorado, Inc.
Colorado ¹	Denver	10552 Stable Lane Littleton, CO <u>Colorado</u> 80125	303-957-2555	Ed Wagner Simple Transitions, INC
Colorado ¹	Littleton	10552 Stable Lane Littleton, CO <u>Colorado</u> 80125	303-957-2555	Ed Wagner Simple Transitions, INC
Colorado ¹	North Denver	10552 Stable Lane Littleton, COColorado 80125	303-957-2555	Ed Wagner Simple Transitions, INC
Colorado / Wyoming ¹	Fort Collins / Cheyenne	471 Expedition Lane Johnstown, COColorado 80534	970-660-8898	Maureen Walker ALL NOCO Seniors, LLC
Connecticut	Greenwich	174 Round Hill Road Greenwich, CTConnecticut 06831	203-651-6866	Mimi Santry Sanctuary Senior Living, LLC
Connecticut ¹	Hartford	111 Elm Street Milton, <u>MAMassachusetts</u> 02186	413-329-6730	Kevin Smith The People Group, Inc.
Delaware	Delaware	8731 Sunny Gallop Dr. Tomball, <u>TX Texas</u> 77375	901-356-4716	Leia Lynch-Franklin & Stephen Franklin Individual
District of Columbia ¹	Washington	6005 Harland St. Lanham, MDMaryland 20706	240-472-9220	Patricia Russell TLC Nurse Enterprise, LLC
Florida	Augustine	135 Jenkins St Ste 105 B St. Augustine, FL 32086	904-417-3292	Victor Burke & Kelly Graham

State	City	Address	Phone	Owner Name(s)
Florida ¹	Boca Raton	111 Gibraltar St Royal Palm Beach, FL 33411 Florida 33411	561-692-4905	Sabrena Rodriguez, FSMAM Services Incorporated
Florida	Clearwater	3400 Cove Cay Dr 1A Clearwater, FL 33760	727-608-8959	Kristin Downer
Florida ¹	Daytona	2862 Sonoma Way Rockledge FL 329552573 Temple St NE Palm Bay, FL 32905	772 783 9449 <u>321-367-</u> 3520	Shane & Misty Clark Individual Terese and Graham Garson
Florida ¹	Ft. Myers-Naples	3337 Baltic Drive Naples, FLFlorida 34119	941-479-3500	Marcus & Chrissy Moser, Moser Group LLC
Florida	Fort Lauderdale	490 SW 62nd Avenue Plantation, PLFlorida 33317	754-600-3390	Alex Escobar
Florida ¹	Jupitar	2862 Sonoma Way Rockledge FLFlorida 32955	772-783-9449	Shane & Misty Clark Individual
Florida ¹	N. Jacksonville	4845 Algonquin Ave Jacksonville, <u>FLFlorida</u> 32210	904-559-3203	Brian Sheridan The Sheridan Group, Inc.
Florida	N. Tampa	20015 Outpost Point Dr Tampa, FLFlorida 33647	813-906-5664	Tee Sharma
Florida	NE Orlando	1033 Dunraven Dr Winter Park, FLFlorida 32792	407-498-2536	Beau & Mercer Herman B & M Forever, Inc
Florida	Palm Bay	2573 Temple St NE Palm Bay, PL <u>Florida</u> 32905	321-367-3520	Terese Kennedy-Garson
Florida	Palm Harbor	3929 Mullenhurst Drive Palm Harbor, FLFlorida 34685	813-523-3068	James Bullock Safe and Sound Senior Care Corporation
Florida	Pensacola	960 Langley Ave Pensacola FL, Florida 32504	850-542-3088	John Horton Calrns 4ds, INC
Florida	Port Charlotte	P.O. Box 100923 Cape Coral, FL Florida 33910	239-990-1949	Gisel Garcia New World Investment Ventures, LLC
Florida ¹	S. Jacksonville	4845 Algonquin Ave Jacksonville, FLFlorida 32210	904-559-3203	Brian Sheridan The Sheridan Group, Inc.
Florida ¹	Sarasota	3337 Baltic Drive Naples, FLFlorida 34119	941-479-3500	Marcus & Chrissy Moser, Moser Group LLC
<u>Florida</u>	St. Petersburg	6987 Indus Valley Cir Parrish, Florida 34219	<u>727-723-8000</u>	<u>Liz Eurich-Plevin</u>
Florida	Tampa	1100 Old Brewster Rd. Fort Meade, FL Florida 33841	813-922-2500	Mary Curtis Pizzano
Florida ¹	West Palm Beach	111 Gibraltar St Royal Palm Beach Florida, 33411, Florida33411	561-692-4905	Sabrena Rodriguez, FSMAM Services Inc
Georgia	Athens	1561 Whitlow Ridge Dr Bisop, GAGeorgia 30621	706-310-8143	Kendall Kineard
Georgia	North Atlanta	56 Geranium CT Acworth, GAGeorgia 30101	404-921-0064	Tara and Greg Schramm
Georgia	North Atlanta	56 Geranium CT Acworth, GAGeorgia 30101	404-921-0064	Tara and Greg Schramm
Georgia	North Georgia	4435 Rhonda Lane Cumming, GAGeorgia 30040	678-965-3459	Greg Santini Santini Insights, Inc.
Idaho <u>Georgia</u>	SE AtlantaBoise	3884 E 12 N Rigby, ID 83442 <u>694</u> <u>Fraser Street</u> Southeast Atlanta, Georgia 30315	404-936- 9902 208-953 7775	Brian Sullivan Yvonne Carlson, Teton Peaks PTA Services Pe.
Illinois	Arlington Heights	606 Lexington Ct. Mundelein, <u>H_Illinois</u> 60060	847-603-3110	Don Hull, Savant Senior Consulting, LLC

State	City	Address	Phone	Owner Name(s)
<u>Illinois</u>	<u>Des Plaines</u>	551 Roosevelt Rd. @234 Glen Ellyn, Illinois 60137	331-208-5041	Allison Green
Illinois	Naperville	2020 Stoffard Ave Wheaton, <u>Hallinois</u> 60187	331-208-5041	Allison Green ALG Senior Consulting, LLC
<u>Illinois</u>	<u>Plainfield</u>	252 Foster Dr. Oswego, Illinois 60543	630-797-8462	Kym Schoenfeld
Indiana	North Indianapolis	6029 Haverford Ave Indianapolis, <u>INIndiana</u> 46220	317-854-9030	David Holder DM Holder Enterprises, LLC
Indiana	Northwest Indiana	7574 Boardwalk Circle Crown Point, <u>INIndiana</u> 46307	219-226-6665	Laura Certa LTC Enterprise, LLC
Kansas	Kansas City	11001 W 141st Street Overland Park, KSKansas 66221	913-538-5435	Mike Nixon A.L.L Kansas City, LLC
Kansas ¹	Topeka	12756 E Bradford Circle Wichita, <u>KSKansas</u> 67206	316-448-5090	John Weber Midwest Senior Consulting, LLC
Kansas ¹	Wichita	12756 E Bradford Circle Wichita, <u>KSKansas</u> 67206	316-448-5090	John Weber Midwest Senior Consulting, LLC
Kentucky/ Indiana	North Nashville Louisville Clarksville	10003 Foxboro Dr. Louisville, KY 402231244 Drakes Ridge Lane Bowling Green, KY 42103	615-359- 3553 <mark>502 208-</mark> 4072	Joe Douglas Jason Coomer Individual
<u>Louisiana</u>	Baton Rouge	7162 Lakeland Drive Zachary, Louisiana 70791	225-663-5911	Tonja Jackson
Louisiana	New Orleans	1450 Annunciation Street New Orleans, <u>LALouisiana</u> 70130	443-307-9572	Germaine Simon P&G LLC
Maine	Portland	141 Wallis Road Rye, NHNew Hampshire 03870	207-835-8115	Shawn Shambo, Shambo, INC
Maryland	Annapolis	215 Sumner Road Annapolis, MDMaryland 21401	410-610-6343	Jen McAndrews Individual
Maryland ¹	Baltimore Heights	6005 Harland St. Lanham, MD Maryland 20706	240-472-9220	Patricia Russell TLC Nurse Enterprise, LLC
Maryland	Bel Air	1209 Hoskins Terrace Apt. 104 Bel Air, MDMaryland 21014	302-593-1818	Ryan Majchrzak R & J Senior Advisors Inc.
Maryland	<u>Bethesda</u>	7215 White House Dr Springfield, Virginia 22153	703-878-7870	Sean Maxwell
Maryland ²	Towson	6 Houndswood Court Baltimore, MDMaryland 21209	443-266-2757	Joanne Wigod, Individual Capcity
Massachusetts	Andover	11 Rivers St. Amesbury, MAMassachusetts 01913	855-562-7747	Jon Younger Younger Senior Services, Inc.
Massachusetts ¹	Boston	111 Elm Street Milton, <u>MAMassachusetts</u> 02186	617-981-2444	Kevin Smith The People Group, Inc
Massachusetts	Bridgewater	PO Box 762 North Scituate, MAMassachusetts 02060	508-681-3016	Stephen Carrier
Massachusetts ¹	Springfield	111 Elm Street Milton, MAMassachusetts 02186	413-329-6730	Kevin Smith The People Group, Inc
Massachusetts	Worchester- Framingham	29 Magnolia Way Bridgewater, MAMassachusetts 02324	617- 924-9200	Mary Zabian MCZ Enterprises, Inc.
Michigan ¹	Ann Arbor	2911 Lakehurst Court Ann Arbor <u>MIMichigan</u> 48105	734-355-3687	Arlene Caddell Treorai INC
Michigan	Bay City	130 Carroll Blvd Bay City, <mark>MI</mark> Michigan	989-331-6628	Richard Martin, Ram 34 Inc.
Michigan	<u>Detroit</u>	199 Hampshire Drive Troy, Michigan 48085	<u>586-208-9938</u>	<u>Larry Leto</u>

State	City	Address	Phone	Owner Name(s)
Michigan ¹	Downriver	16731 Rosa Lane Southgate, MMichigan 48195	248-702-1492	Carolyn Marengere Strana, INC
Michigan ¹	Grand Rapids	9574 Stone View Dr. NE Rockford, MMichigan 49341	616-425-2244	Tom Carabellese Encore Senior Consulting Inc.
Michigan ¹	Holland	9574 Stone View Dr. NE Rockford, MMichigan 49341	616-425-2244	Tom Carabellese Encore Senior Consulting Inc.
Michigan ¹	Livonia	2911 Lakehurst Court Ann Arbor MI, Michigan 48105	734-355-3687	Arlene Caddell Treorai INC
Michigan	Pontiac	760 2nd St. Fenton, MMichigan 48430	810-643-1182	William Anderson
Michigan ¹	Southfield	16731 Rosa Lane Southgate, MMichigan 48195	248-702-1492	Carolyn Marengere Strana, INC
Michigan ¹	Sterling Heights	16731 Rosa Lane Southgate, <u>MIMichigan</u> 48195	248-702-1492	Carolyn Marengere Strana, INC
Minnesota	<u>Duluth</u>	581 W Redwing St Duluth, Minnestoa 55803	218-321-0030	Kailee Olson
Minnesota	Eden Prairie	19827 Iris Way Lakeville, <u>MNMinnestoa</u> 55044	952-300-3180	Danelle Neudecker
Minnesota	Rochester	19827 Iris Way Lakeville, Minnesota 55044	952-300-3180	Danelle Neudecker
Minnesota	St. Paul	5509 River Bluff Curve Bloomington, MNMinnesota 55437	651-262-4550	Sarah Hallmark-Brower
Minnesota	W. Minneapolis	2004 West 140 th Street Burnsville, MNMinnesota 55337	952-377-8676	Steve & Angie Johnson
Mississippi	Tupelo	384 Hwy. 15 North, Pontotoc, MSMississippi 38863	662-470-3009	Carla Lunn Lunn Locators, LLC
<u>Missouri</u>	Kansas City	7701 NE 108th Terrace Kansas City, Missouri 64157	816-599-0093	Les Whisler
<u>Missouri</u>	Pleasant Hill	7701 NE 108th Terrace Kansas City, Missouri 64157	816-599-0093	<u>Les Whisler</u>
Missouri	St. Charles	539 Fox Pointe Drive St. Charles MO, Missouri 63304	636-577-9543	Sharon Osborn
Missouri	St. Louis	617 Hickory Manor Arnold, MO <u>Missouri</u> 63010	314-380-3030	Rob Howe A.L.L. St. Louis, LLC
Nebraska	Greater Omaha	PO Box 3193 Omaha, <u>NENebraska</u> 68103	402-671-0446	Nikki Wulff Wulff Enterprises, LLC
Nebraska/Iowa	Nebraska – Council Bluffs	19664 Virginia Hills Council Bluffs, <u>IAIowa</u> 51503	402-300-3182	Jean Bohnker
New Hampshire ¹	Manchester	6 Clough St Meredith, NHNew Hampshire 03253	603-393-6542	Glen Badger, ALL in NH LLC
New Hampshire ¹	Portsmouth	6 Clough St Meredith, NHNew Hampshire 03253	603-393-6542	Glen Badger, ALL in NH LLC
New Jersey	Bergen County	6 Maryann Court Paramus, NJ New Jersey 07652	201-345-5006	Joseph Chu Eldercare Living Solutions, LLC
New Jersey ¹	East Orange	6 Riverview Dr. W Montclair, NJNew Jersey 07043	973-413-1158	Richard Lanning Eli Senior Services, LLC
New Jersey	Freehold	47 Clark Street Manasquan, NJ New Jersey 08736	732-837-9203	John (Jack) Stewart

State	City	Address	Phone	Owner Name(s)
New Jersey	Greater Cherry Hill	198 River Road Washington Crossing, PAPennsylvania 18977	856-302-4440	Pat Donnelly P. F. Donnelly Enterprises, LLC
New Jersey	Morristown	136 E. Center Street Midland Park, NJ New Jersey 07432	973-536-1250	Eric Moore Moore Care Solutions, Inc.
New Jersey	New Brunswick	2 Williamsburg N Colts Neck, New Jersey 07722	732-915-5616	Tom Morris
New Jersey ¹	Paterson	6 Riverview Dr. W Montclair, NJNew Jersey 07043	973-413-1158	Richard Lanning Eli Senior Services, LLC
New York	<u>Buffalo</u>	23 Hunt Ave, Hamburg, New York 14075	716-246-1310	Colleen McGuire
New York	Long Island	42 Prospect Avenue Glen Cove, New York 11542	516-254-9481	Mike McClernon
New York	New Rochelle	200 California Rd Unit 1 Bronxville, NYNew York 10708	914-362-1745	Leana Walsh LWalsh Adult Placement Corp.
New York	Rochester	385 French Rd, Pittsford, New York14534	<u>585-888-3032</u>	Neil Russell
North Carolina	Durham	11540 Jordanus Dr. Raleigh, <u>NCNorth Carolina</u> 27617	919-241-3414	Allison Schneider-Gulledge ASG Solutions, Inc.
North Carolina	Raleigh	109 Oaken Place Apex, North Carolina 27539	919-300-4117	Jim & Patty Condon Assisted Living Advisers of North Carolina, Inc.
North Carolina	South-Charlotte	6851 Thornton Oaks Court 5220 Rounding Run Rd Charlotte, NC 28270 North Carolina 28277	704-761- 4595 <u>980-236-</u> 7084	Ken & Laura O'Dea Lifestyle Transitions, INCMark Daley
North Carolina	Wilmington	1238 Sea Lavender Dr. Wilmington, North Carolina 28403	910-742-3797	<u>Lexi Beall</u>
Ohio	Akron	4647 Azalea Lane North Olmstead, OHOhio 44070	330-222-5136	Mindi Bub, MMB Senior Transitions, LLC
Ohio	Dayton	7018 Blackoak Drive Reynoldsburg, OHOhio 43068	614-344-4640	Karen Harrison, KBH Services, LLC
Ohio ¹	East Cincinnati	5935 Cleves Warsaw Pike, Cincinnati, OHOhio 45233	513-914-1980	David Flautt, Kind Enterprises Inc.
Ohio	East Cleveland	17768 Lost Trl Chagrin Falls, OHOhio 44023	330-348-0957	David Koch Herona, LLC
Ohio	East Columbus	421 Woodside Meadows Place Gahanna, Ohio 43230	614-636-4905	Miriam Bogard Senior Care Services of Ohio, LLC
<u>Ohio</u>	<u>Toledo</u>	1240 Champe RD, Toledo, Ohio 43615	213-550-9186	Mohammad Alfadhel
Ohio ¹	West Cincinnati	5935 Cleves Warsaw Pike, Cincinnati, OHOhio 45233	513-914-1980	David Flautt, Kind Enterprises Inc.
Ohio	West Columbus	7652 Sawmill Rd. Suite 306 Dublin, OHOhio 43016	614-347-6140	Jasilika Davidson, Davidson Holdings, Inc.
Oregon	East Portland	5515 SW Boundary St Portland, OR 97221	503 560 0405	Russell Morgan Eastside ALL, INC
Oregon	SW Portland	2110 Southwest Nevada Court, Portland, OROregon 97219	503-610-9226	Peter Wilhelm Caring Next Steps, LLC
Pennsylvania	Berwyn	229 Biddle Drive Exton, PAPennsylvania 19341	484-873-8735	John Sherlock, Senior Living Advisors of Chester County, Inc
Pennsylvania	Harrisburg-York	36 Woburn Abbey Ave Camp Hill, PAPennsylvania 17011	717-580-2043	Rick Campomizzi, RC Media LLC
Pennsylvania	N. Pittsburgh	1419 Macon Ave Pittsburgh, PAPennsylvania 15218	724-373-2133	Jennifer Thayer JBT Cares, Inc.

State	City	Address	Phone	Owner Name(s)
Pennsylvania	Pittsburgh	109 Brittany Drive Dunlevy, PAPennsylvania 15432- 0653	724-909-0018	Audie Eddy
Rhode Island ¹	Providence	111 Elm Street Milton, MAMassachusetts 02186	401-787-1940	Kevin Smith The People Group, Inc
South Carolina	Charleston	1668 Seignious Dr. 4 Asbury Place Charleston, SC. South Carolina 29407	<u>843-746-9404</u>	Allison Lanford Leslie Jackson Jaxson Enterprises, Inc.
South Carolina	Greenville	14 Middlemarch Drive Simpsonville, SCSouth Carolina 29681	864-900-2525	Mark & Renee Milstead, Senior Care Advisor, LLC
South Carolina ¹	North Columbia	317 Fernview Dr Columbia, SC South Carolina 29229	803-470-1011	Ramona Britt ALL Midlands, LLC
South Carolina ¹	South Columbia	317 Fernview Dr- <u></u> Columbia, SC South Carolina 29229	803-470-1011	Ramona Britt ALL Midlands, LLC
Tennessee ¹	North Nashville	204 Cambridge Pl Franklin, TN37067 Tennessee 37067	615-375-3553	Tim Tuttle, The Caring Bridge, Inc.
Tennessee ¹	South Nashville	204 Cambridge Pl Franklin, TN37067 Tennessee 37067	615-375-3553	Tim Tuttle, The Caring Bridge, Inc.
Texas	Amarillo-Lubbock	5526 79th St. Lubbock, TX Texas 79424	806-310-4007	Brad Hodge
Texas	Flower Mound	241 S Heartz Rd. Coppell, TX 75019	908-447-1042	Richard Malcolm Tuesday Blast, LLC
Texas	Fort Worth	8490 Old Springtown Rd. Springtown, TX Texas 76082	817-888-8648	Karen Mitchell Cornerstone Hard Surface Solutions, Inc.
Texas	Houston	15722 Gettyburg Drive Houston, TXTexas 77377	832-646-9880	Alane Roberts Assisted Living Locators Houston, LLC
Texas	Irving	3905 Treemont Circle Colleyville, TXTexas 76034	469-250-5220	Mitzie & Jerry Watson DFW Area Eldercare Advisors LLC
Texas	North Austin	4108 Promontory Point Trl Georgetown, TX 78626	512-575-3357	Laura Langley
Texas	Plano	18947 Westwood Place Dallas, TX Texas 75287	972-805-8849	Traci Duff, Compassionate Partners LLC
Texas	SW Houston	1421 Georgina Street Rosenberg, TX 774712722 Glen Haven Blvd Houston, Texas 77025	<u>281-691-6321</u>	Dan & Hannah Wier Robert Wolter Brazos Concierge, LLC
Utah	N. Salt Lake City	4266 S 1325 E Ogden, UT Utah 84403	801-316-0046	Nicholas Gittins North Utah Senior Advisors, LLC
<u>Utah</u>	Wasatch Front South	6461 S. Canyon Cove Dr. Holladay, Utah 84121	801-997-8607	Matt Larson
<u>Virginia</u>	Arlington Alexandri	2359 Battery Hill Circle, Woodbridge, VA 221917215 White House Dr. Springfield, Virginia 22153	<u>703-878-7870</u>	Sean Maxwell Kris & Alisha Jones, KAS Business Group, LLC
Virginia	Fredericksburg	17949 Main Street, #363, Dumfries, VA <u>Virginia</u> 22026	202-302-5740	Henrietta Young RYoung Enterprices, Inc.

State	City	Address	Phone	Owner Name(s)
Virginia	Tuckahoe	10440 Greglynn Rd. North Chesterfield, VA <u>Virginia</u> 23236	804-643-2600	Bill Thompson Assisted Living Locators of Greater Richmond LLC
Washington ¹	Bellevue	38423 251st PL24304 SE 30th St. Enumelaw, WA 98022Sammamish, Washington 98075	425-310- 2665253 319- 2428	Sarfaraz Reddy James & Cindy Holston Sterling Resources Northwest Inc.
Washington ¹	Federal Way	38423 251st PL SE Enumclaw, WAWashington 98022	425-243-9750	James & Cindy Holston Sterling Resources Northwest Inc.
Washington- [‡]	<u>Kingsgate</u> Olympia	11503 11 th 24304 SE 30th St Ct E ₂ Edgewood, WA 98372Sammamish, Washington 98075	425-310- 2665253 319- 2428	Sarfaraz Reddy Tracy Raymond TR Health Care Advisors, Inc.
Washington ¹	Tacoma	11503 11 th St Ct E Edgewood, WA 98372	253 319 2428	Tracy Raymond TR Health Care Advisors, Inc.
Wisconsin	Eau Clair	N8148 577 th St Colfax, WIWisconsin 54730	715-628-2011	Jody Shea
Wisconsin ¹	Lake Geneva	1043 N Osborne Blvd Racine, <u>WIWisconsin</u> 53405	262-289-4020	Jennifer Eastman
Wisconsin ¹	Milwaukee	1043 N Osborne Blvd Racine, WIWisconsin 53405	262-289-4020	Jennifer Eastman

Notes

1. These franchisees operate multiple territories.

FRANCHISEES NOT OPEN AS OF DECEMBER 31, 2023 2024					
State	City	Address	Phone	Owner Name(s)	
Florida Maryland	<u>Laurel</u> St. Petersburg	5565 Suffield Ct. Columbia, Maryland 210446987 Indus Valley Cir Parrish, FL 34219	410-928- 8049 727 723-8000	Shannon Ross Liz Eurich-Plevin	
Georgia New York	New York CitySE Atlanta	694 Fraser 59 East 78th Street, Southeast Atlanta, GA 30315 New York, New York 10075	212-653- 0152404- 936-9902	Michael Wolfson & Howie HorowitzBrian Sullivan	
Washington	<u>Seattle</u> Kingsgate	24304 SE 30th St. Sammamish, WA 980755818 Greenwood Ave N, Seattle, Washington 98103	206-395- 2659 425-310- 2665	Amanda Ebbert and Kathryn Johnson Sarfaraz Reddy	

<u>Part B</u> (Former Franchisees Who Left System During Prior Fiscal Year)

State	City	Current Business Phone or Last Known Home Phone	Owner Name(s)
<u>Alabama</u>	<u>Huntsville</u>	<u>256-542-0223</u>	Kim Krause

State	City	Current Business Phone or Last Known Home Phone	Owner Name(s)
Arizona*	Phoenix Gilbert	602-254-7500 <u>480-649-0008</u>	<u>Dustin Baker</u> <u>Grace-ful Living</u> Sarah & Rijan Shrestha Flarpy , LLC
California	Santa Barbara	805 953 5511	Tracey & John Chudacoff TRAJOH SERVICES INC
Georgia* Arizona*	Metro Atlanta Mesa East	404 921 0064 <u>480-649-0008</u>	<u>Dustin Baker</u> <u>Grace-ful Living</u> Sarah Reese Arrow T Peabody's, LLC
Georgia Arizona*	North Atlanta Chandler Tempe	404-921-0064 <u>480-649-0008</u>	<u>Dustin Baker</u> <u>Grace-ful Living</u> Sarah Reese Arrow T Peabody's, LLC
Minnesota Arizona*	Scottsdale Eden Prairie	<u>952 679 3981</u> 480-454-5611	Greg Holcomb GDCRachel Wasserstrom Pamikis, LLC
<u>Arkansas</u>	Northwest Arkansas	<u>479-362-7504</u>	David Van Der Byl
<u>California</u>	Huntington Beach	<u>949-405-6937</u>	Simone Kelly
<u>California*</u>	Palm Springs	<u>760-576-3000</u>	<u>David Capra</u> <u>Individual</u>
<u>Florida</u>	<u>Augustine</u>	904-417-3292	Victor Burke & Kelly Graham
<u>Idaho</u>	<u>Boise</u>	<u>208-953-7775</u>	<u>Yvonne Carlson,</u> <u>Teton Peaks PTA Services Pc.</u>
Kentucky/Indiana	Louisville-Clarksville	<u>502-208-4072</u>	Jason Coomer
North Carolina*	N.South Charlotte	704- 312-3032 <u>761-4595</u>	Richard Fielding & Brian Stoup, Tarpack Holdings, LLCKen & Laura O'Dea Lifestyle Transitions, INC
Oregon	East Portland Salem	<u>503-</u> 967-7788 <u>560-0405</u>	Steven Trahan Trahan Consulting LLCRussell Morgan, Eastside ALL, Inc.
South Carolina*	<u>Charleston</u>	<u>843-746-9404</u>	<u>Leslie Jackson</u> <u>Jaxson Enterprises, Inc.</u>
<u>Texas</u>	Flower Mound	<u>908-447 1042</u>	Richard Malcolm Tuesday Blast, LLC
<u>Texas</u>	North Austin	<u>512-575-3357</u>	<u>Laura Langley</u>
Texas*	NESW Houston	<u>281-691-</u> 6450 <u>6321</u>	Robert Wolter Brazos Concierge, LLCBrent & Pauline Veazey, 2 Big Brains Inc
<u>Virginia</u>	<u>Alexandria</u>	<u>703-878-7870</u>	Kris & Alisha Jones, KAS Business Group, LLC
Washington 1	<u>Olympia</u>	<u>253-319-2428</u>	<u>Tracy Raymond</u> <u>TR Health Care Advisors, Inc.</u>
Washington 1	<u>Tacoma</u>	<u>253-319-2428</u>	<u>Tracy Raymond</u> <u>TR Health Care Advisors, Inc.</u>

^{*} Franchisee resold their franchise to new franchisees through transfer

If you buy this franchise, your contact information may be disclosed to other buyers when you leave the franchise system.

EXHIBIT "F"

TO DISCLOSURE DOCUMENT

FINANCIAL STATEMENTS

[See Attached]

Exhibit F-1 Audited Financial Statements

Consolidated Financial Report December 31, 2023

Contents

Independent auditor's report	1-2
Financial statements	
Consolidated balance sheets	3
Consolidated statements of operations	4
Consolidated statements of changes in members' equity	5
Consolidated statements of cash flows	6-7
Notes to consolidated financial statements	8-21



RSM US LLP

Independent Auditor's Report

Board of Directors EHC Holding Company, LLC and Subsidiaries

Opinion

We have audited the consolidated financial statements of EHC Holding Company, LLC and its subsidiaries (the Company), which comprise the consolidated balance sheets as of December 31, 2023 and 2022, the related consolidated statements of operations, changes in members' equity, and cash flows for the years ended December 31, 2023 and 2022, and the related notes to the consolidated financial statements (collectively, the financial statements).

In our opinion, the accompanying financial statements present fairly, in all material respects, the financial position of the Company as of December 31, 2023 and 2022, and the results of its operations and its cash flows for the years then ended in accordance with accounting principles generally accepted in the United States of America.

Basis for Opinion

We conducted our audits in accordance with auditing standards generally accepted in the United States of America (GAAS). Our responsibilities under those standards are further described in the Auditor's Responsibilities for the Audit of the Financial Statements section of our report. We are required to be independent of the Company and to meet our other ethical responsibilities, in accordance with the relevant ethical requirements relating to our audits. We believe that the audit evidence we have obtained is sufficient and appropriate to provide a basis for our audit opinion.

Responsibilities of Management for the Financial Statements

Management is responsible for the preparation and fair presentation of the financial statements in accordance with accounting principles generally accepted in the United States of America, and for the design, implementation, and maintenance of internal control relevant to the preparation and fair presentation of financial statements that are free from material misstatement, whether due to fraud or error.

In preparing the financial statements, management is required to evaluate whether there are conditions or events, considered in the aggregate, that raise substantial doubt about the Company's ability to continue as a going concern within one year after the date that the financial statements are issued or available to be issued

THE POWER OF BEING UNDERSTOOD AUDIT | TAX | CONSULTING

1

RSM US LLP is the U.S. member firm of RSM International, a global network of independent audit, tax, and consulting firms. Visit rsmus.com/aboutus for more information regarding RSM US LLP and

Auditor's Responsibilities for the Audit of the Financial Statements

Our objectives are to obtain reasonable assurance about whether the financial statements as a whole are free from material misstatement, whether due to fraud or error, and to issue an auditor's report that includes our opinion. Reasonable assurance is a high level of assurance but is not absolute assurance and, therefore, is not a guarantee that an audit conducted in accordance with GAAS will always detect a material misstatement when it exists. The risk of not detecting a material misstatement resulting from fraud is higher than for one resulting from error, as fraud may involve collusion, forgery, intentional omissions, misrepresentations, or the override of internal control. Misstatements are considered material if there is a substantial likelihood that, individually or in the aggregate, they would influence the judgment made by a reasonable user based on the financial statements.

In performing an audit in accordance with GAAS, we:

- Exercise professional judgment and maintain professional skepticism throughout the audit.
- Identify and assess the risks of material misstatement of the financial statements, whether due to
 fraud or error, and design and perform audit procedures responsive to those risks. Such procedures
 include examining, on a test basis, evidence regarding the amounts and disclosures in the financial
 statements.
- Obtain an understanding of internal control relevant to the audit in order to design audit procedures
 that are appropriate in the circumstances, but not for the purpose of expressing an opinion on the
 effectiveness of the Company's internal control. Accordingly, no such opinion is expressed.
- Evaluate the appropriateness of accounting policies used and the reasonableness of significant
 accounting estimates made by management, as well as evaluate the overall presentation of the
 financial statements.
- Conclude whether, in our judgment, there are conditions or events, considered in the aggregate, that
 raise substantial doubt about the Company's ability to continue as a going concern for a reasonable
 period of time.

We are required to communicate with those charged with governance regarding, among other matters, the planned scope and timing of the audit, significant audit findings, and certain internal control-related matters that we identified during the audit.

RSM US LLP

Detroit, Michigan April 18, 2024

Consolidated Balance Sheets December 31, 2023 and 2022

	2023		2022
Assets			
Current assets:			
Cash and cash equivalents	\$ 8,258,46	4 \$	7,906,497
Accounts receivable, net allowance for credit losses	1,950,26	5	595,028
Inventory	47,08	9	
Prepaid expenses and other current assets	490,72	9	146,386
Total current assets	10,746,54	7	8,647,911
Property, plant, and equipment, net	2,039,65	64	1,671
Other assets:			
Operating lease right-of-use assets, net	1,025,13	4	121,846
Goodwill, net	48,652,99		16,450,078
Intangibles, net	31,456,69		5,157,183
Other assets	129,18		131,896
Contract costs	12,163,66		1,865,412
Total other assets	93,427,66		23,726,415
Total assets	\$ 106,213,86	8 \$	32,375,997
Liabilities and Members' Equity			
Current liabilities:			
Current liabilities: Accounts payable	\$ 972,36	8 \$	391,621
	\$ 972,38 964,76		391,621 1,230,707
Accounts payable	11-11	7	
Accounts payable Accrued expenses	964,76	57 57	1,230,707
Accounts payable Accrued expenses Current portion of deferred franchise fees	964,76 2,422,95	57 57 50	1,230,707 422,746
Accounts payable Accrued expenses Current portion of deferred franchise fees Current portion operating lease liabilities Total current liabilities	964,76 2,422,95 198,15 4,558,24	67 67 60 12	1,230,707 422,746 29,701
Accounts payable Accrued expenses Current portion of deferred franchise fees Current portion operating lease liabilities Total current liabilities Deferred franchise fees, net of current portion	964,76 2,422,95 198,15	67 67 60 12	1,230,707 422,746 29,701 2,074,775
Accounts payable Accrued expenses Current portion of deferred franchise fees Current portion operating lease liabilities Total current liabilities	964,76 2,422,96 198,15 4,558,24	67 67 60 12 72	1,230,707 422,746 29,701 2,074,775 4,869,898
Accounts payable Accrued expenses Current portion of deferred franchise fees Current portion operating lease liabilities Total current liabilities Deferred franchise fees, net of current portion	964,76 2,422,95 198,15 4,558,24 16,479,27 834,04	67 67 60 22 24 6	1,230,707 422,746 29,701 2,074,775 4,869,898 95,071
Accounts payable Accrued expenses Current portion of deferred franchise fees Current portion operating lease liabilities Total current liabilities Deferred franchise fees, net of current portion Operating lease liability, net of current portion	964,76 2,422,98 198,18 4,558,24 16,479,27 834,04 17,313,31	67 67 60 62 22 44 66	1,230,707 422,746 29,701 2,074,775 4,869,898 95,071 4,964,969

Consolidated Statements of Operations Years Ended December 31, 2023 and 2022

	2023	2022
Revenues:		
Franchise fee revenue	\$ 6,507,714	\$ 2,171,527
Support services	666,668	12
Total revenues	7,174,382	2,171,527
Operating expenses:		
Cost of services	306,538	~=
General and administrative expenses	5,531,685	2,986,145
Payroll and benefits	3,885,484	1,755,242
Transaction expenses	1,579,290	1,502,800
Amortization and depreciation expense	3,040,926	919,800
Other operating expenses	546,426	15,406
Total operating expenses	14,890,349	7,179,393
Net loss	\$ (7,715,967)	\$ (5,007,866)

Consolidated Statements of Changes in Members' Equity Years Ended December 31, 2023 and 2022

Balance at December 31, 2023	 84.342.310
Net loss	(7.715.967)
Contributed capital	66,722,024
Balance at December 31, 2022	25,336,253
Net loss	 (5,007,866)
Contributed capital	22,500,000
Balance, January 1, 2022	\$ 7,844,119

Consolidated Statements of Cash Flows Years Ended December 31, 2023 and 2022

	2023	2022
Cash flows from operating activities:		
Net loss	\$ (7,715,967)	\$ (5,007,866)
Adjustments to reconcile net loss to net cash used in operating activities:		
Depreciation and amortization	3,040,926	919,800
Changes in assets and liabilities, net of acquisitions:		
Accounts receivable	146,847	(453,957)
Prepaid expenses and other current assets	(235,413)	(44,821)
Other assets	2,716	(6,575)
Contract costs	(1,143,324)	(117,661)
Accounts payable and accrued expenses	(538,706)	142,018
Deferred franchise fees	1,532,134	300,099
Operating lease assets and liabilities	95,605	2,926
Net cash used in operating activities	(4,815,182)	(4,266,037)
Cash flows from investing activities: Acquisition of businesses, net of cash acquired	(46,199,803)	(13,055,236)
Purchase of intangible assets	· -	(40,512)
Purchase of property and equipment	(1,355,072)	(1,270)
Net cash used in investing activities	(47,554,875)	(13,097,018)
Cash flows from financing activities:		
Proceeds from capital contributions	52,722,024	20,000,000
Net cash provided by financing activities	52,722,024	20,000,000
Net increase in cash and cash equivalents	351,967	2,636,945
Cash and cash equivalents, beginning	7,906,497	5,269,552
Cash and cash equivalents, ending	\$ 8,258,464	\$ 7,906,497

(Continued)

Consolidated Statements of Cash Flows (Continued) Years Ended December 31, 2023 and 2022

	2023	2022
Supplemental schedule of noncash operating, investing and		
financing activities:		
Acquisition of businesses:		
Assets acquired	\$ 40,044,046	\$ 7,257,883
Liabilities assumed	(13,661,493)	(5,078,960)
Net identifiable assets acquired	26,382,553	2,178,923
Goodwill	34,390,411	14,545,344
Net assets acquired	60,772,964	16,724,267
Less cash acquired	(992,260)	(620,715
Add due from seller	419,099	45,924
Less contingent consideration		(594,240
Less units issued as consideration	(14,000,000)	(2,500,000
Cash purchase price	\$ 46,199,803	\$ 13,055,236

Notes to Consolidated Financial Statements

Note 1. Summary of Significant Accounting Policies

Nature of business: EHC Holding Company, LLC and Subsidiaries (the Company) through its wholly owned subsidiaries, including Executive Home Care Franchising, LLC (EH), ALL Franchising, LLC (ALL), ALL Licensing, LLC (Licensing), B&P Burke, LLC (Grasons), Brothers Franchising, LLC (Brothers), and Brothers That Just do Gutters, LLC (BG).

EH is a franchisor that provides home care services to the elderly, physically handicapped and injured, allowing them to live at home. Executive Homecare began operations in 2004.

ALL is a franchisor that provides senior care placement and referral service for in-home companion care, independent retirement options, assisted living, memory care, and skilled nursing facilities. Assisted Living Locators began operations in 2006 and provides services nationwide.

Licensing is an operating company that owns technology used by ALL.

Grasons is a franchisor that provides services for estate sales and business liquidation services. Grasons was established in 2011 and conducts operations from its principal office in California.

Brothers is a franchisor that provides services associated with gutter instillation, repair, and maintenance.

BG operates a Brothers franchise.

Significant accounting policies:

Basis of presentation: The consolidated balance sheets is presented as of December 31, 2023 and 2022. The consolidated statements of operations, changes in members' equity, and cash flows are presented for the years ended of December 31, 2023 and 2022. The accompanying consolidated financial statements of the Company include its wholly owned subsidiaries.

All intercompany transactions have been eliminated. The accompanying consolidated financial statements have been prepared in accordance with accounting standards set by the Financial Accounting Standards Board (FASB). The FASB sets accounting principles generally accepted in the United States of America (U.S. GAAP) that the Company follows to ensure its financial condition, results of operations, and cash flows are consistently reported. References to U.S. GAAP issued by the FASB in these notes to the consolidated financial statements are to the FASB Accounting Standards Codification (ASC).

Revenue recognition policy: The Company recognizes revenue in accordance with ASC 606, Revenue from Contracts with Customers, which provides a five-step model for recognizing revenue from contracts with customers as follows: identify the contract with a customer, identify the performance obligations in the contract, determine the transaction price, allocate the transaction price to the performance obligations in the contract, and recognize revenue when or as performance obligations are satisfied.

Notes to Consolidated Financial Statements

Note 1. Summary of Significant Accounting Policies (Continued)

Nature of services - The Company's franchise agreements include (a) the right to use its symbolic intellectual property over the term of each franchise agreement, (b) preopening services, such as training, (c) ongoing services, such as management of the national brand fund contributions and support services for the franchisees and (d) for certain subsidiaries, a license to use the Company's internal-use software, which is hosted on the Company's software as a service (SaaS) platform. These promises are highly dependent upon and interrelated with the franchise right granted in the franchise agreement, so they are not considered to be individually distinct and therefore are accounted for as a single performance obligation. The performance obligation under the franchise agreement is the promise to provide daily access to the symbolic intellectual property over the term of each franchise agreement, which is a series of distinct services that represents a single performance obligation. Although the franchisor's underlying activities associated with the symbolic intellectual property will vary both within a day and day to day, the symbolic intellectual property is accessed over time and the customer (the franchisee) simultaneously receives and consumes the benefit from the franchisor's performance of providing access to the symbolic intellectual property (including other related activities). Revenue earned from providing the services is collectively referred to as franchise fee revenue.

The Company's franchise fee revenue includes franchise royalties, franchise fees, advertising fund contributions and support services performed for franchisees. Initial franchise fees are based on the market type selected and are paid at the time an individual franchise agreement is signed.

The Company also operates a franchise location. The revenue for this consists of revenue recognized at a point in time as the service is completed.

Payment terms - The transaction price is the amount of consideration to which the Company expects to be entitled in exchange for transferring goods and services to the customer. Agreements may include initial and renewal franchise fees, sales-based royalties, and fees for administrative services performed for the franchisee.

The Company believes its franchising agreements do not contain a significant financing component because (a) the timing of the upfront payment does not arise for the reason of provision of financing to the Company and (b) the sales-based royalty is variable and based on factors outside the Company or the franchisee's control.

Revenue recognition - Initial and renewal franchise fees are recognized as revenue on a straight-line basis over the term of the respective agreement beginning when the agreement is signed. Franchise agreements typically have a term of 10 years with the option to renew for an additional five years if the franchisee is in compliance with the terms of the franchise agreement.

Continuing royalties are calculated as a percentage of franchisees' reported sales that are related entirely to the Company's performance obligation under the franchise agreement. These royalties are considered variable consideration, but because they relate to a license of intellectual property, they are not included in the transaction price. Instead, royalty revenue is recognized as franchisee sales occur. Advertising contributions received from the Company's franchisees are recorded as a component of franchise fee revenue in the consolidated statements of operations.

Contract balances - The Company records accounts receivable when it has the unconditional right to issue an invoice and receive payment, regardless of whether revenue has been recognized. If revenue has not yet been recognized, a contract liability (deferred franchise fees) also is recorded. Revenue is recognized on a straight-line basis over the life of the franchise agreement.

Notes to Consolidated Financial Statements

Note 1. Summary of Significant Accounting Policies (Continued)

Commission costs - The Company defers those direct and incremental costs associated with the sale of franchises. Contract costs are charged to earnings when the related deferred franchise and territory fees are recognized as revenue over the term of the respective agreement. The Company has determined the period of benefit for direct and incremental costs associated with the sale of franchises to be the initial term of the franchise agreement. Expense is recognized on a straight-line basis commensurate with the pattern of revenue recognition. Contract costs are recorded in other assets in the accompanying consolidated balance sheets.

Advertising funds - The Company collects funds from its franchisees for advertising pursuant to the Company's franchise agreements at a percentage of franchisee sales. These advertising services are not considered distinct because they are highly dependent and interrelated to the franchise right. Advertising contributions are considered part of the transaction price for the franchise right and recognized as revenue as the underlying sales occur. The advertising costs incurred for franchisees will be expensed in accordance with the Company's normal policy.

Cash and cash equivalents: The Company considers all short-term, highly liquid investments with original maturities of 90 days or less to be cash and cash equivalents. Cash equivalents consist of money market accounts.

Accounts receivable: Accounts receivables represent amounts due from franchisees pursuant to their individual franchise agreements. Accounts receivables are stated at historical value which approximates fair value. The allowance for credit losses on accounts receivables represents the Company's estimate of expected credit losses over the lifetime of the receivables. This estimation process is based on historical experience, current conditions, asset-specific risk characteristics and reasonable and supportable forecasts about future economic and market conditions. Accounts receivables are written off when deemed uncollectible. Recoveries of royalty receivables previously written off are recorded when received. The allowance for credit loss was approximately \$144,000 and \$37,000 for the years ended December 31, 2023 and 2022, respectively. The Company will continue to monitor and evaluate the adequacy of the allowance for credit losses on accounts receivable on a regular basis and make adjustments as necessary in response to changes in economic conditions and credit quality indicators.

The Company adopted Accounting Standards Update (ASU) 2016-13, Financial Instruments—Credit Losses (Topic 326), on January 1, 2023. This accounting standard requires companies to measure expected credit losses on financial instruments based on the total estimated amount to be collected over the lifetime of the instruments which would include accounts receivables. Prior to the adoption of this accounting standard, the Company recorded incurred loss reserves against account receivable balances based on current and historical information. The adoption of this ASU did not have a material effect on the Company's financial statements.

Concentration of credit risk: The Company's financial instruments that are exposed to concentrations of credit risk consist primarily of cash and cash equivalents and accounts receivable. The Company maintains its cash and cash equivalents in bank deposit accounts, which at times may exceed federally insured limits. The Company has not experienced any losses in such accounts. The Company believes it is not exposed to any significant credit risk on cash and cash equivalents. The Company grants credit to its franchisees and customers. Consequently, the Company's ability to collect the amounts due from franchisees and customers is affected by economic fluctuations. The Company routinely assesses the financial strength of its franchisees and customers and believes that its accounts receivable credit risk exposure is limited.

Notes to Consolidated Financial Statements

Note 1. Summary of Significant Accounting Policies (Continued)

Software development costs: Costs for software developed for internal use are accounted for in accordance with ASC 350, Intangibles—Goodwill and Other - Internal-Use Software. ASC 350 requires the capitalization of certain costs incurred in connection with developing or obtaining internal-use software. In accordance with ASC 350, the Company expenses costs incurred in the preliminary project stage of developing or acquiring internal use software, such as research and feasibility studies, as well as costs incurred in the post-implementation/operational stage, such as maintenance and training. Capitalization of software development costs occurs only after the preliminary project stage is complete, management authorizes the project, and it is probable that the project will be completed and the software will be used for the function intended. Costs associated with the purchase and development of computer software are capitalized and amortized on a straight-line basis over the estimated useful life of the related asset. Software development costs are recorded in property and equipment in the accompanying consolidated balance sheets. The Company capitalized software development costs of approximately \$1,271,000 and \$0 for the years ended December 31, 2023 and 2022, respectively.

Property and equipment: Property and equipment are stated at cost, net of accumulated depreciation and amortization. Expenditures for additions and improvements are capitalized while maintenance and repair expenditures are charged to operations as incurred. When assets are sold or otherwise retired from service, their cost and related accumulated depreciation and amortization are removed from the accounts and any gain or loss is included in the results of operations. Depreciation and amortization is computed using the straight-line method. Property and equipment have estimated useful lives of three to ten years. Depreciation expense for the years ended December 31, 2023 and 2022, was approximately \$98,000 and \$700, respectively.

Goodwill and intangibles: Goodwill is recognized for the excess of the fair value of an acquired business over the fair value of the identifiable net assets acquired. Under ASC 350, Intangibles—Goodwill and Other, the Company elected the accounting alternative to amortize goodwill on a straight-line basis over 10 years.

The Company has elected the provisions of ASU 2014-18, *Business Combinations (Topic 805):*Accounting for Identifiable Intangible Assets in a Business Combination. ASU 2014-18 specifies that a private company that elects the accounting alternative to recognize or otherwise consider the fair value of intangible assets as a result of any in-scope transactions should no longer recognize separately from goodwill: (1) customer-related intangible assets unless they are capable of being sold or licensed independently from the other assets of the business and (2) noncompetition agreements.

The Company tests its recorded goodwill for impairment upon a triggering event. Factors that could trigger an impairment test include, but are not limited to, underperformance relative to historical or projected future operating results, significant changes in the manner of use of the acquired assets or the overall business, significant negative industry or economic trends and a sustained period where market capitalization, plus an appropriate control premium, is less than member's equity. Goodwill is tested using a fair-value approach at the entity level. No indicators of impairment were identified for the years ended December 31, 2023 and 2022.

Intangible assets include franchise agreements, trade names, and developed technology. Intangible assets are amortized on a straight-line basis over their estimated useful lives, which range between five to 14 years.

Notes to Consolidated Financial Statements

Note 1. Summary of Significant Accounting Policies (Continued)

Long-lived assets: Long-lived assets are reviewed for impairment whenever events or changes in circumstances indicate that the carrying amount of the asset may not be recoverable. If impairment is considered, recoverability of these assets is measured by a comparison of the carrying amount of the asset to estimated future undiscounted cash flows expected to be generated by the asset. If the carrying amount of an asset exceeds its estimated future cash flows, an impairment charge is recognized by the amount of which the carrying amount of the asset exceeds the fair value of the asset. No impairment loss has been recognized by the Company as of December 31, 2023 and 2022.

Leases: In February 2016, the FASB issued ASC 842, Leases, to increase transparency and comparability among organizations related to their leasing arrangements. The update requires lessees to recognize most leases on their balance sheets as a right-of-use (ROU) asset representing the right to use an underlying asset and a lease liability representing the obligation to make lease payments over the lease term, measured on a discounted basis. ASC 842 also requires additional disclosure of key quantitative and qualitative information for leasing arrangements. Similar to the previous lease guidance, the update retains a distinction between finance leases (similar to capital leases in ASC 840, Leases) and operating leases, with classification affecting the pattern of expense recognition in the income statement. The Company adopted ASC 842 on January 1, 2022, using the optional transition method to the modified retrospective approach, which eliminates the requirement to restate the prior-period consolidated financial statements. Under this transition provision, the Company has applied ASC 842 to reporting periods beginning on January 1, 2022, while prior periods continue to be reported and disclosed in accordance with the Company's historical accounting treatment under ASC 840, Leases.

The Company elected the package of practical expedients under the transition guidance within ASC 842, in which the Company does not reassess (1) the historical lease classification, (2) whether any existing contracts at transition are or contain leases or (3) the initial direct costs for any existing leases. The Company has not elected to adopt the hindsight practical expedient, and therefore measured the ROU asset and lease liability using the remaining portion of the lease term upon the adoption of ASC 842 on January 1, 2022.

The Company determines if an arrangement is or contains a lease at inception, which is the date on which the terms of the contract are agreed to, and the agreement creates enforceable rights and obligations. A contract is or contains a lease when (i) explicitly or implicitly identified assets have been deployed in the contract and (ii) the Company obtains substantially all of the economic benefits from the use of that underlying asset and directs how and for what purpose the asset is used during the term of the contract. The Company also considers whether its service arrangements include the right to control the use of an asset.

The Company made an accounting policy election available under ASC 842 not to recognize ROU assets and lease liabilities for leases with a term of 12 months or less. For all other leases, ROU assets and lease liabilities are measured based on the present value of future lease payments over the lease term at the commencement date of the lease (or January 1, 2022, for existing leases upon the adoption of ASC 842). The ROU assets also include any initial direct costs incurred and lease payments made at or before the commencement date and are reduced by any lease incentives. To determine the present value of lease payments, the Company made an accounting policy election available to nonpublic companies to utilize a risk-free borrowing rate, which is aligned with the lease term at the lease commencement date (or remaining term for leases existing upon the adoption of ASC 842).

Notes to Consolidated Financial Statements

Note 1. Summary of Significant Accounting Policies (Continued)

The Company has made an accounting policy election to account for lease and nonlease components in its contracts as a single lease component for its various asset classes. The nonlease components typically represent additional services transferred to the Company, such as common area maintenance for real estate, which are variable in nature and recorded in variable lease expense in the period incurred.

Adoption of ASC 842 resulted in the recording of additional ROU assets and lease liabilities related to the Company's operating leases of approximately \$153,000 at January 1, 2022. The adoption of the new lease standard did not materially impact consolidated net earnings or consolidated cash flows and did not result in a cumulative-effect adjustment to the opening balance of retained earnings.

Fair value measurements: The Company uses the fair value measurement and disclosure guidance for all assets and liabilities that are recognized or disclosed at fair value in the consolidated financial statements. The guidance defines fair value as the price that would be received to sell an asset or paid to transfer a liability in an orderly transaction between market participants at the measurement date. In determining fair value, the Company uses various methods including market, income and cost approaches. Based on these approaches, the Company often utilizes certain assumptions that management believes market participants would use in pricing the asset or liability, including assumptions about risk and or the risks inherent in the inputs to the valuation technique. These inputs can be readily observable, market corroborated, or generally unobservable inputs. The Company utilizes valuation techniques that maximize the use of observable inputs and minimize the use of unobservable inputs. The fair value hierarchy ranks the quality and reliability of the information used to determine fair values. Assets and liabilities carried at fair value will be classified and disclosed in one of the following three categories:

- **Level 1:** Inputs to the valuation methodology are unadjusted quoted prices for identical assets or liabilities in active markets that are accessible at the measurement date.
- Level 2: Inputs to the valuation methodology include quoted prices in markets that are not active or quoted prices for similar assets and liabilities in active markets, and inputs that are observable for the asset or liability, either directly or indirectly, for substantially the full term of the financial instrument.
- **Level 3:** Inputs to the valuation methodology are unobservable, reflecting the entity's own assumptions about assumptions market participants would use in pricing the asset or liability.

Income taxes: As a limited liability company, the Company is treated as a partnership for federal and state income tax purposes. As such, the taxable income of the Company is allocated in the tax returns of its members for federal and state tax purposes in accordance with their respective percentage ownership. Accordingly, no provision for federal income taxes is included in the consolidated financial statements. Entity-level, composite state and local income taxes (benefits) are accrued at the applicable rates, if any, and are included in the consolidated statements of operations.

The FASB provides guidance for how uncertain tax provisions should be recognized, measured, disclosed, and presented in the consolidated financial statements. The Company identifies its tax positions taken or expected to be taken in the course of preparing its tax returns and determines whether any tax positions are more likely than not of being sustained when challenged or when examined by the applicable tax authority. Management has determined that there are no uncertain tax positions at December 31, 2023 and 2022.

Notes to Consolidated Financial Statements

Note 1. Summary of Significant Accounting Policies (Continued)

Use of estimates: The preparation of consolidated financial statements in conformity with U.S. GAAP requires management to make estimates and assumptions that affect the reported amounts of assets and liabilities and disclosure of contingent assets and liabilities at the date of the consolidated financial statements and the reported amounts of revenues and expenses during the reporting period. Actual results could differ from those estimates.

Subsequent events: The Company evaluated subsequent events for potential required disclosure through April 18, 2024, which is the date the consolidated financial statements were available to be issued

Note 2. Acquisition of Businesses

Assisted Living Locators: On August 24, 2022, the Company acquired ALL and Licensing. ALL is a franchisor and the franchises provide senior placement and referral services. Licensing is an operating company that owns technology used by ALL. As a result of the transaction, the Company gains penetration into this industry. The Company obtained control of the business by acquiring 100% of the membership interests of ALL and Licensing, therefore the acquisition was accounted for as a business combination. The results of this acquisition have been included in the consolidated financial statements since the date of the acquisition.

The acquisition was funded through cash contributions of equity.

The goodwill arising from the above acquisition is largely due to the ALL's and Licensing's fair value of certain intangible assets along with the assembled workforce subsumed into goodwill, ALL's and Licensing's presence in the marketplace and their long-term expected revenue growth. The goodwill is deductible for income tax purposes.

The following table summarizes the consideration paid and the amounts of the assets acquired, and liabilities assumed at the date of acquisition:

Consideration:

Cash	\$	10,604,497
3,409 Class A shares of EHC Holding Company, LLC		2,500,000
Total invested capital	\$	13,104,497
Recognized amount of identifiable assets acquired and liabilities assumed:		
Cash	\$	554,552
Receivables		90,587
Prepaid assets		63,265
Contract assets		1,535,871
Intangible assets		4,212,000
Other asset		20,232
Accounts payable		(198,630)
Accrued expenses and other liabilities		(246,569)
Contract liabilities		(4,004,575)
Total identifiable net assets acquired	30.	2,026,733
Goodwill		11,077,764
	\$	13,104,497

0 40 004 407

Notes to Consolidated Financial Statements

Note 2. Acquisition of Businesses (Continued)

The fair value of the 3,409 Class A Units was determined based on the value of the Company at the acquisition date, using unobservable inputs.

Transaction expenses of approximately \$1,077,000 were incurred as a result of the business combination and have been expensed by the Company and included in the consolidated statement of operations within the transaction expenses line item for the year ended December 31, 2022

Of the \$4,212,000 of identified intangible assets, \$2,736,000 was assigned to franchise agreements, \$1,034,000 was assigned to trade name, and \$442,000 was assigned to developed technology. Each were determined to have useful lives of 10 years, 14 years and five years, respectively. The weighted-average useful life in years for the identified intangible assets in aggregate was 10.45 years. Franchise agreements, trade names, and developed technology were valued using the multiperiod excess earnings method, relief from royalty method, and cost-to-recreate method, respectively.

Grasons: On November 21, 2022, the Company acquired Grasons. Grasons is a franchisor and the franchises provides downsizing and estate sale services, business liquidation services, senior transition assistance, home staging, and home organization services. As a result of the transaction, the Company gains penetration into this industry. The Company obtained control of the business by acquiring 100% of the membership interests of Grasons, therefore the acquisition was accounted for as a business combination. The results of this acquisition have been included in the consolidated financial statements since the date of the acquisition.

The acquisition was funded through cash contributions of equity.

The goodwill arising from the above acquisition is largely due to the Grasons' fair value of certain intangible assets along with the assembled workforce subsumed into goodwill, Grasons' presence in the marketplace and its long-term expected revenue growth. The goodwill is deductible for income tax purposes.

The following table summarizes the consideration paid and the amounts of the assets acquired, and liabilities assumed at the date of acquisition:

Notes to Consolidated Financial Statements

Note 2.	Acquisition of Businesses	(Continued)	١

Consideration: Cash Contingent consideration Due from seller Total invested capital	\$	3,071,454 594,240 (45,924) 3,619,770
Recognized amount of identifiable assets acquired and liabilities assumed:		
Cash	\$	66,163
Accounts receivable		25,255
Other current assets		12,508
Deposits		1,450
Intangible assets		676,000
Accounts payable		(792)
Accrued expenses and other liabilities		(23,777)
Deferred revenue		(604,617)
Total identifiable net assets acquired	-	152,190
Goodwill	J	3,467,580
	\$	3,619,770

The Company engaged an independent valuation firm to assist with the valuation of intangible assets using the relief from royalty method.

Contingent consideration in the amount of \$594,240 was recognized at the date of acquisition and is included in the balance sheet within the accrued expenses line item at December 31, 2022. The contingent consideration is based on annual royalties for the 12 months ended December 31, 2022 multiplied by an agreed upon rate. The contingent consideration was settled for approximately \$492,000 and goodwill was adjusted for approximately \$102,000 in 2023.

Transaction expenses of approximately \$404,000 were incurred as a result of the business combination and have been expensed by the Company and included in the consolidated statement of operations within the transaction expenses line item for the year ended December 31, 2022

Of the \$676,000 of identified intangible assets, \$295,000 was assigned to franchise agreements and \$381,000 was assigned to trade name. Each were determined to have useful lives of 10 years and 14 years, respectively. The weighted-average useful life in years for the identified intangible assets in aggregate was 12.25 years. Franchise agreements and trade names were valued using the multiperiod excess earnings method and relief from royalty method, respectively.

Brothers: on November 22, 2023, the Company acquired Brothers and BG (collectively Brothers Gutters). Brothers Gutters includes a franchisor and the franchises provide services associated with gutter installation, repair and maintenance. In addition, one franchisee was acquired. As a result of the transaction, the Company gains penetration into this industry. The Company obtained control of the business by acquiring 100% of the membership interests of Brothers Gutters, therefore the acquisition was accounted for as a business combination. The results of this acquisition have been included in the financial statements since the date of the acquisition.

The acquisition was funded through cash contributions of equity.

Notes to Consolidated Financial Statements

Note 2. Acquisition of Businesses (Continued)

The goodwill arising from the above acquisition is largely due to the Brothers Gutters' fair value of certain intangible assets along with the assembled workforce subsumed into goodwill, Brothers Gutters' presence in the marketplace and its long-term expected revenue growth. The goodwill is deductible for income tax purposes.

The following table summarizes the consideration paid and the amounts of the assets acquired, and liabilities assumed at the date of acquisition:

Consideration:

Cash	\$ 47,192,063
14,709 Class A shares of EHC Holding Company, LLC	14,000,000
Due from seller	(419,099)
Total purchase consideration	\$ 60,772,964
Fair value of identifiable assets acquired and liabilities assumed:	
Cash	\$ 992,260
Accounts receivable	1,082,985
Inventory	47,089
Prepaid assets	108,930
Fixed asset	780,949
Contract assets	9,154,927
Intangible assets	27,157,000
Right-of-use asset and other assets	719,906
Accounts payable	(497,365)
Accrued expenses and other	(458,242)
Lease liabilities	(628,435)
Contract liabilities	(12,077,451)
Total identifiable net assets acquired	26,382,553
Goodwill	34,390,411
	\$ 60,772,964

The fair value of the 14,709 Class A Units was determined based on the value of the Company at the acquisition date, using unobservable inputs.

The primary area of the preliminary valuation that is not yet finalized relates to the on-going negotiation with the Company and the seller on the working capital adjustment. An estimate of \$419,099 was recorded at December 31, 2023.

Transaction expenses of approximately \$1,579,000 were incurred as a result of the business combination and have been expensed by the Company and included in the statement of operations within the transaction expenses line item for the year ended December 31, 2023.

Of the \$27,157,000 of identified intangible assets, \$21,578,000 was assigned to franchise agreements and \$5,579,000 was assigned to trade name. Each were determined to have useful lives of 10 years and 14 years, respectively. The weighted-average useful life in years for the identified intangible assets in aggregate was 10.8 years. Franchise agreements and trade names were valued using the multi-period excess earnings method and relief from royalty method, respectively.

Notes to Consolidated Financial Statements

Note 3. Property and Equipment

Property and equipment is summarized as follows:

	-	2023	2022
Furniture and fixtures	\$	386,373	\$ 2,367
Computer equipment and software		1,450,561	
Equipment		50,890	80
Leasehold improvements		250,564	¥t
Total property and equipment		2,138,388	2,367
Less accumulated depreciation and amortization		(98,734)	(696)
Property and equipment, net	\$	2,039,654	\$ 1,671

Note 4. Intangible Assets and Goodwill

Following is a summary of intangible assets:

	December 31, 2023
	Gross Carrying Accumulated Net Carrying
	Amount Amortization Amount
Franchise agreements	\$ 24,937,000 \$ 711,414 \$ 24,225,586
Trade names	7,126,000 248,785 6,877,215
Developed technology	482,512 128,614 353,898
	\$ 32,545,512 \$ 1,088,813 \$ 31,456,699
Goodwill	\$ 51,475,640 \$ 2,822,649 \$ 48,652,991
	December 31, 2022
	Gross
	Carrying Accumulated Net Carrying
	Amount Amortization Amount
Franchise agreements	\$ 3,359,000 \$ 139,043 \$ 3,219,957
Trade names	1,547,000 60,174 1,486,826
Developed technology	482,512 32,112 450,400
	\$ 5,388,512 \$ 231,329 \$ 5,157,183
Goodwill	\$ 17,187,323 \$ 737,245 \$ 16,450,078

Notes to Consolidated Financial Statements

Note 4. Intangible Assets and Goodwill (Continued)

The change in the carrying value of goodwill for the year ended December 31, 2023 and 2022, is as follows:

Balance at December 31, 2021	\$ 2,602,433
Additions of goodwill	14,545,344
Amortization expense	(697,699)
Balance at December 31, 2022	\$ 16,450,078
Additions of goodwill	34,390,411
Reduction of contingent consideration	(102,094)
Amortization expense	(2,085,404)
Balance at December 31, 2023	\$ 48,652,991

Amortization expense recognized on intangible assets and goodwill totaled approximately \$2,943,000 and \$919,000 as of December 31, 2023 and 2022, respectively.

The future estimated aggregate amortization expense for intangibles and goodwill is as follows for each of the next five years ending December 31:

	· · · · · · · · · · · · · · · · · · ·	Goodwill	į.	Intangibles	
Years ending December 31:	st 				
2024	\$	5,147,564	\$	3,275,574	_
2025		5,147,564		3,275,574	
2026		5,147,564		3,275,574	
2027		5,147,564		3,239,050	
2028		5,147,564		3,152,671	

Note 5. Leases

The Company leases three offices under operating lease agreements that have an initial term of 3 - 6 years. Some leases include one or more option to exercise renewal terms that can extend the lease term for more years, at the Company's sole discretion. In addition, some leases contain rights to terminate whereby those termination options are held by either the Company, the lessor, or both parties. These options to extend or terminate the lease are included in the lease terms only when it is reasonably certain that the Company will exercise that option. The Company's leases generally do not contain any material restrictive covenants.

Operating lease cost is recognized on a straight-line basis over the lease term. Lease expense is approximately \$106,000 and \$29,000 for the years ended December 31, 2023 and 2022, respectively. For the years ended December 31, 2023 and 2022, weighted average remaining on lease term is 4.7 years and 3.8 years and the weighted average discount rate is 2.64% and 1.37%, respectively.

Notes to Consolidated Financial Statements

Note 5. Leases (Continued)

Supplemental cash flow information related to leases is as follows for the years ended December 31, 2023 and 2022:

		2023		2022
Operating leases: Operating cash outflows—payments on operating leases	\$	91,762	\$	30,150
Right-of-use assets in exchange for new lease obligations: Operating leases	\$	1,084,556	\$	
The future minimum rentals under this lease for the years subseque follows:	nt to	December 31	, 202	3, are as
Years ending December 31:				
2024			\$	222,245
2025				230,942
2026				228,719
2027				203,733
2028				139,664
Thereafter				53,632
# 3 TY 3				4 070 005

Total lease payments 1,078,935

Less imputed interest (46,741)

Total present value of lease liabilities \$ 1,032,194

Note 6. Related-Party Transactions

A company related to the Company's majority member, charges the Company for financial and management services under a management services agreement for reimbursement of reasonable direct expenses, which is included in general and administrative expenses on the accompanying consolidated statements of operations. The total expense for the years ended December 31, 2023 and 2022, is approximately \$568,000 and \$500,000, respectively.

Notes to Consolidated Financial Statements

Note 7. Members' Equity

Members' equity consisted of the following membership units:

	2	2023		
	Units	Units Issued		
	Authorized	and Outstanding		
Class A Units	1,000,000	109,546		
Class B Units	1,000,000	1,661		
	2	022		
	Units	Units Issued		
	Authorized	and Outstanding		
Class A Units	1,000,000	39,316		
Class B Units	1,000,000	823		

Class A Units have voting rights on all matters requiring the consent, approval or vote of the Members. The Class A Units receive preference on distributions.

Class B Units are deferred units that do not have voting rights and have been issued to designated management employees of the Company without any corresponding capital contribution. The holders of these units are entitled to share in the appreciation of the Company's assets that occur subsequent to the date of grant. The Class B Units are dilutive to the participating preferred units. The units substantially vest upon a change in control of the Company, if still employed.

The Company has issued 2,062 and 823 units to certain management employees as of December 31, 2023 and 2022 and had 1,224 units forfeited as of December 31, 2023.

No compensation expense is recognized on the Class B units as their vesting condition is not considered probable until a change in control occurs.

Note 8. Phantom Stock

During 2023, the Company adopted a phantom deferred unit plan. Each share of phantom deferred unit awarded to eligible individuals represents a contractual right to receive an amount in cash equal to the fair value of the unit upon the occurrence of a significant sale. The Company has authorized 1,000,000 units and 3,286 units are outstanding at December 31, 2023. No compensation expense is recognized on the phantom stock deferred units as their vesting condition is not considered probable until a significant sale occurs.

CALLRN Franchise, LLC

FINANCIAL STATEMENTS December 31, 2021 and 2020

TABLE OF CONTENTS

	Pages
INDEPENDENT AUDITOR'S REPORT	
FINANCIAL STATEMENTS	
Balance Sheets	1
Statements of Income and Changes in Shareholder's Equity	2
Statements of Cash Flows	3
Notes to Financial Statements	4-10

RC Consulting LLC

15560 N Frank Lloyd Wright Blvd. Scottsdale, Arizona 85260 480-330-1101

INDEPENDENT AUDITOR'S REPORT

To the Shareholder's CALLRN Franchise, LLC Scottsdale, Arizona

Report on the Financial Statements

We have audited the accompanying financial statements of CALLRN Franchise, LLC, which comprise the balance sheets as of December 31, 2021 and 2020, and the related statements of income and changes in shareholder's equity and cash flows for the years then ended, and the related notes to the financial statements.

Management's Responsibility for the Financial Statements

Management is responsible for the preparation and fair presentation of these financial statements in accordance with accounting principles generally accepted in the United States of America; this includes the design, implementation, and maintenance of internal control relevant to the preparation and fair presentation of financial statements that are free from material misstatement, whether due to fraud or error.

Auditor's Responsibility

Our responsibility is to express an opinion on these financial statements based on our audits. We conducted our audits in accordance with auditing standards generally accepted in the United States of America. Those standards require that we plan and perform the audit to obtain reasonable assurance about whether the financial statements are free of material misstatement. An audit involves performing procedures to obtain audit evidence about the amounts and disclosures in the financial statements. The procedures selected depend on the auditor's judgment, including the assessment of the risks of material misstatement of the financial statements, whether due to fraud or error. In making those risk assessments, the auditor considers relevant to the entity's preparation and fair presentation of the financial statements in order to design audit procedures that are appropriate in the circumstances, but not for the purpose of expressing an opinion on the effectiveness of the appropriateness of accounting policies used and the reasonableness of significant accounting estimates made by management, as well as evaluating the overall presentation of the financial statements. We believe that the audit evidence we have obtained is sufficient and appropriate to provide a basis for our audit opinion.

Member American Institute of Certified Public Accountants
Licensed Arizona (15876-R)

RC Consulting LLC

15560 N Frank Lloyd Wright Blvd. Scottsdale, Arizona 85260 480-330-1101

Opinion

In our opinion, the financial statements referred to above present fairly, in all material respects, the financial position of CALLRN Franchise, LLC as of December 31, 2021 and 2020 and the results of its income and its cash flows for the years then ended in conformity with accounting principles generally accepted in the United States of America.

Scottsdale, Arizona March 25, 2022

RC Consucroso 200

Member American Institute of Certified Public Accountants Licensed Arizona (15876-R)

CALLRN FRANCHISE, LLC BALANCE SHEETS DECEMBER 31, 2021 AND 2020

	2021		2020		
ASSETS					
CURRENT ASSETS					
Cash and Cash Equivalents	S	407,061	\$	225,590	
Accounts Receivable, Less Allowance \$50,000 - 2021 and \$200,000 - 2020		330,167		325,662	
Note Receivable, Franchisee - Current		8,421		7,932	
Contract Assets - Current		462,524		404,638	
Due From Related Party		277,685		265,925	
Prepaid Expenses		24,798		20,640	
Total Current Assets		1,510,656		1,250,387	
PROPERTY AND EQUIPMENT					
Office Furniture, Equipment & Vehicles		165,111		29,686	
Accumulated Depreciation		(51,081)		(29,686)	
Property and Equipment, Net		114,030		-	
OTHER ASSETS					
Trademarks		5,232		5,232	
Accumulated Amortization		(3,839)		(3,490)	
Contract Assets - Less Current		2,440,602		2,381,908	
Note Receivable, Franchisee - Less Current		13,504		23,925	
Other Assets, Net		2,455,499		2,407,575	
TOTAL ASSETS	<u>s</u>	4,080,185	s	3,657,962	
LIABILITIES AND SHAREHOLDER	'S EQUITY				
CURRENT LIABILITIES					
Accounts Payable	\$	341,603	S	198,310	
Contract Liabilities - Current		838,208		747,045	
Total Current Liabilities		1,179,811		945,355	
LONG-TERM DEBT					
Contract Liabilities - Less Current		4,142,222		4,116,469	
Total Long-Term Liabilities	_	4,142,222		4,116,469	
SHAREHOLDER'S EQUITY (DEFICIT)		(1,241,848)		(1,403,862)	
STARLING DECT (DETICIT)		(1,2-1,0-10)		,1,100,000	
TOTAL LIABILITIES AND SHAREHOLDER'S EQUITY	\$	4,080,185	S	3,657,962	

The accompanying notes are an integral part of these financial statements.

CALLRN FRANCHISE, LLC STATEMENTS OF INCOME AND CHANGES IN SHAREHOLDER'S EQUITY DECEMBER 31, 2021 AND 2020

	2021	2020	
REVENUES			
Franchise sales	\$ 838,208	\$ 747,045	
Placement fees and royalties	1,571,528	1,422,177	
Total Revenues	2,409,736	2,169,222	
COST OF REVENUES			
Franchise sales, commissions and marketing	462,524	404,590	
Total Cost Of Revenues	462,524	404,590	
GROSS PROFIT	1,947,212	1,764,632	
OPERATING EXPENSES			
Advertising	520,485	320,916	
Amortization	349	349	
Automobile	20,616	17,442	
Bad Debt	83,699	74,000	
Consultants	262,464	183,496	
Depreciation	21,395		
Insurance	22,307	41,488	
Office supplies	63,710	71,801	
Professional fees	82,489	121,199	
Rent	39,317	37,589	
Salaries, taxes and benefits	794,549	735,173	
Utilities	10,826	11,669	
Total Operating Expenses	1,922,207	1,615,122	
NET OPERATING INCOME	25,005	149,510	
OTHER INCOME			
PPP loan forgiveness	137,009	134,403	
NET INCOME	162,014	283,913	
SHAREHOLDER'S EQUITY, Beginning	(1,403,862)	(1,272,775)	
Shareholder's Distributions		(415,000)	
SHAREHOLDER'S EQUITY, Ending	\$ (1,241,848)	\$ (1,403,862)	

The accompanying notes are an integral part of these financial statements.

CALLRN FRANCHISE, LLC STATEMENTS OF CASH FLOWS DECEMBER 31, 2021 AND 2020

\$ 25,005 349 21,395	\$ 149,510 349
349	
2.17	349
2.17	349
21,395	
	-
(4,505)	113,297
(4,158)	(16,740)
9,932	53,564
(116,580)	(184,980)
116,916	215,902
143,293	(48,858)
191,647	282,044
(135,425)	
-	(62,800)
(135,425)	(62,800)
137,009	134,403
(11,760)	
-	(415,000)
125,249	(280,597)
181,471	(61,353)
225,590	286,943
\$ 407,061	s 225,590
	(4,158) 9,932 (116,580) 116,916 143,293 191,647 (135,425) (135,425) (135,425) 137,009 (11,760) 125,249 181,471 225,590

The accompanying notes are an integral part of these financial statements.

NOTES TO FINANCIAL STATEMENTS

December 31, 2021 and 2020

NOTE 1: Organization and Nature of Business

CALLRN Franchise, LLC ("the Company") is a Delaware limited liability company that was formed on February 3, 2006. The Company is an Assisted Living Referral ("ALR") franchising network that assists seniors and their families by locating assisted living facilities that meet their geographic, financial, emotional, spiritual and medical needs. It has developed a proprietary system including logo, method of operation and provisions of services, confidential operations manual, marketing and advertising, and techniques and know-how for imparting quality services for establishing, operating, managing and marketing a franchised ALR company.

On April 3, 2020, Assisted Living Locators Holdings, LLC was formed. All of the stock of CALLRN Franchise, LLC was contributed to Assisted Living Locators Holdings, LLC in exchange for an equal ownership in new entity by the shareholders. In addition, Assisted Living Locators Licensing, LLC was formed on April 3, 2020. Its purpose is to maintain all Database activities and Licensing activities of the Franchisor.

NOTE 2: Summary of Significant Accounting Policies

Basis of Accounting

The financial statements are prepared according to the accrual method of accounting in conformity with accounting principles generally accepted in the United States of America (U.S. GAAP). Under this method, revenues are recorded when earned, and expenses are recorded when incurred.

Revenue Recognition

On January 1, 2019, the Company recognized revenues in accordance with the provisions of FASB Accounting Standards Codification ("ASC") 606, Revenue from Contracts with Customers, which provides guidance on the recognition, presentation and disclosure of revenue in financial statements.

ASC 606 outlines the basic criteria that must be met to recognize revenue and provides guidance for disclosure related to revenue recognition policies. The Company recognizes revenue upon transfer of control of promised services to customers in an amount that reflects the consideration the Company expects to receive in exchange for those services. The Company at times may enter into contracts that can include various combinations of services, which are generally capable of being distinct and accounted for as separate performance obligations. In such cases, revenue would be recognized at the time of delivery or over time for each performance of service. Revenue is recognized net of allowance for returns and any taxes collected from customers, which are subsequently remitted to governmental authorities.

NOTES TO FINANCIAL STATEMENTS December 31, 2021 and 2020

NOTE 2: Summary of Significant Accounting Policies (Continued)

Revenue Recognition - continued

The Company's revenues consist of sales to individual franchises. When a franchise is sold, the Company agrees to provide multiple performance promises in each contract, primarily the delivery of its initial custom training program and the rights to use the associated business processes, systems, access to preferred vendors training and access to proprietary software systems and advertising. Franchises are also required to contribute to a national advertising campaign sponsored by the Company. Revenue from contributions to the national advertising campaign are recognized when funds are expended.

ASC 606 provides that revenues are to be recognized when control of promised goods or services is transferred to a customer in an amount that reflects the consideration expected to be received for those goods or services. This standard does not impact the Company's recognition for recurring revenues, such as advertising, royalties, licensing and other recurring fees. The standard does change the timing in which the Company recognizes initial fees from new franchisees and new franchise terms. The Company's accounting policy through December 31, 2018, was to recognize initial franchise fees when received and upon the completion of required franchisee training. Beginning January 1, 2019, the initial franchise fees are allocated to the performance obligations completed during the initial follow up training. Allocation of the initial franchise fee is based on the term of the franchise agreement, which is 10 years.

The Company adopted ASC 606 as of January 1, 2019, using the modified retrospective method on all contracts. This method allows the standard to be applied retrospectively through a cumulative catch-up adjustment recognized upon adoption. As such, comparative information in the Company's financial statements has not been restated and continues to be reported under the accounting standards in effect for those periods. The cumulative adjustment recorded through shareholder's equity (deficit) on January 1, 2019 upon adoption of ASC 606 consisted of Contract Liabilities of \$3,474,707 and Contract Assets of \$1,888,702.

The Company recognized a Contract Asset for the incremental costs of obtaining the contract arising from the sales commissions to outside parties and employees as well as fees paid to franchising organizations for membership and travel to these organizations for training. The Company amortizes these Contract Assets over the term of the franchise agreement which is 10 years.

NOTES TO FINANCIAL STATEMENTS

December 31, 2021 and 2020

NOTE 2: Summary of Significant Accounting Policies (Continued)

Cash and Cash Equivalents

Cash and cash equivalents include cash on hand; cash in banks, and all highly liquid investments that are readily convertible to cash with a maturity of three months or less at the time of purchase.

Property and Equipment

Property and Equipment are reported at cost. Depreciation is calculated using the straight-line method over the estimated useful lives of the assets. For the years ended December 31, 2021 and 2020, depreciation expense was \$21,395 and \$0, respectively.

Intangible Assets

Intangible assets consist of trademark costs. The Company has established intangible assets for the cost of developing certain trademarks.

The Company has adopted Financial Accounting Standards Board ("FASB") Accounting Standards Codification ("ASC") 350, Intangibles - Goodwill and Other that requires that goodwill and intangible assets with indefinite lives (such as intellectual property) no longer be amortized to earnings but be tested for impairment at least annually. Intangible assets that have finite lives (such as franchise documents, non-compete agreements and trademarks are amortized over their estimated useful lives.

Use of Estimates

Preparation of the Company's financial statements in accordance with generally accepted accounting principles requires management to make estimates and assumptions that affect the reported amounts of assets and liabilities, disclosure of any contingent assets and liabilities at the date of the financial statements and reported amounts of revenues and expenses during the reporting period. Actual results could differ from those estimates.

Advertising Costs

The Company expenses advertising costs as incurred. Advertising expense for the years ended December 31, 2021 and 2020 were \$520,485 and \$320,916, respectively.

NOTES TO FINANCIAL STATEMENTS

December 31, 2021 and 2020

NOTE 2: Summary of Significant Accounting Policies (Continued)

Income Taxes

The shareholders of the Company have elected to be taxed under the provisions of Subchapter S of the Internal Revenue Code. Under those provisions, the Company does not pay income taxes on its income. Instead, the shareholders are responsible for individual income taxes on their respective share of the Company's taxable income.

Current Assets and Current Liabilities

Current assets and current liabilities in the accompanying financial statements are based on a normal one-year operating cycle.

Concentrations

The Company maintains cash balances in one financial institution participating in the Federal Deposit Insurance Corporation ("FDIC") program whereby cash deposits are insured for up to \$250,000. Cash balances in their financial institutions have at times exceeded the FDIC insured amounts. However, the Company has not experienced any related losses in such accounts, and management believes there is no significant credit risk associated with these accounts.

Fair Value of Financial Instruments

For certain of the Company's financial instruments, including cash and cash equivalents, accounts receivable, accounts payable and accrued expenses, the carrying amounts approximate fair value due to their short maturities. The amounts shown for notes payable also approximate fair value because current interest rates and terms offered to the Company for similar debt are substantially the same.

Brand and System Development Fund

The Company receives funds from each franchisee and is required to use those funds to promote the franchisees, the CALLRN brand and franchise network. The Brand and System Development Fund began in 2019. The balance of the fund account as of December 31, 2021 was \$68,261 and is included in cash and cash equivalents. The Company expended \$259,901 and \$179,137 from this fund during 2021 and 2020, respectively which is included in advertising expense.

NOTES TO FINANCIAL STATEMENTS

December 31, 2021 and 2020

NOTE 3: Accounts Receivable

Account receivables are recorded at amounts due based on the terms of executed franchise agreements for franchise sales, placement royalties and advertising fees. The accounts receivable is carried at original invoice amount less an estimate for doubtful receivables based on a review of outstanding amounts. Management regularly evaluates individual customer's receivables considering their financial condition, credit history and current economic conditions. Write-offs are charged to the allowance for doubtful when deemed uncollectible. Recoveries of accounts receivable previously written off are recorded as income when received.

The Company had an allowance for doubtful accounts of \$50,000 and \$200,000 for the years ended December 31, 2021 and 2020, respectively.

The assisted living facilities requested that the Company complete all administrative billing processes for the Franchisee's for placement commissions earned. Included in accounts receivable is \$95,825 that are due from assisted living facilities for placement commissions earned and completed by franchisees. Also, included in accounts payable is \$145,215 due to Franchisee's as of December 31, 2021.

NOTE 4: Note Receivable

The Company has a note receivable from a Franchisee that require monthly payments of \$802, including interest at 6.0%. The note matures on August 4, 2024. Principal payments are as follows for the years ended December 31, - 2022 - \$8,421; 2023 - \$8,941 and 2024 - \$4,563.

NOTE 5: Related Party Transactions

On April 3, 2020 Assisted Living Locators Licensing LLC was formed. All of the Website and Data Base systems of the Company were transferred into that entity at its net book value, resulting in a due from Assisted Living Locators Licensing LLC in the amount of \$265,925. The Company advanced \$11,760 for the year ended December 31, 2021. This amount is due on demand with no interest.

NOTE 6: Operating Lease Agreement

The Company entered into an agreement for office space. Terms of the agreement require monthly base rental payments of \$2,173 per month through December 31, 2021, plus common area maintenance. Rent expense for the years ended December 31, 2021 and 2020 was \$39,317 and \$37,589, respectively.

NOTES TO FINANCIAL STATEMENTS December 31, 2021 and 2020

NOTE 9: Commitments and Contingencies

Litigation

The Company may be subject to various claims, legal actions and complaints arising in the ordinary course of business. In the opinion of management, all matters are of such kind, or involve such amounts, that unfavorable disposition, if any, would not have a material effect on the financial position of the Company.

Exhibit F-39	Franchise Disclosure Document (2025)

EHC Holding Company, LLC and Subsidiaries Consolidated Financial Report December 31, 2024

	Contents
Independent Auditor's Report	1-2
Consolidated Financial Statements	
Balance Sheet	3
Statement of Operations	4
Statement of Members' Equity	5
Statement of Cash Flows	6
Notes to Consolidated Financial Statements	7-13



Plante & Moran, PLLC Suite 900. 2601 Cambridge Court Aubum Hills MI 48326 Tel: 249.375,7100 Fax: 249.375,7100 Pax: 249.375,7100

Independent Auditor's Report

To the Board of Directors EHC Holding Company, LLC and Subsidiaries

Opinion

We have audited the consolidated financial statements of EHC Holding Company, LLC and Subsidiaries (the "Company"), which comprise the consolidated balance sheet as of December 31, 2024 and the related consolidated statements of operations, members' equity, and cash flows for the year then ended, and the related notes to the consolidated financial statements.

In our opinion, the accompanying consolidated financial statements present fairly, in all material respects, the financial position of the Company as of December 31, 2024 and the results of its operations and its cash flows for the year then ended in accordance with accounting principles generally accepted in the United States of America.

Basis for Opinion

We conducted our audit in accordance with auditing standards generally accepted in the United States of America (GAAS). Our responsibilities under those standards are further described in the Auditor's Responsibilities for the Audit of the Consolidated Financial Statements section of our report. We are required to be independent of the Company and to meet our ethical responsibilities in accordance with the relevant ethical requirements relating to our audit. We believe that the audit evidence we have obtained is sufficient and appropriate to provide a basis for our audit opinion.

Responsibilities of Management for the Consolidated Financial Statements

Management is responsible for the preparation and fair presentation of the consolidated financial statements in accordance with accounting principles generally accepted in the United States of America and for the design, implementation, and maintenance of internal control relevant to the preparation and fair presentation of consolidated financial statements that are free from material misstatement, whether due to fraud or error.

In preparing the consolidated financial statements, management is required to evaluate whether there are conditions or events, considered in the aggregate, that raise substantial doubt about the Company's ability to continue as a going concern within one year after the date that the consolidated financial statements are issued or available to be issued.

Auditor's Responsibilities for the Audit of the Consolidated Financial Statements

Our objectives are to obtain reasonable assurance about whether the consolidated financial statements as a whole are free from material misstatement, whether due to fraud or error, and to issue an auditor's report that includes our opinion. Reasonable assurance is a high level of assurance but is not absolute assurance and, therefore, is not a guarantee that an audit conducted in accordance with GAAS will always detect a material misstatement when it exists. The risk of not detecting a material misstatement resulting from fraud is higher than for one resulting from error, as fraud may involve collusion, forgery, intentional omissions, misrepresentations, or the override of internal control. Misstatements are considered material if there is a substantial likelihood that, individually or in the aggregate, they would influence the judgment made by a reasonable user based on the consolidated financial statements.



To the Board of Directors EHC Holding Company, LLC and Subsidiaries

In performing an audit in accordance with GAAS, we:

- Exercise professional judgment and maintain professional skepticism throughout the audit.
- Identify and assess the risks of material misstatement of the consolidated financial statements, whether due to
 fraud or error, and design and perform audit procedures responsive to those risks, Such procedures include
 examining, on a test basis, evidence regarding the amounts and disclosures in the consolidated financial
 statements.
- Obtain an understanding of internal control relevant to the audit in order to design audit procedures that are
 appropriate in the circumstances but not for the purpose of expressing an opinion on the effectiveness of the
 Company's internal control. Accordingly, no such opinion is expressed.
- Evaluate the appropriateness of accounting policies used and the reasonableness of significant accounting estimates made by management, as well as evaluate the overall presentation of the consolidated financial statements.
- Conclude whether, in our judgment, there are conditions or events, considered in the aggregate, that raise substantial doubt about the Company's ability to continue as a going concern for a reasonable period of time.

We are required to communicate with those charged with governance regarding, among other matters, the planned scope and timing of the audit, significant audit findings, and certain internal control-related matters that we identified during the audit.

Plante + Moren, PLLC

March 25, 2025

Consolidated Balance Sheet

December 31, 2024

	December 41, 2424
Assets	
Current Assets Cash Accounts receivable - Net Inventory Prepaid expenses and other current assets	\$ 7,200,071 2,685,757 47,090 1,032,791
Total current assets	10,965,709
Property and Equipment - Net	2,069,949
Operating Lease Right-of-use Assets - Net	792,487
Goodwill - Net	52,665,564
Intangible Assets - Net	28,380,024
Other Assets Deferred commission costs - Net of current portion Other noncurrent assets	5,332,257 489,961
Total other assets	5,822,218
Total assets	\$ 100,695,951
Liabilities and Member	rs' Equity
Current Liabilities Accounts payable Current portion of operating lease liabilities Deferred franchise fees Other current liabilities: Accrued compensation Other accrued liabilities	\$ 868,428 212,365 3,010,287 596,645 1,279,257
Total current liabilities	5,966,982
Operating Lease Liabilities - Net of current portion	603,246
Deferred Franchise Fees - Net of current portion	16,498,710
Total liabilities	23,068,938
Members' Equity	77,627,013
Total liabilities and members' equity	\$ 100,695,951

See notes to consolidated financial statements.

3

Consolidated Statement of Operations

Year Ended December 31, 2024

-2 To No.			
Net	Pa	Inn	110

tt Revenue
Royalty fees
Initial franchise fees
National brand fund fees
Service fees
Other

Total net revenue

Operating Expenses
Consolidated Net Loss

\$ 9,732,577 4,547,181 2,706,984 7,467,150 1,283,105 25,736,997

32,452,294

\$ (6,715,297)

See notes to consolidated financial statements,

Consolidated Statement of Members' Equity

Year Ended December 31, 2024

Balance - January 1, 2024 Consolidated net loss \$ 84,342,310

(6,715,297)

Balance - December 31, 2024

\$ 77,627,013

See notes to consolidated financial statements. 5

Consolidated Statement of Cash Flows

Year Ended December 31, 2024

Cash Flows from Operating Activities		
Consolidated net loss	\$	(6,715,297)
Adjustments to reconcile consolidated net loss to net cash from operating acti	vities:	0.007.500
Depreciation and amortization		9,997,566
Noncash lease expense		16,064
Bad debt expense		219,820
Changes in operating assets and liabilities that (used) provided cash:		225-5750
Accounts receivable		(955,312)
Prepaid expenses and other assets		(980,991)
Deferred commission costs		(3,364,692)
Accounts payable and other accrued liabilities		807,195
Deferred franchise fees	_	606,768
Net cash used in operating activities		(368,879)
Cash Flows Used in Investing Activities - Purchase of property and equipment	t	(689,514)
Net Decrease in Cash		(1,058,393)
Cash - Beginning of year		8,258,464
Cash - End of year	\$	7,200,071

See notes to consolidated financial statements.

Notes to Consolidated Financial Statements

December 31, 2024

Note 1 - Nature of Business

EHC Holding Company, LLC and Subsidiaries (the "Company") includes its wholly owned subsidiaries, Evive Brands, LLC (Evive); Executive Home Care Franchising, LLC (EHC); ALL Franchising, LLC (ALL); ALL Licensing, LLC (Licensing); B&P Burke, LLC (Grasons); Brothers Parsons Franchising, LLC (Brothers); Brothers Parsons HV, LLC (BPHV); and Brothers Parsons IP, LLC (BPIP).

EHC is a franchisor that provides home care services to the elderly, physically handicapped, and injured, allowing them to live at home. EHC began operations in 2004.

ALL is a franchisor that provides senior care placement and referral service for in-home companion care, independent retirement options, assisted living, memory care, and skilled nursing facilities. ALL began operations in 2006 and provides services nationwide. Licensing is an operating company that owns the intellectual property used by ALL.

Grasons is a franchisor that provides services for estate sales and business liquidation services and was established in 2011.

Brothers is a franchisor that provides services associated with gutter installation, repair, and maintenance BPIP is an operating company that owns the intellectual property used by Brothers BPHV operates a Brothers franchise.

Note 2 - Significant Accounting Policies

Basis of Accounting

The consolidated financial statements have been prepared in accordance with accounting principles generally accepted in the United States of America. The Company has elected to adopt certain accounting alternatives for private companies developed by the Private Company Council, including the alternative for accounting for goodwill and intangibles.

Principles of Consolidation

The consolidated financial statements include the accounts of the Company and all of its wholly owned subsidiaries. All material intercompany accounts and transactions have been eliminated in consolidation.

Accounts Receivable

Trade accounts receivable are stated at invoice amounts. An allowance for credit losses is established for amounts expected to be uncollectible over the contractual life of the receivables. At December 31, 2024, the Company had recorded an allowance for credit losses in the amount of \$131,037. The Company evaluates the collectibility of its accounts receivable and determines the appropriate allowance for expected credit losses based on a combination of factors, including the aging of the receivables, historical collection trends, and charge-offs, and includes adjustments for current economic conditions and reasonable and supportable forecasts. When the Company is aware of a franchisee or customer's inability to meet its financial obligation, the Company may individually evaluate the related receivable to determine the allowance for expected credit losses. Uncollectible amounts are written off against the allowance for credit losses in the period they are determined to be uncollectible. Recoveries of amounts previously written off are recognized when received. The net accounts receivable balance as of January 1, 2024 was \$1,950,265.

Property and Equipment

Property and equipment are recorded at cost. Depreciation and amortization are computed using the straight-line method. Assets are depreciated over their estimated useful lives, which range from 3 to 10 years. The cost of leasehold improvements is depreciated (amortized) over the lesser of the length of the related leases or the estimated useful lives of the assets. Costs of maintenance and repairs are charged to expense when incurred.

Notes to Consolidated Financial Statements

December 31, 2024

Note 2 - Significant Accounting Policies (Continued)

Capitalized Software Costs

The Company capitalizes significant costs incurred in the acquisition or development of software for internal use, including the costs of the software, materials, consultants, interest, payroll, and payroll-related costs for employees incurred in developing internal-use computer software, once final selection of the software is made. Costs incurred prior to the final selection of software and costs not qualifying for capitalization are charged to expense. Capitalized software amortization expense was approximately \$203,000 in 2024. The net book value of capitalized software costs included in property and equipment at December 31, 2024 was approximately \$1,032,000. The estimated useful life of the capitalized software is five years.

Leases

The Company has operating leases for corporate office space. The Company recognizes expense for operating leases on a straight-line basis over the lease term. The Company made a policy election not to separate lease and nonlease components for all operating leases. Therefore, all payments are included in the calculation of the right-of-use asset and lease liability.

The Company has an operating lease with a lease term of one year or less that the Company elected to account for as a short-term lease. As this lease is a short-term lease, it is not included in the right-of-use asset and lease liability. Total expense related to short-term leases is *de minimis*.

The Company elected to use the risk-free rate as the discount rate for calculating the right-of-use asset and lease liability in place of the incremental borrowing rate for all operating leases.

Intangible Assets

Acquired intangible assets subject to amortization are stated at cost and are amortized using the straightline method over the estimated useful lives of the assets, which range from 5 to 14 years. Intangible assets that are subject to amortization are reviewed for potential impairment whenever events or circumstances indicate that carrying amounts may not be recoverable. Assets not subject to amortization are tested for impairment at least annually.

The Company has elected to apply the private company accounting alternative for intangible assets acquired in a business combination developed by the Private Company Council. Under the accounting alternative, certain acquired customer-related intangible assets and noncompetition agreements are not separately recognized apart from goodwill.

Goodwill

The recorded amounts of goodwill from prior business combinations are based on management's best estimates of the fair values of assets acquired and liabilities assumed at the date of acquisition.

The Company has elected to apply the private company accounting alternative for goodwill developed by the Private Company Council. Under the accounting alternative, goodwill is amortized on a straight-line basis over a 10-year period. Additionally, goodwill is assessed for potential impairment if events occur or circumstances change that indicate the fair value of the Company may be less than its carrying value. The Company has elected to test goodwill for impairment at the entitywide level.

No impairment charge was recognized during the year ended December 31, 2024.

Notes to Consolidated Financial Statements

December 31, 2024

Note 2 - Significant Accounting Policies (Continued)

Revenue Recognition

The Company's franchise agreements include (a) the right to use its symbolic intellectual property over the term of each franchise agreement (typically 10 years), (b) preopening services, such as training, (c) ongoing services, such as management of the national brand fund contributions and support services for the franchisees; and (d) for certain subsidiaries, a license to use the Company's internal-use software, which is hosted on the Company's software as a service (SaaS) platform. These promises are highly dependent upon and interrelated with the franchise right granted in the franchise agreement, so they are not considered to be individually distinct and, therefore, are accounted for as a single performance obligation. The performance obligation under the franchise agreement is the promise to provide daily access to the symbolic intellectual property over the term of each franchise agreement, which is a series of distinct services that represents a single performance obligation. Although the franchisor's underlying activities associated with the symbolic intellectual property will vary both within a day and day to day, the symbolic intellectual property is accessed over time and the customer (the franchisee) simultaneously receives and consumes the benefit from the franchisor's performance of providing access to the symbolic intellectual property (including other related activities). Revenue earned from providing these services is identified as royalty fees, initial franchise fees, and national brand fund fees on the accompanying consolidated statement of operations.

The Company also operates a franchise location. The revenue for this consists of revenue recognized at a point in time as the service is completed. This revenue is identified as service fees on the accompanying consolidated statement of operations.

Payment Terms

Initial franchise fees are due and typically paid when a franchise agreement is executed and are nonrefundable. These fees are collected prior to the satisfaction of the Company's performance obligations, resulting in the Company recognizing deferred revenue contract liabilities. The portion of contract liabilities that is expected to be recognized as revenue within one year is classified as current on the consolidated balance sheet. Deferred initial franchise fees as of January 1, 2024 equaled \$18,902,229, initial franchise fees are also received pursuant to area development agreements, which grant the right to develop franchised territories in future periods in specific geographic areas. Royalties and advertising fees are paid on a monthly basis based upon a percentage of franchisee gross sales. Technology fees are paid on a monthly basis based upon a fixed amount. Service fees are due 30 days from when the service is performed.

Allocating the Transaction Price

The transaction price is the amount of consideration to which the Company expects to be entitled in exchange for providing franchisees with the franchise rights to service customers. To determine the transaction price, the Company considers its customary business practices and the terms of the underlying agreement. For the purpose of determining transaction prices, the Company assumes performance obligations will be satisfied as promised in accordance with franchise agreements and that the agreements will not be canceled, renewed, or modified.

The Company's franchise agreements with franchisees have transaction prices that contain a fixed and variable component. Variable consideration includes revenue related to royalties and advertising fees, as the transaction price is based on the franchisees' sales. The variable consideration is recognized based on the actual amounts incurred each month.

Costs to Obtain a Franchise Agreement

The Company incurs commission costs to obtain franchise agreements with franchisees. The commissions are related to franchise fee revenue, which is recognized over time. As a result, the commission costs are capitalized as deferred commission costs and are expensed over the term of the respective franchise agreement, which is determined to be the period of benefit.

Notes to Consolidated Financial Statements

December 31, 2024

Note 2 - Significant Accounting Policies (Continued)

Advertising Expense

In accordance with the Company's franchise agreements, franchisees pay a percentage of monthly sales to an advertising fund to be used for advertising, marketing, and other promotional purposes. Advertising expense is charged to income during the year in which it is incurred. Advertising expense for 2024 was \$3,755,611.

Income Taxes

The Company is treated as a partnership for federal income tax purposes. Consequently, federal income taxes are not payable or provided for by the Company. Members are taxed individually on their pro rata ownership share of the Company's earnings. The Company's net income or loss is allocated among the members in accordance with the Company's operating agreement.

Concentrations of Credit Risk

The Company maintains cash in bank deposit accounts that at times may exceed federally insured limits. The Company has not experienced any losses in such accounts.

Use of Estimates

The preparation of consolidated financial statements in conformity with accounting principles generally accepted in the United States of America requires management to make estimates and assumptions that affect the reported amounts of assets and liabilities and disclosure of contingent assets and liabilities at the date of the consolidated financial statements and the reported amounts of revenue and expenses during the reporting period. Actual results could differ from those estimates.

Subsequent Events

The consolidated financial statements and related disclosures include evaluation of events up through and including March 25, 2025, which is the date the financial statements were available to be issued.

Note 3 - Property and Equipment

Property and equipment are summarized as follows:

Machinery and equipment	\$ 224.961
Vehicles	281,658
Furniture and fixtures	140,755
Computer equipment and software	1,678,755
Leasehold improvements	251,913
The second section of the second seco	-
Total cost	2,578,042
A community and a sound leaf on	500 000
Accumulated depreciation	508,093
Net property and equipment	\$ 2,069,949
(100 bitch call calculation)	2,000,010

Depreciation expense was \$328,825 for the year ended December 31, 2024.

Notes to Consolidated Financial Statements

December 31, 2024

Note 4 - Intangible Assets and Goodwill

Intangible assets and goodwill of the Company at December 31, 2024 are summarized as follows.

	G	ross Carrying Amount	-	Accumulated Amortization
Amortized intangible assets and goodwill:				
Franchise agreements	S	24,937,000	\$	3,205,115
Trade names		7,131,232		934,157
Developed technology		807,674		356,610
Goodwill	100	62,057,306		9,391,742
Total amortized intangible assets and goodwill	S	94,933,212	\$	13,887,624

Amortization expense for intangible assets and goodwill totaled \$9,668,741 for the year ended December 31, 2024.

Estimated amortization expense for the years ending December 31 is as follows:

Years Ending	Amount	
2025	\$	9,515,106
2026		9,515,106
2027		9,515,106
2028		9,515,106
2029		9,515,106
Thereafter	_	33,470,058
Total	S	81,045,588

Note 5 - Leases

The Company leases three offices under operating lease agreements that have an initial term of three to six years. Some leases include one or more options to exercise renewal terms that can extend the lease term at the Company's sole discretion. In addition, some leases contain rights to terminate whereby those termination options are held by the Company, the lessor, or both parties. These options to extend or terminate the lease are included in the lease terms only when it is reasonably certain that the Company will exercise that option. The Company's leases generally do not contain any material restrictive covenants.

Notes to Consolidated Financial Statements

December 31, 2024

Note 5 - Leases (Continued)

Future minimum annual commitments under these operating leases are as follows:

Years Ending December 31		Amount
2025 2026 2027 2028 2029	59	230,942 228,719 203,733 139,664 53,632
Total		856,690
Less amount representing interest		41,079
Present value of net minimum lease payments		815,611
Less current obligations		212,365
Long-term obligations under leases	5	603,246

Lease cost - Operating lease cost	S	235,997
Other information:		
Cash paid for amounts included in the measurement of lease liabilities - Operating cash flows from operating leases	s	222,245
Weighted-average remaining lease term (years) - Operating leases Weighted-average discount rate - Operating leases		3.7 2.6 %

Note 6 - Related Party Transactions

The majority member charges the Company for financial and management services under a management services agreement for reimbursement of reasonable direct expenses, which is included in operating expenses on the accompanying consolidated statement of operations. The total expense for the year ended December 31, 2024 is \$701,563.

Note 7 - Members' Equity

Class A units have voting rights on all matters requiring the consent, approval, or vote of the members. The Class A units receive preference on distributions. There were 1,000,000 units authorized and 109,546 units issued and outstanding as of December 31, 2024.

Class B units do not have voting rights and are issued to designated management employees of the Company without any corresponding capital contribution. The holders of these units are entitled to share in the appreciation of the Company's assets that occur subsequent to the date of grant. The Class B units are dilutive to the participating preferred units. There were 1,000,000 units authorized and no units issued or outstanding as of December 31, 2024.

Notes to Consolidated Financial Statements

December 31, 2024

Note 7 - Members' Equity (Continued)

As defined in the EHC Holding Company, LLC Amended and Restated Limited Liability Company Agreement, a deferred unit provides the right to be issued a Class B unit prior to a significant sale, assuming the fair market value of the Company exceeds the threshold amount, as defined in the Deferred Unit Agreements. The deferred units are time-vesting units that generally vest one-seventh each continuous year of service, becoming fully vested on the seventh anniversary of the grant date. These units allow for accelerated vesting upon the occurrence of a significant sale. As of December 31, 2024, there were 3,803 deferred units issued, 2,540 deferred units outstanding, and 502 deferred units vested. No compensation expense was recognized during 2024, as the fair value of the units is *de minimis*.

Consolidated Financial Report December 31, 2023

Contents

Independent auditor's report	1-2
Financial statements	
Consolidated balance sheets	3
Consolidated statements of operations	4
Consolidated statements of changes in members' equity	5
Consolidated statements of cash flows	6-7
Notes to consolidated financial statements	8-21



RSM U5 LLP

Independent Auditor's Report

Board of Directors EHC Holding Company, LLC and Subsidiaries

Opinion

We have audited the consolidated financial statements of EHC Holding Company, LLC and its subsidiaries (the Company), which comprise the consolidated balance sheets as of December 31, 2023 and 2022, the related consolidated statements of operations, changes in members' equity, and cash flows for the years ended December 31, 2023 and 2022, and the related notes to the consolidated financial statements (collectively, the financial statements).

In our opinion, the accompanying financial statements present fairly, in all material respects, the financial position of the Company as of December 31, 2023 and 2022, and the results of its operations and its cash flows for the years then ended in accordance with accounting principles generally accepted in the United States of America.

Basis for Opinion

We conducted our audits in accordance with auditing standards generally accepted in the United States of America (GAAS). Our responsibilities under those standards are further described in the Auditor's Responsibilities for the Audit of the Financial Statements section of our report. We are required to be independent of the Company and to meet our other ethical responsibilities, in accordance with the relevant ethical requirements relating to our audits. We believe that the audit evidence we have obtained is sufficient and appropriate to provide a basis for our audit opinion.

Responsibilities of Management for the Financial Statements

Management is responsible for the preparation and fair presentation of the financial statements in accordance with accounting principles generally accepted in the United States of America, and for the design, implementation, and maintenance of internal control relevant to the preparation and fair presentation of financial statements that are free from material misstatement, whether due to fraud or error.

In preparing the financial statements, management is required to evaluate whether there are conditions or events, considered in the aggregate, that raise substantial doubt about the Company's ability to continue as a going concern within one year after the date that the financial statements are issued or available to be issued.

THE POWER OF BEING UNDERSTOOD AUDIT | TAX | CONSULTING

1

RSM US.LP is the U.S. member firm of RSM international, a global network of independent audit, tax, and consulting firms. Visit rsmus.com/aboutus for more information regarding RSM US.LLP and

Auditor's Responsibilities for the Audit of the Financial Statements

Our objectives are to obtain reasonable assurance about whether the financial statements as a whole are free from material misstatement, whether due to fraud or error, and to issue an auditor's report that includes our opinion. Reasonable assurance is a high level of assurance but is not absolute assurance and, therefore, is not a guarantee that an audit conducted in accordance with GAAS will always detect a material misstatement when it exists. The risk of not detecting a material misstatement resulting from fraud is higher than for one resulting from error, as fraud may involve collusion, forgery, intentional omissions, misrepresentations, or the override of internal control. Misstatements are considered material if there is a substantial likelihood that, individually or in the aggregate, they would influence the judgment made by a reasonable user based on the financial statements.

In performing an audit in accordance with GAAS, we:

- Exercise professional judgment and maintain professional skepticism throughout the audit.
- Identify and assess the risks of material misstatement of the financial statements, whether due to
 fraud or error, and design and perform audit procedures responsive to those risks. Such procedures
 include examining, on a test basis, evidence regarding the amounts and disclosures in the financial
 statements.
- Obtain an understanding of internal control relevant to the audit in order to design audit procedures
 that are appropriate in the circumstances, but not for the purpose of expressing an opinion on the
 effectiveness of the Company's internal control. Accordingly, no such opinion is expressed.
- Evaluate the appropriateness of accounting policies used and the reasonableness of significant
 accounting estimates made by management, as well as evaluate the overall presentation of the
 financial statements.
- Conclude whether, in our judgment, there are conditions or events, considered in the aggregate, that
 raise substantial doubt about the Company's ability to continue as a going concern for a reasonable
 period of time.

We are required to communicate with those charged with governance regarding, among other matters, the planned scope and timing of the audit, significant audit findings, and certain internal control-related matters that we identified during the audit.

RSM US LLP

Detroit, Michigan April 18, 2024

Consolidated Balance Sheets December 31, 2023 and 2022

	2023		2022
Assets			
Current assets:			
Cash and cash equivalents	\$ 8,258,464	\$	7,906,497
Accounts receivable, net allowance for credit losses	1,950,265		595,028
Inventory	47,089		-
Prepaid expenses and other current assets	490,729		146,386
Total current assets	10,746,547	ererererere	8,647,911
Property, plant, and equipment, net	2,039,654	£	1,671
Other assets:			
Operating lease right-of-use assets, net	1,025,134		121,846
Goodwill, net	48,652,991		16,450,078
Intangibles, net	31,456,699		5,157,183
Other assets	129,180		131,896
Contract costs	12,163,663		1,865,412
Total other assets	93,427,667		23,726,415
Total assets	\$ 106,213,868	\$	32,375,997
Liabilities and Members' Equity			
Current liabilities:			
Accounts payable	\$ 972,368	\$	391,621
Accrued expenses	964,767		1,230,707
Current portion of deferred franchise fees	2,422,957		422,746
Current portion operating lease liabilities	198,150		29,701
Total current liabilities	4,558,242	? :-	2,074,775
Deferred franchise fees, net of current portion	16,479,272		4,869,898
Operating lease liability, net of current portion	834,044		95,071
7),	17,313,316		4,964,969
Total liabilities	21,871,558		7,039,744
Members' equity	84,342,310	3	25,336,253
Total liabilities and members' equity	\$ 106,213,868	\$	32,375,997

See notes to consolidated financial statements.

Consolidated Statements of Operations Years Ended December 31, 2023 and 2022

	2023	2022
Revenues:		
Franchise fee revenue	\$ 6,507,714	\$ 2,171,527
Support services	666,668	1=
Total revenues	7,174,382	2,171,527
Operating expenses:		
Cost of services	306,538	~
General and administrative expenses	5,531,685	2,986,145
Payroll and benefits	3,885,484	1,755,242
Transaction expenses	1,579,290	1,502,800
Amortization and depreciation expense	3,040,926	919,800
Other operating expenses	546,426	15,406
Total operating expenses	14,890,349	7,179,393
Net loss	\$ (7,715,967)	\$ (5,007,866)

See notes to consolidated financial statements.

Consolidated Statements of Changes in Members' Equity Years Ended December 31, 2023 and 2022

Balance, January 1, 2022	\$ 7,844,11	19
Contributed capital	22,500,00	00
Net loss	(5,007,86	36)
Balance at December 31, 2022	25,336,25	53
Contributed capital	66,722,02	24
Net loss	(7,715,96	37)
Balance at December 31, 2023	\$ 84,342,31	10

See notes to consolidated financial statements.

Consolidated Statements of Cash Flows Years Ended December 31, 2023 and 2022

		2023	 2022
Cash flows from operating activities:			
Net loss	\$	(7,715,967)	\$ (5,007,866)
Adjustments to reconcile net loss to net cash used in operating activities:			
Depreciation and amortization		3,040,926	919,800
Changes in assets and liabilities, net of acquisitions:			
Accounts receivable		146,847	(453,957)
Prepaid expenses and other current assets		(235,413)	(44,821)
Other assets		2,716	(6,575)
Contract costs		(1,143,324)	(117,661)
Accounts payable and accrued expenses		(538,706)	142,018
Deferred franchise fees		1,532,134	300,099
Operating lease assets and liabilities		95,605	2,926
Net cash used in operating activities		(4,815,182)	(4,266,037)
Cash flows from investing activities: Acquisition of businesses, net of cash acquired Purchase of intangible assets Purchase of property and equipment Net cash used in investing activities	~	(46,199,803) - (1,355,072) (47,554,875)	(13,055,236) (40,512) (1,270) (13,097,018)
Cash flows from financing activities:			
Proceeds from capital contributions		52,722,024	 20,000,000
Net cash provided by financing activities	_	52,722,024	20,000,000
Net increase in cash and cash equivalents		351,967	2,636,945
Cash and cash equivalents, beginning		7,906,497	5,269,552
Cash and cash equivalents, ending	\$	8,258,464	\$ 7,906,497

(Continued)

Consolidated Statements of Cash Flows (Continued) Years Ended December 31, 2023 and 2022

		2023	2022
Supplemental schedule of noncash operating, investing and			
financing activities:			
Acquisition of businesses:			
Assets acquired	\$	40,044,046	\$ 7,257,883
Liabilities assumed		(13,661,493)	(5,078,960)
Net identifiable assets acquired		26,382,553	2,178,923
Goodwill		34,390,411	14,545,344
Net assets acquired		60,772,964	16,724,267
Less cash acquired		(992,260)	(620,715
Add due from seller		419,099	45,924
Less contingent consideration			(594,240
Less units issued as consideration		(14,000,000)	(2,500,000
Cash purchase price	_\$_	46,199,803	\$ 13,055,236

See notes to consolidated financial statements.

Notes to Consolidated Financial Statements

Note 1. Summary of Significant Accounting Policies

Nature of business: EHC Holding Company, LLC and Subsidiaries (the Company) through its wholly owned subsidiaries, including Executive Home Care Franchising, LLC (EH), ALL Franchising, LLC (ALL), ALL Licensing, LLC (Licensing), B&P Burke, LLC (Grasons), Brothers Franchising, LLC (Brothers), and Brothers That Just do Gutters, LLC (BG).

EH is a franchisor that provides home care services to the elderly, physically handicapped and injured, allowing them to live at home. Executive Homecare began operations in 2004.

ALL is a franchisor that provides senior care placement and referral service for in-home companion care, independent retirement options, assisted living, memory care, and skilled nursing facilities. Assisted Living Locators began operations in 2006 and provides services nationwide.

Licensing is an operating company that owns technology used by ALL.

Grasons is a franchisor that provides services for estate sales and business liquidation services. Grasons was established in 2011 and conducts operations from its principal office in California.

Brothers is a franchisor that provides services associated with gutter instillation, repair, and maintenance.

BG operates a Brothers franchise.

Significant accounting policies:

Basis of presentation: The consolidated balance sheets is presented as of December 31, 2023 and 2022. The consolidated statements of operations, changes in members' equity, and cash flows are presented for the years ended of December 31, 2023 and 2022. The accompanying consolidated financial statements of the Company include its wholly owned subsidiaries.

All intercompany transactions have been eliminated. The accompanying consolidated financial statements have been prepared in accordance with accounting standards set by the Financial Accounting Standards Board (FASB). The FASB sets accounting principles generally accepted in the United States of America (U.S. GAAP) that the Company follows to ensure its financial condition, results of operations, and cash flows are consistently reported. References to U.S. GAAP issued by the FASB in these notes to the consolidated financial statements are to the FASB Accounting Standards Codification (ASC).

Revenue recognition policy: The Company recognizes revenue in accordance with ASC 606, Revenue from Contracts with Customers, which provides a five-step model for recognizing revenue from contracts with customers as follows: identify the contract with a customer, identify the performance obligations in the contract, determine the transaction price, allocate the transaction price to the performance obligations in the contract, and recognize revenue when or as performance obligations are satisfied.

Notes to Consolidated Financial Statements

Note 1. Summary of Significant Accounting Policies (Continued)

Nature of services - The Company's franchise agreements include (a) the right to use its symbolic intellectual property over the term of each franchise agreement, (b) preopening services, such as training, (c) ongoing services, such as management of the national brand fund contributions and support services for the franchisees and (d) for certain subsidiaries, a license to use the Company's internal-use software, which is hosted on the Company's software as a service (SaaS) platform. These promises are highly dependent upon and interrelated with the franchise right granted in the franchise agreement, so they are not considered to be individually distinct and therefore are accounted for as a single performance obligation. The performance obligation under the franchise agreement is the promise to provide daily access to the symbolic intellectual property over the term of each franchise agreement, which is a series of distinct services that represents a single performance obligation. Although the franchisor's underlying activities associated with the symbolic intellectual property will vary both within a day and day to day, the symbolic intellectual property is accessed over time and the customer (the franchisee) simultaneously receives and consumes the benefit from the franchisor's performance of providing access to the symbolic intellectual property (including other related activities). Revenue earned from providing the services is collectively referred to as franchise fee revenue.

The Company's franchise fee revenue includes franchise royalties, franchise fees, advertising fund contributions and support services performed for franchisees. Initial franchise fees are based on the market type selected and are paid at the time an individual franchise agreement is signed.

The Company also operates a franchise location. The revenue for this consists of revenue recognized at a point in time as the service is completed.

Payment terms - The transaction price is the amount of consideration to which the Company expects to be entitled in exchange for transferring goods and services to the customer. Agreements may include initial and renewal franchise fees, sales-based royalties, and fees for administrative services performed for the franchisee.

The Company believes its franchising agreements do not contain a significant financing component because (a) the timing of the upfront payment does not arise for the reason of provision of financing to the Company and (b) the sales-based royalty is variable and based on factors outside the Company or the franchisee's control.

Revenue recognition - Initial and renewal franchise fees are recognized as revenue on a straight-line basis over the term of the respective agreement beginning when the agreement is signed. Franchise agreements typically have a term of 10 years with the option to renew for an additional five years if the franchisee is in compliance with the terms of the franchise agreement.

Continuing royalties are calculated as a percentage of franchisees' reported sales that are related entirely to the Company's performance obligation under the franchise agreement. These royalties are considered variable consideration, but because they relate to a license of intellectual property, they are not included in the transaction price. Instead, royalty revenue is recognized as franchisee sales occur. Advertising contributions received from the Company's franchisees are recorded as a component of franchise fee revenue in the consolidated statements of operations.

Contract balances - The Company records accounts receivable when it has the unconditional right to issue an invoice and receive payment, regardless of whether revenue has been recognized. If revenue has not yet been recognized, a contract liability (deferred franchise fees) also is recorded. Revenue is recognized on a straight-line basis over the life of the franchise agreement.

Notes to Consolidated Financial Statements

Note 1. Summary of Significant Accounting Policies (Continued)

Commission costs - The Company defers those direct and incremental costs associated with the sale of franchises. Contract costs are charged to earnings when the related deferred franchise and territory fees are recognized as revenue over the term of the respective agreement. The Company has determined the period of benefit for direct and incremental costs associated with the sale of franchises to be the initial term of the franchise agreement. Expense is recognized on a straight-line basis commensurate with the pattern of revenue recognition. Contract costs are recorded in other assets in the accompanying consolidated balance sheets.

Advertising funds - The Company collects funds from its franchisees for advertising pursuant to the Company's franchise agreements at a percentage of franchisee sales. These advertising services are not considered distinct because they are highly dependent and interrelated to the franchise right. Advertising contributions are considered part of the transaction price for the franchise right and recognized as revenue as the underlying sales occur. The advertising costs incurred for franchisees will be expensed in accordance with the Company's normal policy.

Cash and cash equivalents: The Company considers all short-term, highly liquid investments with original maturities of 90 days or less to be cash and cash equivalents. Cash equivalents consist of money market accounts.

Accounts receivable: Accounts receivables represent amounts due from franchisees pursuant to their individual franchise agreements. Accounts receivables are stated at historical value which approximates fair value. The allowance for credit losses on accounts receivables represents the Company's estimate of expected credit losses over the lifetime of the receivables. This estimation process is based on historical experience, current conditions, asset-specific risk characteristics and reasonable and supportable forecasts about future economic and market conditions. Accounts receivables are written off when deemed uncollectible. Recoveries of royalty receivables previously written off are recorded when received. The allowance for credit loss was approximately \$144,000 and \$37,000 for the years ended December 31, 2023 and 2022, respectively. The Company will continue to monitor and evaluate the adequacy of the allowance for credit losses on accounts receivable on a regular basis and make adjustments as necessary in response to changes in economic conditions and credit quality indicators.

The Company adopted Accounting Standards Update (ASU) 2016-13, Financial Instruments—Credit Losses (Topic 326), on January 1, 2023. This accounting standard requires companies to measure expected credit losses on financial instruments based on the total estimated amount to be collected over the lifetime of the instruments which would include accounts receivables. Prior to the adoption of this accounting standard, the Company recorded incurred loss reserves against account receivable balances based on current and historical information. The adoption of this ASU did not have a material effect on the Company's financial statements.

Concentration of credit risk: The Company's financial instruments that are exposed to concentrations of credit risk consist primarily of cash and cash equivalents and accounts receivable. The Company maintains its cash and cash equivalents in bank deposit accounts, which at times may exceed federally insured limits. The Company has not experienced any losses in such accounts. The Company believes it is not exposed to any significant credit risk on cash and cash equivalents. The Company grants credit to its franchisees and customers. Consequently, the Company's ability to collect the amounts due from franchisees and customers is affected by economic fluctuations. The Company routinely assesses the financial strength of its franchisees and customers and believes that its accounts receivable credit risk exposure is limited.

Notes to Consolidated Financial Statements

Note 1. Summary of Significant Accounting Policies (Continued)

Software development costs: Costs for software developed for internal use are accounted for in accordance with ASC 350, Intangibles—Goodwill and Other - Internal-Use Software. ASC 350 requires the capitalization of certain costs incurred in connection with developing or obtaining internal-use software. In accordance with ASC 350, the Company expenses costs incurred in the preliminary project stage of developing or acquiring internal use software, such as research and feasibility studies, as well as costs incurred in the post-implementation/operational stage, such as maintenance and training. Capitalization of software development costs occurs only after the preliminary project stage is complete, management authorizes the project, and it is probable that the project will be completed and the software will be used for the function intended. Costs associated with the purchase and development of computer software are capitalized and amortized on a straight-line basis over the estimated useful life of the related asset. Software development costs are recorded in property and equipment in the accompanying consolidated balance sheets. The Company capitalized software development costs of approximately \$1,271,000 and \$0 for the years ended December 31, 2023 and 2022, respectively.

Property and equipment: Property and equipment are stated at cost, net of accumulated depreciation and amortization. Expenditures for additions and improvements are capitalized while maintenance and repair expenditures are charged to operations as incurred. When assets are sold or otherwise retired from service, their cost and related accumulated depreciation and amortization are removed from the accounts and any gain or loss is included in the results of operations. Depreciation and amortization is computed using the straight-line method. Property and equipment have estimated useful lives of three to ten years. Depreciation expense for the years ended December 31, 2023 and 2022, was approximately \$98,000 and \$700, respectively.

Goodwill and intangibles: Goodwill is recognized for the excess of the fair value of an acquired business over the fair value of the identifiable net assets acquired. Under ASC 350, Intangibles—Goodwill and Other, the Company elected the accounting alternative to amortize goodwill on a straight-line basis over 10 years.

The Company has elected the provisions of ASU 2014-18, *Business Combinations (Topic 805)*: Accounting for Identifiable Intangible Assets in a Business Combination. ASU 2014-18 specifies that a private company that elects the accounting alternative to recognize or otherwise consider the fair value of intangible assets as a result of any in-scope transactions should no longer recognize separately from goodwill: (1) customer-related intangible assets unless they are capable of being sold or licensed independently from the other assets of the business and (2) noncompetition agreements.

The Company tests its recorded goodwill for impairment upon a triggering event. Factors that could trigger an impairment test include, but are not limited to, underperformance relative to historical or projected future operating results, significant changes in the manner of use of the acquired assets or the overall business, significant negative industry or economic trends and a sustained period where market capitalization, plus an appropriate control premium, is less than member's equity. Goodwill is tested using a fair-value approach at the entity level. No indicators of impairment were identified for the years ended December 31, 2023 and 2022.

Intangible assets include franchise agreements, trade names, and developed technology. Intangible assets are amortized on a straight-line basis over their estimated useful lives, which range between five to 14 years.

Notes to Consolidated Financial Statements

Note 1. Summary of Significant Accounting Policies (Continued)

Long-lived assets: Long-lived assets are reviewed for impairment whenever events or changes in circumstances indicate that the carrying amount of the asset may not be recoverable. If impairment is considered, recoverability of these assets is measured by a comparison of the carrying amount of the asset to estimated future undiscounted cash flows expected to be generated by the asset. If the carrying amount of an asset exceeds its estimated future cash flows, an impairment charge is recognized by the amount of which the carrying amount of the asset exceeds the fair value of the asset. No impairment loss has been recognized by the Company as of December 31, 2023 and 2022.

Leases: In February 2016, the FASB issued ASC 842, Leases, to increase transparency and comparability among organizations related to their leasing arrangements. The update requires lessees to recognize most leases on their balance sheets as a right-of-use (ROU) asset representing the right to use an underlying asset and a lease liability representing the obligation to make lease payments over the lease term, measured on a discounted basis. ASC 842 also requires additional disclosure of key quantitative and qualitative information for leasing arrangements. Similar to the previous lease guidance, the update retains a distinction between finance leases (similar to capital leases in ASC 840, Leases) and operating leases, with classification affecting the pattern of expense recognition in the income statement. The Company adopted ASC 842 on January 1, 2022, using the optional transition method to the modified retrospective approach, which eliminates the requirement to restate the prior-period consolidated financial statements. Under this transition provision, the Company has applied ASC 842 to reporting periods beginning on January 1, 2022, while prior periods continue to be reported and disclosed in accordance with the Company's historical accounting treatment under ASC 840, Leases.

The Company elected the package of practical expedients under the transition guidance within ASC 842, in which the Company does not reassess (1) the historical lease classification, (2) whether any existing contracts at transition are or contain leases or (3) the initial direct costs for any existing leases. The Company has not elected to adopt the hindsight practical expedient, and therefore measured the ROU asset and lease liability using the remaining portion of the lease term upon the adoption of ASC 842 on January 1, 2022.

The Company determines if an arrangement is or contains a lease at inception, which is the date on which the terms of the contract are agreed to, and the agreement creates enforceable rights and obligations. A contract is or contains a lease when (i) explicitly or implicitly identified assets have been deployed in the contract and (ii) the Company obtains substantially all of the economic benefits from the use of that underlying asset and directs how and for what purpose the asset is used during the term of the contract. The Company also considers whether its service arrangements include the right to control the use of an asset.

The Company made an accounting policy election available under ASC 842 not to recognize ROU assets and lease liabilities for leases with a term of 12 months or less. For all other leases, ROU assets and lease liabilities are measured based on the present value of future lease payments over the lease term at the commencement date of the lease (or January 1, 2022, for existing leases upon the adoption of ASC 842). The ROU assets also include any initial direct costs incurred and lease payments made at or before the commencement date and are reduced by any lease incentives. To determine the present value of lease payments, the Company made an accounting policy election available to nonpublic companies to utilize a risk-free borrowing rate, which is aligned with the lease term at the lease commencement date (or remaining term for leases existing upon the adoption of ASC 842).

Notes to Consolidated Financial Statements

Note 1. Summary of Significant Accounting Policies (Continued)

The Company has made an accounting policy election to account for lease and nonlease components in its contracts as a single lease component for its various asset classes. The nonlease components typically represent additional services transferred to the Company, such as common area maintenance for real estate, which are variable in nature and recorded in variable lease expense in the period incurred.

Adoption of ASC 842 resulted in the recording of additional ROU assets and lease liabilities related to the Company's operating leases of approximately \$153,000 at January 1, 2022. The adoption of the new lease standard did not materially impact consolidated net earnings or consolidated cash flows and did not result in a cumulative-effect adjustment to the opening balance of retained earnings.

Fair value measurements: The Company uses the fair value measurement and disclosure guidance for all assets and liabilities that are recognized or disclosed at fair value in the consolidated financial statements. The guidance defines fair value as the price that would be received to sell an asset or paid to transfer a liability in an orderly transaction between market participants at the measurement date. In determining fair value, the Company uses various methods including market, income and cost approaches. Based on these approaches, the Company often utilizes certain assumptions that management believes market participants would use in pricing the asset or liability, including assumptions about risk and or the risks inherent in the inputs to the valuation technique. These inputs can be readily observable, market corroborated, or generally unobservable inputs. The Company utilizes valuation techniques that maximize the use of observable inputs and minimize the use of unobservable inputs. The fair value hierarchy ranks the quality and reliability of the information used to determine fair values. Assets and liabilities carried at fair value will be classified and disclosed in one of the following three categories:

- **Level 1:** Inputs to the valuation methodology are unadjusted quoted prices for identical assets or liabilities in active markets that are accessible at the measurement date.
- **Level 2:** Inputs to the valuation methodology include quoted prices in markets that are not active or quoted prices for similar assets and liabilities in active markets, and inputs that are observable for the asset or liability, either directly or indirectly, for substantially the full term of the financial instrument.
- **Level 3:** Inputs to the valuation methodology are unobservable, reflecting the entity's own assumptions about assumptions market participants would use in pricing the asset or liability.

Income taxes: As a limited liability company, the Company is treated as a partnership for federal and state income tax purposes. As such, the taxable income of the Company is allocated in the tax returns of its members for federal and state tax purposes in accordance with their respective percentage ownership. Accordingly, no provision for federal income taxes is included in the consolidated financial statements. Entity-level, composite state and local income taxes (benefits) are accrued at the applicable rates, if any, and are included in the consolidated statements of operations.

The FASB provides guidance for how uncertain tax provisions should be recognized, measured, disclosed, and presented in the consolidated financial statements. The Company identifies its tax positions taken or expected to be taken in the course of preparing its tax returns and determines whether any tax positions are more likely than not of being sustained when challenged or when examined by the applicable tax authority. Management has determined that there are no uncertain tax positions at December 31, 2023 and 2022.

Notes to Consolidated Financial Statements

Note 1. Summary of Significant Accounting Policies (Continued)

Use of estimates: The preparation of consolidated financial statements in conformity with U.S. GAAP requires management to make estimates and assumptions that affect the reported amounts of assets and liabilities and disclosure of contingent assets and liabilities at the date of the consolidated financial statements and the reported amounts of revenues and expenses during the reporting period. Actual results could differ from those estimates.

Subsequent events: The Company evaluated subsequent events for potential required disclosure through April 18, 2024, which is the date the consolidated financial statements were available to be issued

Note 2. Acquisition of Businesses

Assisted Living Locators: On August 24, 2022, the Company acquired ALL and Licensing. ALL is a franchisor and the franchises provide senior placement and referral services. Licensing is an operating company that owns technology used by ALL. As a result of the transaction, the Company gains penetration into this industry. The Company obtained control of the business by acquiring 100% of the membership interests of ALL and Licensing, therefore the acquisition was accounted for as a business combination. The results of this acquisition have been included in the consolidated financial statements since the date of the acquisition.

The acquisition was funded through cash contributions of equity.

The goodwill arising from the above acquisition is largely due to the ALL's and Licensing's fair value of certain intangible assets along with the assembled workforce subsumed into goodwill, ALL's and Licensing's presence in the marketplace and their long-term expected revenue growth. The goodwill is deductible for income tax purposes.

The following table summarizes the consideration paid and the amounts of the assets acquired, and liabilities assumed at the date of acquisition:

Cons	ider	ation

Cash	\$	10,604,497
3,409 Class A shares of EHC Holding Company, LLC		2,500,000
Total invested capital	\$	13,104,497
Recognized amount of identifiable assets acquired and liabilities assumed:		
Cash	\$	554,552
Receivables		90,587
Prepaid assets		63,265
Contract assets		1,535,871
Intangible assets		4,212,000
Other asset		20,232
Accounts payable		(198,630)
Accrued expenses and other liabilities		(246,569)
Contract liabilities		(4,004,575)
Total identifiable net assets acquired	5.0	2,026,733
Goodwill		11,077,764
	\$	13.104.497

Notes to Consolidated Financial Statements

Note 2. Acquisition of Businesses (Continued)

The fair value of the 3,409 Class A Units was determined based on the value of the Company at the acquisition date, using unobservable inputs.

Transaction expenses of approximately \$1,077,000 were incurred as a result of the business combination and have been expensed by the Company and included in the consolidated statement of operations within the transaction expenses line item for the year ended December 31, 2022

Of the \$4,212,000 of identified intangible assets, \$2,736,000 was assigned to franchise agreements, \$1,034,000 was assigned to trade name, and \$442,000 was assigned to developed technology. Each were determined to have useful lives of 10 years, 14 years and five years, respectively. The weighted-average useful life in years for the identified intangible assets in aggregate was 10.45 years. Franchise agreements, trade names, and developed technology were valued using the multiperiod excess earnings method, relief from royalty method, and cost-to-recreate method, respectively.

Grasons: On November 21, 2022, the Company acquired Grasons. Grasons is a franchisor and the franchises provides downsizing and estate sale services, business liquidation services, senior transition assistance, home staging, and home organization services. As a result of the transaction, the Company gains penetration into this industry. The Company obtained control of the business by acquiring 100% of the membership interests of Grasons, therefore the acquisition was accounted for as a business combination. The results of this acquisition have been included in the consolidated financial statements since the date of the acquisition.

The acquisition was funded through cash contributions of equity.

The goodwill arising from the above acquisition is largely due to the Grasons' fair value of certain intangible assets along with the assembled workforce subsumed into goodwill, Grasons' presence in the marketplace and its long-term expected revenue growth. The goodwill is deductible for income tax purposes.

The following table summarizes the consideration paid and the amounts of the assets acquired, and liabilities assumed at the date of acquisition:

Notes to Consolidated Financial Statements

Note 2.	Acquisition of Businesses	(Continued)	
---------	---------------------------	-------------	--

Consideration:		
Cash	\$	3,071,454
Contingent consideration		594,240
Due from seller		(45,924)
Total invested capital	\$	3,619,770
Recognized amount of identifiable assets acquired and liabilities assumed:		
Cash	\$	66,163
Accounts receivable		25,255
Other current assets		12,508
Deposits		1,450
Intangible assets		676,000
Accounts payable		(792)
Accrued expenses and other liabilities		(23,777)
Deferred revenue		(604,617)
Total identifiable net assets acquired	-	152,190
Goodwill	4	3,467,580
	\$	3,619,770

The Company engaged an independent valuation firm to assist with the valuation of intangible assets using the relief from royalty method.

Contingent consideration in the amount of \$594,240 was recognized at the date of acquisition and is included in the balance sheet within the accrued expenses line item at December 31, 2022. The contingent consideration is based on annual royalties for the 12 months ended December 31, 2022 multiplied by an agreed upon rate. The contingent consideration was settled for approximately \$492,000 and goodwill was adjusted for approximately \$102,000 in 2023.

Transaction expenses of approximately \$404,000 were incurred as a result of the business combination and have been expensed by the Company and included in the consolidated statement of operations within the transaction expenses line item for the year ended December 31, 2022

Of the \$676,000 of identified intangible assets, \$295,000 was assigned to franchise agreements and \$381,000 was assigned to trade name. Each were determined to have useful lives of 10 years and 14 years, respectively. The weighted-average useful life in years for the identified intangible assets in aggregate was 12.25 years. Franchise agreements and trade names were valued using the multiperiod excess earnings method and relief from royalty method, respectively.

Brothers: on November 22, 2023, the Company acquired Brothers and BG (collectively Brothers Gutters). Brothers Gutters includes a franchisor and the franchises provide services associated with gutter installation, repair and maintenance. In addition, one franchisee was acquired. As a result of the transaction, the Company gains penetration into this industry. The Company obtained control of the business by acquiring 100% of the membership interests of Brothers Gutters, therefore the acquisition was accounted for as a business combination. The results of this acquisition have been included in the financial statements since the date of the acquisition.

The acquisition was funded through cash contributions of equity.

Notes to Consolidated Financial Statements

Note 2. Acquisition of Businesses (Continued)

The goodwill arising from the above acquisition is largely due to the Brothers Gutters' fair value of certain intangible assets along with the assembled workforce subsumed into goodwill, Brothers Gutters' presence in the marketplace and its long-term expected revenue growth. The goodwill is deductible for income tax purposes.

The following table summarizes the consideration paid and the amounts of the assets acquired, and liabilities assumed at the date of acquisition:

Consideration:

Cash 14,709 Class A shares of EHC Holding Company, LLC Due from seller Total purchase consideration	\$ 47,192,063 14,000,000 (419,099) \$ 60,772,964
Fair value of identifiable assets acquired and liabilities assumed:	
Cash	\$ 992,260
Accounts receivable	1,082,985
Inventory	47,089
Prepaid assets	108,930
Fixed asset	780,949
Contract assets	9,154,927
Intangible assets	27,157,000
Right-of-use asset and other assets	719,906
Accounts payable	(497,365)
Accrued expenses and other	(458,242)
Lease liabilities	(628, 435)
Contract liabilities	(12,077,451)
Total identifiable net assets acquired	26,382,553
Goodwill	34,390,411
	\$ 60,772,964

The fair value of the 14,709 Class A Units was determined based on the value of the Company at the acquisition date, using unobservable inputs.

The primary area of the preliminary valuation that is not yet finalized relates to the on-going negotiation with the Company and the seller on the working capital adjustment. An estimate of \$419,099 was recorded at December 31, 2023.

Transaction expenses of approximately \$1,579,000 were incurred as a result of the business combination and have been expensed by the Company and included in the statement of operations within the transaction expenses line item for the year ended December 31, 2023.

Of the \$27,157,000 of identified intangible assets, \$21,578,000 was assigned to franchise agreements and \$5,579,000 was assigned to trade name. Each were determined to have useful lives of 10 years and 14 years, respectively. The weighted-average useful life in years for the identified intangible assets in aggregate was 10.8 years. Franchise agreements and trade names were valued using the multi-period excess earnings method and relief from royalty method, respectively.

Notes to Consolidated Financial Statements

Note 3. Property and Equipment

Property and equipment is summarized as follows:

	2023	2022
Furniture and fixtures	\$ 386,373	\$ 2,367
Computer equipment and software	1,450,561	=:
Equipment	50,890	#8
Leasehold improvements	250,564	¥t
Total property and equipment	2,138,388	2,367
Less accumulated depreciation and amortization	(98,734)	(696)
Property and equipment, net	\$ 2,039,654	\$ 1,671

Note 4. Intangible Assets and Goodwill

Following is a summary of intangible assets:

	December 31, 2023						
	3	Gross Carrying Accumulated Amount Amortization			Net Carrying Amount		
Franchise agreements Trade names Developed technology	\$	24,937,000 7,126,000 482,512 32,545,512	\$	711,414 248,785 128,614 1,088,813	\$	24,225,586 6,877,215 353,898 31,456,699	
Goodwill	\$	51,475,640	\$	2,822,649	\$	48,652,991	
	December 31, 2022						
		, 0			Net Carrying Amount		
Franchise agreements Trade names Developed technology	\$	3,359,000 1,547,000 482,512	\$	139,043 60,174 32,112	\$	3,219,957 1,486,826 450,400	
	\$	5,388,512	\$	231,329	\$	5,157,183	
Goodwill	^	17,187,323	\$	737,245	\$	16,450,078	

Notes to Consolidated Financial Statements

Note 4. Intangible Assets and Goodwill (Continued)

The change in the carrying value of goodwill for the year ended December 31, 2023 and 2022, is as follows:

Balance at December 31, 2021	\$ 2,602,433
Additions of goodwill	14,545,344
Amortization expense	(697,699)
Balance at December 31, 2022	\$ 16,450,078
Additions of goodwill	34,390,411
Reduction of contingent consideration	(102,094)
Amortization expense	(2,085,404)
Balance at December 31, 2023	\$ 48,652,991

Amortization expense recognized on intangible assets and goodwill totaled approximately \$2,943,000 and \$919,000 as of December 31, 2023 and 2022, respectively.

The future estimated aggregate amortization expense for intangibles and goodwill is as follows for each of the next five years ending December 31:

		Goodwill	Intangibles		:S	
Years ending December 31:	,					
2024	\$	5,147,564	\$	3,275,574		
2025		5,147,564		3,275,574		
2026		5,147,564		3,275,574		
2027		5,147,564		3,239,050		
2028		5,147,564		3,152,671		

Note 5. Leases

The Company leases three offices under operating lease agreements that have an initial term of 3 - 6 years. Some leases include one or more option to exercise renewal terms that can extend the lease term for more years, at the Company's sole discretion. In addition, some leases contain rights to terminate whereby those termination options are held by either the Company, the lessor, or both parties. These options to extend or terminate the lease are included in the lease terms only when it is reasonably certain that the Company will exercise that option. The Company's leases generally do not contain any material restrictive covenants.

Operating lease cost is recognized on a straight-line basis over the lease term. Lease expense is approximately \$106,000 and \$29,000 for the years ended December 31, 2023 and 2022, respectively. For the years ended December 31, 2023 and 2022, weighted average remaining on lease term is 4.7 years and 3.8 years and the weighted average discount rate is 2.64% and 1.37%, respectively.

Notes to Consolidated Financial Statements

Note 5. Leases (Continued)

Supplemental cash flow information related to leases is as follows for the years ended December 31, 2023 and 2022:

	2023		2022	
Operating leases: Operating cash outflows—payments on operating leases	\$	91,762	\$ 30,150	
Right-of-use assets in exchange for new lease obligations: Operating leases	\$	1,084,556	\$ _	

The future minimum rentals under this lease for the years subsequent to December 31, 2023, are as follows:

follows:	
Years ending December 31:	
2024	\$ 222,245
2025	230,942
2026	228,719
2027	203,733
2028	139,664
Thereafter	 53,632
Total lease payments	1,078,935
Less imputed interest	 (46,741)
Total present value of lease liabilities	\$ 1,032,194

Note 6. Related-Party Transactions

A company related to the Company's majority member, charges the Company for financial and management services under a management services agreement for reimbursement of reasonable direct expenses, which is included in general and administrative expenses on the accompanying consolidated statements of operations. The total expense for the years ended December 31, 2023 and 2022, is approximately \$568,000 and \$500,000, respectively.

Notes to Consolidated Financial Statements

Note 7. Members' Equity

Members' equity consisted of the following membership units:

		2023		
	_	Units	Units Issued	
	· <u></u>	Authorized	and Outstanding	
Class A Units		1,000,000	109,546	
Class B Units		1,000,000	1,661	
	· · · · · · · · · · · · · · · · · · ·	2022		
		Units	Units Issued	
		Authorized	and Outstanding	
Class A Units		1,000,000	39,316	
Class B Units		1,000,000	823	

Class A Units have voting rights on all matters requiring the consent, approval or vote of the Members. The Class A Units receive preference on distributions.

Class B Units are deferred units that do not have voting rights and have been issued to designated management employees of the Company without any corresponding capital contribution. The holders of these units are entitled to share in the appreciation of the Company's assets that occur subsequent to the date of grant. The Class B Units are dilutive to the participating preferred units. The units substantially vest upon a change in control of the Company, if still employed.

The Company has issued 2,062 and 823 units to certain management employees as of December 31, 2023 and 2022 and had 1,224 units forfeited as of December 31, 2023.

No compensation expense is recognized on the Class B units as their vesting condition is not considered probable until a change in control occurs.

Note 8. Phantom Stock

During 2023, the Company adopted a phantom deferred unit plan. Each share of phantom deferred unit awarded to eligible individuals represents a contractual right to receive an amount in cash equal to the fair value of the unit upon the occurrence of a significant sale. The Company has authorized 1,000,000 units and 3,286 units are outstanding at December 31, 2023. No compensation expense is recognized on the phantom stock deferred units as their vesting condition is not considered probable until a significant sale occurs.

Exhibit F-2

<u>Unaudited Interim Financial Information Statements</u>

[See Attached]

THESE FINANCIAL STATEMENTS ARE PREPARED WITHOUT AN AUDIT. PROSPECTIVE FRANCHISES OR SELLERS OF FRANCHISES SHOULD BE ADVISED THAT NO CERTIFIED PUBLIC ACCOUNTANT HAS AUDITED THESE FIGURES OR EXPRESSED HIS/HER OPINION WITH REGARD TO THE CONTENT OR FORM.

EHC Holding Company, LLC and Subsidiaries Consolidated Balance Sheets May 31, 2024 (UNAUDITED)

Assets:	May 31, 2024
Assets.	
Current Assets:	
Cash and Cash equivalents	7,737,039
Accounts receivable, net of allowance for credit losses	2,141,489
Inventory	47,089
Prepaid expenses and other current assets	810,629
Total current assets	10,736,246
Property, plant, and equipment, net	1,972,747
Other assets:	
Operating lease right-of-use assets, net	1,008,819
Goodwill and intangibles, net	76,640,086
Other assets	12,492
Contract costs	12,727,059_
Total other assets	90,388,456
Total assets	103,097,450
Liabilities and Member's Equity	
Current liabilities	
Accounts payable	667,962
Accrued expenses	1,307,394_
Total current libilities	1,975,356
Deferred franchise fees	19,512,125
Operating lease liability	998,147
Total liabilities	22,485,629
Manufaced and the	
Members' equity	80,611,821
Total liabilities and members' equity	103,097,450

EHC Holding Company, LLC and Subsidiaries Consolidated Statement of Operations January 1, 2024 thru May 31, 2024 UNAUDITED

	January 1, 2024 thru May 31, 2024		
Revenues:			
Franchise fee revenue	\$	7,495,475	
Support services	\$	1,604,778	
Total revenues	\$	9,100,253	
Operating expenses:			
Cost of services	\$	878,487	
General and administrative expenses	\$	3,462,834	
Payroll and benefits	\$	4,233,579	
Amortization and depreciation expense	\$	3,820,317	
Other operating expenses	\$	854,625	
Total operating expenses	\$	13,249,841	
Net loss	\$	(4,149,588)	

EHC Holding Company, LLC and Subsidiaries Consolidated Statements of Cash Flows January 1, 2024 thru May 31, 2024 UNAUDITED

	January 1, 2024 ช่าง May 31, 2024		
Cash Flows from operating activities	-	siei d May 01, 2024	
Net Loss	\$	(4,149,588)	
Adjustments to reconcile net loss to net cash used in operating activities:			
Depreciation and amorization	\$	3,820,317	
Changes in assets and liabilities			
Accounts receivable	\$	(191,224)	
Prepaid expenses and other current assets	\$ \$	(319,900)	
Other assets	\$	116,688	
Contract costs	\$ \$	(563,396)	
Accounts payable and accrued expenses	\$	(105,026)	
Deferred franchise fees	\$	609,896	
Operating lease assets and liabilities	\$	(17,732)	
Net cash used in operating activities	\$	(799,964)	
Cash Flows from investing activities			
Sale of intangible assets	\$	419,099	
Purchase of property and equipment	\$	(140,560)	
Net cash from investing activities	\$	278,539	
Cash Flows from financing activities	\$		
Net cash from financing activities	\$	-	
Net decrease in cash and cash equivalents	\$	(521,425)	
Cash and cash equivalents, beginning	_\$	8,258,464	
Cash and cash equivalents, ending	\$	7,737,039	

EHC Holding Company, LLC Balance Sheet April 30, 2025 (Unaudited)

	Month Ending 04/30/2025
Assets	
Current Assets	
Cash and Cash Equivalents	4,988,775
Accounts Receivable, Net	3,776,307
Inventory	23,968
Prepaid Expenses	607,517
Intercompany Receivable	(7,111)
Other Current Assets	8.281,516
Total Current Assets	17,670,971
Fixed Assets, Net	
Fixed Assets	1,183,394
Total Fixed Assets, Net	1,183,394
Intangible Assets, Net	
Intangible Assets	16,872,797
Goodwill	59,971,512
Amortization	(220,427)
Total Intangible Assets, Net	77,064,736
Other Assets	
Other Assets	3,612,121
Total Other Assets	3,612,121
Total Assets	\$ 99,531,223
Liabilities and Equity	
Current Liabilities	
Accounts Payable	952,730
Accrued Liabilities	2,580,062
Accrued Taxes	19,593
Deferred Revenue	4,114,942
Other Current Liabilities	187,372
Total Current Liabilities	7,854,699
Other Liabilities	200
Deferred Revenue	17,185,523
Other Liabilities	559,475
Total Other Liabilities	17,744,997
Stockholders Equity	Letterita
Member's Equity	73,931,526
Total Liabilities and Equity	\$ 90,531,223

EHC Holding Company, LLC Profit and Loss For the Year Ending April 30, 2025 (Unaudited)

Year To Date
04/30/2025
841,436
5,847,269
1,040,418
7,729,123
1,387,611
1,387,611
6,341,512
420,374
1,706,681
3,038,743
3,396,212
187,775
871,009
60,832
9,681,625
601
16,832
(372,807)
(355,374)
\$ (3,695,487)

Guaranty of Performance				

VA/ARANTEE OF PERFORMANCE

For value tweaved, EBC Helding Company, LLC, a Delaware funited fishibly company (the Coursesor'), located at 630 Fifth Avenue, Suite 400, New York, NY 19111, absolutely and accordinatedly guarantees to assume the duties and obligations of ALL Functioning, LLC, located at 8100 E. Indian School Road, Suite 201, Scottadale, AZ 8523) (the "Function"), indice its functione registered as a suite where the frunchise is registered, and under its Funchise Agreement identified in its 2024. Function Document, as it may be unwinded, and as that Frunchise Agreement may be used as with franchise and amended, mod-fiel or extended from time to time. This guarantee continues until all such obligations of the Frunchise under its frunchise registrations and the Frunchise Agreement as satisfied or until the liability of Frunchises in its frunchises under the Frunchise Agreement has been completely discharged, whichever first occurs. The Ocumentor is not discharged from liability if o claim by a franchise against the Frunchises received in the first of default on the part of the Franchise. This guarantee is building on the Course of the C

The University of the guessing of Standale: All zons on the 2nd day of May, 2024.

Chapman

RRIC Holding Company LLC

10 mayor K

GUARANTEE OF PERFORMANCE

Tor value received, EHC Holding Company, LLC, a Delaware limited liability company (the "Guarantor"), located at 630 Fifth Avenue, Suite 400, New York, NY 10111, absolutely and unconditionally guarantees to assume the duties and obligations of ALL Franchising, LLC, located at 8100 E. Indian School Road, Suite 201, Scottsdale, AZ 85251 (the "Franchisor"), under its franchise registration in each state where the franchise is registered, and under its Franchise Agreement identified in its 2025 Franchise Disclosure Document, as it may be amended, and as that Franchise Agreement may be entered into with franchisecs and amended, modified or extended from time to time. This guarantee continues until all such obligations of the Franchisor under its franchise registrations and the Franchise Agreement are satisfied or until the liability of Franchisor to its franchisees under the Franchise Agreement has been completely discharged, whichever first occurs. The Guarantor is not discharged from liability if a claim by a franchise against the Franchisor remains outstanding. Notice of acceptance is waived. The Guaranter does not waive receipt of notice of default on the part of the Franchisor. This guarantee is binding on the Guarantor and its successors and assigns.

The Guarantor signs this guarantee at Scottsdale, Arizona on the 21st day of April, 2025.

Guarantor:

EHC Holding Company, LLC

By: Mayor R Songory R. Esgar, 1900

EXHIBIT "G"

TO DISCLOSURE DOCUMENT

OTHER AGREEMENTS

EXHIBIT G-1

FRANCHISEE DISCLOSURE QUESTIONNAIRE

[See Attached]

MAY NOT BE SIGNED OR USED IF FRANCHISEE RESIDES WITHIN, OR THE FRANCHISED BUSINESS WILL BE LOCATED WITHIN, A FRANCHISE REGISTRATION STATE³

FRANCHISEE DISCLOSURE QUESTIONNAIRE

As you know ALL Franchising, LLC ("we" or "us), and you are preparing to enter into a Franchise Agreement for the operation of an Assisted Living Locators franchise. We require that you complete this Questionnaire (a) so that we can determine whether our franchise sales team followed proper sales procedures and (b) to provide us with reasonable assurance that, prior to signing the Franchise Agreement, you have had an adequate opportunity to review the Franchise Disclosure Document and its attachments, consult with legal and/or business advisors of your choosing, and ask us questions about any disclosures or terms that you do not understand. You cannot sign or date this Questionnaire the same day as the Receipt for the Franchise Disclosure Document but you must sign and date it the same day you sign the Franchise Agreement and pay your franchise fee. Please review each of the following questions carefully and provide honest responses to each question.

Yes	No	l.	Have you received from us and personally reviewed the Franchise Agreement and, if applicable, Area Development Agreement ("ADA"), together with all attachments to those agreements?
			[If you answer "no," please explain in Explanation Section]
Yes	No	2.	Have you received from us and personally reviewed a Franchise Disclosure Document ("FDD")?
			[If you answer "no," please explain in Explanation Section]
Yes	No	3.	Did you sign a receipt for the FDD indicating the date you received it?
Yes	No	4.	Do you understand all the information contained in the FDD, Franchise Agreement and ADA (if applicable)?
			[If you answer "no," please identify any information you don't understand in Explanation Section]
Yes	No	5.	Did you receive the FDD at least 14 calendar days before signing any agreement relating to the franchise (other than an NDA) or paying any money?
Yes	No	6.	Did you receive a complete execution copy of the Franchise Agreement at least seven (7) calendar days before you signed it?
Yes	No	7.	Have you reviewed the FDD, Franchise Agreement and ADA (if applicable) with a lawyer, accountant or other professional advisor?
Yes	No	8.	Have you discussed the benefits and risks of developing and operating an Assisted Living Locators franchise with an existing Assisted Living Locators franchisee?
Yes	No	9.	Do you understand the risks of developing and operating an Assisted Living Locators franchise?
Yes	No	10.	Do you understand the success or failure of your franchise will depend in large part upon your skills, abilities and efforts and those of the persons you employ as well as many factors beyond your control such as competition, interest rates, the economy, inflation, labor and supply costs and other relevant factors?
Yes	No	11.	Do you understand all disputes or claims you may have arising out of or relating to the Franchise Agreement or ADA (if applicable) must be arbitrated in Arizona if not resolved informally or by mediation?

Registration states include California, Hawaii, Illinois, Indiana, Maryland, Michigan, Minnesota, New York, North Dakota, Rhode Island, South Dakota, Virginia, Washington and Wisconsin.

Yes	No	12.	Do you understand that the Franchise Agreement, ADA (if applicable) and the attachments those agreements contain the entire agreement between us and you concerning the Assiste Living Locators franchise, meaning any prior oral or written statements not set out in the Franchise Agreement, ADA or the attachments will not be binding?			
Yes	No	13.	Did any of our employees or representatives, or any person speaking on our behalf, make an statement or promise regarding the costs involved in operating an Assisted Living Locator franchise that is not contained in the FDD or that is contrary to, or different from, the informatio contained in the FDD?			
			[If you answer "yes," please describe the statement or promise in Explanation Section]			
Yes	No	14.	Did any of our employees or representatives, or any person speaking on our behalf, make an statement or promise regarding the training, assistance or support that will be provided to you that is not contained in the FDD or that is contrary to, or different from, the information in the FDD?			
			[If you answer "yes," please describe the statement or promise in Explanation Section]			
Yes	No	15.	Did any of our employees or representatives, or any person speaking on our behalf, make a statement or promise regarding the actual, average, projected or hypothetical profits or earning the likelihood of success, the amount of money you may earn, or the total amount of revenue Assisted Living Locators business may generate, other than any information included in Item of the FDD?			
			[If you answer "yes," please describe the statement or promise in Explanation Section]			
SIGNING	THIS QUI	ESTION	TYOUR ANSWERS ARE IMPORTANT TO US AND THAT WE WILL RELY ON THEM. BY NAIRE, YOU ARE REPRESENTING THAT YOU HAVE CONSIDERED EACH QUESTION NDED TRUTHFULLY TO THE ABOVE QUESTIONS.			
he franch n the indi on behalf	ise relations ucement, or of the francl	ship shal (ii) disc hisor. T AIRE M	or acknowledgement signed or agreed to by a franchisee in connection with the commencement of a law the effect of (i) waiving any claims under any applicable state franchise law, including frauthaming reliance on any statement made by any franchisor, franchise seller, or other person acting this provision supersedes any other term of any document executed in connection with the franchise made NOT BE SIGNED OR USED IF THE FRANCHISEE RESIDES WITHIN, OR IF THE WILL BE LOCATED WITHIN THE STATE OF MARYLAND.			
Signature	of Franchise	e Applic	ant Signature of Franchise Applicant			
Name (ple	ease print)		Name (please print)			
Dated			Dated			
Signature	of Franchise	e <u>Applic</u>	ant Signature of Franchise Applicant			
Name (ple	ease print)		Name (please <u>print</u>)			
Dated			Dated			
EXPLAN.	ATION SEC	CTION				
Please inc	lude any ex	planatio	ns below and refer to the applicable question number.			

EXHIBIT G-2

GENERAL RELEASE

[See Attached]

WAIVER AND RELEASE OF CLAIMS

This Waiver and Release of Claims (the	his "Agreement") is made as	of, 202 (the " <u>Ef</u>	fective
<u>Date</u> ") by	, a(n)	_ ("you") and each individual hol	lding a
direct or indirect ownership interest i	n you (collectively "Owner")	in favor of ALL Franchising, I	LLC, a
Delaware limited liability company ("us	," and together with you and O	wner, the "Parties").	

Background

- A. We signed a Franchise Agreement with you, dated ______, 202____ (the "Franchise Agreement") pursuant to which we granted you the right to own and operate an Assisted Living Locators business;
- B. You have notified us of your desire to transfer the Franchise Agreement and all rights related thereto, or an ownership interest in the franchisee entity, to a transferee, [enter into a successor franchise agreement] and we have consented to such transfer [agreed to enter into a successor franchise agreement]; and
- C. As a condition to our consent to the transfer [your ability to enter into a successor franchise agreement], you and Owner have agreed to execute this Agreement upon the terms and conditions stated below.
- D. In consideration of our consent to the transfer [our entering into a successor franchise agreement], and for other good and valuable consideration, the sufficiency and receipt of which are hereby acknowledged, and intending to be legally bound, you and Owner hereby agree as set forth below.

Agreement

- 1. Release. Owner, you, and each of your officers, directors, shareholders, members, owners, employees, agents, representatives, affiliates, parents, divisions, successors and assigns, and all persons or firms claiming by, through, under, or on behalf of any or all of them (the "Franchisee Parties"), hereby release, acquit and forever discharge us, any and all of our past and present affiliates, parents, subsidiaries and related companies, divisions and partnerships, consultants, advisors and franchise sellers and its and their respective past and present officers, directors, shareholders, members, owners, employees, agents, representatives, affiliates, parents, divisions, successors and assigns, and the spouses of such individuals (collectively, the "Franchisor Parties"), from any and all claims, liabilities, damages, expenses, actions or causes of action which any of the Franchisee Parties may now have or has ever had, whether known or unknown, past or present, absolute or contingent, suspected or unsuspected, of any nature whatsoever, directly or indirectly arising out of or relating to the execution and performance (or lack thereof) of the Franchise Agreement or the offer, sale or acceptance of the franchise related thereto (including, but not limited to any disclosures and representations made in connection therewith). The foregoing release shall not be construed to apply with respect to any obligations contained within this Agreement.
- 2. <u>California Law</u>. You and Owner hereby express your intention to release all existing claims, whether known or unknown, against the Franchisor Parties. Accordingly, you and Owner hereby waive Section 1542 of the California Civil Code, which provides the following:
 - "A general release does not extend to claims which the creditor does not know or suspect to exist in his or her favor at the time of executing the release, which if known by him or her must have materially affected his or her settlement with the debtor."

[Section 2 only applies for California franchisees; otherwise it is omitted]

3. <u>Washington Franchise Law</u>. The General Release does not apply with respect to claims arising under the Washington Franchise Investment Protection Act, RCW 19.100, or the rules adopted thereunder.

[Section 3 only applies for Washington franchisees; otherwise it is omitted]

4. Non-disparagement. Each of the Franchisee Parties expressly covenant and agree not to make any false

representation of facts, or to defame, disparage, discredit or deprecate any of the Franchisor Parties or otherwise communicate with any person or entity in a manner intending to damage any of the Franchisor Parties, the business conducted by any of the Franchisor Parties or the reputation of any of the Franchisor Parties. For purposes of clarity, the obligations in this Section apply to all methods of communications, including the making of statements or representations through direct verbal or written communication as well as the making of statements or representations on the Internet, through social media sites or through any other verbal, digital or electronic method of communication. The obligations in this Section also prohibit the Franchisee Parties from indirectly violating this Section by influencing or encouraging third parties to engage in activities that would constitute a violation of this Section if conducted directly by a Franchisee Party.

5. Representations and Warranties. You and Owner each represent and warrant that: (a) [Insert franchisee entity name] is duly authorized to execute this Agreement and perform its obligations hereunder; (b) neither you nor Owner has assigned, transferred or conveyed, either voluntarily or by operation of law, any of their rights or claims against any of the Franchisor Parties or any of the rights, claims or obligations being terminated or released hereunder; (c) you and Owner have not and shall not (i) institute or cause to be instituted against any of the Franchisor Parties any legal proceeding of any kind, including the filing of any claim or complaint with any state or federal court or regulatory agency, alleging any violation of common law, statute, regulation or public policy premised upon any legal theory or claim whatsoever relating to the matters released in this Agreement or (ii) make any verbal, written or other communication that could reasonably be expected to damage or adversely impact any Franchisor Party's reputation or goodwill; and (d) the individuals identified as Owners on the signature pages hereto together hold 100% of the legal and beneficial ownership interests in [Insert franchisee entity name].

6. Miscellaneous.

- (a) The Parties agree that each has read and fully understands this Agreement and that the opportunity has been afforded to each Party to discuss the terms and contents of said Agreement with legal counsel and/or that such a discussion with legal counsel has occurred.
- (b) This Agreement shall be construed and governed by the laws of the State of Arizona.
- (c) In the event that it shall be necessary for any Party to institute legal action to enforce, or for the breach of, any of the terms and conditions or provisions of this Agreement, the prevailing Party in such action shall be entitled to recover all of its reasonable costs and attorneys' fees.
- (d) All of the provisions of this Agreement shall be binding upon and inure to the benefit of the Parties and their respective current and future directors, officers, partners, attorneys, agents, employees, shareholders and the spouses of such individuals, successors, affiliates, and assigns.
- (e) This Agreement contains the entire agreement and understanding between the Parties with respect to the subject matter hereof and supersedes and is in lieu of all prior and contemporaneous agreements, understandings, inducements and conditions, expressed or implied, oral or written, of any nature whatsoever with respect to the subject matter hereof. This Agreement may not be modified except in a writing signed by each of the Parties.
- (f) If one or more of the provisions of this Agreement shall for any reason be held invalid, illegal or unenforceable in any respect, such invalidity, illegality or unenforceability shall not affect or impair any other provision of this Agreement, but this Agreement shall be construed as if such invalid, illegal or unenforceable provision had not been contained herein.
- (g) The Parties agree to do such further acts and things and to execute and deliver such additional agreements and instruments as any Party may reasonably require to consummate, evidence, or confirm the transactions contemplated hereby.
- (h) This Agreement may be executed in counterparts, each of which shall be deemed an original, but all of which together shall constitute but one document.

(i) THIS RELEASE DOES NOT APPLY WITH RESPECT TO CLAIMS ARISING UNDER THE WASHINGTON FRANCHISE INVESTMENT PROTECTION ACT, RCW 19.100, AND RULES ADOPTED THEREUNDER.

In witness whereof, the Parties have executed this Agreement as of the date first written above.

FRANCHISEE	<u>:</u>	
		
By:		
Name:		
Its:		
FRANCHISE (OWNERS:	
Name:		 -
Name:		 -
		 _
Name:		-

EXHIBIT G-3

ALL-IN USER AGREEMENT AND PRIVACY POLICY

[See Attached]

Acceptable Use Policy

Date: 3/22/2018

1 Overview

The purpose of this policy is to establish acceptable and unacceptable use of electronic devices, network resources and information platforms at Assisted Living Locators ("ALL") and by their franchisees in conjunction with its established culture of ethical and lawful behavior, openness, trust, and integrity.

ALL provides networks, and other electronic information systems to meet missions, goals, and initiatives and must manage them responsibly to maintain the confidentiality, integrity, and availability of its information assets. This policy requires the users of these information assets to comply with company policies and protects the company against damaging legal issues.

2 Scope

All franchisees, as well as their employees, contractors, consultants, temporary and other workers ("You"), along with employees, contractors, consultants, temporary and other works at ALL, including all personnel affiliated with third parties must adhere to this policy. This policy applies to information assets owned or leased by ALL. You are responsible for the security of your own devices, and those associated with your franchise(s) used in conducting your business, however ALL has the right to block your use of company assets with these devices in the event that it determines, at it's sole discretion, that your devices are creating harm to company assets.

There shall be no exceptions to this policy.

General Requirements

- 2.1.1 You are responsible for exercising good judgment regarding appropriate use of ALL resources in accordance with ALL policies, standards, and guidelines. ALL resources may not be used for any unlawful or prohibited purpose.
- 2.1.2 For security, compliance, and maintenance purposes, authorized personnel may monitor and audit equipment, systems, and network traffic per Audit Procedures developed by ALL and updated from time to time. Devices that interfere with other devices or users on the ALL network, or that harm ALL's data or the asset itself in any way may be disconnected. You are prohibited from taking measures to actively block our ability to monitor and audit your use of ALL information assets.

System Accounts

2.1.3 You are responsible for the security of data, accounts, and systems under your control. Keep

- passwords secure and do not share account or password information with anyone, including other personnel, family, or friends. Providing access to another individual, either deliberately or through failure to secure its access, is a violation of this policy.
- 2.1.4 You must maintain system-level and user-level passwords in accordance with our established Password Policies associated with our various information assets.
- 2.1.5 You must ensure through legal or technical means that proprietary information remains within the control of Assisted Living Locators and its franchisees at all times. Conducting ALL business that results in the storage of proprietary information on personal or non-ALL controlled environments, including devices maintained by you or a third party with whom ALL does not have a contractual agreement, is prohibited. This specifically prohibits the use of an e-mail account that is not provided by ALL, or its customer and partners, for company business.

Network Use

You are responsible for the security and appropriate use of ALL network resources under your control. Using ALL resources for the following is strictly prohibited:

- 2.1.6 Causing a security breach to either ALL or other network resources, including, but not limited to, accessing data, servers, or accounts to which you are not authorized; circumventing user authentication on any device; or sniffing network traffic.
- 2.1.7 Causing a disruption of service to either ALL or other network resources, including, but not limited to, ICMP floods, packet spoofing, denial of service, heap or buffer overflows, and forged routing information for malicious purposes.
- 2.1.8 Introducing honeypots, honeynets, or similar technology on the ALL network.
- 2.1.9 Violating copyright law, including, but not limited to, illegally duplicating or transmitting copyrighted pictures, music, video, and software.
- 2.1.10 Exporting or importing software, technical information, encryption software, or technology in violation of international or regional export control laws.
- 2.1.11 Use of the Internet or ALL's network that violates the ALL policies or local laws.
- 2.1.12 Intentionally introducing malicious code, including, but not limited to, viruses, worms, Trojan horses, e-mail bombs, spyware, adware, and keyloggers.
- 2.1.13 Port scanning or security scanning on a production network unless authorized in advance by Information Security.

Electronic Communications

The following are strictly prohibited:

- 2.1.14 Inappropriate use of communication vehicles and equipment, including, but not limited to, supporting illegal activities, and procuring or transmitting material that violates ALL policies against harassment or the safeguarding of confidential or proprietary information.
- 2.1.15 Sending Spam via e-mail, text messages, pages, instant messages, voice mail, or other forms of electronic communication.

- 2.1.16 Forging, misrepresenting, obscuring, suppressing, or replacing a user identity on any electronic communication to mislead the recipient about the sender.
- 2.1.17 Posting the same or similar non-business-related messages to large numbers of Usenet newsgroups (newsgroup spam).
- 2.1.18 Use of an ALL e-mail or IP address to engage in conduct that violates ALL policies or guidelines. Posting to a public newsgroup, bulletin board, or listserv with an ALL e-mail or IP address represents ALL to the public; therefore, you must exercise good judgment to avoid misrepresenting or exceeding your authority in representing the opinion of the company.

3 Enforcement

A franchisee shall be accountable for the performance of this policy by all their employees, contractors, consultants, temporary and other workers. A franchisee found to have violated this policy may be subject to disciplinary action, up to and including termination of their franchise agreement. An ALL employee found to have violated this policy may be subject to disciplinary action, up to and including termination of employment. A violation of this policy by an ALL temporary worker, contractor or vendor may result in the termination of their contract or assignment with ALL.

4 Definitions

Term	Definition
honeypot, honeynet	Network decoys that serve to distract attackers from valuable machines on a network. The decoys provide an early warning for intrusion detection and detailed information on vulnerabilities.
Spam	Electronic junk mail or junk newsgroup postings. Messages that are unsolicited, unwanted, and irrelevant.

5 Revision History

Date of Change	Responsible	Summary of Change
3/22/18		Policy created

Privacy Policy

Date: 3/22/2018

Assisted Living Locators ("ALL") is dedicated to protecting the privacy rights of our users ("you") and the clients, partners and properties of our users. This Privacy Policy (this "Policy") describes the ways we collect, store, use, and manage the information, including personal information, and data that you provide or we collect in connection with our service on the ALL-INTM platform ("Service"). Please note that the scope of this Policy is limited to information and data collected or received by ALL through your use of the Service. ALL is not responsible for the actions of third party people or companies not associated with the Service, the content of their sites, the use of information or data you provide to them, or any products or services they may offer. Any link to those sites does not constitute our sponsorship of, or affiliation with, those people or companies for privacy policy purposes.

HIPAA privacy rules apply to All-INTM and we take security and information privacy matters very seriously. We engage in ongoing efforts to ensure that ALL-INTM complies with all applicable laws and regulations, including HIPAA. If you become aware of a data breach, circumstances or conditions that could leave to a data breach, or have concerns about ALL-INTM security or compliance, please contact us immediately.

By using the Service, you are expressing your agreement to this Policy and the processing of your information and data, including your personal information, in the manner provided in this Policy. If you do not agree to these terms, please do not use the Service.

If you have any questions or concerns, please contact the ALL Corporate Office, c/o <u>Angela Angelica</u> Olea.

Information and Data We Collect from You

We may collect the following information and data from you:

- Contact information (such as name and email address)
- Username and password
- Profile information that you volunteer
- Automatically collected data (such as IP Address, your device's operating system, browser type and language)
- Mobile device identifiers (such as your unique device ID (persistent/non-persistent, hardware type, medial access control ("MAC") address, international mobile equipment identity ("IMEI"), and your device name
- Client Information for the purpose of facilitating a match to a property
- Property Information, including pictures and contracts for the purpose of facilitating client matches
- Referral Partner, Marketing Resource, and Contact Information for the purpose of identifying sources of business and to maintain your franchise operation

Tracking Technologies

We use technologies such as cookies, beacons, scripts, and tags to identify a user's computer/device and to "remember" things about your visit, such as your preferences or a username and password. Information contained in a cookie may be linked to your personal information, such as your user ID, for purposes such as improving the quality of our Services or making the Service easier to use. You can disable cookies at any time, although you may not be able to access or use features of the Service.

How We Use Your Information

ALL's primary goals in using information are to create your account, provide Services to you, improve our Service or contact you. We store information on servers located in the United States via IBM Softlayer and/or Amazon Web Services.

We use information collected through our Service for purposes described in this Policy or disclosed to you in connection with our Service.

Website Services

While you are browsing the Service, your computer's operating system, Internet Protocol (IP) address, access times, browser type and language and referring Web site addresses may be logged automatically. We may use this information to monitor, develop and analyze your use of the Service.

We may ask you to submit and we may process data that is personal to you, including but not limited to your name, profile photo, links to your profiles on social networking websites and other third-party sites, user names and e-mail and mail addresses. We may use this information to create a user profile and provide you Services.

We may use your email address to contact you about your experience with the Sites and notify you about ALL news and activities.

Customer Service

We may use your email address when you contact our customer service group to contact you about your experience with our Service and to notify you about news and activities.

Third Party Services

Our Services may contain third party tracking and data collection and analytics tools from our service providers for the purpose of determining actions that could improve our service. Such third parties may use cookies, APIs, and SDKs in our Service to enable them to collect and analyze user and device related data and information on our behalf. The third parties may have access to and collect data and information about you, such as your device identifier, MAC address, IMEI, locale (specific location where a given language is spoken), geo-location information, IP address, application usage, access and session times, and applications present on the device or in use at a particular time on the device

ALL does not share your personal information or the information on your clients, properties or contacts, referral partners and marketing resources outside of the ALL franchise system and only to facilitate placements in other franchisee territories.

- We may release your information as permitted by law, such as to comply with a subpoena, or when we believe that release is appropriate to comply with the law; investigate fraud, respond to a government request, enforce or apply our rights; or protect the rights, property, or safety of us or our users, or others. This includes exchanging information with other companies and organizations for fraud protection. In certain situations, ALL may be required to disclose personal information in response to lawful requests by public authorities or for local law enforcement requirements.
- ALL may share your information in connection with any merger, sale of our assets, or a financing or acquisition of all or a portion of our business to another company.

• We may share aggregate or anonymous information about you with potential franchisees, business partners, sponsors, and other third parties.

Security Safeguards

ALL takes reasonable measures to protect your information from unauthorized access or against loss, misuse or alteration by third parties.

Although we make good faith efforts to store the information collected on the Service in a secure operating environment that is not available to the public, we cannot guarantee the absolute security of that information during its transmission or its storage on our systems. Further, while we attempt to ensure the integrity and security of our network and systems, we cannot guarantee that our security measures will prevent third-party "hackers" from illegally obtaining access to this information. We do not warrant or represent that your information will be protected against, loss, misuse, or alteration by third parties. No method of transmission over the Internet, or method of electronic storage, is 100% secure, however. Therefore, we cannot guarantee its absolute security.

Data Retention

We retain your information for as long as your account is active or as needed to provide you Services. We will also retain and use your information as necessary to comply with our legal obligations, resolve disputes, and enforce our agreements.

International Transfer

We may transfer information that we collect about you to affiliated entities, or to other third parties across borders and from your country or jurisdiction to other countries or jurisdictions around the world. Please note that these countries and jurisdictions may not have the same data protection laws as your own jurisdiction, and we take steps to ensure adequate safeguards are in place to enable transfer of information to the U.S. and the use and disclosure of information about you, including personal information, as described in this Policy.

Revision History

Date of Change	Responsible	Summary of Change
3/22/18		Policy created

EXHIBIT G-4

HIPAA BUSINESS ASSOCIATE AGREEMENT

[See Attached]

HIPAA BUSINESS ASSOCIATE AGREEMENT (45 CFR §164.504(e))

This Business Associate Agreement (this "Agreement") is entered into as of	_, 202,
between ALL Franchising, LLC, a Delaware limited liability company with a principal place of	business
located at 8100 E. Indian School Road, Suite 201, Scottsdale, Arizona 85251 ("Business Associa	ate") and
, a(n) with a principal place of business lo	cated at
("Covered Entity").	

I. RECITALS

- A. Business Associate performs, or assists in the performance of, a function or activity or provides services of a type for Covered Entity that makes Business Associate a "business associate" for purposes of the HIPAA Privacy Regulations. {45 CFR §160.103}
- B. Covered Entity will disclose Protected Health Information to Business Associate in conjunction with the function, activity or services performed or provided by Business Associate. {45 CFR §160.103}
- C. Covered Entity will disclose electronic health information to Business Associate in conjunction with the function, activity or services performed or provided by Business Associate.
- D. Covered Entity and Business Associate desire to enter into this Agreement as required by the HIPAA Privacy and Security Regulations to provide satisfactory assurance to Covered Entity that Business Associate will appropriately safeguard that Protected Health Information. {45 CFR §164.502(e)(l)}

II. AGREEMENT

NOW THEREFORE, Covered Entity and Business Associate agree as follows:

- 1. **Definitions.** The following terms are defined as set forth below. Any terms used but not otherwise defined in this Agreement have the definitions set forth in the Regulations and the Health Information Technology for Economic and Clinical Health Act ("HITECH"), found in Title XIII of the American Recovery and Reinvestment Act of 2009, Public Law 111-005, and any regulations promulgated thereunder.
 - a. "Breach" shall have the meaning set forth in 45 C.F.R. § 164.402.
 - b. "Electronic Protected Health Information" shall mean individually identifiable health information that is transmitted by or maintained by electronic media. It includes devices in computers and any removable/transportable digital memory medium. Transmission media include the internet, xtranet or intranet, leased lines, dial-up lines, private networks, and physical movement of removable/transportable media. Certain transmissions, including of paper, via facsimile, and of voice, via telephone, are not considered to be transmissions via electronic media if the information being exchanged did not exist in electronic form immediately before the transmission. {45 CFR §160.103}
 - c. "HIPAA Privacy Regulations" shall mean the regulations at 45 CFR §160 and §164, subparts A and E.
 - d. "HIPAA Security Regulations" shall mean the regulations at 45 CFR \$160 and \$164, subpart C.
 - e. "HIPAA Breach Notification Rule" shall mean the regulations at 45 CFR §164, subpart D.

- f. "HIPAA Rules" shall mean the HIPAA Privacy Regulations, the HIPAA Security Regulations, the HIPAA Breach Notification Rule and the HIPAA enforcement rule at 45 CFR §160, subpart C.
- g. "Individual" shall have the same meaning as the term "individual" in 45 C.F.R. § 160.103 and shall include a person who qualifies as a personal representative in accordance with 45 C.F.R. § 164.502(g).
- h. "Protected Health Information" or "PHI" means, subject to the definition provided at 45 C.F.R. § 160.103, individually identifiable health information that Business Associate receives from Covered Entity or creates, receives, transmits or maintains on behalf of Covered Entity for purposes of performing the services under the Engagement. Unless otherwise stated in this Agreement, any provision, restriction or obligation in this Agreement related to the use of PHI shall apply equally to EPHI.
- *i.* Regulations" shall mean collectively the HIPAA Security Regulations, the HIPAA Privacy Regulations, the HIPAA Breach Notification Rule and the HIPAA enforcement rule at 45 CFR §160, subpart C.
- j. "Required by Law" shall have the same meaning as the term "required by law" in 45 C.F.R. § 164.103.
- k. "Secretary" shall mean the Secretary of the Department of Health and Human Services or their designee.
- "Security Incident" means the attempted or successful unauthorized access, use, disclosure, modification or destruction of information or interference with the system operations in an information system. {45 CFR §164.304}
- m. "Subcontractor" means a person to whom a business associate delegates a function, activity or service, other than in the capacity of a member of the workforce of such business associate.
- n. "Unsecured Protected Health Information" means Protected Health Information that is not rendered unusable, unreadable, or indecipherable to unauthorized individuals through the use of a technology or methodology specified by the Secretary. {45 CFR 164.402}

Business Associate acknowledges and agrees that all PHI that is created or received by Covered Entity and disclosed or made available in any form by Covered Entity to Business Associate, or is created, received, maintained or transmitted by Business Associate on Covered Entity's behalf, will be subject to this Agreement. This Agreement will commence upon the Effective Date and will continue as long as Business Associate has use, custody or access to PHI subject to this Agreement, and thereafter for the period required by the Regulators.

- 2. **Restriction on Use and Disclosure of Protected Health Information.** Except as permitted or required by this Agreement or as required by law, Business Associate shall not use, de-identify, or further disclose any Protected Health Information disclosed or otherwise made available to it by Covered Entity. {45 CFR §164.504(e)(2)(i) and (e)(2)(ii)(A)}
- 3. **Authorized Uses and Disclosures**. Except as otherwise limited in this Agreement, Business Associate is hereby authorized to use and disclose Protected Health Information for the following purposes:
 - a. Generally. Business Associate may use or disclose Protected Health Information on behalf of, or to provide services to, Covered Entity for the following purposes, if such use or disclosure of Protected Health Information would not violate the HIPAA Privacy

- Regulations if done by Covered Entity or the minimum necessary policies and procedures of Covered Entity: Provide ongoing assistance with the operations of the Covered Entity's franchised business. {45 CFR §164.504(e)(2)}
- b. Management and Administration. Business Associate may use and disclose Protected Health Information for the proper management and administration of Business Associate or to carry out the legal responsibilities of Business Associate, provided:
 - i. The disclosure is required by law; or
 - ii. Business Associate obtains reasonable assurances from the person to whom the information is disclosed that it will be held confidentially and used or further disclosed only as required by law or for the purpose for which it was disclosed to the person and the person will immediately notify the Business Associate of any instances of which it is aware in which the confidentiality of the information has been breached. {45 CFR§164.504(e)(2)(i)(A) and 45 CFR §164.504(e)(4)}
- c. Violations of Law. Business Associate may use Protected Health Information to report violations of law to appropriate Federal and State authorities, consistent with 20 CFR §164.502(j)(l).

4. Business Associate's Obligations.

- a. Representation and Acknowledgment. Business Associate represents that is has complied and will comply with the requirements of the Regulations applicable to it and acknowledges that it is aware that it is subject to the civil and criminal penalties of section 1176 and 1177 of the Social Security Act.
- b. Safeguards. Business Associate shall use appropriate safeguards, and comply, where applicable, with the HIPAA Security Regulations with respect to electronic Protected Health Information, to prevent use or disclosure of Protected Health Information other than as permitted or required by this Agreement or as required by law. {45 CFR §164.504(e)(2)(ii)(B)}
- c. Security of Electronic Protected Health Information. Business Associate shall implement administrative, physical and technical safeguards that reasonably and appropriately protect the confidentiality, integrity and availability of the electronic Protected Health Information that it creates, receives, maintains or transmits on behalf of Covered Entity. {45 CFR §164.314(a)(2)(i)(A)}
- d. Reporting. Business Associate shall report to Covered Entity any use or disclosure of Protected Health Information not permitted by this Agreement of which it becomes aware, including Breaches of Unsecured Protected Health Information as required by the HIPAA Breach Notification Rule.
- e. Subcontractors. Business Associate shall ensure that any subcontractors that create or receive Protected Health Information on behalf of Business Associate agree to the same restrictions and conditions that apply to Business Associate with respect to such information. {45 CFR §164.314(a)(2)(i)(B)}
- f. Providing Electronic Protected Health Information to Agents or Subcontractors. BUSINESS ASSOCIATE shall ensure that any agent, including a subcontractor, to whom it provides electronic Protected Health Information, agrees to implement reasonable and appropriate safeguards to protect the electronic Protected Health Information. {45 CFR §164.314(a)(2)(i)(B)}

- g. Individual's Access to Information. Business Associate shall make available and permit access to Protected Health Information about an individual by that individual in accordance with 45 CFR §164.524. {45 CFR §164.504(e)(2)(ii)(E)}
- h. Amendment of Protected Health Information. Business Associate shall make available to Covered Entity Protected Health Information for amendment and incorporate any amendments to Protected Health Information in accordance with 45 CFR §164.526. {45 CFR §164.504(e)(2)(ii)(P)}
- i. Accounting of Disclosures. Business Associate shall document such disclosures of Protected Health Information and information related to such disclosures as would be required for Covered Entity to respond to a request by an individual for an accounting of disclosures of Protected Health Information in accordance with 42 CFR §164.528.

Business Associate shall make available the information required to provide an accounting of disclosures in accordance with 42 CFR §164.528. Such information shall be given to Covered Entity by Business Associate within 20 days after Covered Entity notifies Business Associate of Covered Entity need for the information. {45 CFR §164.504(e)(2)(ii)(G)}

5. Covered Entity's Obligations.

- a. Provisions for Covered Entity to Inform Business Associate of Privacy Practices and Restrictions. Covered Entity shall notify Business Associate of any limitation(s) in its Notice of Privacy Practices of Covered Entity in accordance with 45 CFR §164.520, to the extent that such limitation may affect Business Associate's use or disclosure of Protected Health Information.
- b. Permissible Requests by Covered Entity. Covered Entity shall not request Business Associate to use or disclose protected health information in any manner that would not be permissible under the HIPAA privacy regulations if done by Covered Entity.

6. **Breach Notification**.

- a. Notice to Covered Entity. Business Associate will report to Covered Entity any suspected Breach of Unsecured PHI by Business Associate or any of its officers, directors, employees, Subcontractors or agents. All notifications required under this Section will be made by Business Associate without unreasonable delay and in no event later than ten (10) days of discovery. Business Associate will use the standard at 45 C.F.R. § 164.410(a) to determine when the suspected Breach is treated as discovered. Covered Entity shall have discretion to determine whether a suspected Breach has given rise to a Breach. Business Associate will cooperate with Covered Entity and provide such information as Covered Entity reasonably requires in making this determination. In notifying Covered Entity of a suspected Breach, Business Associate will provide, to the extent reasonably possible, as much of the information it has that would be required in notifying a Covered Entity of a Breach, under 45 C.F.R. § 164.410. If Covered Entity determines that a Breach has occurred, Business Associate will provide any other available information that Covered Entity is required to include in its notification to individuals pursuant to 45 C.F.R.§164.404(c).
- b. Notice to Individuals and Media. In the event Covered Entity determines a Breach has occurred that was caused by the acts or omissions of Business Associate, its Subcontractors, officers, directors, employees or agents, Business Associate will cooperate with Covered Entity to notify, (i) individuals whose Unsecured PHI has been, or is reasonably believed by Covered Entity to have been, accessed, acquired, used or disclosed, and (ii) the media, as required pursuant to 45 C.F.R. § 164.406, if the legal requirements for media notification are triggered by the circumstances of such Breach. Business Associate will cooperate in

Covered Entity's Breach analysis process and procedures, if requested. Covered Entity will at all times have the final decision about the content of any notification required to be given under the Regulations.

c. Proof of Encryption. In the event of a Breach of Secured Protected Health Information, Business Associate shall notify Covered Entity of the Breach as stated in subparagraph (6)(a), above, and within 20 calendar days after giving such notice to Covered Entity, provide proof satisfactory to Covered Entity that such Protected Health Information was not Unsecured Protected Health Information. {42 CFR §164.402}

7. **Term and Termination**.

- a. Generally. This Agreement shall be effective when executed on behalf of both of the parties hereto and shall terminate when all of the Protected Health Information provided by Covered Entity to Business Associate, or created or received by Business Associate on behalf of Covered Entity, is destroyed or returned to Covered Entity, or, if it is not feasible to return or destroy Protected Health Information, protections are extended to such information, in accordance with the termination provisions in this Paragraph (7).
- b. Mutual Agreement. This Agreement may be terminated by mutual written agreement of the parties.
- c. Termination for Cause. Upon Covered Entity's knowledge of a material breach of this Agreement by Business Associate. Covered Entity shall either
 - i. Provide an opportunity for Business Associate to cure the breach or end the violation and terminate the Agreement if Business Associate does not cure the breach or end the violation within the time specified by Covered Entity;
 - ii. Immediate terminate the Agreement if Business Associate has breached a material term of this Agreement and cure is not possible.

d. Effect of Termination.

- i. Except as provided in paragraph (2) below, upon termination of this Agreement for any reason, Business Associate shall return or destroy all Protected Health Information received from Covered Entity, or created or received by Business Associate on behalf of Covered Entity that Business Associate maintains in any form. This provision also shall apply to Protected Health Information that is in the possession of subcontractors of Business Associate. Business Associate shall retain no copies of the Protected Health Information.
- ii. In the event that Business Associate determines that returning or destroying the protected information is not feasible, Business Associate shall provide to Covered Entity notification of the conditions that make return or destruction not feasible. Business Associate shall extend the protections of this Agreement to such Protected Health Information and limit further uses and disclosures of such Protected Health Information to those purposes that make the return or destruction not feasible, for so long as Business Associate maintains such Protected Health Information.
- 8. **Subpoena.** In the event Business Associate receives a subpoena for any Protected Health Information in Business Associate's possession, Business Associate shall immediately notify Covered Entity of the subpoena and deliver a copy of the subpoena to Covered Entity. Business Associate shall respond to the subpoena only in accordance with the HIPAA Privacy Regulations.
- 9. **Notices.** Any notices required or permitted to be given under this Agreement shall be in writing

and shall be personally delivered or sent by certified or registered mail, first class postage prepaid, return receipt requested, or by prepaid overnight delivery services such that proof of delivery will be obtained, and shall be addressed as set forth below or to such other address as may be specified in a prior written notice to the other party: To those addresses indicated in the Covered Entity's Franchise Agreement with Business Associate. Such notice shall be deemed to be given on the date it is deposited in the mail as stated above, on the date it is given to the overnight delivery service or the date it is given personally to the party to whom it is directed. A notice shall be deemed to have been given personally to a party if it is handed to the representative of the party to whom the notice must be addressed or if left at his or her office located at the street address to which a notice would be mailed.

- 10. **Amendment.** This Agreement may not be changed, modified or amended except by a written agreement executed on behalf of each of the parties.
- 11. **No Waiver.** No waiver of one or more of the provisions of this Agreement or the failure to enforce any provision of this Agreement by either party shall be construed as a waiver of any subsequent breach of this Agreement, nor a waiver of the right at any time thereafter to require strict compliance with all of its terms.
- 12. **Entire Agreement.** This Agreement sets forth the entire agreement and understanding between the parties as to the matters contained in it, and supersedes all prior discussions, agreements and understandings of every kind and nature between them.
- 13. **Headings.** The headings placed between the various paragraphs and subparagraphs of this Agreement are inserted for ease of reference only, do not constitute a part of this Agreement, and shall not be used in any way whatsoever in the construction or interpretation of this Agreement.
- 14. **Interpretation.** Any ambiguity in this Agreement shall be resolved to permit Covered Entity to comply with the HIPAA Privacy Rule, 45 CFR §164.500 et seq., the HIPAA Security Rule, 45 CFR §164.302 et seq., and the HIPAA Breach Notification Rule, 45 CFR §164.400 et seq., as each may be amended from time to time.
- 15. **Governing Law.** This Agreement shall be construed and enforce in accordance with, and governed by, the laws of the State of Arizona.

In witness whereof, the parties hereto have caused this Agreement to be executed by their duly authorized representatives on the date set forth below.

Covered Entity/Franchices

Dusiness Associate/Franchisor.	Covered Entity/Franchisee.
ALL Franchising, LLC, a Delaware limited liability company	a(n)
nuomity company	_
By:	By:
Name:	Name:
Its:	Its:

Puginaga Aggagiata/Franchigare

EXHIBIT G-5

POWER OF ATTORNEY – TELEPHONE AND INTERNET

[See Attached]

POWER OF ATTORNEY - TELEPHONE AND INTERNET

STATE OF	-
COUNTY OF	
KNOW ALL MEN BY THESE PRESENTS	
LLC, a Delaware limited liability company ("Franchisee and in Franchisee's name, place and execute, acknowledge, certify, deliver, accept, redocuments as, in the sole discretion of Franchisesigning to Franchisor all of Franchisee's right, the Franchisee's franchisee and all related Yellow Pagand other business listings including, but not list Service Agreement and any other transfer documents are company providing services to Franchises and perform all acts and things which, in the sole be done as fully to all intents and purposes	irrevocably constitutes and appoints ALL Franchising, anchisor"), true and lawful attorney-in-fact and agent for stead, to do or cause to be done all things and to sign, cord and file all agreements, certificates, instruments and isor, is necessary or advisable for the sole purpose of tle and interest in and to any and all telephone numbers of es, White Pages, internet listings, networking media sites mited to, the execution and delivery of any Transfer of tentation required by the applicable telephone or internet es, and grants to Franchisor full power and authority to do the discretion of Franchisor, are necessary or advisable to as Franchisee might or could itself do, ratifying and for cause to be done by virtue of this Power of Attorney
person to act as its attorney-in-fact and agent, no required to ascertain the authority of Franchiso responsible in any way for the proper application. Any person, firm or corporation dealing with F certificate of Franchisor that this Power of Attornance in full force and effect, and Franchisee corporation acting in reliance on a certificate or a	egardless of whether Franchisee has designated any other to person, firm or corporation dealing with Franchisor is r, nor to see to the performance of the agency, nor be on of funds or property paid or delivered to Franchisor. Franchisor is fully protected in acting and relying on a rney on the date of the certificate has not been revoked must not take any action against any person, firm or copy of this Power of Attorney. Any document executed to include a certificate on the part of Franchisor, whether y termination of this Power of Attorney.
Agreement dated	following the expiration or termination of the Franchise by and between Franchisor and Franchisee. The sy of any act or deed that Franchisor may have affected
	ed as an irrevocable power of attorney coupled with an of Arizona and the laws of the State of Arizona govern all ney and construction of its provisions.
In Witness Whereof, the undersigned, 202	has executed this Power of Attorney as of
FRANCHISOR:	FRANCHISEE:
ALL Franchising, LLC, a Delaware limited liability company	a(n),
By: Name:	By:
Its:	Its:

EXHIBIT G-6

SAMPLE CLIENT CONTRACT

[See Attached]



CLIENT AGREEMENT

Assisted Living Locators appreciates the opportunity to locate care options for your loved one. Our goal is to locate the best assisted living home, community or living situation to meet your needs. While we can advise you on your choices, the final decision will be yours to make.

We do not have any ownership or management relationship with any assisted living home or community. We have a referral fee arrangement with certain assisted living homes and communities under which we receive a fee for referring a client if the client moves into the assisted living home or community. Regardless of any referral fee arrangement, our referral and placement fees are free to you and your loved one. No portion of our fees are assessed to you or to our client and the fees charged by the community are the same to you no matter whether they pay us a referral fee or not.

Date of Disclosure: ___

Assisted Living Locators	Client Name:
Signature	Signature
Name:	Name:
Title:	Relationship to Prospective Resident
	□ Self
	☐ Family Member
	□ POA
	□Other
	Please Identify Relationship

Assisted Living Locators | (877) 275-5991 | assisted living locators, com



CLIENT AGREEMENT

Assisted Living Locators appreciates the opportunity to locate care options for your loved one. Our goal is to locate the best assisted living home, community or living situation to meet your needs. While we can advise you on your choices, the final decision will be yours to make.

We do not have any ownership or management relationship with any assisted living home or community. We have a referral fee arrangement with certain assisted living homes and communities under which we receive a fee for referring a client if the client moves into the assisted living home or community. Regardless of any referral fee arrangement, our referral and placement fees are free to you and your loved one. No portion of our fees are assessed to you or to our client and the fees charged by the community are the same to you no matter whether they pay us a referral fee or not.

Date of Disclosure:	-
Assisted Living Locators	Client Name:
Signature	Signature
Name:	Name:
Title:	Relationship to Prospective Resident:
	☐ Self
	☐ Family Member
	□POA
	□Other
	Please Identify Relationship

Assisted Living Locators | (877) 275-5991 | assisted living locators, com

EXHIBIT G-7

FRANCHISE RESALE AGREEMENT

[See Attached]

FRANCHISE RESALE AGREEMENT

This Franchise Resale	Agreement (the "Agreement") is enter	ered into as of [],
202320 (the "Effective I	Date") between ALL Franchising, LLC,	, a Delaware limited liability company as
successor-in-interest by	merger with CALLRN Franchise LLC	C, a Delaware limited liability company
(" <u>we</u> " or " <u>us</u> ") and [], a(n) [] (" <u>you</u> " and together with us, the
"Parties").		
	D	
	<u>BACKGROUND</u>	
A On [1 202[] the Parties entered in	to an Assisted Living Locators Franchise

- A. On ________, 202[__], the Parties entered into an Assisted Living Locators Franchise Agreement (the "<u>Franchise Agreement</u>"), pursuant to which we granted you the right and license to open and operate an Assisted Living Locators business within the Territory defined therein (your "<u>ALL Business</u>").
- B. You have notified us that you desire to sell your ALL Business and you have requested our assistance in effectuating the sale and transfer therefore.
- C. We have agreed to assist you in your efforts to sell your ALL Business subject to the terms and conditions set forth in this Agreement.

AGREEMENT

- 1. <u>Background Recitals</u>. The statements made in the Recitals above are true and accurate and are incorporated herein.
- 2. <u>Defined Terms</u>. Any capitalized term that is not defined herein shall have the meaning ascribed to it in the Franchise Agreement.
- 3. <u>Term.</u> The term of this Agreement (the "<u>Term</u>") begins on the Effective Date and expires on the expiration or earlier termination of the current term of the Franchise Agreement. If the Parties renew the term of the Franchise Agreement, then the Parties may (but need not) renew the Term of this Agreement upon mutually agreeable terms and conditions.
- 4. **Engagement**. You hereby engage us to provide franchise brokerage services during the Term on a non-exclusive basis. We hereby accept the engagement and agree to assist you in your efforts to: (a) locate one or more qualified buyers for your ALL Business; and (b) sell your ALL Business to a qualified buyer that we approve as meeting our minimum qualifications and eligibility requirements for a franchisee. You agree that we may offer (and publicly list) your ALL Business for sale at any price equal to or greater than \$[49,900].
- 5. <u>Sales Assistance</u>. During the Term, we agree to provide reasonable assistance in connection with your efforts to sell your ALL Business. We will utilize our current franchise recruiting system and our internal and external franchise sales network (our "<u>Franchise Recruitment Program</u>") in an effort to identify qualified candidates to purchase your ALL Business. We shall pre-qualify all leads to ensure they meet our minimum qualifications and eligibility requirements for franchisees. We do not represent or guaranty that our efforts to identify a qualified buyer willing to purchase your ALL Business will be successful. Under no circumstances will we have any liability to you based on the services we render pursuant to this Agreement.
- 6. Your Obligations. We will expend valuable time and resources attempting to market and sell your ALL Business. For this reason, you may not refuse any purchase offer that we present to you for a purchase price of at least \$[49,900]. However, you will have an opportunity to negotiate a higher price after we introduce the potential buyer to you. You agree to cooperate with us in good faith and provide all reasonable assistance and information we request in order to effectuate the purposes of this Agreement and facilitate the sale of your ALL Business. You agree to promptly provide all

potential buyers with full access to your books and records for due diligence purposes. Throughout the Term, you agree to remain in full compliance with all terms of the Franchise Agreement. You understand that all terms in the Franchise Agreement governing transfer of your ALL Business (including payment of the transfer fee) shall apply to any sale of your ALL Business notwithstanding the execution of this Agreement.

- 7. <u>Purchase Agreement</u>. We assume no responsibility for preparing or negotiating the asset or stock purchase agreement between you and the buyer. You and the buyer are solely responsible for preparing and negotiating the asset or stock purchase agreement pursuant to which you will transfer ownership of your ALL Business to the buyer.
- 8. Fees and Costs. If either (a) you sell your ALL Business to a buyer solicited through our Franchise Recruitment Program or (b) we materially assist you in your efforts to sell your ALL Business to a buyer solicited by you or an unaffiliated third party, then you agree (as applicable) to: (a) reimburse us for all referral fees, commissions and other compensation we must pay to a third-party broker, referral agent, sales agent, development agent or lead generation source (a "Broker") in connection with the sale of your ALL Business (a "Broker Commission"); or (b) pay us a \$10,000 resale assistance fee (the "Resale Assistance Fee") if the buyer is not procured by a Broker to whom a Broker Commission is owed in connection with the sale. You agree to pay us the Broker Commission or Resale Assistance Fee, as applicable, in one lump sum concurrently with the closing of the sale of your ALL Business to the buyer. The Broker Commission or Resale Assistance Fee, as applicable, shall be in addition to any transfer fee imposed under the Franchise Agreement.
- 9. <u>Indemnification</u>. Your indemnification obligation under the Franchise Agreement shall extend to and apply with respect to any damages we incur as a result of or in connection with: (a) your breach of this Agreement; or (b) your discussions, negotiations or other relationship with any potential buyer solicited through the Franchise Recruitment Program, including your breach of any agreement between you and such buyer.
- 10. <u>No Liability</u>. You hereby agree to hold harmless, and not assert any claims against, us, our affiliates, any of our (or our affiliates') owners, officers, employees or other representatives, or any member of our Franchise Recruitment Program, in connection with any dispute or disagreement between you and any prospective or actual buyer of your ALL Business.
- 11. **<u>Default & Termination</u>**. If you breach any obligation under this Agreement and fail to cure the breach within 10 days after notice from us, we may immediately terminate this Agreement upon notice to you and pursue any and all remedies available to us under this Agreement, the Franchise Agreement, at law or in equity. Your default under this Agreement constitutes a default under the Franchise Agreement. Similarly, your default under the Franchise Agreement constitutes a default under this Agreement.
- 12. Effect of Expiration or Termination. Upon expiration of the Term, we have no further responsibility or obligation to assist you with the sale of your ALL Business unless otherwise agreed to by both Parties in writing. If, following the expiration or termination of this Agreement, you sell your ALL Business to any person: (a) who was initially solicited through our Franchise Recruitment Program during the Term of this Agreement; and/or (b) with respect to whom we provided material assistance in connection with your sales efforts during the Term of this Agreement, then you agree to pay us the Broker Commission or Resale Assistance Fee, as applicable, set forth in §8 notwithstanding the prior termination or expiration of this Agreement.

13. Miscellaneous.

(a) <u>No Waiver</u>. By executing this Agreement, we shall not be deemed to have: (a) waived or impaired any right, power or option granted to us under the Franchise Agreement; or (b) waived or consented to any default or breach by you under the Franchise Agreement.

- (b) <u>Effect on Franchise Agreement</u>. All terms, conditions, covenants and representations set forth in the Franchise Agreement shall remain in full force and effect during the Term of this Agreement.
- (c) <u>Binding Nature</u>. This Agreement shall be binding upon the Parties hereto and their respective heirs, personal representatives, successors and assigns.
- (d) <u>Time of Essence</u>. Time is of the essence of this Agreement and of every term, covenant and condition hereof.
- (e) <u>Headings</u>. The headings in this Agreement are for convenience only and are not to be construed as a part of this Agreement or in any way defining, limiting or amplifying the provisions hereof.
- (f) <u>Governing Law</u>. This Agreement shall be governed by, construed and enforced under the laws of the State of Arizona.
- (g) <u>Dispute Resolution</u>. Any dispute between the Parties relating to this Agreement shall be resolved in accordance with the dispute resolution procedures set forth in the Franchise Agreement, all of which are incorporated herein by this reference.
- (h) Entire Agreement; Modification. This Agreement constitutes the entire agreement and understanding between the Parties regarding the subject matter hereof. This Agreement may not be modified except in a writing signed by both Parties.
- (i) <u>Counterparts</u>. This Agreement may be executed in any number of counterparts, each of which shall be deemed to be an original, and all of which together shall constitute one and the same document.

The Parties have executed this Agreement effective as of the Effective Date first above written.

"FRANCHISOR"

ALL Franchising, LLC, a Delaw	are limited liability company
By:	
Name:	
Title:	
"FRANCHISEE"	
[]	
By:	
Name:	
Title·	

EXHIBIT "H"

TO DISCLOSURE DOCUMENT

STATE ADDENDA AND AGREEMENT RIDERS

[See Attached]

STATE ADDENDA AND AGREEMENT RIDERS

ADDENDUM TO FRANCHISE AGREEMENT, SUPPLEMENTAL AGREEMENTS,

AND FRANCHISE DISCLOSURE DOCUMENT FOR CERTAIN STATES FOR

ALL Franchising, LLC

BACKGROUND AND PURPOSE

Certain states have laws governing the franchise relationship and franchise documents. Certain states require modifications to the FDD, Franchise Agreement, Supplemental Agreements and other documents related to the sale of a franchise. This State-Specific Addendum ("State Addendum") will modify these agreements to comply with the applicable state's laws. The terms of this State Addendum will only apply if you meet the requirements of the applicable state independently of your signing of this State Addendum. The terms of this State Addendum (but only the State Addendum for the applicable State) will override any inconsistent provision of the FDD, Franchise Agreement or any Supplemental Documents. This State Addendum only applies to the following states: California, Hawaii, Illinois, Indiana, Maryland, Michigan, Minnesota, New York, North Dakota, Rhode Island, South Dakota, Virginia, Washington, and Wisconsin.

If your state requires these modifications, you will sign this State Addendum along with the Franchise Agreement and any Supplemental Agreements. If you sign this State Addendum, only the terms applicable to the state or states whose franchise laws apply to your transaction will govern. If you sign this State Addendum, but none of the state franchise laws listed above applies because their jurisdictional requirements have not been met, then this State Addendum will be void and inapplicable to you.

CALIFORNIA

- 1. The California Franchise Investment Law requires a copy of all proposed agreements relating to the sale of the Franchise be delivered together with the Disclosure Document.
- 2. Section 31125 of the California Corporations Code requires us to give you a disclosure document, in a form containing the information that the Commissioner may by rule or order require, before a solicitation of a proposed material modification of an existing franchise.
- 3. Neither the franchisor nor any person or franchise broker in Item 2 of the FDD is subject to any currently effective order of any national securities association or national securities exchange, as defined in the Securities Exchange Act of 1934, 15 U.S.C.A. 78a et seq., suspending or expelling such persons from membership in such association or exchange.
- 4. The earnings claims figures do not reflect the costs of sales, operating expenses, or other costs or expenses that must be deducted from the gross revenue or gross sales figures to obtain your net income or profit. You should conduct an independent investigation of the costs and expenses you will incur in operating your franchise business. Franchisees or former franchisees listed in the Disclosure Document may be one source of this information.
- 5. The Franchise Agreement and Supplemental Agreements require binding you to litigate in Maricopa County, Arizona with each party bearing its own costs of litigation.
- 6. Prospective franchisees are encouraged to consult private legal counsel to determine the applicability of California and federal laws (such as Business and Professions Code Section 20040.5, Code of Civil Procedure Section 1281, and the Federal Arbitration Act) to any provisions of a Franchise Agreement or Supplemental Agreement restricting venue to a forum outside the State of California.
- 7. The Franchise Agreement and Supplemental Agreements require application of the laws of Arizona. This provision may not be enforceable under California law.
- 8. The Franchise Agreement and Supplemental Agreements may provide for termination upon bankruptcy. Any such provision may not be enforceable under federal bankruptcy law (11 U.S.C.A. Sec. 101 et seq.).
- 9. The Franchise Agreement and Supplemental Agreements may contain a covenant not to compete which extends beyond the termination of the franchise. This provision may not be enforceable under California law.
- 10. Under California Civil Code Section 1671, certain liquidated damages clauses are unenforceable. Any such provisions contained in the Franchise Agreement or Supplemental Agreements may not be enforceable.
- 11. California Business and Professions Code Sections 20000 through 20043 provide rights to you concerning termination, transfer, or non-renewal of a franchise. If the Franchise Agreement or Supplemental Agreements contain a provision that is inconsistent with the California Franchise Investment Law, the California Franchise Investment Law will control.
- 12. You must sign a general release of claims if you renew or transfer your Franchise. California Corporations Code Section 31512 voids a waiver of your rights under the Franchise Investment Law (California Corporations Code Sections 31000 through 31516). Business and Professions Code Section 20010 voids a waiver of your rights under the Franchise Relations Act (Business and Professions Code Sections 20000 through 20043).

13. OUR WEBSITE HAS NOT BEEN REVIEWED OR APPROVED BY THE CALIFORNIA DEPARTMENT OF FINANCIAL PROTECTION & INNOVATION. ANY COMPLAINTS CONCERNING THE CONTENT OF THIS WEBSITE MAY BE DIRECTED TO THE CALIFORNIA DEPARTMENT OF FINANCIAL PROTECTION & INNOVATION AT https://dfpi.ca.gov/.

HAWAII

1. The following is added to the Cover Page:

THIS FRANCHISE WILL BE/HAS BEEN FILED UNDER THE FRANCHISE INVESTMENT LAW OF THE STATE OF HAWAII. FILING DOES NOT CONSTITUTE APPROVAL, RECOMMENDATION OR ENDORSEMENT BY THE DIRECTOR OF COMMERCE AND CONSUMER AFFAIRS OR A FINDING BY THE DIRECTOR OF COMMERCE AND CONSUMER AFFAIRS THAT THE INFORMATION PROVIDED IN THIS FRANCHISE DISCLOSURE DOCUMENT IS TRUE, COMPLETE AND NOT MISLEADING.

THE FRANCHISE INVESTMENT LAW MAKES IT UNLAWFUL TO OFFER OR SELL ANY FRANCHISE IN THIS STATE WITHOUT FIRST PROVIDING TO YOU OR SUBFRANCHISOR AT LEAST SEVEN DAYS PRIOR TO THE EXECUTION BY YOU OR SUBFRANCHISOR OF ANY BINDING FRANCHISE OR OTHER AGREEMENT, OR AT LEAST SEVEN DAYS PRIOR TO THE PAYMENT OF ANY CONSIDERATION BY YOU, WHICHEVER OCCURS FIRST, A COPY OF THE FRANCHISE DISCLOSURE DOCUMENT, TOGETHER WITH A COPY OF ALL PROPOSED AGREEMENTS RELATING TO THE SALE OF THE FRANCHISE.

THIS FRANCHISE DISCLOSURE DOCUMENT CONTAINS A SUMMARY ONLY OF CERTAIN MATERIAL PROVISIONS OF THE FRANCHISE AGREEMENT. THE CONTRACT OR AGREEMENT SHOULD BE REFERRED TO FOR A STATEMENT OF ALL RIGHTS, CONDITIONS, RESTRICTIONS AND OBLIGATIONS OF BOTH US AND YOU.

2. Our registered agent in the state authorized to receive service of process:

Commissioner of Securities of the State of Hawaii Department of Commerce and Consumer Affairs Business Registration Division 335 Merchant Street, Room 203 Honolulu, Hawaii 96813

- 3. The states in which this filing is effective are listed on the Exhibit to the FDD titled "State Effective Dates".
- 4. The states in which this filing is or will be shortly on file include the following: California, Hawaii, Illinois, Indiana, Maryland, Michigan, Minnesota, New York, Rhode Island, Virginia, Washington and Wisconsin.
- 5. The states, if any, which have refused, by order or otherwise, to register these franchises include the following: None.
- 6. The states, if any, which have revoked or suspended the right to offer these franchises include the following: None.
- 7. The states, if any, in which the filing of these franchises has been withdrawn include the following: None.
- 8. No statement, questionnaire, or acknowledgment signed or agreed to by a franchisee in connection with the commencement of the franchise relationship shall have the effect of (i) waiving any claims under any applicable state franchise law, including fraud in the inducement, or (ii) disclaiming reliance on any

statement made by any franchisor, franchise seller, or other person acting on behalf of the franchisor. This provision supersedes any other term of any document executed in connection with the franchise.

ILLINOIS

In recognition of the requirements of the Illinois Franchise Disclosure Act, 815 ILCS 705, the Disclosure Document and the Franchise Agreement and Supplemental Agreements are amended as follows:

- 1. Illinois law shall apply to and govern the Franchise Agreement and Supplemental Agreements.
- 2. In accordance with <u>Section 4</u> of the Illinois Franchise Disclosure Act, any provision in the Franchise Agreement and Supplemental Agreements that designated jurisdiction and venue in a forum outside of the State of Illinoi is void.
- 3. Your rights upon Termination and Non-Renewal are set forth in sections 19 and 20 of the Illinois Franchise Disclosure Act.
- 4. In conformance with Section 41 of the Illinois Franchise Disclosure Act, any condition, stipulation, or provision purporting to bind any person acquiring any franchise to waive compliance with the Illinois Franchise Disclosure Act or any other law of Illinois is void.
- 5. The Franchise Agreement and Supplemental Agreements are amended to state the following:

To the extent that any provision in the Illinois State Addendum is inconsistent with any provision in this Agreement, the provision in the Illinois State Addendum shall control.

INDIANA

In recognition of the requirements of the Indiana Franchise Disclosure Law, IC 23-2-2-2.5, the Franchise Agreement and Supplemental Agreements are amended as follows:

- 1. The laws of the State of Indiana supersede any provisions of the Disclosure Document, Franchise Agreement and Supplemental Agreements if such provisions are in conflict with Indiana law.
- 2. The Franchise Agreement and Supplemental Agreements are amended to provide that such agreements will be construed in accordance with the laws of the State of Indiana.
- 3. Any provision in the Franchise Agreement which designates jurisdiction or venue, or requires the franchisee to agree to jurisdiction or venue, in a forum outside of Indiana, is deleted from any Franchise Agreement and Supplemental Agreement issued in the State of Indiana.
- 4. The prohibition by Indiana Code § 23-2-2.7-1(7) against unilateral termination of the franchise without good cause or in bad faith, good cause being defined therein as material breach of the Franchise Agreement or Supplemental Agreement (as applicable), shall supersede the provisions of the Franchise Agreement or Supplemental Agreement (as applicable) in the State of Indiana to the extent they may be inconsistent with such prohibition.
- 5. Liquidated damages and termination penalties are prohibited by law in the State of Indiana and, therefore, the Disclosure Document, the Franchise Agreement and Supplemental Agreements are amended by the deletion of all references to liquidated damages and termination penalties and the addition of the following language to the original language that appears therein:

Notwithstanding any such termination, and in addition to the obligations of the franchisee as otherwise provided, or in the event of termination or cancellation of the Franchise Agreement under any of the other provisions therein, the franchisee nevertheless shall be, continue and remain liable to franchisor for any and all damages which franchisor has sustained or may sustain by reason of such default or defaults and the breach of the Franchise Agreement on the part of the franchisee for the unexpired Term of the Franchise Agreement.

At the time of such termination of the Franchise Agreement, the franchisee covenants to pay to franchisor within 10 days after demand as compensation all damages, losses, costs and expenses (including reasonable attorney's fees) incurred by franchisor, and/or amounts which would otherwise be payable thereunder but for such termination for and during the remainder of the unexpired Term of the Franchise Agreement. This Agreement does not constitute a waiver of the franchisee's right to a trial on any of the above matters.

6. No release language set forth in the Disclosure Document or Franchise Agreement or Supplemental Agreement shall relieve franchisor or any other person, directly or indirectly, from liability imposed by the laws concerning franchising of the State of Indiana. Any provision in the Franchise Agreement or Supplemental Agreement that would require you to prospectively assent to a release, assignment, novation, waiver or estoppel which purports to relieve any person from liability imposed by the Indiana Deceptive Franchise Practices Law is void to the extent that such provision violates such law.

MARYLAND

In recognition of the requirements of the Maryland Franchise Registration and Disclosure Law (the "Maryland Franchise Law"), the Disclosure Document and Franchise Agreement is amended as follows:

1. Item 5 of the Disclosure Document is amended as follows:

The Maryland Securities Commissioner has imposed a fee deferral requirement because of the Franchisor's financial condition. Accordingly, Item 5 of the Franchise Disclosure Document and Section 7.1 of the Franchise Agreement are amended to provide that payment of all initial fees is deferred until after franchisor's pre-opening obligations are complete and franchisee is open for business.

- 2. Item 17 of the Disclosure Document is amended to add the following:
 - a. The general release required as a condition of renewal, sale and/or assignment/transfer shall not apply any liability under the Maryland Franchise Registration and Disclosure Law.
 - b. A franchisee may bring a lawsuit in Maryland for claims arising under the Maryland Franchise Registration and Disclosure Law.
 - c. Any claims arising under the Maryland Franchise Registration and Disclosure Law must be brought within 3 years after the grant of the franchise.
 - d. In the event of a conflict of laws to the extent required by the Maryland Franchise Registration and Disclosure Law, Maryland law shall prevail.
 - e. The Franchise Agreement and Supplemental Agreements provide for termination upon bankruptcy. This provision may not be enforceable under federal bankruptcy law (11 U.S.C. Section 101, et seq.).
- 3. The Franchise Disclosure Questionnaire, which is attached as an Exhibit to the Disclosure Document, is amended as follows:

All representations requiring prospective franchisees to assent to the release, estoppel or waiver of liability are not intended to nor shall they act as a release, estoppel, or waiver of any liability incurred under the Maryland Franchise Registration and Disclosure Law.

In recognition of the requirements of the Maryland Franchise Law, the Franchise Agreement and Supplemental Agreements are amended to add the following:

- 1. Any claims arising under the Maryland Franchise Law must be brought within three (3) years after the grant of the franchise.
- 2. Pursuant to COMAR 02.02.08.16L, the general release required as a condition of renewal, sale, and/or assignment/transfer shall not apply to any liability under the Maryland Franchise Law.
- 3. You may bring a lawsuit in Maryland for claims arising under the Maryland Franchise Law.
- 4. No statement, questionnaire, or acknowledgment signed or agreed to by a franchisee in connection with the commencement of the franchise relationship shall have the effect of (i) waiving any claims under any applicable state franchise law, including fraud in the inducement, or (ii) disclaiming reliance on any statement made by any franchisor, franchise seller, or other person acting on behalf of the franchisor. This provision supersedes any other term of any document executed in connection with the franchise..."

- 5. Any acknowledgements or representations by you that disclaim the occurrence and/or acknowledge the non-occurrence of acts that would constitute a violation of the Maryland Law are not intended to nor shall they act as a release, estoppel or waiver of any liability incurred under the Maryland Law.
- 6. Nothing in the Franchise Agreement, Supplemental Agreement or in any related agreement is intended to disclaim the representations made in the Franchise Disclosure Document.
- 7. The Maryland Securities Commissioner has imposed a fee deferral requirement because of the Franchisor's financial condition. Accordingly, Item 5 of the Franchise Disclosure Document and Section 7.1 of the Franchise Agreement are amended to provide that payment of all initial fees is deferred until after franchisor's pre-opening obligations are complete and franchisee is open for business.

MICHIGAN

THE STATE OF MICHIGAN PROHIBITS CERTAIN UNFAIR PROVISIONS THAT ARE SOMETIMES IN FRANCHISE DOCUMENTS. IF ANY OF THE FOLLOWING PROVISIONS ARE IN THESE FRANCHISE DOCUMENTS, THE PROVISIONS ARE VOID AND CANNOT BE ENFORCED AGAINST YOU.

Each of the following provisions is void and unenforceable if contained in any document relating to a franchise:

- (a) A prohibition on the right of a franchisee to join an association of franchisees.
- (b) A requirement that a franchisee assent to a release, assignment, novation, waiver, or estoppel which deprives a franchisee of rights and protections provided in this act. This shall not preclude a franchisee, after entering into a franchise agreement, from settling any and all claims.
- (c) A provision that permits a franchisor to terminate a franchise prior to the expiration of its term except for good cause. Good cause shall include the failure of the franchisee to comply with any lawful provision of the franchise agreement and to cure such failure after being given written notice thereof and a reasonable opportunity, which in no event need be more than 30 days, to cure such failure.
- (d) A provision that permits a franchisor to refuse to renew a franchise without fairly compensating the franchisee by repurchase or other means for the fair market value at the time of expiration of the franchisee's inventory, supplies, equipment, fixtures, and furnishings. Personalized materials which have no value to the franchisor and inventory, supplies, equipment, fixtures, and furnishings not reasonably required in the conduct of the franchise business are not subject to compensation. This subsection applies only if: (i) The term of the franchise is less than 5 years and (ii) the franchisee is prohibited by the franchise or other agreement from continuing to conduct substantially the same business under another trademark, service mark, trade name, logotype, advertising, or other commercial symbol in the same area subsequent to the expiration of the franchise or the franchisee does not receive at least 6 months advance notice of franchisor's intent not to renew the franchise.
- (e) A provision that permits the franchisor to refuse to renew a franchise on terms generally available to other franchisees of the same class or type under similar circumstances. This section does not require a renewal provision.
- (f) A provision requiring that arbitration or litigation be conducted outside this state. This shall not preclude the franchisee from entering into an agreement, at the time of arbitration, to conduct arbitration at a location outside this state.
- (g) A provision which permits a franchisor to refuse to permit a transfer of ownership of a franchise, except for good cause. This subdivision does not prevent a franchisor from exercising a right of first refusal to purchase the franchise. Good cause shall include, but is not limited to:
- (i) The failure of the proposed transferee to meet the franchisor's then current reasonable qualifications or standards.
 - (ii) The fact that the proposed transferee is a competitor of the franchisor or subfranchisor.
- (iii) The unwillingness of the proposed transferee to agree in writing to comply with all lawful obligations.

- (iv) The failure of the franchisee or proposed transferee to pay any sums owing to the franchisor or to cure any default in the franchise agreement existing at the time of the proposed transfer.
- (h) A provision that requires the franchisee to resell to the franchisor items that are not uniquely identified with the franchisor. This subdivision does not prohibit a provision that grants to a franchisor a right of first refusal to purchase the assets of a franchise on the same terms and conditions as a bona fide third party willing and able to purchase those assets, nor does this subdivision prohibit a provision that grants the franchisor the right to acquire the assets of a franchise for the market or appraised value of such assets if the franchisee has breached the lawful provisions of the franchise agreement and has failed to cure the breach in the manner provided in subdivision (c).
- (i) A provision which permits the franchisor to directly or indirectly convey, assign, or otherwise transfer its obligations to fulfill contractual obligations to the franchisee unless provision has been made for providing the required contractual services.

If the franchisor's most recent financial statements are unaudited and show a net worth of less than \$100,000.00, the franchise may request the franchisor to arrange for the escrow of initial investment and other funds paid by the franchise until the obligations, if any, of the franchisor to provide real estate, improvements, equipment, inventory, training or other items included in the franchise offering are fulfilled. At the option of the franchisor, a surety bond may be provided in place of escrow.

THE FACT THAT THERE IS A NOTICE OF THIS OFFERING ON FILE WITH THE ATTORNEY GENERAL DOES NOT CONSTITUTE APPROVAL, RECOMMENDATION, OR ENDORSEMENT BY THE ATTORNEY GENERAL.

Any questions regarding this notice should be directed to:

State of Michigan
Department of Attorney General
CONSUMER PROTECTION DIVISION
Attention: Franchise Section
G. Mennen Williams Building, 1st Floor
525 West Ottawa Street
Lansing, Michigan 48913
Telephone Number: (517) 373-7117

MINNESOTA

In recognition of the Minnesota Franchise Law, Minn. Stat., Chapter 80C, Sections 80C.01 through 80C.22, and the Rules and Regulations promulgated pursuant thereto by the Minnesota Commission of Securities, Minnesota Rule 2860.4400, et. seq., the Disclosure Document, Franchise Agreement and Supplemental Agreements are amended as follows:

- 1. Minnesota Rule 2860.4400(D) prohibits us from requiring you to assent to a general release.
- 2. We will comply with Minnesota Statute Section 80C.14, Subds. 3, 4 and 5 which require, except in certain specified cases, that you be given 90 days' notice of termination (with 60 days to cure) and 180 days' notice for non-renewal of the Franchise Agreement or Supplemental Agreement; and that consent to the transfer of the franchise will not be unreasonably withheld.
- 3. Minnesota Statute Section 80C.21 and Minnesota Rule 2860.4400(J) prohibit us from requiring litigation to be conducted outside Minnesota, requiring waiver of a jury trial, or requiring you to consent to liquidated damages, termination penalties or judgment notes. In addition, nothing in the Franchise Disclosure Document or agreement(s) can abrogate or reduce any of your rights as provided for in Minnesota Statues, chapter 80C, or your rights to any procedure, forum, or remedies provided for by the laws of the jurisdiction. In addition, we will comply with the provisions of Minnesota Rule 2860.4400(J), which state that you cannot waive any rights, you cannot consent to our obtaining injunctive relief, we may seek injunctive relief, and a court will determine if a bond if required.
- 4. We will comply with Minnesota Statute Section 80C.12, Subd. 1(g), which requires that we protect your right to use the trademarks, service marks, trade names, logotypes or other commercial symbols or indemnify you from any loss, costs or expenses arising out of any claim, suit or demand regarding the use of the name.
- 5. We will comply with Minnesota Statute Section 80C.17, Subd. 5 regarding limitation of claims.
- 6. No statement, questionnaire, or acknowledgement signed or agreed to by a franchisee in connection with the commencement of the franchise relationship shall have the effect of (i) waiving any claims under any applicable state franchise law, including, fraud in the inducement, or (ii) disclaiming reliance on any statement made by any franchisor, franchise seller, or other person acting on behalf of the franchisor. This provision supersedes any other term of any document executed with the franchise.

NEW YORK

In recognition of the requirements of the General Business Laws of the State of New York, Article 33, §§680 through 695, the Disclosure Document, Franchise Agreement and Supplemental Agreements are amended as follows:

1. The following information is added to the cover page of the Disclosure Document:

INFORMATION COMPARING FRANCHISORS IS AVAILABLE. CALL THE STATE ADMINISTRATORS LISTED IN EXHIBIT A OR YOUR PUBLIC LIBRARY FOR SOURCES OF INFORMATION. REGISTRATION OF THIS FRANCHISE BY NEW YORK STATE DOES NOT MEAN THAT NEW YORK STATE RECOMMENDS IT OR HAS VERIFIED THE INFORMATION IN THIS FRANCHISE DISCLOSURE DOCUMENT. IF YOU LEARN THAT ANYTHING IN THE FRANCHISE DISCLOSURE DOCUMENT IS UNTRUE, CONTACT THE FEDERAL TRADE COMMISSION AND NEW YORK STATE DEPARTMENT OF LAW, BUREAU OF INVESTOR PROTECTION, 28 LIBERTY STREET, 21ST FLOOR, NEW YORK, NEW YORK 10005. THE FRANCHISOR MAY, IF IT CHOOSES, NEGOTIATE WITH YOU ABOUT ITEMS COVERED IN THE FRANCHISE DISCLOSURE DOCUMENT. HOWEVER, THE FRANCHISOR CANNOT USE THE NEGOTIATING PROCESS TO PREVAIL UPON A PROSPECTIVE FRANCHISEE TO ACCEPT TERMS WHICH ARE LESS FAVORABLE THAN THOSE SET FORTH IN THIS FRANCHISE DISCLOSURE DOCUMENT.

2. The following is added at the end of Item 3 of the Disclosure Document:

Except as provided above, with regard to the franchisor, its predecessor, a person identified in Item 2, or an affiliate offering franchises under the franchisor's principal trademark:

A. No such party has an administrative, criminal or civil action pending against that person alleging: a felony, a violation of a franchise, antitrust, or securities law, fraud, embezzlement, fraudulent conversion, misappropriation of property, unfair or deceptive practices, or comparable

civil or misdemeanor allegations.

- B. No such party has pending actions, other than routine litigation incidental to the business, which are significant in the context of the number of franchisees and the size, nature or financial condition of the franchise system or its business operations.
- C. No such party has been convicted of a felony or pleaded nolo contendere to a felony charge or, within the 10 year period immediately preceding the application for registration, has been convicted of or pleaded nolo contendere to a misdemeanor charge or has been the subject of a civil action alleging: violation of a franchise, antifraud, or securities law; fraud; embezzlement; fraudulent conversion or misappropriation of property; or unfair or deceptive practices or comparable allegations.
- D. No such party is subject to a currently effective injunctive or restrictive order or decree relating to the franchise, or under a Federal, State, or Canadian franchise, securities, antitrust, trade regulation or trade practice law, resulting from a concluded or pending action or proceeding brought by a public agency; or is subject to any currently effective order of any national securities association or national securities exchange, as defined in the Securities and Exchange Act of 1934, suspending or expelling such person from membership in such

association or exchange; or is subject to a currently effective injunctive or restrictive order relating to any other business activity as a result of an action brought by a public agency or department, including, without limitation, actions affecting a license as a real estate broker or sales agent.

3. The following is added to the end of the "Summary" sections of Item 17(c) of the Disclosure Document, titled "Requirements for franchisee to renew or extend," and Item 17(m) of the Disclosure Document, entitled "Conditions for franchisor approval of transfer":

However, to the extent required by applicable law, all rights you enjoy and any causes of action arising in your favor from the provisions of Article 33 of the General Business Law of the State of New York and the regulations issued thereunder shall remain in force; it being the intent of this proviso that the non-waiver provisions of General Business Law Sections 687.4 and 687.5 be satisfied.

4. The following language replaces the "Summary" section of Item 17(d) of the Disclosure Document, titled "Termination by franchisee":

You may terminate the agreement on any grounds available by law.

5. The following is added to the end of the "Summary" sections of Item 17(v) of the Disclosure Document, titled "Choice of forum", and Item 17(w) of the Disclosure Document, titled "Choice of law":

The foregoing choice of law should not be considered a waiver of any right conferred upon the franchisor or upon the franchisee by Article 33 of the General Business Law of the State of New York.

STATE SPECIFIC ADDENDUM TO ALL FRANCHISING, LLC FRANCHISE AGREEMENT AS REQUIRED BY THE NORTH DAKOTA FRANCHISE INVESTMENT LAW

Notwithstanding anything to the contrary set forth in the ALL Franchising, LLC Franchise Disclosure Document, the following provisions shall supersede any inconsistent provisions and apply to all Assisted Living Locators franchises offered and sold in the state of North Dakota:

The North Dakota Addendum is only applicable if you are a resident of North Dakota or if your Assisted Living Locators franchised business will be operated in North Dakota.

1. Section 4.2 of the Franchise Agreement is hereby amended by the addition of the following language that appears therein:

Provisions requiring North Dakota franchisees to sign a general release upon renewal of the franchise agreement are not enforceable in North Dakota.

2. Section 15.4 of the Franchise Agreement is hereby amended by the addition of the following language to the original language that appears therein:

Covenants not to compete such as those mentioned above are generally considered unenforceable in the state of North Dakota.

3. Section 23.3 of the Franchise Agreement is hereby amended by the addition of the following language to the original language that appears therein:

The provision requires that the franchisee consent to the jurisdiction of courts where Franchisor's headquarters is then located. This requirement is deleted from all Franchise Agreements used in the State of North Dakota.

4. Section 18.8 of the Franchise Agreement is hereby amended by the addition of the following language to the original language that appears therein:

For North Dakota Franchisees, North Dakota law shall apply.

5. Section 25.1 of the Franchise Agreement is amended by the addition of the following language to the original language that appears therein:

For North Dakota franchises, arbitration will be agreeable to all parties and may not be remote from the franchisee's place of business.

6. Section 23.4 of the Franchise Agreement is hereby amended as follows:

Section 17.3.B of the Franchise Agreement requires the franchisee to consent to a limitation of claims within one year. The Commissioner has determined this to be unfair, unjust, and inequitable within the intent of Section 51-19-09 of the North Dakota Franchise Investment Law. The provision shall be changed to read the statute of limitations under North Dakota Law will apply.

7. Section 23.6 of the Franchise Agreement is hereby amended by the addition of the following language to the original language that appears therein:

Provisions requiring the franchisee to consent to a waiver of exemplary and punitive damages are not enforceable under Section 51-19-09 of the North Dakota Franchise Investment Law.

8. Section 23.5 of the Franchise Agreement is hereby amended by the addition of the following language to the original language that appears therein:

Provisions requiring a franchisee to consent to a waiver of trial by jury are not enforceable under Section 51-19-09 of the North Dakota Franchise Investment Law.

9. Section 23.3 of the Franchise Agreement is hereby amended by the addition of the following language to the original language that appears therein:

Section 23.3 of the Franchise Agreement stipulates that the franchisee shall pay all costs and expenses incurred by the franchisor in enforcing the agreement. The Commissioner has determined this to be unfair, unjust, and inequitable within the intent of Section 51-19-09 of the North Dakota Franchise Investment Law. The provision shall be changed to read that the prevailing party in any enforcement action is entitled to recover all costs and expenses including attorney's fees.

10. No statement, questionnaire, or acknowledgement signed or agreed to by a franchisee in connection with the commencement of the franchise relationship shall have the effect of (i) waiving any claims under any applicable state franchise law, including fraud in the inducement, or (ii) disclaiming reliance on any statement made by any franchisor, franchise seller, or other person acting on behalf of the franchisor. This provision supersedes any other term of any document executed in connection with the franchise.

FRANCHISOR:

ALL Franchising, LLC	
By:	
Title:	
Signature:	
FRANCHISEE:	
By:	
Title:	
Signature:	

ADDENDUM TO ALL FRANCHISING, LLC FRANCHISE DISCLOSURE DOCUMENT REQUIRED BY THE STATE OF NORTH DAKOTA

For franchises and franchisees subject to the North Dakota Franchise Investment Law, the following information supersedes or supplements, as the case maybe, the corresponding disclosures in the main body of the text of the ALL FRANCHISING, LLC Franchise Disclosure Document ("FDD").

- 1. Item 17 is amended by the addition of the following language to the original language that appears therein:
- (a) Covenants not to compete upon termination or expiration of a franchise agreement are generally unenforceable in North Dakota, except in certain instances as provided by law.
- (b) Any provision in the Franchise Agreement which designates jurisdiction or venue or requires the franchisee to agree to jurisdiction or venue in a forum outside of North Dakota is void with respect to any cause of action which is otherwise enforceable in North Dakota.
- (c) Any provision in the Franchise Agreement which requires a franchisee to waive his or her right to a jury trial has been determined to be unfair, unjust and inequitable within the intent of Section 51-19-09 of the North Dakota Franchise Investment Law.
- (d) Any provision requiring a franchisee to sign a general release upon renewal of the franchise agreement has been determined to be unfair, unjust and inequitable within the intent of Section 51-19-09 of the North Dakota Franchise Investment Law.
- (e) Any provision in the Franchise Agreement requiring a franchisee to agree to the arbitration or mediation of disputes at a location that is remote from the site of the franchisee's business has been determined to be unfair, unjust and inequitable within the intent of Section 51-19-09 of the North Dakota Franchise Investment Law.
- (f) Apart from civil liability as set forth in Section 51-19-12 of the N.D.C.C., which is limited to violations of the North Dakota Franchise Investment Law (registration and fraud), the liability of the franchisor to a franchisee is based largely on contract law. Despite the fact that those provisions are not contained in the franchise investment law, those provisions contain substantive rights intended to be afforded to North Dakota residents and it is unfair to franchise investors to require them to waive their rights under North Dakota Law.
- (g) Any provision in the Franchise Agreement requiring that the Franchise Agreement be construed according to the laws of a state other than North Dakota are unfair, unjust or inequitable within the intent of Section 51-19-09 of the North Dakota Franchise Investment Law.
- (h) Any provision in the Franchise Agreement requiring a franchisee to consent to termination or liquidated damages is unfair, unjust or inequitable within the intent of Section 51-19-09 of the North Dakota Franchise Investment Law.

- (i) Any provision in the Franchise Agreement requiring a franchisee to consent to a waiver of exemplary and punitive damages is unfair, unjust or inequitable within the intent of Section 51-19-09 of the North Dakota Franchise Investment Law.
- 2. No statement, questionnaire, or acknowledgement signed or agreed to by a franchisee in connection with the commencement of the franchise relationship shall have the effect of (i) waiving any claims under any applicable state franchise law, including fraud in the inducement, or (ii) disclaiming reliance on any statement made by any franchisor, franchise seller, or other person acting on behalf of the franchisor. This provision supersedes any other term of any document executed in connection with the franchise.

Agreed to by:
FRANCHISEE:
By:
Name:
Title:
Date:
FRANCHISOR:
ALL Franchising, LLC
ALL Pranchishing, LLC
By:
Name:
Title:
Date:

RHODE ISLAND

In recognition of the requirements of the Rhode Island Franchise Investment Act (the "Rhode Island Franchise Law"), the Disclosure Document, Franchise Agreement and Supplemental Agreements are amended as follows:

- 1. We will not require that you prospectively assent to a waiver, condition, stipulation, or provision that purports to relieve any person from liability imposed by the Rhode Island Franchise Law. This provision does not apply to the settlement of disputes, claims, or civil lawsuits brought under the Rhode Island Franchise Law.
- 2. Section 19-28.1-14 of the Rhode Island Franchise Law provides that "A provision in a franchise agreement restricting jurisdiction or venue to a forum outside this state or requiring the application of the laws of another state is void with respect to a claim otherwise enforceable under this Act." If a claim is enforceable under the Rhode Island Franchise Law, we will not restrict jurisdiction or venue to a forum outside the State of Rhode Island or require the application of the laws of another state.
- 3. We will not prohibit you from joining a trade association or association of franchisees. We will not retaliate against you for engaging in these activities.
- 4. Any provision in the Franchise Agreement that limits the time period in which you may assert a legal claim against us under the Rhode Island Franchise Law is amended to provide for a four (4) year statute of limitations for purposes of bringing a claim arising under the Rhode Island Franchise Law. Notwithstanding the foregoing, if a rescission offer has been approved by the Rhode Island director of business registration, then the statute of limitations is ninety (90) days after your receipt of the rescission offer.

VIRGINIA

In recognition of the requirements of the Virginia Retail Franchising Act, the Disclosure Document, Franchise Agreement and Supplemental Agreements are amended as follows:

1. Item 17 of the Disclosure Document is amended to add the following:

Pursuant to Section 13.1-564 of the Virginia Retail Franchising Act, it is unlawful for a franchisor to cancel a franchise without reasonable cause. If any grounds for default or termination stated in the Franchise Agreement or Supplemental Agreement does not constitute "reasonable cause," as that term may be defined in the Virginia Retail Franchising Act or the laws of Virginia, that provision may not be enforceable.

Pursuant to Section 13.1-564 of the Virginia Retail Franchising Act, it is unlawful for a franchisor to use undue influence to induce a franchisee/area developer to surrender any right given to him under the applicable agreement.

- 2. If any provision of the Franchise Agreement or any Supplemental Agreement involves the use of undue influence by the franchisor to induce a franchisee/area developer to surrender any rights given to him under the applicable agreement, that provision may not be enforceable.
- 3. We will not require that you prospectively assent to a waiver, condition, stipulation, or provision that purports to relieve any person from liability imposed by the Virginia Retail Franchising Act.
- 4. Any provision in the Franchise Agreement or Supplemental Agreement that limits the time period in which you may assert a legal claim against us under the Virginia Retail Franchising Act is amended to provide for a four (4) year statute of limitations for purposes of bringing a claim arising under the Virginia Retail Franchising Act.
- 5. Pursuant to Section 13.1-564 of the Virginia Retail Franchising Act, it shall be unlawful for a franchisor to cancel a franchise without reasonable cause. If any grounds for default or termination stated in the Franchise Agreement or Supplemental Agreement does not constitute "reasonable cause," as that term may be defined in the Virginia Retail Franchising Act or the laws of Virginia, that provision may not be enforceable.
- 6. The Virginia State Corporation Commission's Division of Securities and Retail Franchising requires us to defer payment of the initial franchise fee and other initial payments owed by franchisees to the franchisor until the franchisor has completed its pre-opening obligations under the franchise agreement.

WASHINGTON ADDENDUM TO THE FRANCHISE DISCLOSURE DOCUMENT, THE FRANCHISE AGREEMENT, AND ALL RELATED AGREEMENTS

In recognition of the requirements of the Washington Franchise Investment Protection Act, Chapter 19.100 RCW, the Disclosure Document, Franchise Agreement and Supplemental Agreements are amended as follows:

The provisions of this Addendum form an integral part of, are incorporated into, and modify the Franchise Disclosure Document, the franchise agreement, and all related agreements regardless of anything to the contrary contained therein. This Addendum applies if: (a) the offer to sell a franchise is accepted in Washington; (b) the purchaser of the franchise is a resident of Washington; and/or (c) the franchised business that is the subject of the sale is to be located or operated, wholly or partly, in Washington.

- 1. <u>Conflict of Laws.</u> In the event of a conflict of laws, the provisions of the Washington Franchise Investment Protection Act, <u>Chapterchapter</u> 19.100 RCW₇ will prevail.
- 2. Franchisee Bill of Rights. RCW 19.100.180 may supersede the Franchise Agreement and Supplemental Agreements provisions in the franchise agreement or related agreements concerning your relationship with the franchisor, including in the areas of termination and renewal of your franchise. There may also be court decisions which maythat supersede the Franchise Agreement and Supplemental Agreements in franchise agreement or related agreements concerning your relationship with the franchisor. Franchise agreement provisions, including the areas of termination and renewal of your franchise. those summarized in Item 17 of the Franchise Disclosure Document, are subject to state law.
- 3. <u>Site of Arbitration, Mediation, and/or Litigation.</u> In any <u>arbitration or mediation involving a franchise purchased in Washington, the <u>arbitration or mediation site will be either in the state of Washington, or in a place mutually agreed upon at the time of the <u>arbitration or mediation</u>, or as determined by the <u>arbitrator or mediator at the time of arbitration or mediation.</u> In addition, if litigation is not precluded by the franchise agreement, a franchise may bring an action or proceeding arising out of or in connection with the sale of franchises, or a violation of the Washington Franchise Investment Protection Act, in Washington.</u></u>
- 4. General Release. A release or waiver of rights executed by a in the franchise agreement or related agreements purporting to bind the franchisee may not include rights to waive compliance with any provision under the Washington Franchise Investment Protection Act or any rule rules or order orders thereunder is void except when executed pursuant to a negotiated settlement after the agreement is in effect and where the parties are represented by independent counsel, in accordance with RCW 19.100.220(2). In addition, any such release or waiver executed in connection with a renewal or transfer of a franchise is likewise void except as provided for in RCW 19.100.220(2).
- 5. Statute of Limitations and Waiver of Jury Trial. Provisions contained in the franchise agreement or related agreements that unreasonably restrict or limit the statute of limitations period for claims under the Washington Franchise Investment Protection Act, or rights or remedies under the Act such as a right to a jury trial, may not be enforceable.
- <u>6. Transfer Fees. Transfer fees are collectable only to the extent that they reflect the franchisor's reasonable estimated or actual costs in effecting a transfer.</u>
- 7. **Termination by Franchisee.** The franchisee may terminate the franchise agreement under any grounds permitted under state law.

- 8. Certain Buy-Back Provisions. Provisions in franchise agreements or related agreements that permit the franchisor to repurchase the franchisee's business for any reason during the term of the franchise agreement without the franchisee's consent are unlawful pursuant to RCW 19.100.180(2)(j), unless the franchise is terminated for good cause.
- 9. **Fair and Reasonable Pricing.** Any provision in the franchise agreement or related agreements that requires the franchisee to purchase or rent any product or service for more than a fair and reasonable price is unlawful under RCW 19.100.180(2)(d).
- 4.10. Waiver of Exemplary & Punitive Damages. RCW 19.100.190 permits franchisees to seek treble damages under certain circumstances. Accordingly, provisions contained in the franchise agreement or elsewhere requiring franchisees to waive exemplary, punitive, or similar damages are void, except when executed pursuant to a negotiated settlement after the agreement is in effect and where the parties are represented by independent counsel. Provisions such as those which unreasonably restrict or limit the statute of limitations period for claims under the Act, or rights or remedies under the Act such as a right to a jury trial, may not be enforceable. in accordance with RCW 19.100.220(2).
- 5. Transfer fees are collectable to the extent that they reflect the franchisor's reasonable estimated or actual costs in effecting a transfer.
 - 11. **Franchisor's Business Judgement.** Provisions in the franchise agreement or related agreements stating that the franchisor may exercise its discretion on the basis of its reasonable business judgment may be limited or superseded by RCW 19.100.180(1), which requires the parties to deal with each other in good faith.
 - 12. **Indemnification.** Any provision in the franchise agreement or related agreements requiring the franchisee to indemnify, reimburse, defend, or hold harmless the franchisor or other parties is hereby modified such that the franchisee has no obligation to indemnify, reimburse, defend, or hold harmless the franchisor or any other indemnified party for losses or liabilities to the extent that they are caused by the indemnified party's negligence, willful misconduct, strict liability, or fraud.
 - 13. Attorneys' Fees. If the franchise agreement or related agreements require a franchisee to reimburse the franchisor for court costs or expenses, including attorneys' fees, such provision applies only if the franchisor is the prevailing party in any judicial or arbitration proceeding.
 - 6.14. Noncompetition Covenants. Pursuant to RCW 49.62.020, a noncompetition covenant is void and unenforceable against an employee, including an employee of a franchisee, unless the employee's earnings from the party seeking enforcement, when annualized, exceed \$100,000 per year (an amount that will be adjusted annually for inflation). In addition, a noncompetition covenant is void and unenforceable against an independent contractor of a franchisee under RCW 49.62.030 unless the independent contractor's earnings from the party seeking enforcement, when annualized, exceed \$250,000 per year (an amount that will be adjusted annually for inflation). As a result, any provisions provision contained in the franchise agreement or elsewhere that conflicts with these limitations are is void and unenforceable in Washington.
 - 7.15. Nonsolicitation Agreements. RCW 49.62.060 prohibits a franchisor from restricting, restraining, or prohibiting a franchisee from (i) soliciting or hiring any employee of a franchisee of the same franchisor or (ii) soliciting or hiring any employee of the franchisor. As a result, any such provisions contained in the franchise agreement or elsewhere are void and unenforceable in Washington.

- 16. -Questionnaires and Acknowledgments. No statement, questionnaire, or acknowledgment signed or agreed to by a franchisee in connection with the commencement of the franchise relationship shall have the effect of (i) waiving any claims under any applicable state franchise law, including fraud in the inducement, or (ii) disclaiming reliance on any statement made by any franchisor, franchise seller, or other person acting on behalf of the franchisor. This provision supersedes any other term of any document executed in connection with the franchise.
- 17. **Prohibitions on Communicating with Regulators.** Any provision in the franchise agreement or related agreements that prohibits the franchisee from communicating with or complaining to regulators is inconsistent with the express instructions in the Franchise Disclosure Document and is unlawful under RCW 19.100.180(2)(h).
- 18. Advisory Regarding Franchise Brokers. Under the Washington Franchise Investment Protection Act, a "franchise broker" is defined as a person that engages in the business of the offer or sale of franchises. A franchise broker represents the franchisor and is paid a fee for referring prospects to the franchisor and/or selling the franchise. If a franchisee is working with a franchise broker, franchisees are advised to carefully evaluate any information provided by the franchise broker about a franchise.
- 8.19. The following Risk Factor is added to the State Cover Page: THE FRANCHISOR HAS LIMITED FINANCIAL RESOURCES WHICH MIGHT NOT BE ADEQUATE TO FUND ITS PRE-OPENING OBLIGATIONS TO EACH FRANCHISEE AND PAY OPERATING EXPENSES.
- 9.20. Section 14.1 of the Franchise Agreement is hereby amended to include the following language:
 - "In lieu of an impound of franchise fees, the Franchisor will not require or accept the payment of any initial franchise fees until the franchisee has (a) received all pre-opening and initial training obligations that it is entitled to under the franchise agreement or offering circular, and (b) is open for business."
- 10.21. Section 19 of the Franchise Agreement is hereby amended to include the following language:
 - "Franchisees have no obligation to indemnify or hold harmless an indemnified party for losses to the extent that they are determined to have been caused solely and directly by the indemnified party's gross negligence, willful misconduct, strict liability, or fraud."
- 11.22. No statement, questionnaire, or acknowledgment signed or agreed to by a franchisee in connection with the commencement of the franchise relationship shall have the effect of (a) waiving any claims under any applicable state franchise law, including fraud in the inducement, or (b) disclaiming reliance on any statement made by any franchisor, franchise seller, or other person acting on behalf of the franchisor. This provision supersedes any other term of any document executed in connection with the franchise.

WISCONSIN

The Wisconsin Fair Dealership Law, Chapter 135 of the Wisconsin Statutes supersedes any provision of the Franchise Agreement and Supplement Agreements (if applicable) if such provision is in conflict with that law. The Franchise Disclosure Document, the Franchise Agreement and the Supplemental Agreements are amended accordingly.

APPLICABLE ADDENDA

If any one of the preceding Addenda for specific states ("Addenda") is checked as an "Applicable Addenda" below, then that Applicable Addenda shall be incorporated into the Franchise Disclosure Document, Franchise Agreement, Supplemental Agreements (if applicable) and any other specified agreement(s) entered into by us and the undersigned franchisee. To the extent any terms of an applicable Addenda conflict with the terms of the Franchise Disclosure Document, Franchise Agreement, Supplemental Agreement (if applicable) and other specified agreement(s), the terms of the Applicable Addenda shall supersede the terms of the Franchise Agreement.

	California		Michigan		South Dakota
	Hawaii		Minnesota		Virginia
	Illinois		New York		Washington
	Indiana		North Dakota		Wisconsin
	Maryland		Rhode Island		
Dated:		_, 202_			
			FRANCHISOR:		
	ALL Franchising, LLC				
	By:				
			Title:		
			FRANCHISEE:		
			By:		
			Title:		

EXHIBIT "I"

TO DISCLOSURE DOCUMENT

STATE EFFECTIVE DATES

State Effective Dates

The following states have franchise laws that require that the Franchise Disclosure Document be registered or filed with the state, or be exempt from registration: California, Hawaii, Illinois, Indiana, Maryland, Michigan, Minnesota, New York, North Dakota, Rhode Island, South Dakota, Virginia, Washington, and Wisconsin.

This document is effective and may be used in the following states, where the document is filed, registered or exempt from registration, as of the Effective Date stated below:

State	Effective Date
California	Exempt
Hawaii	
Illinois	Exempt
Indiana	Exempt
Maryland	Exemption effective June 5, 2024
Michigan	January 25, 2024 <u>2025</u> (amended May 9,
	2024 2025)
Minnesota	
New York	Exempt
North Dakota	May 21, 2024
Rhode Island	May 27, 2024
South Dakota	Exempt
Virginia	
Washington	Pending
Wisconsin	May 20, 2024 <u>14, 2025</u>

Other states may require registration, filing, or exemption of a franchise under other laws, such as those that regulate the offer and sale of business opportunities or seller-assisted marketing plans.

EXHIBIT "J"

TO DISCLOSURE DOCUMENT

RECEIPTS

[See Attached]

RECEIPT

This Disclosure Document summarizes certain provisions of the franchise agreement and other information in plain language. Read this Disclosure Document and all agreements carefully.

If ALL Franchising, LLC offers you a franchise, it must provide this Disclosure Document to you 14 days before you sign a binding agreement or make a payment with the franchisor or an affiliate in connection with the proposed franchise sale. New York requires that we give you this Disclosure Document at the earlier of the first personal meeting or 10 business days before the execution of the franchise or other agreement or the payment of any consideration that relates to the franchise relationship. You must also receive a franchise agreement containing all material terms at least 5 business days before you sign a franchise agreement.

If ALL Franchising, LLC does not deliver this Disclosure Document on time, or if it contains a false or misleading statement or a material omission, a violation of federal and state law may have occurred and should be reported to the Federal Trade Commission, Washington, DC 20580, and the appropriate state agency listed in EXHIBIT "A" to this Disclosure Document.

The franchise seller(s) involved with the sale of this franchise is/are:

<u>AngelaAn</u> 5832	gelica Olea; 8100 E. Indian School Road, Suite 201, Scottsdale, Arizona 85251; (480) 332 5884277			
Tom Ingle	e; 8100 E. Indian School Road, Suite 201, Scottsdale, Arizona 85251; (949) 498-4104			
Jason Wie	dder; 8100 E. Indian School Road, Suite 201, Scottsdale, Arizona 85251 (800) 267-7816			
Issuance Date: May 9,	2024 <u>2025</u>			
ALL Franchising, LLC	"s agent to receive service of process is listed in EXHIBIT "B" to this Disclosure Document.			
I received a Franchise	Disclosure Document dated May 9, 2024—that included the following Exhibits:			
EXHIBIT "A"	STATE AGENCIES AND ADMINISTRATORS			
EXHIBIT "B"	FRANCHISORS AGENT FOR SERVICE OF PROCESS			
EXHIBIT "C"	FRANCHISE AGREEMENT			
EXHIBIT "D"	TABLE OF CONTENTS OF BRAND STANDARDS MANUAL			
EXHIBIT "E"	LIST OF FRANCHISEES			
EXHIBIT "F"	FINANCIAL STATEMENTS AND GUARANTEE OF PERFORMANCE			
EXHIBIT "G"	OTHER AGREEMENTS			
EXHIBIT "G"-1	FRANCHISEE DISCLOSURE QUESTIONNAIRE			
EXHIBIT "G"-2	GENERAL RELEASE			
EXHIBIT "G"-3	ALL-IN USER AGREEMENT AND PRIVACY POLICY			
EXHIBIT "G"-4	HIPAA BUSINESS ASSOCIATE AGREEMENT			
EXHIBIT "G"-5	POWER OF ATTORNEY – TELEPHONE AND INTERNET			
EXHIBIT "G"-6	SAMPLE CLIENT CONTRACT			
EXHIBIT "G"-7	FRANCHISE RESALE AGREEMENT			
EXHIBIT "H"	STATE ADDENDA AND AGREEMENT RIDERS			
EXHIBIT "I"	STATE EFFECTIVE DATES			
EXHIBIT "J"	RECEIPTS			
	Print Name			
Date	(Signature) Prospective Franchise Owner			

(This Receipt should be executed in duplicate. One Receipt must be signed and remains in the Franchise Disclosure Document as the prospective franchise owner's copy. The other Receipt must be signed and returned to ALL Franchising, LLC.)

RECEIPT

This Disclosure Document summarizes certain provisions of the franchise agreement and other information in plain language. Read this Disclosure Document and all agreements carefully.

If ALL Franchising, LLC offers you a franchise, it must provide this Disclosure Document to you 14 days before you sign a binding agreement or make a payment with the franchisor or an affiliate in connection with the proposed franchise sale. New York requires that we give you this Disclosure Document at the earlier of the first personal meeting or 10 business days before the execution of the franchise or other agreement or the payment of any consideration that relates to the franchise relationship. You must also receive a franchise agreement containing all material terms at least 5 business days before you sign a franchise agreement.

If ALL Franchising, LLC does not deliver this Disclosure Document on time, or if it contains a false or misleading statement or a material omission, a violation of federal and state law may have occurred and should be reported to the Federal Trade Commission, Washington, DC 20580, and the appropriate state agency listed in EXHIBIT "A" to this Disclosure Document.

The franchise seller(s) involved with the sale of this franchise is/are:

<u>Angela</u> Ange 5832	lica Olea; 8100 E. Indian School Road, Suite 201, Scottsdale, Arizona 85251; (480) 332 5884277-			
Tom Ingle; 8	3100 E. Indian School Road, Suite 201, Scottsdale, Arizona 85251; (949) 498-4104			
Jason Wiedd	ler; 8100 E. Indian School Road, Suite 201, Scottsdale, Arizona 85251; (800) 267-7816			
Issuance Date: MayMay	9, 2024. <u>2025</u>			
ALL Franchising, LLC's	agent to receive service of process is listed in EXHIBIT "B" to this Disclosure Document.			
I received a Franchise Di	sclosure Document dated May 9, 2024 that included the following Exhibits:			
EXHIBIT "A" EXHIBIT "B" EXHIBIT "C" EXHIBIT "D" EXHIBIT "E" EXHIBIT "F" EXHIBIT "G" —EXHIBIT "G"-1 —EXHIBIT "G"-2 —EXHIBIT "G"-3 —EXHIBIT "G"-4 —EXHIBIT "G"-5 —EXHIBIT "G"-6 —EXHIBIT "G"-7 EXHIBIT "H" EXHIBIT "I" EXHIBIT "J"	STATE AGENCIES AND ADMINISTRATORS FRANCHISORS AGENT FOR SERVICE OF PROCESS FRANCHISE AGREEMENT TABLE OF CONTENTS OF BRAND STANDARDS MANUAL LIST OF FRANCHISEES FINANCIAL STATEMENTS AND GUARANTEE OF PERFORMANCE OTHER AGREEMENTS FRANCHISEE DISCLOSURE QUESTIONNAIRE GENERAL RELEASE ALL-IN USER AGREEMENT AND PRIVACY POLICY HIPAA BUSINESS ASSOCIATE AGREEMENT POWER OF ATTORNEY – TELEPHONE AND INTERNET SAMPLE CLIENT CONTRACT FRANCHISE RESALE AGREEMENT STATE ADDENDA AND AGREEMENT RIDERS STATE EFFECTIVE DATES RECEIPTS			
Print Name				
1 THE PAINC				
Date	(Signature) Prospective Franchise Owner			

(This Receipt should be executed in duplicate. One Receipt must be signed and remains in the Franchise Disclosure Document as the prospective franchise owner's copy. The other Receipt must be signed and returned to ALL Franchising, LLC.)